

CHARACTERISTICS OF CHRONIC ANKLE INSTABILITY AND THE ROLE OF JOINT MOBILISATION

Rajapaksha Mudiyanselage Ishanka Madhurangani Weerasekara

MPhil, BSc (Physiotherapy)

A thesis submitted in fulfilment of the requirements for the

degree of Doctor of Philosophy (Physiotherapy)

The University of Newcastle

August 2019

This research was supported by an Australian Government Research Training Program (RTP) Scholarship This is to certify that the thesis entitled **'Characteristics of Chronic Ankle Instability and the Role of Joint Mobilisation'** submitted in fulfilment of the requirements for the degree of Doctor of Philosophy (Physiotherapy) is in a form ready for examination.

Signed _____

Date 16 August 2019

Ishanka Weerasekara

School of Health Sciences

Faculty of Health and Medicine

The University of Newcastle

DECLARATION

I hereby certify that the work embodied in the thesis is my own work, conducted under normal supervision. The thesis contains no material which has been accepted, or is being examined, for the award of any other degree or diploma in any university or other tertiary institution and, to the best of my knowledge and belief, contains no material previously published or written by another person, except where due reference has been made. I give consent to the final version of my thesis being made available worldwide when deposited in the University's Digital Repository, subject to the provisions of the Copyright Act 1968 and any approved embargo.

I hereby certify that this thesis is in the form of a series of papers. I have included as part of the thesis a written declaration from each co-author, endorsed in writing by the Faculty Assistant Dean (Research Training), attesting to my contribution to any jointly authored papers.

Ishanka Weerasekara

Signature

Date

16 August 2019

ACKNOWLEDGEMENTS

This thesis marks the end of my candidature as a PhD student and I could not have accomplished it alone. There are numerous people to whom I owe my gratitude.

First and foremost, I express my sincere thanks to Professor Darren Rivett, Dr Peter Osmotherly and Associate Professor Suzanne Snodgrass for their patience, guidance and support during my PhD candidature. I am also extremely grateful to John Tessier and Dr Sarah Walmsley for their enormous assistance with some of the studies contained in this thesis.

My heartfelt thanks goes to Dr Jodie Marquez, Lucia Chambers, Associate Professor Coralie English and all the Discipline of Physiotherapy staff members at The University of Newcastle. I found a very lovely friend in each of you, and appreciate the research opportunities you provided and the support you gave me during my stay in Newcastle.

Thank you very much to my kind study participants who gave their time willingly to participate in my research. Without your generosity this thesis would not have been possible.

I also wish to acknowledge my dear friends, 'the PhD warriors' of the School of Health Sciences. We shared many things from birthday cake to debriefs about our progress, and helped each other overcome challenges in our studies. You made my time in Australia very memorable. Katherine, special thanks to you for being a beautiful friend to me and my family. Also to all of my lovely Sri Lankan friends and families in Newcastle. Thank you so much for your company, love and time spent with me and my family. I did not miss home much when I was with you all.

My sincere respect goes to Professor Chula Goonasekera, Dr Hilary Suraweera, Professor Shyama Banneheka and the physiotherapy academic staff members of the University of Peradeniya, Sri Lanka. I am proud to say that I am from the first cohort of the BSc (Physiotherapy) programme of Sri Lanka that you established.

I would also like to share the pride of this achievement with my Amma, Thaththa, Nangi, Malli, Achchila, and also to my Seeya looking down from heaven. This journey started from the day you taught me 'A, E, I, O, U', Thaththa. Thank you so much.

I would like to share the tears of joy with nobody else, but with you my better half Chamara. Thank you for your big heart, your shoulder to cry on, and listening to me throughout this whole process. I love you to the moon and back. And to you Shenara and Evan - I will share 100% of my time with you two after this I promise.

Lastly, I want to share this message I found somewhere on Twitter. It captures most of my feelings and I cannot write it any better: "I am a person of colour. I am an immigrant. I came from a third world country. English is my second language. I am minority. Australia has given me the opportunity to establish a wonderful life for my family. I am grateful."

PUBLICATIONS AND PRESENTATIONS

Papers published and submitted for publication in peer-reviewed journals as part of this thesis:

- Weerasekara, I., Osmotherly, P., Snodgrass, S., Marquez, J., de Zoete, R., & Rivett, D. A. (2018). Clinical benefits of joint mobilisation on ankle sprains: a systematic review and metaanalysis. *Archives of Physical Medicine & Rehabilitation*, 99(7), 1395-1412.e1395. doi:https://doi.org/10.1016/j.apmr.2017.07.019
- Weerasekara, I., Osmotherly, P. G., Snodgrass, S. J., Tessier, J., & Rivett, D. A. (2019). Effects of mobilisation with movement (MWM) on anatomical and clinical characteristics of chronic ankle instability: a randomised controlled trial protocol. *BMC Musculoskeletal Disorders*, 20(1), 75. doi:10.1186/s12891-019-2447-x
- Weerasekara, I., Osmotherly, P., Snodgrass, S., Tesseir, J., & Rivett, D. A. (2019). Is the fibula positioned anteriorly in weight-bearing in individuals with chronic ankle instability? (under review).
- Weerasekara, I., Snodgrass, S., Osmotherly, P., Tessier, J., & Rivett, D. A. (2019). Clinical characteristics of chronic ankle instability: a case-control study (under review).

Published abstracts from peer-reviewed conferences:

 Weerasekara, I., Osmotherly, P., Snodgrass, S., de Zoete, R., & Rivett, D. (2016, October 12-15). Clinical benefits of passive joint mobilisation on ankle sprains. In *Journal of Science &* *Medicine in Sport*. Paper presented at the Sports Medicine Australia Conference, Sydney, Australia. *20*, e49. doi:10.1016/j.jsams.2017.01.133

- Weerasekara, I., Osmotherly, P., Snodgrass, S., Marquez, J., de Zoete, R., & Rivett, D. (2017, October 19-21). A systematic review and meta-analysis of the clinical benefits of passive joint mobilisation on ankle sprains. Paper presented at the *Australian Physiotherapy Association Conference*, Sydney, Australia.
- Weerasekara, I., Osmotherly, P., Snodgrass, S., Tessier, J., & Rivett, D. (2019, May 10-13).
 Fibular position in chronic ankle instability (CAI) and the reliability of weight-bearing radiographic measurements of fibular position. Paper presented at the *World Confederation for Physical Therapy Congress*, Geneva, Switzerland.
- Weerasekara, I., Osmotherly, P., Snodgrass, S., Tessier, J., & Rivett, D. (2019, October 3-4). Fibular position in chronic ankle instability radiographically measured in weight-bearing (accepted). Paper will be presented at the 8th International Ankle Symposium, Amsterdam, Netherlands.

LIST OF TABLES

Table 2.1	Reported abnormalities of fibula position in lateral ankle ligament injuries	48
Table 3.1	Search strategy	65
Table 3.2	Description of the eligible studies	71
Table 3.3	Number of outcome evaluations investigating at each time point of interest, listed by the reported positive effects	101
Table 3.4	Assessment of the quality of evidence	106
Table 4.1	Characteristics of the participants	125
Table 4.2	Comparison of fibular position between individuals with unstable ankles and healthy ankles	126
Table 4.3	Characteristics of receiver operating characteristic (ROC) curve for fibular position	127
Table 4.4	Intra-class correlation coefficient (ICC _{2,1}) and standard error of measurement (SEM) with 95% confidence intervals (CI) for inter-rater and intra-rater reliability of measurements of fibular position	128
Table 5.1	Characteristics of the participants in each group	145
Table 5.2	Comparison of characteristics between individuals with CAI and healthy individuals, and between the affected ankle and other ankle of CAI participants	148
Table 5.3	Bivariate correlations identified as being statistically significant (p<.05) for the CAI affected ankle	153
Table 5.4	Results of final multivariate regression model (backwards Wald method, R ² =0.413) indicating variables predicting group membership of CAI (n=66)	156

Inclusion and exclusion criteria	170
Participants characteristics	218
Effect of MWM with taping on outcomes in individuals with CAI	207
Effect of MWM with taping on the changes in outcomes at each time point in individuals with CAI	213
Effects of MWM with taping on individuals with a displaced fibula (i.e. normalised fibular position \geq 27), compared to individuals with a normally positioned fibular on the changes of primary outcomes at each time point	220
Effects of MWM with taping on individuals with a mechanically unstable ankle compared to individuals with a mechanically stable ankle on the changes of primary outcomes at each time point	222
	 Inclusion and exclusion criteria Participants characteristics Effect of MWM with taping on outcomes in individuals with CAI Effect of MWM with taping on the changes in outcomes at each time point in individuals with CAI Effects of MWM with taping on individuals with a displaced fibula (i.e. normalised fibular position ≥ 27), compared to individuals with a normally positioned fibular on the changes of primary outcomes at each time point Effects of MWM with taping on individuals with a mechanically unstable ankle compared to individuals with a mechanically stable ankle on the changes of primary outcomes at each time point

LIST OF FIGURES

Figure 2.1	A schematic representation of the lateral ligaments of the ankle complex	27
Figure 3.1	Flow chart of study selection	90
Figure 3.2	Funnel plots	91
Figure 3.3	Percentage and number of outcome evaluations with and without positive findings following each technique combination of mobilisation for any clinically relevant outcome at any time point	93
Figure 3.4	PEDro scores for assessment of quality of individual criteria	95
Figure 3.5	Quality assessment tool for observational cohort and cross-sectional studies	96
Figure 3.6	PEDro scores for assessment of quality of individual intervention studies	98
Figure 3.7	Quality assessment tool for observational cohort and cross-sectional studies scores for assessment of quality of individual cohort studies	99
Figure 3.8	MD (95% CI) of the immediate effect of joint mobilisation on dynamic balance by pooling data from 5 studies (n=180)	103
Figure 3.9	SMD (95% CI) of the immediate effect of joint mobilisation on static balance by pooling data from 3 studies (n=100)	103
Figure 3.10	SMD (95% CI) of the immediate effect of joint mobilisation on weight-bearing DFROM by pooling data from 7 studies (n=249)	104
Figure 3.11	SMD (95% CI) of the immediate effect of joint mobilisation on pain intensity by pooling data from 2 studies (n=47)	104
Figure 3.12	SMD (95% CI) of the short-term effect of joint mobilisation on weight-bearing DFROM by pooling data from 2 studies (n=94)	105

Figure 4.1	Measurement of normalised fibular position	121
Figure 4.2	Overview of the recruitment process	123
Figure 4.3	Receiver operating characteristic (ROC) curve used to calculate the area under the curve and cut-off score for normalised fibular position	127
Figure 5.1	Inclusion and exclusion criteria for individuals with CAI	138
Figure 5.2	Flow chart of participants in the selection process	144
Figure 6.1	Flow of the study	174
Figure 6.2	Fibular position measurement; the distance between the anterior edge of the distal fibula and the anterior edge of the distal tibia	176
Figure 6.3	Anterior translation of the talus during the anterior drawer stress test is measured as the distance on X-ray from the posterior lip of the tibial joint surface to the nearest point of the talar dome	184
Figure 7.1	Design and flow of participants through the trial	195

LIST OF ABBREVIATIONS

ADL	activities of daily living
AMI	axial malleolar index
ANZCTR	Australian New Zealand Clinical Trial Registry
AP	antero-posterior
APTA	American Physical Therapy Association
ATFL	anterior talofibular ligament
AUC	area under the curve
BMI	body mass index
CAI	chronic ankle instability
CAIT	Cumberland ankle instability tool
CFL	calcaneofibular ligament
CI	confidence interval
СоР	centre of pressure
СТ	computed tomography
df	degrees of freedom
DFROM	dorsiflexion range of motion
FAAM	foot and ankle ability measure
FAOS	foot and ankle outcome score

GRADE	Grading of recommendations, assessment, development and evaluation
HVLA	high velocity low amplitude
IAC	International Ankle Consortium
ICC	intra-class correlation coefficient
ICF	International classification of functioning, disability and health
IMI	intermalleolar index
ITT	intention to treat
IV	inverse variance
LR	likelihood ratio
MCID	minimal clinically important difference
MD	mean difference
MDC	minimal detectable change
ML	medio-lateral
MRI	magnetic resonance imaging
MWM	mobilisation with movement
N/A	not applicable
OR	odds ratio
PL	postero-lateral
РМ	postero-medial
POLICE	protection, optimal loading, ice, compression and elevation

- pressure pain threshold PPT posterior talofibular ligament PTFL quality of life QOL randomised controlled trial RCT RICE rest, ice, compression and elevation receiver operating characteristic ROC range of motion ROM SD standard deviation SE standard error SEBT star excursion balance test SEM standard error of measurement standard mean difference **SMD** talocrural joint TCJ TFJ tibiofibular joint time to stabilisation TTS visual analogue scale VAS
- WHO World Health Organisation

TABLE OF CONTENTS

DECLARATION ii	ii
ACKNOWLEDGEMENTSi	v
PUBLICATIONS AND PRESENTATIONS	'n
LIST OF TABLES vii	ii
LIST OF FIGURES	X
LIST OF ABBREVIATIONS xi	ii
TABLE OF CONTENTS	1
Abstract	8
Chapter 1 Introduction1	1
1.1 Background and context1	1
1.1.1 Chronic ankle instability (CAI)1	1
1.1.2 Prevalence, effects and burden of CAI1	2
1.1.3 Rehabilitation of CAI1	4
1.1.4 Principles of MWM1	5
1.2 Research aims1	7
1.3 Significance1	7
1.4 Overview of the thesis	9
1.5 Scope/delimitations	0
Chapter 2 Literature review	2

2.1 Chronic ankle instability (CAI)	22
2.1.1 General description of CAI	22
2.1.2 Defining CAI	23
2.1.3 Mechanism of CAI	24
2.2 Anatomy and pathophysiology of CAI	25
2.2.1 Talocrural joint	25
2.2.2 Subtalar joint	27
2.2.3 Distal tibiofibular syndesmosis	
2.3 Clinical evaluation and diagnosis of CAI	
2.3.1 Patient reported findings	
2.3.2 Physical examination and tests	
2.3.3 Radiological examination	46
2.3.4 Other tools to assess CAI	54
2.4 Management of CAI	54
2.4.1 Conservative interventions	55
2.4.2. Surgical interventions	
2.5 Summary	
Chapter 3 Clinical benefits of joint mobilisation on ankle sprains: a systemati	c review and meta-
analysis	61
Overview	61
3.1 Introduction	63
3.2 Method	65
3.2.1 Registration	65

3.2.2 Search strategy	65
3.2.3 Identification and selection of studies	67
3.2.4 Assessment of methodological quality	70
3.2.5 Assessment of the quality of evidence	70
3.2.6 Data extraction and statistical analysis	70
3.3 Results	89
3.3.1 Selection and characteristics of included studies	89
3.3.2 Common mobilisation techniques used in rehabilitation of ankle sprains	93
3.3.3 Quality of studies	95
3.3.4 Effects of mobilisation on sub-acute/chronic ankle sprains	101
3.3.5 Immediate effects of mobilisation on ankle sprains	103
3.3.6 Short-term effects of mobilisation on ankle sprains	106
3.3.7 Long-term effects of mobilisation on ankle sprains	107
3.3.8 Quality of evidence	107
3.4 Discussion	109
3.4.1 Study limitations	113
3.5 Conclusion	114
Chapter 4 Is the fibula positioned anteriorly in weight-bearing in individuals with c	hronic ankle
instability?	115
Overview	115
4.1 Introduction	117
4.2 Materials and methods	119
4.2.1 Analysis	122

4.3 Results	124
4.3.1 Participant characteristics	125
4.3.2 Comparison of fibular position	125
4.3.3 ROC analysis	125
4.3.4 Reliability of the weight-bearing X-Ray measures of fibular position	129
4.5 Discussion	130
Chapter 5 Clinical characteristics of chronic ankle instability: a case-control study	136
Overview	136
5.1 Introduction	137
5.2 Methods	139
5.2.1 Participants	139
5.2.3 Procedures	141
5.2.3 Analysis	144
5.3 Results	146
5.3.1 Participant characteristics	146
5.3.2 Comparison of the characteristics	148
5.3.3 Regression analysis	155
5.4 Discussion	159
5.5 Conclusion	164
Chapter 6 Effects of mobilisation with movement (MWM) on anatomical and clinical cha	aracteristics of
chronic ankle instability: a randomised controlled trial protocol	166
Overview	166

6.1 Introduction168
6.2 Methods171
6.2.1 Design
6.2.2 Participants171
6.2.3 Procedure
6.2.4 Outcome measures
6.2.5 Application of the intervention
6.2.6 Sample size and data analysis
6.3 Discussion
Chapter 7 Mobilisation with movement with taping is not effective in changing fibular position or
improving clinical outcomes in chronic ankle instability: a randomised controlled trial193
Overview193
7.1 Introduction195
7.2 Materials and Methods197
7.2.1 Design
7.2.2 Participants, therapists, centres
7.2.2 Intervention

7.2.3 Intervention	199
7.2.4 Outcome measures	201
7.2.5 Data analysis	204
7.3 Results	205
7.3.1 Flow of participants, therapists, centres through the study	205
7.3.2 Compliance with trial method	207
7.3.3 Effect of 'Mulligan's MWM with fibular repositioning taping' intervention	207

7.3.4 Effect on fibular position in individuals with CAI	208
7.3.5 Effect on common clinical measures in individuals with CAI	208
7.3.6 Effect on subgroups	209
7.4 Discussion	227
Chapter 8 Discussion and conclusions	231
8.1 Summary of study findings	232
8.2 Significance of the findings of the thesis	235
8.3 Strengths and limitations of the studies	237
8.3.1 Systematic review	237
8.3.2 Case-control study	238
8.3.3 Randomised controlled trial	239
8.4 Clinical implications of findings	241
8.5 Future research	242
8.6 Conclusions	243
References	245
Appendices	289
Appendix 1 Statements of collaboration from co-authors	289
Statement from Darren A. Rivett relating to papers published with Ishanka Weerasekara	290
Statement from Peter G. Osmotherly relating to papers published with Ishanka Weerasekara	292
Statement from Suzanne J. Snodgrass relating to papers published with Ishanka Weerasekara.	294
Statement from John Tessier relating to papers published with Ishanka Weerasekara	296
Statement from Jodie Marquez relating to a paper published with Ishanka Weerasekara	298

Statement from Rutger de Zoete relating to a paper published with Ishanka Weerasekara	300
Appendix 2 Human research ethics committee approval: case-control study	303
Appendix 3 Human research ethics committee approval: randomised controlled trial	307
Appendix 4 Safety review notification	311
Appendix 5 Dosimetry report: case-control study	314
Appendix 6 Dosimetry report: randomised controlled trial	316
Appendix 7 PROSPERO systematic review registration notification	318
Appendix 8 ANZCTR clinical trial registration notification	320
Appendix 9 Information statement: case-control study	322
Appendix 10 Consent form: case-control study	327
Appendix 11 Information statement: randomised controlled trial	329
Appendix 12 Consent form: randomised controlled trial	335
Appendix 13 Participant recruitment flyers: case-control study	337
Appendix 14 Participant recruitment flyers: randomised controlled trial	340
Appendix 15 Data collection forms (both case-control and RCT studies) including questionna	ires343
Appendix 16 Journal publications	359

Abstract

Chronic ankle instability (CAI) is the occurrence of giving way and/or recurrent sprain and/or feelings of instability of a previously injured ankle. Compared to a simple ankle sprain, the additional morbidity and additional costs incurred due to diagnostic imaging related to CAI create significant economic and other societal consequences. Direct costs can include consultations with physicians, physiotherapists and other health professionals, diagnostic imaging and various hospital expenses, while indirect costs may arise from productivity losses, absenteeism (from school, work and sports) and diminished levels of physical activity resulting in economic burden on the healthcare system and for the individual patient. Rehabilitation of CAI commonly involves manual therapy procedures applied to joints in the ankle region, such as non-thrust passive joint mobilisation, high-velocity thrust manipulation and mobilisation with movement (MWM) as described by Brian Mulligan. These techniques largely involve a continuum of skilled passive movements of joints that are applied at different speeds and amplitudes. The overall aims of this thesis were to explore the clinical characteristics of CAI and to determine the effects of joint mobilisation in CAI. This thesis comprises four studies designed to meet these aims, presented as five manuscripts which have either been published in peer-reviewed journals or are in the process of review or submission.

Study 1 is a systematic review and meta-analysis undertaken to evaluate the current evidence for joint mobilisation in the clinical rehabilitation of ankle sprains, using the previously published body of literature. Based upon this investigation, joint mobilisation appears to be beneficial for improving dynamic balance immediately after application and dorsiflexion range of movement in the short-term

in chronic ankle sprains. The results of this study also suggested that the combination of MWM and associated fibular repositioning taping is more likely to exhibit a clinical benefit than other assessed mobilisation techniques, and potentially supporting the hypothesis of Brian Mulligan that the distal fibula is displaced anteriorly in CAI.

Study 2 explored the position of the distal fibula in relation to the position of the distal tibia in CAI using weight-bearing radiographs. The findings of this study indicated that there is a more anteriorly positioned fibula in individuals with CAI compared to individuals with healthy ankles when assessed in a weight-bearing position, consistent with Mulligan's hypothesis. Notably, this study was the first to use weight-bearing radiographs to measure the fibula position in CAI.

Study 3 investigated other clinical characteristics of CAI including ankle dorsiflexion range of motion, balance, self-reported function, pain and pressure pain threshold which have not been consistently reported, or for which little research has been published in the previous literature. It was found that individuals with CAI exhibit a multi-factorial presentation including impaired ankle dorsiflexion range, reduced static and dynamic balance, lower self-reported function and greater pain intensity, compared to individuals with healthy ankles. Despite the persistence of pain often reported in CAI, no evidence was found to suggest maladaptive central nervous system sensitisation (nociplastic pain).

Study 4 involved a randomised controlled trial (RCT) to investigate the effects of MWM of the distal tibio-fibular joint with fibular repositioning taping on fibular position and other clinical characteristics of CAI. The protocol for this study was published, and in contrast to almost all previously published studies of this manual therapy technique, highlighted the long-term nature of the trial with a one year follow-up. The immediate and short-term findings revealed no significant differences in any of the outcomes measured except for improvements in two self-reported function subscales (pain and sports on the foot and ankle outcome score (FAOS) questionnaire) in the placebo (detuned laser) intervention group. The long-term results were still being collected at the time of the submission of this thesis so are not reported or considered in the discussion.

The body of work contained in this thesis extends our current understanding of CAI and its clinical management. The results have implications for the identification of the features of CAI and for improving its treatment using joint mobilisation. Reproduction of the contained research using more homogeneous samples of CAI may permit a greater understanding of this persistent condition and how it responds to manual therapy. In addition, the use of weight-bearing radiographs in future studies to assess fibula position may be a more functional method. Other future directions for research and implications for clinical practice are discussed in detail in relation to the results of each of the studies in this thesis.

Chapter 1 Introduction

1.1 Background and context

1.1.1 Chronic ankle instability (CAI)

Lateral ankle sprains are among the most common sports-related injuries and also have high recurrence rates (Attenborough et al., 2014; Hiller et al., 2012; Simon et al., 2014). A large percentage of individuals with lateral ankle sprains develop chronic ankle instability (CAI) (Kobayashi & Gamada, 2014; Vicenzino et al., 2006; Wikstrom & Brown, 2014). Chronic ankle instability, chronic lateral ankle instability, ankle instability, residual ankle instability, chronic instability, recurrent instability, recurrent lateral ankle instability, and chronic ankle sprain have all been used interchangeably in the literature to describe the condition referred to as CAI (Hiller et al., 2011).

Two main subgroups of CAI are widely accepted; those with mechanical instability and those with functional ankle instability (Freeman, 1965; Hertel, 2002; Hiller et al., 2011). These are not mutually exclusive but part of a continuum (Hertel, 2002; Hiller et al., 2011). Functional instability is proposed to result from functional insufficiencies such as impaired proprioceptive and neuromuscular control, while mechanical instability is thought to result from isolated or combined anatomical changes occurring after an initial sprain (Hertel, 2002; Hiller et al., 2011). These anatomical changes include impaired arthrokinematics, pathological laxity, synovial tissue changes, and the development of degenerative joint disease. Altered fibular position at the inferior tibio-fibular joint is one arthrokinematic restriction proposed to be related to recurrent ankle sprains in CAI (Hertel, 2002).

Due to inconsistency in participant selection criteria for CAI in published studies, and the associated limitations in generalising to the target population, the International Ankle Consortium (IAC) has published standard criteria for inclusion and exclusion of participants with CAI when recruiting for research (Gribble et al., 2013). These inclusion and exclusion criteria, based on the ankle injury history and the timing of the ankle sprain, provide a broader description of the condition despite not quantifying any associated disability.

1.1.2 Prevalence, effects and burden of CAI

Epidemiological studies investigating ankle sprains in various countries highlight the high incidence during sports training and games, with rates reported in person years as 7 per 1000 in Denmark, 6.09 per 1000 in the United Kingdom, and 2.15 per 1000 in the United States (Bridgman et al., 2003; Holmer et al., 1994; Waterman, Belmont, et al., 2010). In Sri Lanka in 2015, 4.6% of adults in the community were affected by ankle sprains which lead to chronicity, suggesting a similar picture overall between developing and developed countries (Weerasekara & Hiller, 2017). Persistent residual symptoms of an initial ankle sprain including recurrent sprains, episodes of giving way at the ankle joint, pain, swelling, and decreased function have been termed CAI (Delahunt, Coughlan, et al., 2010; Gribble et al., 2013; Hubbard & Hicks-Little, 2008). It has been reported that up to 70% of these ankle sprains lead to CAI in the short-term (Gribble et al., 2016a). CAI may lead to articular degeneration of the ankle joint, increasing the risk of secondary osteoarthritis (Fong et al., 2009; Gribble et al., 2016a; Hiller et al., 2012; Hubbard & Hicks-Little, 2008), which can adversely impact on quality of life (QOL) (Hiller et al., 2012).

The high prevalence and long-term deficits associated with CAI produce a large societal and economic burden (Fong et al., 2007; Gribble et al., 2016a). Generally, most ankle injury direct costs are spent on consultations with physicians and physiotherapists, diagnostic imaging and hospital facility usage (Bielska et al., 2017; Gribble et al., 2016a). The combined direct and indirect costs of ankle sprain rehabilitation has been reported as ranging from \$1809 to \$5271 per person (2016 USD) (Bielska et al., 2017). The overall healthcare cost of the management of CAI is thought to be greater than this, as it is associated with additional morbidity and additional diagnostic testing, such as advanced imaging procedures (Gribble et al., 2016b; Soboroff et al., 1984). Moreover, with CAI being the second leading cause of post-traumatic ankle osteoarthritis (Saltzman et al., 2005), it contributes substantially to osteoarthritic surgical cases (Delco et al., 2017). The estimated 4400 total ankle replacements and 25,000 ankle fusions carried out in the United States each year reflects the economic burden associated with ankle osteoarthritis (Delco et al., 2017). Further, productivity loss (with a mean duration of unemployment of 7-29 days), absenteeism (from school, work and sports), loss of leisure time, diminished level of physical activity, and cost of travel all contribute to the indirect costs associated with CAI (Bielska et al., 2017; Fong et al., 2007; Gribble et al., 2016a; Waterman, Belmont, et al., 2010).

Impairments related to CAI such as increased ligament laxity and proprioceptive deficits may lead to activity limitations, and consequently participation restrictions occupationally or in some sports. This reflects the potential impact of CAI in line with the International classification of functioning, disability and health (ICF) of the World health organisation (WHO) (Hiller et al., 2011).

1.1.3 Rehabilitation of CAI

Clinical practice guidelines linked to the ICF from the orthopaedic section of the American Physical Therapy Association (APTA) recommends manual therapy, therapeutic exercises and activities, and sports related activity training in the progressive loading rehabilitation phase of ankle sprain, based on the scientific literature accepted for publication prior to April 2012 (Martin et al., 2013). Further, this conservative treatment may be supported by proprioceptive training and external ankle support such as taping and braces (De Vries et al., 2003). Various manual therapy procedures, including graded passive joint mobilisation, joint manipulation, and non weight-bearing and weight-bearing mobilisations with movement (MWM) have all been recommended during the progressive phase loading of ankle sprain rehabilitation (Cruz-Diaz et al., 2015; Doherty et al., 2017; Loudon et al., 2014).

1.1.3.1 Joint mobilisation

Mobilisation is defined as "comprising a continuum of skilled passive movements to the joints and/or related soft tissues that are applied at varying speeds and amplitudes, including a small-amplitude and high-velocity therapeutic movement" (American Physical Therapy Association, 2003, p2). A variety of

mobilisation techniques are commonly used by physiotherapists based upon varying approaches and paradigms for the treatment of ankle sprains and ankle instability.

A number of studies have investigated the effects of joint mobilisation and manipulation on ankle sprains using a variety of outcome measures including pain, range of motion (ROM), and restricted function from the acute to chronic stages of recovery, and these have shown mixed findings (Collins et al., 2004; Cosby et al., 2011; Cruz-Diaz et al., 2015; Delahunt et al., 2013; Gilbreath et al., 2014; Gómez et al., 2015; Green et al., 2001; O'brien & Vicenzino, 1998; Penso, 2008; Vicenzino et al., 2006; Woodman et al., 2013). These studies have mostly investigated MWM, the mobilisation intervention which was introduced by New Zealand physiotherapist Brian Mulligan in the latter part of the 20th century (Mulligan, 1993; Mulligan, 1995).

1.1.4 Principles of MWM

MWM techniques as described in the Mulligan Concept of manual therapy, have been reported to be effective in improving clinically important outcomes of CAI including dorsiflexion ROM (DFROM), postural control, and self-reported instability (Cruz-Diaz et al., 2015; Reid et al., 2007; Vicenzino et al., 2006). The original proposed mechanism of action of MWM is known as the 'positional fault hypothesis.' In this hypothesis, it is postulated that MWM works by correcting a minor bony incongruity which is at the source of the patient's presenting painful movement problem (Hing et al., 2015).

In the application of MWM, a pain-free passive accessory joint mobilisation/glide is sustained while the patient performs an active or passive physiological movement/task which was previously painfully limited (Hoch & Mckeon, 2010; Vicenzino et al., 2011) . This problematic movement or task is also referred to as the client specific impairment measure, and most frequently this is a movement or muscle contraction performed to the onset of pain, or to the end of the available ROM or maximum muscle contraction if no pain is produced during the technique. Vicenzino et al (2011) have proposed a model to explain the effects of MWM. These authors propose that the underpinning mechanism has both a biomechanical component (which results in a corrective transient bony displacement) and a neurophysiological component (which results in non-opioid hyperalgesia), together producing the observable clinical benefits to the patient (Vicenzino et al., 2011). The capacity of MWM to reverse any bony positional displacement has not been well investigated and remains controversial, with longterm effects unknown.

Similarly, the effects of MWM on investigated clinical deficits in individuals with CAI compared to individuals with healthy ankles, such as in dorsiflexion range of motion (DFROM), balance and function have not been consistently detected at any time point, including from immediately after application to the longer term.

1.2 Research aims

The overall aim of this thesis is to determine the clinical benefits of ankle joint mobilisation on CAI. Specifically, the aims of the planned studies are as follows:

- To perform a systematic review of the literature on the clinical benefits of joint mobilisation including MWM on ankle sprains and CAI
- 2. To determine the presence of fibular displacement as an anatomical characteristic of ankles with chronic instability
- 3. To determine the clinical characteristics of ankles with chronic instability
- 4. To assess the effect of MWM on fibular position in CAI
- 5. To evaluate the effectiveness of MWM on the immediate, short and long term clinical outcomes of CAI.

1.3 Significance

This thesis is significant in addressing several deficiencies in the literature related to CAI. First, it remains unclear exactly which anatomical and clinical characteristics contribute to CAI, and whether, and in what circumstances, joint mobilisation (including MWM) is effective in the management of CAI. It is also unknown which joint mobilisation techniques might provide the best balance of beneficial and adverse effects. Considering this, a systematic and up to date review of the scientific evidence is important. Further, studies investigating the anatomical and clinical characteristics of CAI might be helpful in understanding the underpinning mechanisms by which joint mobilisation including MWM may produce any effects. Finally, there is a dearth of sufficiently powered, high quality and

well reported RCTs with a long-term follow-up, evaluating the effects of MWM in the treatment of CAI and providing trustworthy guiding evidence for clinicians.

This PhD project comprises a systematic review and meta-analysis, a case-control study and a randomised controlled trial (RCT). The systematic review and meta-analysis is the first review of the literature in which joint mobilisation is investigated as the sole intervention to determine its clinical benefits in the management of ankle sprains. In particular, evidence for the effects of MWM are evaluated for individuals with ankle sprains, including CAI. The next study uses a case-control design to develop a better understanding of the anatomical and clinical characteristics of CAI. This study explores fibular displacement in relation to the tibia in CAI in the more functional weight-bearing position, and for the first time in the literature. It reports a discrimination (or cut-off) scores for an abnormally positioned fibula in relation to the tibia that best differentiates individuals with and without ankle instability. Further, it assesses specific clinical outcome measures, including that of pressure pain threshold (PPT) to investigate whether there is any centrally driven pain response in CAI. As the final study of the thesis, the RCT aims to determine the effectiveness of MWM on various clinically relevant outcomes in CAI, including long-term benefits which have not previously been investigated. Moreover, the capacity of MWM to reverse any displacement of the fibula is assessed and this is the first study to investigate the effect of joint mobilisation on fibular displacement. The overall thesis will summarise the current research about the effects of joint mobilisation on ankle sprains, report the anatomical and clinical characteristics of CAI compared to healthy ankles, and measure the effects of MWM on various clinical outcomes of CAI.

1.4 Overview of the thesis

This thesis investigates the clinical benefits of the MWM joint mobilisation on CAI. It comprises a narrative literature review followed by a series of research papers (both published peer-reviewed articles and journal manuscripts under review) which describe the four studies undertaken, the overall findings of which are consolidated in a concluding discussion chapter.

The chapter outlines are as follows:

- Chapter 1 provides introductory information regarding CAI, joint mobilisation and details the aims and significance of the thesis
- Chapter 2 describes the available literature on the characteristics of CAI and the potential role of joint mobilisation in the rehabilitation of CAI
- Chapter 3 details Study 1 which reviews the current evidence for the clinical benefits of joint mobilisation for ankle sprains, in the form of a systematic review and meta–analysis (Weerasekara et al., 2018)
- Chapter 4 details Study 2, a case-control study exploring changes in fibular position compared to the position of the tibia as an anatomical characteristic of CAI, in order to investigate whether there is a difference in the fibular positon in CAI compared to healthy individuals.
- Chapter 5 details Study 3, a case-control study exploring clinical characteristics of CAI, in
 order to investigate whether individuals with CAI have specific deficits in terms of ankle
 DFROM, balance, self-reported function, pain and PPT compared to healthy matched controls,
 and do any of these differentiate between individuals with CAI and healthy individuals.

- Chapter 6 reports the protocol for Study 4, an interventional study to assess the effect of MWM on the anatomical and clinical properties of CAI (Weerasekara, Osmotherly, Snodgrass, et al., 2019b). This RCT is designed to investigate the immediate, short and long term effects of MWM on fibular position and clinical outcomes (ankle DFROM, balance, self- reported function, PPT, and pain intensity) in individuals with CAI, as well as identify any prognostic factors indicating recovery from CAI using MWM.
- Chapter 7 reports the actual findings of Study 4, that is the results of the interventional study to assess the effects of MWM on the anatomical and clinical properties of CAI. The immediate and short-term effects are reported in this chapter, but at the time of submission of the thesis the long-term data were still being collected.
- Chapter 8, the final chapter, provides a concluding summary and overarching discussion of the key findings and limitations of the various studies comprising the thesis, as relevant to the potential role of joint mobilisation in CAI and its pathophysiology, with recommendations for future research and implications for clinical practice.

1.5 Scope/delimitations

Whilst many joint mobilisation/manual therapy techniques have been suggested as useful in the rehabilitation of individuals with CAI, the scope of this thesis became increasingly focussed on the MWM joint mobilisation technique. This decision was made because the initial study, the systematic review and meta-analysis of previously published studies, suggested that the combination of Mulligan's MWM and associated fibular repositioning taping is more likely to produce a clinical

benefit than the other assessed joint mobilisation techniques (Weerasekara et al., 2018). Other physical interventions commonly employed in the treatment of CAI were not within the scope of this thesis. Finally, the decision was made to focus most of the research conducted in the thesis on CAI, because typically joint mobilisation including MWM is not the preferred choice of treatment in the acute stage of ankle sprain (Van Den Bekerom et al., 2013), and because CAI poses a greater burden to society and the individual.

Chapter 2 Literature review

2.1 Chronic ankle instability (CAI)

2.1.1 General description of CAI

The most frequently occurring acute sporting injury is reported to be the acute ankle sprain, accounting for about 14% of all sporting injuries (Fong et al., 2009; Garrick, 1977; Hertel, 2002; Powden et al., 2016). The acute ankle sprain is regularly under-treated, or even not treated at all as it is often not considered a severe or debilitating injury (Hertel, 2002; Mckay et al., 2001; Weerasekara & Hiller, 2017; Weerasekara et al., 2016). However, twenty to forty percent of individuals with acute ankle sprains are considered to be at risk of developing CAI due to varying factors such as inadequate rehabilitation or persistent deficits developed as a long-term sequela of the initial sprain (Hershkovich et al., 2015; Kobayashi & Gamada, 2014; Miklovic et al., 2018; Weerasekara, Tennakoon, et al., 2019).

The occurrence of repetitive bouts of lateral ankle instability, resulting after a single severe sprain or after numerous minor sprains is clinically considered to indicate CAI (Guillo et al., 2013; Hertel, 2002). Persistent pain, 'giving way' and feelings of instability in the ankle, and recurrent sprains, potentially leading to persistent disability are some of the key features of CAI (Attenborough et al., 2014; Gribble et al., 2016a). Although acute lateral ankle sprains occur at approximately the same rate between males (78.8%) and females (77.3%) (Tummala et al., 2018). CAI has been shown to have a comparatively greater prevalence among males (males, 1.1%; females, 0.7%) (Hershkovich et al., 2015).
Soccer, basketball and volleyball are the most represented sports in cases of CAI (Attenborough et al., 2014), and increased body mass index (BMI) and greater body height have also been associated with an increased incidence of CAI (Hershkovich et al., 2015). Commonly the mechanism of ankle sprains involves a sudden violent inversion or supination movement (Fong et al., 2009; Powden et al., 2016), however the mechanism by which CAI develops remains somewhat unclear.

While the direct costs for rehabilitation of acute lateral ankle sprains are relatively low, the indirect costs from follow-up rehabilitation and injury-associated time loss result in a higher financial impact (Gribble et al., 2016a; Weerasekara & Hiller, 2017). Importantly, the impacts on patient QOL, including on leisure time and work productivity are considerable (Gribble et al., 2016a).

2.1.2 Defining CAI

CAI has been defined in many ways but is mostly described as "an encompassing term used to classify an individual with both mechanical and functional instability of the ankle joint" (Gribble, Philip A et al., 2013, p. 583; Hertel, 2002). The original model, as proposed by Hertel explains CAI as resulting from both mechanical instability and functional instability (Hertel, 2002). Hertel's model of CAI has been commonly used in research, with modifications later added by Hiller et al (Hiller et al., 2011). According to the revised Hertel model, CAI is a heterogeneous condition of perceived instability, which may include mechanical or functional instability or a combination of both (Hertel, 2002; Hiller et al., 2011). Notably, no correlation has been found between mechanical and functional instabilities, though they may both be present as parallel phenomena (Peters et al., 1991). Recently, selection criteria for patients with CAI in controlled research have been published, focused on defining a history of acute lateral ankle sprains and subsequent functional limitations as indicated by episodes of 'giving way' and validated self-reported outcome tools (Gribble et al., 2013). Interestingly, measures to evaluate mechanical instability were not considered amongst these criteria (Lohrer et al., 2015).

Functional instability has been proposed to result from functional insufficiencies such as impaired proprioception, neuromuscular control, postural control or strength, while mechanical instability is thought to result from isolated or combined structural changes after the initial sprain (Hertel, 2002). These structural changes may include impaired arthrokinematics, pathological laxity, synovial tissue changes, and the development of degenerative joint disease which can cause alteration of the mechanics of one or more joints within the ankle complex (Hertel, 2002; Hiller et al., 2011).

2.1.3 Mechanism of CAI

The initial tissue damage during the first episode of a lateral ankle sprain may develop into a combination of deficits such as sensory impairments and motor impairments that eventually can influence the clinical outcomes in a patient with CAI (Hertel & Corbett, 2019). The development of CAI following a lateral ankle sprain may be related to both personal (e.g. age and sex) and environmental (e.g. physical activity and sports participation) factors (Mccann & Gribble, 2016). Extensive research has been carried out to explore the alterations of anatomical structures and function in CAI, but the reasons for persistence of the impairments in some people but not others have not been

clearly explained (Mccann & Gribble, 2016). Further, development of CAI is impacted by patient resilience and self-efficacy, social attitudes, and rehabilitation adherence (Mccann & Gribble, 2016). Underlying central nervous system changes related to pain in CAI remain unclear, but central sensitisation may be a possible factor for the persistence of pain.

A recent update to the Hertel model has attempted to explain the development of CAI according to the ICF classification of the WHO, utilising some more contemporary injury models such as the biopsychosocial model, dynamic systems theory and the neuromatrix of pain theory (Hertel & Corbett, 2019). Essentially, it is a longstanding failure to cope with specific impairments resulting from the initial trigger of sensorimotor changes (via inflammatory and pain mediators) which occurred during the acute injury (Hertel & Corbett, 2019).

2.2 Anatomy and pathophysiology of CAI

The ankle joint complex comprises three major articulations: the talocrural joint, the subtalar joint, and the distal tibiofibular syndesmosis (Fong et al., 2009).

2.2.1 Talocrural joint

The talocrural joint behaves like a hinge joint and is formed by the articulation of the dome of talus, the tibial plafond, the medial malleolus and the lateral malleolus. The talocrural joint is supported laterally by three main ligaments; the anterior talofibular ligament (ATFL), the calcaneofibular ligament (CFL)

and the posterior talofibular ligament (PTFL) (Figure 2.1). Medially it is supported by the deltoid ligament (Fong et al., 2009; Hintermann, 2005). The ATFL runs from the anterior rim of the lateral malleolus to the lateral aspect of the talar neck. The CFL originates from tip of the lateral malleolus and inserts onto the lateral face of the calcaneus where it lies between the calcaneus and the peroneal tendons. The PTFL originates from the posterior part of the lateral malleolus and inserts into the posterior part of the talus. Two other ligaments are also located laterally; the anterior inferior tibiofibular ligament (origin, anterior part of the distal fibula; insertion, anterior part of the tibia) and the posterior inferior tibiofibular ligament (origin, posterior tibia; insertion, posterior malleolus) (Mckiernan et al., 2017). Each of the lateral ligaments has a role in stabilising the talocrural joint and/or the subtalar joint, depending on the position of the ankle (Hintermann, 2005). The medial ligament complex is composed of superficial and deep portions of the deltoid ligament. The tibiocalcaneal, tibiospring and tibionavicular ligaments comprise the superficial portion, and span from the medial malleolus to insert broadly onto the calcaneus, navicular, talar neck and spring ligament. The deep anterior tibiotalar, superficial posterior tibiotalar and deep posterior tibiotalar ligaments constitute the deep deltoid complex (Hintermann, 2005).



Figure 2. 1 The lateral ligaments of the foot and ankle joint. The anterior talofibular ligament is typically composed of two separate bands (1) Tip of the lateral malleolus; (2) tibia; (3) anterior tibiofibular ligament; (4) distal fascicle of the anterior tibiofibular ligament; (5) superior band of the anterior talofibular ligament; (6) inferior band of the anterior talofibular ligament; (7) lateral articular surface of the talus; (8) neck of the talus; (9) head of the talus; (10) calcaneofibular ligament; (11) talocalcaneal interosseous ligament; (12) cervical ligament; (13) talonavicular ligament; (14) navicular (Golanó et al., 2010).

2.2.2 Subtalar joint

The subtalar joint is a gliding joint formed by the articulation between the talus and the calcaneus (Norkus & Floyd, 2001). The anterior compartment of the subtalar joint is formed by the talocalcaneonavicular joint and the posterior component is formed by the talocalcaneal joint, separated by the canalis and sinus tarsi. The talocalcaneonavicular joint may be characterised as a ball-and-socket

joint (Bartoníček et al., 2018; Schuenke et al., 2009). The talocalcaneal joint of the subtalar complex is reinforced by anterior, posterior, lateral, medial, and interosseous talocalcaneal ligaments (Norkus & Floyd, 2001). The talocalcaneonavicular joint of the subtalar complex is supported by the spring ligament, the medial part of the bifurcate ligament (calcaneonavicular ligament) and the dorsal talonavicular ligament. The spring ligament is formed collectively by the superomedial ligament (largest), the medial plantar oblique ligament (thinnest) and the inferior plantar ligament (shortest) (Bartoníček et al., 2018). The bifurcate ligament is assumes a typical Y or V shape, and originating from the anteromedial edge of the anterior calcaneal process and inserting onto the lateral pole of the navicular close to the edge of its talar articular surface. The talonavicular ligament runs from the superior and partly from the lateral aspect of the neck of the talus into the upper surface of the navicular (Bartoníček et al., 2018). The canalis and sinus tarsi contain a complex system of ligaments and the roots of the inferior extensor retinaculum, the cervical ligament, the interosseous talocalcaneal ligament, and the anterior talocalcaneal ligament can be seen (Bartoníček et al., 2018).

2.2.3 Distal tibiofibular syndesmosis

The distal tibiofibular syndesmosis is formed by the articulation between the distal tibia and fibula (Fong et al., 2009; Schuenke et al., 2009). This articulation is further subdivided according to region into the superior tibiofibular joint, the interosseous membrane, and the inferior tibiofibular joint (Norkus & Floyd, 2001). The superior and inferior tibiofibular joints of distal tibiofibular syndesmosis are syndesmotic articulations between the convex surface of the fibula and the concave surface of the tibia, where the superior tibiofibular joint is located proximally and the inferior tibiofibular joint is located distally (Norkus & Floyd, 2001). The interosseous membrane is a thick osseofascial structure

extending from the tibial periosteum to the fibula and anchors the fibula and tibia together (Norkus & Floyd, 2001). The superior tibiofibular joint of the distal tibiofibular syndesmosis is stabilised by the anterior superior tibiofibular and posterior superior tibiofibular ligaments (Norkus & Floyd, 2001). The interosseous ligament binds the tibia and fibula, and is continuous with the interosseous membrane at the syndesmosis (De Vries, 2009; Norkus & Floyd, 2001). The inferior tibiofibular joint is reinforced by the anterior inferior tibiofibular ligament, the posterior inferior tibiofibular ligament, and the interosseous ligament (Norkus & Floyd, 2001).

With regard to the lateral ligaments which are relevant for inversion injuries, the ATFL is on maximum tension when the foot is in plantarflexion and supination, the CFL in dorsiflexion and supination, and the PTFL in dorsiflexion (De Vries, 2009). Lateral ankle sprains account for 80-85% of all ankle sprain injuries (Ferran & Maffulli, 2006) and 65-73% involve an isolated injury to the ATFL (Fong et al., 2009; Lynch, 2002). Combined injury of the ATFL and CFL is reported as being 20-25% of cases, and the PTFL has a lesser involvement of around 10% (Lynch, 2002; Renström & Konradsen, 1997). Another 3-10% of all ankle sprains are syndesmosis injuries or 'high ankle' sprains (Dubin et al., 2011; Renström & Konradsen, 1997), which may also involve a partial tear of the anterior deltoid ligament or malleolus fractures (Lynch, 2002; Renström & Konradsen, 1997). Only 3-33% of all ankle sprains involve the medial aspect of the ankle as the strong medial deltoid ligament is quite resistant to tearing (Fallat et al., 1998; Lynch, 2002; Ribbans & Garde, 2013). Most ankle sprains recover within several weeks to 2-3 months depending on the grade and site of the sprain, although 20-40% result in chronic sequelae (Krabak & Baima, 2008).

2.3 Clinical evaluation and diagnosis of CAI

A history of an ankle sprain preludes the signs and symptoms of CAI. A lateral ankle sprain is usually associated with an episode of an acute inversion/supination injury to the ankle, with or without an audible 'snapping' sound or feeling of a 'tearing' sensation (Lynch, 2002). Generalised pain and swelling are also present (Guillo et al., 2013; Lynch, 2002). CAI is accompanied by the perception of an abnormal ankle with a combination of symptoms reported, including recurrent sprains, pain, swelling, and avoidance of activities (Guillo et al., 2013). Provisional diagnosis is reliant on the patient's reported findings, with physical and radiological examinations used to confirm the diagnosis of CAI. During the assessment of patient reported findings, standard questions aiming to capture the presentation of CAI are used to establish a provisional diagnosis clinically. The key criteria to establish in the questioning include recurrence of acute ankle sprains, giving way of the ankle, avoidance or adaptation to daily or sporting activities, perception of an unstable ankle, and a perceived abnormal ankle indicated by factors such as pain or swelling (Guillo et al., 2013). There are also some selfreported scoring tools available to quantify the severity of ankle instability (including the Cumberland ankle instability tool [CAIT]), and to assess the functional impairments associated with CAI (including the foot and ankle ability measure [FAAM], and the foot and ankle outcome score [FAOS]), however they may not appropriate for everyday clinical use (Guillo et al., 2013).

During the physical examination, a comparative assessment of both ankles is undertaken, including assessing the alignment of the lower leg and foot while standing and during gait, and isolating the location of any tenderness. In addition, assessment of the range of ankle movements, ligament stress tests such as the talar tilt test and anterior drawer test, and stability measures such as single-leg stance

are usually important components of the physical examination (Gribble, 2019; Guillo et al., 2013). Standard plain radiographs, including anteroposterior, lateral and mortise views, as well as a comparative Saltzmann view to assess hindfoot alignment, may be ordered if indicated (Guillo et al., 2013). Comparative stress radiographs of the anterior drawer test and talar tilt test are also commonly performed. Advanced imaging procedures such as magnetic resonance imaging (MRI), ultrasonography and computed tomography (CT) may be helpful in some cases, though they are not routinely performed (Gribble, 2019; Guillo et al., 2013). Each of these assessment procedures are further discussed in the following sections.

2.3.1 Patient reported findings

Usually reports of pain, swelling, a feeling of giving way, or actual re-injury predominate the patient interview. These may occur during sporting activities, walking on uneven ground, or activities of daily living (ADL) (Peters et al., 1991). Questioning may reveal a history of recurrent acute ankle sprains, giving way of the ankle without a new sprain, perceptions of an insecure or unstable ankle, avoidance of or adaptation to daily or sporting activities, and the general perception of an abnormal ankle (Guillo et al., 2013). Several self-reported questionnaires may be used to quantify the functional impairments associated with CAI including the FAAM questionnaire, the FAOS questionnaire, and the CAIT questionnaire to assess the severity of the disorder and monitor progress following treatment.

2.3.1.1 Foot and ability measure (FAAM) questionnaire

The FAAM questionnaire is one of two general self-reported foot and ankle function questionnaires used to describe the level of disability recommended by the IAC in the selection of participants with CAI in controlled research (Gribble et al., 2013). This 29-item questionnaire was developed by Martin et al., and consists of two subscales: a 21-item ADL subscale and a 8-item sports subscale (Martin et al., 2005). Each item is scored on a 5 point Likert scale whereby 4 indicates 'no difficulty at all' and 0 indicates 'unable to do'. Total scores range from 0 to 84 for the ADL subscale and from 0 to 32 for the sports subscale, and these scores are converted to percentages. A higher percentage represents a higher level of function (Carcia et al., 2008), with 0% indicating an inability to perform the task and 100% indicating a level of function equivalent to before the injury. In addition, the perception of the participant about their ankle is rated as normal, nearly normal, abnormal, or severely abnormal on a categorical scale.

This instrument has been examined for reliability and validated in various languages including Brazilian, Dutch, English, French, German, Italian, Japanese, Persian, Spanish, Thai and Turkish (Arunakul et al., 2015; Borloz et al., 2011; Celik et al., 2016; Cervera-Garvi et al., 2017; Mazaheri et al., 2010; Moreira et al., 2016; Nauck & Lohrer, 2011; Sartorio et al., 2014; Uematsu et al., 2015; Weel et al., 2016). Responsiveness of this instrument has been evaluated for various ankle conditions including diabetes, achilles tendon diseases, and CAI (Carcia et al., 2008; Hoch et al., 2016; Kivlan et al., 2011; Reb et al., 2017). The questions were formed from a review of the literature and input from an expert panel. Initial item reduction was carried out with expert clinicians from the APTA. The participants involved in final item reduction included 1027 patients referred to physical therapy by a physician and who were receiving treatment for a leg, ankle, or foot musculoskeletal disorder (Martin et al., 2005). The FAAM was validated for CAI using 30 young athletes and it is suggested that this instrument may be used to detect self-reported functional deficits related to CAI (Carcia et al., 2008). The minimal detectable change (MDC) and minimal clinically important difference (MCID) calculated for this questionnaire are 5.7 and 8 points, and 12.3 and 9 points, respectively, for the ADL and sports subscales (Martin et al., 2005).

2.3.1.2 Foot and ankle outcome score (FAOS) questionnaire

The FAOS questionnaire is the other of the two IAC endorsed general self-reported foot and ankle function questionnaires used to describe the level of disability in the selection of participants with CAI in controlled research (Gribble et al., 2013). This is a 42-item questionnaire assessing five separate subscales: pain (nine items), other symptoms (seven items), ADL (17 items), sport and recreational activities (five items), and QOL (four items) (Roos et al., 2001). Each item is scored on a 5 point Likert scale ranging from 0 to 4. A normalised score is calculated with lower percentage scores representing extreme symptoms for each subscale (Roos et al., 2001).

This instrument has been tested for reliability and validated in various languages including Dutch, Iranian and Thai (Angthong; Negahban et al., 2010; Van Den Akker-Scheek et al., 2013). Responsiveness of this instrument has been tested for various ankle conditions including acquired flatfoot deformity, ankle osteoarthritis, and hallux rigidus (Hogan et al., 2016; Mani et al., 2013; Mani et al., 2015). Questions were adapted from the knee injury and osteoarthritis outcome score and validated using the responses of 213 participants who had undergone an anatomical reconstruction of the lateral ankle ligaments on average 12 years prior to study enrolment (Roos et al., 2001). The MDC and MCID have not been reported.

2.3.1.3 Cumberland ankle instability tool (CAIT) questionnaire

The CAIT has been designed to evaluate the severity of functional ankle instability (Hiller et al., 2006). The CAIT questionnaire is one of the three recommended questionnaires used to confirm the 'presence of self-reported ankle instability' which is one of the IAC inclusion criteria for participants with CAI in controlled research (Gribble et al., 2013). A cut-off score ≤ 24 identifies a 'self-reported unstable ankle' (Gribble et al., 2013). The maximum score is 30 for this 9-item questionnaire, and a low score represents more severe functional ankle instability. This instrument has been tested for reliability and validated in various languages including Dutch, Persian, Japanese, Korean, Spanish and Brazilian-Portuguese (Cruz-Diaz et al., 2013; De Noronha et al., 2008; Hadadi et al., 2017; Ko et al., 2015; Kunugi et al., 2017; Vuurberg et al., 2016). The questions included in the CAIT were identified from previous studies on ankle injury and from focus group interviews with people with CAI. The final version of CAIT was devised with 236 participants from a university campus, the general community and from among dance students at a performing arts high school (Hiller et al., 2006), and was validated with 177 participant responses. Participants with a score of 27 or lower were found to be likely to have functional ankle instability (Hiller et al., 2006). The CAIT has an MDC and MCID of \geq 3 points (Wright et al., 2017).

2.3.2 Physical examination and tests

Comparative assessment of both ankles is necessary during physical examination and testing. Gait, alignment of the lower leg and hind foot, precise location of tenderness, ankle ROM, calf muscle tightness, and strength and pain on resisted function of the peroneal and tibialis posterior tendons are all assessed during the physical examination (Guillo et al., 2013). In addition, the Beighton scale tests are used to identify general ligament laxity (Guillo et al., 2013). The anterior drawer and talar tilt stress tests for the lateral ligaments can be used to establish the presence of mechanical instability (Guillo et al., 2013). However, the descriptions of the amount of translation considered physiologically normal for these tests are conflicting. Published normal responses have ranged from 2mm to 9mm for the anterior drawer test (Gould et al., 1980; Johannsen, 1978; Karlsson et al., 1989; Lynch, 2002), and 5° to 23° for the talar tilt test (Cox, 1985; Lynch, 2002; Rubin & Witten, 1960). Comparison of the amount of pathologic anterior laxity with the opposite normal ankle is generally recommended for the anterior drawer test, with a between limb difference exceeding 3mm being considered a positive anterior drawer test (Jolman et al., 2017; Lynch, 2002). More than 10° difference from the opposite normal side is generally considered abnormal when interpreting the outcome of the talar tilt test (Lynch, 2002). Stability and proprioceptive control of the ankle can also be assessed using single-leg stance with eyes open and then eyes close (Guillo et al., 2013).

The benefits and limitations of the physical examination tests used for outcome measurements within this thesis, including related common methods of measurement, normal ranges of values, reliability and MDC values are discussed below, where they have been previously reported.

2.3.2.1 Weight-bearing DFROM

Dorsiflexion ROM deficits may have implications for a patient sustaining future lower extremity injuries, including plantar fasciopathy, ankle sprains, and patellofemoral pain syndrome (Hoch et al., 2015; Pope et al., 1998; Rabin & Kozol, 2012; Riddle et al., 2003). Relationships between weight-bearing DFROM, sagittal plane kinematics at the knee and hip, and vertical ground reaction forces during single-leg landing have been observed in individuals with CAI (Hoch et al., 2015). Further, relationships have been observed between ankle DFROM and the anterior reach direction of the star excursion balance test (SEBT) which indicate that ankle DFROM may influence dynamic balance (Basnett et al., 2013). Identification of dorsiflexion deficits in individuals with CAI may potentially assist clinicians to modify the landing biomechanics of these individuals, potentially reducing the likelihood of further injury (Hoch et al., 2015).

Several methods and tools are available to measure ankle DFROM in both non weight-bearing and weight-bearing positions. Weight-bearing measures of dorsiflexion have been suggested to be advantageous over non weight-bearing measures as they are easy to perform, provide the ability to obtain full ROM with the participant's own body weight, and because they are more functional (Rabin & Kozol, 2012). A weight-bearing lunge position using a standard goniometer, digital inclinometer, or a tape measure using the distance-to-wall technique are all common, reliable methods to obtain ankle DFROM with low measurement error (Konor et al., 2012). One limitation of this method is the

difficulty of performing this manoeuvre on weight-bearing restricted individuals, such as immobilised patients (Rabin & Kozol, 2012).

Weight-bearing measures including goniometric, inclinometric and tape measure methods may be used to reflect the available ROM during functional activities. Each involves an individual performing a weight-bearing lunge (Konor et al., 2012). The goniometric method is commonly used clinically, however requires a greater degree of technical proficiency (Konor et al., 2012). The inclinometric method is easy to use, however the cost of equipment is usually higher than for a standard goniometer (Konor et al., 2012). The tape measure method is inexpensive, can be used in a variety of settings and does not require the technical proficiency of goniometric or inclinometric methods (Konor et al., 2012). During the tape measure method, the patient is instructed to lunge towards the wall, touching their knee to the wall, whilst keeping their heel in contact with the floor. They are then asked to move their foot away from the wall in 1cm increments until the heel no longer maintains contact with the floor or the knee is no longer in contact with the wall. Maximal dorsiflexion is considered to be the greatest distance between the great toe and wall with the participant's knee maintaining contact with the wall (Gilbreath et al., 2014; Hoch, Staton, et al., 2012; Konor et al., 2012). The same procedure is followed for the opposite side. Each centimetre away from the wall in the lunge test represents approximately 3.6° of ankle dorsiflexion (Bennell et al., 1998).

Individuals with CAI have been shown to have a mean 1.7cm deficit in the weight-bearing lunge test (Hoch, Staton, et al., 2012) compared to healthy matched individuals (Drewes et al., 2009). Both the

inter-rater reliability (intra-class correlation coefficient (ICC)=0.65-0.99) and intra-rater reliability (ICC=0.80-0.99) of the weight-bearing lunge test have been reported as good (Powden et al., 2015). Average MDC scores for inter-clinician and intra-clinician weight-bearing lunge test are 4.6° in goniometric method (1.6cm in tape measure method) and 4.7° (1.9cm) respectively (Powden et al., 2015).

2.3.2.2 Static balance

Postural control impairments in CAI have not been detected consistently with the use of traditional instrumented measures (Mckeon & Hertel, 2008). No differences have been revealed in single-leg stance balance measures between individuals with non-injured ankles after a primary ankle sprain and individuals with CAI (De Vries et al., 2010; Isakov & Mizrahi, 1997). Further, no differences have been demonstrated between individuals with and without functional ankle instability (Ross & Guskiewicz, 2004). However, stabilometry scores were reported as reduced among soccer players with functional instability, but were not apparently affected in those with mechanical instability (Tropp et al., 1985). Similarly, no differences in stabilometry measures were demonstrated between mechanically stable and mechanically unstable ankles in a latter study (Leanderson et al., 1993).

A number of non-instrumented and instrumented single-leg stance balance measures for ankle instability have been reported, with instrumented force plate measures more recently becoming the gold standard of assessment (Mckeon & Hertel, 2008). The balance error scoring system, timed balance test, and foot lift test are some available clinical tests used in measuring static balance. The timed balance test was found to not be ideal in detecting a significant decline in balance until the age range of 60 to 70 years and also for athletes with mild concussion, suggesting a lack of sensitivity in detecting mild changes in balance (Browne & O'hare, 2001; Clark, 2007). Potentiometric displacement transducers, mechanical ataxia meters, sway magnetometry, multi-sensor polymer insoles and threedimensional video analysis are some tools that have been used to record sway patterns of individuals (Browne & O'hare, 2001). Precise measurements of static balance with centre of pressure (CoP) data using force plates is now commonly the method used in research to quantify balance deficits associated with CAI (Linens et al., 2014).

When measuring static balance using CoP data obtained with the participant standing on the centre of a force plate, a standardised single-leg stance position is maintained. The individual is instructed to flex the non-tested leg slightly at the hip, whilst flexing the knee to 90°. The arms are crossed at the chest with each hand resting on the opposite shoulder. Measurements are recorded in both 'eyes open' and 'eyes close' positions. For 'eyes open', the individual is asked to maintain a fixed gaze on a cross marked on the wall five metres in front of them and remain as still as possible for 10 seconds (Karlsson & Frykberg, 2000).

The most commonly assessed CoP variable is sway velocity calculated as the absolute mean value of the instantaneous velocity of the CoP in a given direction during a given period (Ross et al., 2009), which indicates how quickly a person shifts and is able to control their CoP (Childs, 2016). A cut-off score of \geq 1.56cm/s for CoP velocity measures in the single-leg stance position has been used to

distinguish individuals with and without CAI (Linens et al., 2014). Measurement of sway area in weight-bearing has been demonstrated to be reliable with an ICC \geq 0.70 (Golriz et al., 2012). MDC values are estimated at 12mm/sec, 39mm, 4303mm² for CoP mean velocity, average location of CoP, and sway area respectively, for one repetition on the force plate (Golriz et al., 2012).

2.3.2.3 Dynamic balance

Some studies which have used force plate data alone have identified dynamic balance deficits in individuals with CAI (Brown & Mynark, 2007; Groters et al., 2013; Ross & Guskiewicz, 2004; Wikstrom et al., 2007), while other studies have reported no differences (Bernier et al., 1997; Shiravi et al., 2017). These conflicting findings may be due to the various activities researchers have chosen to use to assess dynamic balance on force plates (i.e. multiple hop test, single-leg hop, weight shifting tasks, and perturbations through the use of a tilting platform). In addition, the two studies with nonsignificant findings had comparatively small sample sizes raising the possibility of insufficient statistical power leading to a null result (Bernier et al., 1997; Shiravi et al., 2017). Poor dynamic balance has been observed in individuals with CAI compared to both lateral ankle sprain copers (described as perceiving a successful recovery from initial lateral ankle sprain with no or minimal symptoms within a year post-injury) (Hertel & Corbett, 2019; Wikstrom & Brown, 2014) and healthy controls (Doherty et al., 2016). When considering subgroups of CAI, individuals with mechanical instability have demonstrated increased postural sway compared to those with functionally unstable (and healthy) ankles. Postural control in individuals with functional instability has not been shown to be different to those with healthy ankles (Chen et al., 2014). When SEBT data were combined with force plate measures, method that which excludes any influence by the researcher the velocity of the

CoP was found to be lower in individuals with CAI compared to those without CAI in the mediolateral component (Pionnier et al., 2016), indicating balance deficits are associated with CAI.

Dynamic balance tests have been suggested to provide a better means for identifying CAI over static balance tests as they assess balance during a functional task performance (Linens et al., 2014). There are several instrumented and non-instrumented methods available to measure dynamic balance, including the side hop test, SEBT test, and figure-of-8 hop test (Arnold et al., 2009; Linens et al., 2014). Time to stabilisation (TTS) in a single-leg jumping task and the SEBT are commonly used functional measures to assess dynamic balance in CAI (Arnold et al., 2009; Gribble et al., 2012). TTS provides better measures with larger effect sizes compared to SEBT alone, however it requires a force plate and analytical software (Arnold et al., 2009). Use of CoP measures during the SEBT to assess dynamic postural control has been introduced using movement analysis tools (optoelectronic cameras and force platforms) in some recent studies (Doherty et al., 2015; Pionnier et al., 2016).

Combining SEBT (in anterior, postero-lateral and postero-medial directions) with CoP data from force plate measures could assess balance more objectively than clinical tests alone, and could precisely quantify the balance measures (Bansbach, 2017). To do this, the individual is asked to establish a stable base of support on the stance limb in the middle of the testing grid on a force plate. While standing on a single limb, the individual is instructed to reach as far as possible with the reaching limb along each line (anterior, postero-medial and postero-lateral directions), lightly touching the line with the most distal portion of the reaching foot without shifting weight or coming to rest on the foot of the

reaching limb. The reaching limb is then returned to the starting position on the centre of the grid. If the individual lifts or shifts any part of the foot of the stance limb during the trial, the trial is not considered as complete (Gribble et al., 2012). After performing a maximum of four non-recorded trials for familiarisation, the next trial for each direction is recorded (Pionnier et al., 2016; Robinson & Gribble, 2008). Normalised SEBT values are obtained by dividing the excursion distance by the participant's leg length (the distance between the anterior superior iliac spine and the ipsilateral medial malleolus), and then multiplying by 100 (Gribble & Hertel, 2003; Pionnier et al., 2016). Data for CoP velocity, area and amplitude to quantify spatio-temporal parameters (medio-lateral and anteroposterior) are acquired at 100Hz, under the foot during single-leg stance (Pionnier et al., 2016).

The intra-rater and inter-rater reliability of the SEBT are both reported as good to excellent for both legs (ICC values ranging from 0.87 to 0.90), and the MDC values are calculated as 7.2% for the right leg and 6.2% for the left (Van Lieshout et al., 2016).

2.3.2.4 Pain intensity and pain distribution

One common residual symptom following a lateral ankle injury is reported to be pain. In a recent study of individuals with CAI, 12.4%, 47.7% and 39.9% reported constant pain, pain during physical activities, and no pain respectively (Adal et al., 2019). It is therefore important to understand the pain characteristics and distribution of pain in patients with CAI.

Various pain intensity measures are available including the visual analogue scale (VAS), point numerical rating scale, box scale, behavioural rating scale, verbal rating scale, McGill pain questionnaire and chronic pain grade scale (Hawker et al., 2011; Jensen et al., 1986). The pain scale used does not influence the pain intensity reported (Kremer et al., 1981). However, the numerical pain scale appears to be the most practical index both clinically and in research (Jensen et al., 1986). Further, in a recent study on osteoarthritic knee pain, the VAS was found to be the most reliable measure of pain (Alghadir et al., 2018). The VAS is easy to administer, complete and score (Hawker et al., 2011), and has been widely used in adult populations in research, including those with rheumatic disease. However, difficulty in understanding the concept and requirements of the VAS may impact studies involving older pain populations. Significant correlations have been found between age and incorrect responses to the VAS (Jensen et al., 1986; Kremer et al., 1981). Nonetheless, the VAS is considered to be a valuable instrument to assess pain intensity and changes in pain when respondents are given sound instructions (Haefeli & Elfering, 2006).

The VAS consists of a 100mm horizontal line, with 'no pain' anchoring the left of the line and 'worst possible pain' anchoring the right. The patient is asked to place a line perpendicular to the VAS line at the point that represents the pain intensity at rest. A higher score indicates greater pain intensity.

The recommended categorisation for VAS scores are no pain (0-4mm), mild pain (5-44mm), moderate pain (45-74mm), and severe pain (75-100mm) (Jensen et al., 2003). The validity of the VAS for detecting changes in pain intensity has been supported by several studies (Ferreira-Valente et al., 2011;

Price et al., 1983). The test-retest reliability of the VAS was reported as excellent (0.97), and the MDC as 0.08 (Alghadir et al., 2018).

2.3.2.5 Pressure pain threshold (PPT)

Chronic pain in CAI may result from increased nociceptor sensitivity known as peripheral sensitisation which is restricted to the site of tissue injury (Giesbrecht & Batti , 2005), or as a result of central sensitisation due to an increase in the excitability of neurons within the central nervous system, contributing to diffuse hypersensitivity in regions beyond the damaged tissue (Giesbrecht & Batti , 2005; Starkweather et al., 2016; Woolf, 1983; Wright, 1999). Distinguishing between peripheral (nociceptive) pain mechanisms and central nervous system (nociplastic) pain processing alterations/maladaption in musculoskeletal pain is important as central sensitisation is considered a potential influence in the development and maintenance of chronic/persistent pain (Coronado et al., 2014; Woolf, 1983; Wright, 1999). More personalised methods for monitoring the trajectory of chronic pain and responses to treatment may also improve outcomes for patients with CAI. Such individually specific clinical data may result in better long-term prediction of health outcomes by the potential provision of important diagnostic and prognostic information. Further, these data might also assist in assessing the severity of the condition and its response to a particular intervention (Bedson et al., 2019).

Clinically, central sensitisation results in increased pain sensitivity, spreading of hyperalgesia to surrounding areas, and spontaneous pain (Arendt-Nielsen et al., 2010). Widespread pressure

hypersensitivity is considered an indicator of central sensitisation. Thus, increased pressure pain sensitivity is indicated by decreased pain threshold measures in algometry (Ramiro-Gonzalez et al., 2012). Evidence of central sensitisation is usually tested by assessing PPT (Coronado et al., 2014), with methods including cuff algometry, pressure algometry, and algometry with electric stimulation (Mutlu & Ozdincler, 2015). PPT measurements for ankle conditions are obtained in each leg from two local points (to assess local hypersensitivity) and one remote body area (to assess central sensitisation) (Ramiro-Gonzalez et al., 2012). The recommended points are anterior to the lateral malleolus over the ATFL, inferior to the medial malleolus over the deltoid ligament, and over the proximal third of the tibialis anterior muscle belly. The algometer probe (contact surface of 1cm²) is placed perpendicular to the skin and pressure is applied (40kPa/s). The individual is asked to indicate when the feeling of the stimulus changes from 'pressure only' to 'discomfort' by pressing an indicator switch (Arendt-Nielsen et al., 2010; Rebbeck et al., 2015). This process is performed three consecutive times and a 10 second rest period is allowed between each set of measurements.

Pressure algometry is considered a stable and reliable measure of PPT (Frank et al., 2013). The interrater reliability of pressure algometry has been reported to be high when the algometer pressure is applied at a consistent rate (ICC 0.91, 95% (confidential interval) CI 0.82–0.97) (Chesterton et al., 2007).

2.3.3 Radiological examination

Although the patient reported findings, physical examination and specific clinical tests may provide a diagnosis in certain cases of ankle injury, complementary radiological tests are important in defining joint laxity and to assist in confirming some diagnoses. Stress radiographs, MRI and ultrasonography are often used to assess or to confirm ligamentous or tendinous pathology (Guillo et al., 2013; Peters et al., 1991; Tourné et al., 2010). Comparative stress radiographic images with the anterior drawer test and talar tilt test can be performed to confirm mechanical instability (Guillo et al., 2013; Peters et al., 1991). Commonly hind foot alignment is assessed using standard pain radiographs which may include standing antero-posterior, lateral and mortise views, and a comparative Saltzmann view (or Méary view) (Guillo et al., 2013).

Plain films are used in initial screening in diffuse ankle pain and for detection of gross bony lesions, while ultrasonography is utilised as a primary tool of investigation in imaging focal soft tissue abnormalities. MRI is found to be effective for those cases with an uncertain diagnosis, as it can exclude most clinically relevant pathologies (El-Liethy & Kamal, 2016). Magnetic resonance arthrogram and CT are also recommended in cases of high clinical suspicion meniscal or ligamentous injury. However, in order to localise a particular structure in chronic foot or ankle pain presentations, magnetic resonance imaging is generally recommended because of its ability to detect both soft tissue and bone lesions. Further, ultrasonography is particularly useful in dynamic evaluation (Aagesen & Melek, 2013).

2.3.3.1 Stress radiography tests

Several clinical tests can be used to assess the integrity of the key ligaments. The anterior drawer test is utilised to assess the integrity of the ATFL by assessing the amount of anterior translation of the talus under the tibia, and the talar tilt test stresses the CFL using passive adduction of the calcaneus to detect any excessive movement (Lynch, 2002). Stress radiography techniques are more often used for research purposes or to diagnose ankle joint mechanical instability and this involves specialised instruments and standardised loads (Lynch, 2002; Wheeless, 2016). Radiological evaluation of the amount of anterior translation of the talus, or talar tilt angle during a ligament stress test, is a widely accepted method to quantify any mechanical laxity in CAI (Hoffman et al., 2011; Hubbard & Hicks-Little, 2008).

Known limitations of stress radiography are that the testing may be influenced by pain, as well as the risk of exposure of the subject or provider to ionising radiation. The use of a standardised stress force has been shown to reduce measurement error (Sisson et al., 2011). Stress radiographs have been found to have moderate sensitivity, high specificity and a high positive predictive value for the evaluation of lateral ankle instability (Jolman et al., 2017). An absolute anterior displacement of 10mm or a between-limb difference of 3mm in anterior translation of the talus is considered clinically significant (Jolman et al., 2017; Lynch, 2002). A talar tilt angle of more than 10° of difference between limbs is generally considered abnormal (Lynch, 2002).

2.3.3.2 Position of the fibula in lateral ankle sprains including CAI

The 'positional fault' theory was introduced by Brian Mulligan to explain persistent pain after joint injury. According to this hypothesis, injuries or sprains might result in a minor positional fault (bony incongruence) of a joint causing pain and restriction in physiological movement (Mulligan, 1995). Mulligan specifically hypothesised that following an ankle inversion injury, the distal fibula may in some cases be malpositioned anteriorly due to forces transmitted through the attachment of the ATFL, leading to chronic ankle pain and impairment (Mulligan, 1995). Whether a pre-existing fibular 'positional fault' predisposes the patient to the injury, or whether the injury results in the fibular 'positional fault' is yet to be thoroughly examined (Hubbard et al., 2006).

The presence of fibula displacement has been investigated radiologically (Berkowitz & Kim, 2004; Eren et al., 2003; Hubbard & Hertel, 2008; Hubbard et al., 2006; Kobayashi et al., 2014; Li et al., 2017; Mavi et al., 2002; Merlin et al., 2005; Scranton et al., 2000; Wikstrom, Tillman, et al., 2010) and non-radiologically in ankle ligament injuries, including acute, sub-acute and chronic sprains, including CAI (Table 2.1). The findings of these studies have not been consistent, with evidence reported of an anteriorly positioned fibula, posteriorly positioned fibula, laterally positioned fibula, and no positional abnormality at all.

Radiographic measurement of fibular position is the most common method employed in studies to date, using lateral X-ray images with the measurement being the distance between the most anterior margin of the tibia and the most anterior margin of the fibula (Hubbard & Hertel, 2008; Hubbard et al.,

2006; Wikstrom, Tillman, et al., 2010). Measurements are made perpendicular to a line drawn vertically from the most anterior part of the tibia (Hubbard & Hertel, 2008; Hubbard et al., 2006). Due to individual anatomical variations, a normalisation technique has been proposed which reports fibular position as a percentage of tibial width (Wikstrom, Tillman, et al., 2010). The test-retest reliability ICC_{3,1} has been calculated as 0.98 with a SEM of 0.64mm for this measurement, and for intra-tester reliability the ICC_{3,1} is 0.92 and SEM is 0.72mm (Hubbard et al., 2006). However, these reliability values are reported for non weight-bearing positions. Considering that weight-bearing forces may cause widening of the distal tibiofibular joint, it is plausible that fibular position measures may detect greater differences in weight-bearing radiographs used to measure fibular position, which are arguably more functional. However, there is a potential for rotation of the tibia and fibula in a weight-bearing radiograph which might introduce minor variability in fibular positional measurements, if the individual is not meticulously positioned.

Study	Participants	Measurement method	Fibular positional abnormality	Mean (SD) value of displacement	
				symptomatic	control
Berkowitz & Kim, 2004	CAI (patients with instability, n=65; uninjured controls, n=65)	radiological (CT/MRI scans)	posterior	17 (6)°	9 (4)°
Eren et al., 2003	acute ankle sprains (patients with ankle sprain, n=61; uninjured controls, n=101)	radiological (CT scans)	posterior	11.5 (7)°	5.85 (4.9)°
Fukuhara et al., 2012	sub-acute lateral ankle sprains (patients with unilateral sub- acute lateral ankle sprain, n=10; uninjured controls, n=10)	non-radiological (instrument consists of a pedestal and 2 devices incorporating a caliper)	anterior	43.7 (8.3)mm	37.4 (4.2)mm

Table 2.1 Reported abnormalities of fibula position in lateral ankle ligament injuries

Hubbard et al., 2006	CAI (patients with unilateral CAI, n=30; uninjured controls, n=30)	radiological (fluoroscopic	anterior	14.3 (3.1)mm	16.1 (4.6)mm
		lateral images)			
Hubbard & Hertel, 2008	sub-acute lateral ankle sprains (patients with sub-acute	radiological (fluoroscopic lateral images)	anterior	14.2 (3.4)mm	16.8 (2.3)mm
	uninjured controls, n=11)				
Kavanagh,	acute or chronic ankle	non-radiological	anterior	7.5mm (greatest	10mm (greatest
1999	sprains (patients with ankle sprains, n=8 [chronic, n=2; acute, n=6]; uninjured controls, n=17)	(instrument consisting of two potentiometers, a wooden bar with a strain gauge attached to an amplifier and circuit box)		amount)	amount)

Kobayashi et	CAI	radiological (3-D	lateral	at lateral	not reported
al., 2014	(patients with unilateral CAI,	analysis of CT-		malleolus 0.60	
		based bone		(1.09)mm; at	
	n=17, control-contralateral	models)		5cm 0.57	
	side, II-17)			(0.81)mm; at	
				10cm 0.68	
				(1.15)mm	
Listal 2017	CAL	radialagiaal	inconclusivo	AMI 11.06	A MI 7 80 (4 41) ^o
LI Ct al, 2017	CAI		(A ML significant	AMI 11.00	Alvii 7.89 (4.41)
	(patients with mechanical	(MIKI)	(AMI significant	(3.02)	IMI
	ankle instability, n=54;		and IVII non-	IMI 9.97 (4.48)°	9.23 (4.09)°
	patients for reasons unrelated		significant)		
	to ankle instability, n=51)				
Mavi et al	recurrent sprained ankles	radiological	anterior	male	male
2002	(patients with recurrent	(MRI)		11.8 (1.4)mm:	14 3(3 5)mm:
	sprained ankles n=18:	()		female 11.2	female 12.5
	uninjured controls $n=75$)			(1.3)mm	(2.9)mm
	uninjured controls, ii–75)			(1.5)	(2.7)
Merlin et al.,	lateral ankle sprains (patients	radiological	superior	6.19 (0.28)cm	not reported
2005	with sprained ankles, n=8;	(MRI)			
	uninjured controls, n=30)				
Scranton &	CAI	radiological (CT)	posterior	not reported	not reported
Rogers 2000		radiological (C1)	Posterior	not reported	not reported
Kugers, 2000	(patients undergoing a				

	Brostrom procedure for				
	ankle instability, n=23;				
	uninjured controls, n=100)				
Wikstrom et	CAI	radiological (X-	no abnormality	non-normalised	non-normalised
al., 2010	(patients with CAI, n=24; uninjured controls, n=24; copers, n=24)	ray)		12.8 (4.2)mm; normalised 29.1 (8.2)%	12.2 (3.3)mm; normalised 29.2 (7.1)%

AMI, axial malleolar index; CAI, chronic ankle instability; CT, computed tomography; IMI, intermalleolar index; MRI,

magnetic resonance imaging; SD, standard deviation

2.3.4 Other tools to assess CAI

Due to popularity and convenient application, smart phone-based systems have also been developed to assess postural control ability in individuals with CAI. Acceleration data is used to represent postural control performance during eyes open and close conditions (Chiu et al., 2017). However, the role of smartphone applications in the actual diagnosis of CAI is yet to be ascertained.

In summary, the clinical assessment of CAI should include a combination of factors; (i) patient history including injury mechanism and patient reported outcome measures, (ii) physical examination and tests including a comprehensive assessment of clinical characteristics such as deficits in DFROM and balance, and (iii) conventional radiography examination techniques, as appropriate.

2.4 Management of CAI

A large number of treatment strategies for ankle sprains and CAI are detailed in the literature. An overview of systematic reviews with meta-analysis indicated strong evidence for exercise therapy and bracing in preventing ankle sprain recurrence, although the optimal treatment strategy remains unclear (Doherty et al., 2017). However, studies of manual therapy interventions have been limited by short timeframes for follow-up (Doherty et al., 2017). Neuromuscular training also appears to improve ankle function in patients with CAI (De Vries et al., 2011), along with weight loss, activity appropriate footwear and external restraints recommended in athletic patient populations (Mccriskin et al., 2015). Patients with functional instability are more likely to benefit from non-surgical measures (Mccriskin et al., 2015). Generally, patients with CAI who fail functional rehabilitation progress to surgical

management (Chan et al., 2011). With early mobilisation of the ankle joint (rather than six weeks of immobilisation), functional rehabilitation is recommended in the post-surgical management of CAI to ensure early return to work and sports (De Vries et al., 2011).

2.4.1 Conservative interventions

Acute management of recurrent sprains in CAI has long been considered to involve rest, ice, compression and elevation (RICE) as the standard approach (Ajis & Maffulli, 2006; Herb & Hertel, 2014). Joint protection, optimal loading, ice, compression and elevation (POLICE) was introduced later for acute ankle sprain management (Bleakley et al., 2012), however there is limited evidence to support this regime (Herb & Hertel, 2014). Generally, the clinical rehabilitation of CAI aims to restore the identified deficits such as in ROM, balance and function with differing clinical interventions (Ajis & Maffulli, 2006). A plethora of therapeutic interventions for CAI have been proposed including joint mobilisation (e.g. MWM, manipulation), balance retraining (e.g. single-leg standing balance exercises, dynamic balance tasks), peroneal muscle strengthening (e.g. resistance exercises), proprioceptive retraining (e.g. wobbleboard exercises), soft-tissue mobilisation, passive calf stretching and orthotics (Ajis & Maffulli, 2006; Kosik et al., 2016; Mccriskin et al., 2015). These are generally applied in combination depending on the deficits with which the individual patient presents. For example, joint mobilisation has been shown to improve DFROM (Hoch, Andreatta, et al., 2012), while strength training and proprioceptive retraining have been shown to improve muscle weakness and function in CAI (Kaminski et al., 2003). Various electrophysical modalities (e.g. ultrasound, contrast therapy), oral anti-inflammatory medication, injectable steroid medication, elastic bandaging and strapping are also often clinically used in combination with rehabilitation programs (Ajis & Maffulli, 2006).

A variety of joint mobilisation techniques, including manipulation are commonly practised to manage ankle sprains. Joint mobilisation is suggested as being effective at restoring objective impairments such as reduced ankle ROM or decreased dynamic postural control, and the combination of mobilisation with other conservative interventions is endorsed (Kosik et al., 2016). In a large scale study of patients with lateral ankle sprains in the United States who received physiotherapy, 52% received manual therapy with joint mobilisation/manipulation being the most commonly applied manual therapy technique (51.7%) (Feger et al., 2017). However, there is limited supportive evidence for the use of joint mobilisation in isolation for improving outcomes of patients with CAI, including self-reported function (Kosik et al., 2016).

2.4.1.1 Mobilisation with movement (MWM)

Several trials and systematic reviews have investigated the effects of manual therapy on CAI. No studies have actually assessed joint mobilisation as the sole intervention nor undertaken a metaanalysis, except for one very recent systematic review (Vallandingham et al., 2019). However, this study assessed MWM for only two outcomes of CAI; DFROM and dynamic balance (Vallandingham et al., 2019). Further, it is not clear which joint mobilisation technique is more likely to produce a clinical benefit for people with CAI. MWM is a mobilisation technique recommended in the rehabilitation of CAI (Vicenzino et al., 2011) which concurrently uses a pain-free sustained passive accessory joint mobilisation with an active or passive physiological movement during application (Hoch & Mckeon, 2010). MWM was introduced by New Zealand physiotherapist Brian Mulligan (Mulligan, 1993), and has been applied to improve pain, swelling, function, feelings of instability, postural control, and ankle DFROM following lateral ankle sprains (Collins et al., 2004; Cruz-Diaz et al., 2015; Gilbreath et al., 2014; Gómez et al., 2015; Reid et al., 2007; Vicenzino et al., 2006). Similar symptoms are widely existent individually or in combination in individuals with CAI. According to Mulligan, rather than tearing or rupturing the ATFL, in some cases an inversion injury exerts an anterior and slightly caudad force on the fibula via the ATFL, thus creating a 'positional fault' in that direction. Therefore, the fibula may be mobilised using a MWM technique incorporating a sustained postero-cephalad glide, while asking the patient to simultaneously repeat the previously symptomatic movement, such as active inversion or dorisflexion (Hing et al., 2015; Mulligan, 1995).

For patients with CAI, the inferior tibio-fibular joint may be mobilised using a 'fibular MWM for dorsiflexion and/or inversion in non weight-bearing' according to Mulligan's paradigm. The technique is initially applied in supine lying, with the tibia resting on the treatment table and the ankle and foot unsupported off the edge of the table. The therapist applies a sustained anteroposterior glide to the fibula, with some inclination slightly cephalad and laterally, while stabilising the tibia. While the therapist maintains this glide in a pain-free manner, the patient is instructed to perform active ankle inversion or dorsiflexion to the first onset of pain or end of range. At this point, if there is no pain, the therapist can apply overpressure to the active movement for a few seconds (Hing et al., 2015; Vicenzino et al., 2006). Treatment is recommended to be applied with 6-10 repetitions in a set, with 3-5 sets in a treatment session depending on the pain response of the patient (Hing et al., 2015). The patient may be reassessed using the active physiological movement or functional task with which they

are most painfully restricted, known as the client specific impairment measure. If there is a clinical indication, MWM can be progressed to a weight-bearing position in some cases.

Mulligan ankle taping, or fibular repositioning taping, is an intervention which may be used to enhance the effect of the MWM technique described in the previous section by using tape to maintain the glide component of the MWM (Someeh et al., 2015b). In addition, the potential of fibular repositioning taping to enhance proprioception is suggested to improve stability and confidence, and also decrease feelings of instability in individuals with CAI (Someeh et al., 2015b). Non-stretchable tape is applied to the ankle starting 2cm anterior to the fibula and 1cm proximal to the tip of the lateral malleolus. The tape is spiralled obliquely around the lower leg while applying the pain-free fibula glide and the taping is finished on the anterior aspect of the skin (Hing et al., 2015).

2.4.2. Surgical interventions

If CAI is unresponsive to a three month rehabilitation program, then surgical treatment may be indicated (Giannini et al., 2014). Numerous anatomic and non-anatomic procedures have been described in the surgical management of CAI in the literature. Anatomic techniques generally consist of direct repair of injured ligaments, while non-anatomic procedures consist of tenodesis in order to substitute for the injured ligaments and to address pathological ankle mobility (Chan et al., 2011; Giannini et al., 2014). However, these techniques often result in joint stiffness and limited ROM of the ankle and subtalar joints. Non-anatomic techniques are associated with an increased rate of complications compared to anatomic reconstruction (Giannini et al., 2014).
Management of CAI may vary depending on the impairment, or combination of impairments, that the individual with CAI clinically presents with, including pathomechanical impairments, sensory-perceptual impairments, and/or motor-behavioural impairments. Personal factors such as age, BMI and environmental factors such as social support, access to health service may affect the severity of these impairments. Therefore, the clinical outcome of management could end up being a coper who perceives a full recovery or with CAI who perceives recurrent ankle sprains (Hertel & Corbett, 2019. The management of CAI is generally based both on the clinical judgement of the practitioner from their assessment of the condition, and also the application of evidence-based practice recommendations, as appropriate. The options for management range from conservative (e.g., manual therapies, electrotherapeutic modalities, thermal therapies, orthotics, exercise therapies [including strengthening, stretching, proprioceptive training, range of motion exercises, balance exercises, neuro-muscular training, gait training and functional exercises]), to pharmaceutical interventions and surgery.

2.5 Summary

Ankle sprains are a common musculoskeletal injury and often develop into CAI. Along with its high prevalence, recurring episodes of injury in CAI contribute to its high healthcare cost. The mechanism for development of CAI, whether mechanical or functional remains unclear in the literature. Identification of the anatomical and clinical characteristics specific to CAI, may assist in better defining this debilitating condition and its subgroups. The pain mechanism(s) involved in CAI should also be specifically investigated, and whether there are underlying central changes related to the chronic pain in CAI. While the literature supports that joint mobilisation helps to improve some clinically relevant outcomes in CAI, the underpinning mechanism by which joint mobilisation may work is not clearly understood, nor are the long-term effects known. In particular, the clinical benefits of MWM in the management of CAI and its subgroups remain unclear, including if or how MWM may alter fibular position.

Chapter 3 Clinical benefits of joint mobilisation on ankle sprains: a systematic review and meta–analysis

This chapter has been published as:

Weerasekara, I., Osmotherly, P., Snodgrass, S., Marquez, J., de Zoete, R., & Rivett, D. A. (2018). Clinical benefits of joint mobilisation on ankle sprains: a systematic review and metaanalysis. *Archives of Physical Medicine & Rehabilitation, 99*(7), 1395-1412.e1395. doi:https://doi.org/10.1016/j.apmr.2017.07.019

The work presented in this manuscript was completed in collaboration with the co-authors (Appendix 1). The protocol for this systematic review was registered with the International Prospective Register of Systematic Reviews on January 12, 2016 (CRD42016030194) (Appendices 7 and 16).

Overview

The epidemiologic findings and socio-economic burden associated with lateral ankle sprains and CAI suggest that current rehabilitation approaches may be inadequate. The aim of the study presented in this chapter is to synthesise the best available evidence for joint mobilisation, as a common rehabilitation approach in managing patients with acute to chronic ankle sprains including CAI. Previous systematic reviews which have evaluated the evidence on the effectiveness of manual therapies for ankle sprains or CAI have included multi-modal studies that involved co-interventions such as RICE and home exercise programs, as an adjunct to joint mobilisation. At the time of the

study, no systematic reviews have actually assessed joint mobilisation as the sole intervention for ankle sprains or CAI, or indeed undertaken a meta-analysis, despite it being a common intervention in the rehabilitation of a number of ankle conditions. Therefore, the systematic review presented in this chapter aimed to synthesise and meta-analyse the available evidence for ankle joint mobilisation in grade I or II ankle sprains of the medial or lateral ligaments in the acute to chronic stages of rehabilitation in any ambulant setting.

This chapter describes a systematic investigation of the current published evidence for various joint mobilisation techniques on ankle sprains including CAI, as found in 11 databases from inception to June 2017. At the time of the study, this is the first systematic review and meta-analysis to only include studies in which joint mobilisation is the sole intervention, thus providing a clearer understanding of the effects of mobilisation in isolation. The current review did not identify any studies evaluating the clinical benefits of joint mobilisation on acute ankle sprains. The quantitative analysis was therefore conducted using data from studies involving chronic ankle sprains including CAI, and thus the findings of the meta-analysis may be applicable to the joint mobilisation treatment of chronic ankle sprains including CAI. This chapter also assesses the immediate, short-term and long-term clinical benefits of joint mobilisation on ankle sprains and persistent ankle instability, as described in the reviewed body of literature.

3.1 Introduction

Ankle sprains are a common injury in sports and the general community, and may lead to chronic pain, functional limitations and physical disability (Parker & Jelsma, 2010; Woolf, 2003). Epidemiological studies conducted in various countries highlight the high incidence of ankle sprains during sports training and competition with rates reported as 7 per 1000 in Denmark, 6.09 per 1000 in United Kingdom, and 2.15 per 1000 in the United States in person years (Bridgman et al., 2003; Holmer et al., 1994; Waterman, Owens, et al., 2010). Plantarflexion inversion sprain or lateral ankle sprain, is the most common type of ankle sprain (Doherty et al., 2014). It typically results in either an injury of the inferior tibiofibular ligament, ATFL or the bifurcate ligament (Hing et al., 2015). Eversion injuries often result in damage to the deltoid and spring ligaments of the medial aspect of the ankle (Hing et al., 2015).

According to the clinical practice guidelines linked to the ICF from the Orthopaedic Section of The APTA, manual therapy is recommended for both the acute and progressive loading phases of rehabilitation (Martin et al., 2013). Management of ankle sprains commonly involves mobilisation procedures applied to the joint, such as non-thrust joint mobilisation, high velocity thrust manipulation, and MWM.

The mechanisms by which these techniques are purported to work are biomechanical (such as stretching/tearing tissue, inducing cavitation within the joint, reducing muscle hypertonicity/stiffness)

and neurophysiological, potentially including spinal cord and supra-spinally mediated mechanisms (Mccarthy et al., 2015; Vicenzino et al., 2011).

Several studies have investigated the effects of manual therapy on ankle sprains using a variety of outcome measures including pain, ROM and function from the acute to chronic stages of recovery, with different results reported (Collins et al., 2004; Cosby et al., 2011; Cruz-Diaz et al., 2015; Delahunt et al., 2013; Gilbreath et al., 2014; Green et al., 2001; Marron-Gomez et al., 2015; O'brien & Vicenzino, 1998 ; Penso, 2008; Vicenzino et al., 2006; Woodman et al., 2013). Several systematic reviews have attempted to collate this evidence but have been limited by their narrow focus on lateral ankle sprains and restricted outcome measures (Bleakley et al., 2008; Hoch & Mckeon, 2010; Loudon et al., 2014; Terada et al., 2013; Van Der Wees et al., 2006). Previous systematic reviews have all included some studies which involved other interventions such as RICE and home exercise programs, as an adjunct to mobilisation. Therefore, they have not actually assessed mobilisation as the sole intervention. Moreover, the clinical benefits of joint mobilisation have not yet been evaluated through meta-analysis, despite it being a common intervention used in the rehabilitation of a number of ankle conditions and despite the growing body of empirical literature.

The present systematic review aims to address these limitations by synthesising and meta-analysing the available evidence for ankle joint mobilisation (including high velocity thrust manipulation) in grade I or II ankle sprains of the medial or lateral ligaments in the acute/sub-acute/chronic stages of rehabilitation in any ambulant setting.

3.2 Method

3.2.1 Registration

The protocol for this systematic review was registered with the International Prospective Register of Systematic Reviews (PROSPERO) on January 12, 2016 (CRD42016030194).

3.2.2 Search strategy

A search of electronic databases, including MEDLINE, MEDLINE In Process, Embase, AMED, PsycINFO, CINAHL, Cochrane library, PEDro, Scopus, SPORTDiscus, and Dissertations and Thesis was conducted from inception to June, 2017. In addition to the database search, a hand search of the reference lists of identified studies was also carried out. A search strategy (Table 3.1) was developed for the main search strings of ankle sprain and mobilisation. Keywords used for 'ankle sprain' included sprain, talocrural joint, ligament injuries, lateral ligament, medial ligament, deltoid ligament, collateral ligament, ATFL, PTFL, sprain and strain, and ankle twist. Key words used for 'mobilisation' included manual therapy, joint mobilisation, manipulation, MWM, Maitland, Mulligan, and rehabilitation. These terms were used alone and in combinations during the search.

Table 3.1 Search strategy

#	Searches
1	Ankle Injuries/
2	ankle sprain.mp.
3	(ankle* adj5 injur*).tw.
4	(ankle* adj5 sprain*).tw.
5	(ankle* adj5 twist*).tw.
6	(injur* adj5 ligament*).tw.
7	lateral ligament*.mp. or Collateral Ligaments/
8	Ankle Joint/ or medial ligament*.mp.
9	Ankle Joint/ or deltoid ligament*.mp.
10	ATFL.mp.
11	PTFL.mp.
12	"Sprains and Strains"/
13	talocrural.tw.
14	1 or 2 or 3 or 4 or 5 or 6 or 7 or 8 or 9 or 10 or 11 or 12 or 13
15	Chiropractic/ or Manipulation, Orthopedic/
16	musculoskeletal manipulation.mp. or Musculoskeletal Manipulations/
17	(joint* adj5 manipul*).tw.
18	(ankle* adj5 rehab*).tw.
19	Mulligan*.mp.
20	Maitland*.mp.

21	MWM*.mp.
22	manual therap*.mp.
23	manual technique*.mp.
24	(joint* adj5 mobili?ation*).tw.
25	15 or 16 or 17 or 18 or 19 or 20 or 21 or 22 or 23 or 24
26	Randomized controlled trial.pt.
27	clinical trial.pt.
28	random*.tw.
29	trial*.tw.
30	group*.tw.
31	case series.tw.
32	cross-over studies/
33	Cross-Sectional Studies/
34	exp Cohort Studies/
35	26 or 27 or 28 or 29 or 30 or 31 or 32 or 33 or 34
36	14 and 25 and 35
37	limit 36 to humans

3.2.3 Identification and selection of studies

Full text RCTs, crossover studies, cross-sectional studies, cohort studies, and case series published in peer reviewed journals and dissertations were considered for the present review. Studies were not restricted by language, provided the title and abstract were in English. Studies not involving live

human participants (e.g., model-based, animal and cadaveric investigations) were excluded. Conference proceedings, commentaries, research notes, editorials, and letters were also excluded. To be included, studies were required to meet the following criteria:

Participants

Live humans (without any age limitation) with a grade I or II lateral or medial ligament sprain of the ankle at any stage of recovery (acute to chronic) in any ambulant setting who have been treated with joint mobilisation. Studies involving grade III sprains, fractures (other than Weber type A), and syndesmotic injuries were excluded from this review.

Intervention

Studies reporting any type of joint mobilisation techniques applied to the talocrural joint, subtalar joint, or inferior tibiofibular joint by a physiotherapist, medical practitioner, osteopath, chiropractor or athletic trainer were eligible for inclusion in the review. Interventions other than therapist performed joint mobilisation were excluded from the review.

Comparators

Studies reporting any conservative intervention for comparison, such as exercise therapy, elevation and icing, supportive strapping, sham intervention, or no treatment, were eligible for inclusion. Control

groups with healthy subjects were also eligible as a comparator. Studies which compared mobilisation techniques to surgical interventions were excluded.

Outcome measures

All commonly reported clinical impairments (pain, swelling, balance, proprioception, strength, stability, and gait), activity restriction and self-reported confidence, community participation, QOL, reinjury rate, function, and return to sport were considered for the review. The primary outcomes of interest were ankle ROM, pain, QOL, and function.

Timing of the measurement of the outcomes was categorised as either 'immediate', measured immediately following the intervention (Southerst et al., 2015), 'short-term' measured up to 3 months following the intervention (Van Ochten et al., 2014), and 'long-term' measured at 3 or more months (Bleakley et al., 2008) following the intervention.

Identified studies were exported to reference management software (EndNote X7.3.1, Ontario, Canada) and duplicate records were manually removed. Study titles and abstracts were initially screened by two independent reviewers, followed by screening of full text papers, to determine the eligibility of the identified studies. Disagreement between the reviewers was resolved by consensus or involvement of a third reviewer. The level of agreement between reviewers was assessed using Cohen's Kappa (Viera & Garrett, 2005).

3.2.4 Assessment of methodological quality

The methodological quality of individual studies was assessed using the PEDro scale for RCTs and the Quality Assessment Tool for Observational Cohort and Cross-sectional Studies (Higgins et al., 2011; National Institutes of Health, 2014; Pedro-Scale, 1999). Two independent reviewers assessed the methodological quality and the level of agreement between reviewers was assessed using Cohen's Kappa.

3.2.5 Assessment of the quality of evidence

The overall quality of evidence was assessed at the stage of meta- analysis, using the Grading of Recommendations, Assessment, Development and Evaluation (GRADE) approach (Atkins et al., 2004). The quality of the evidence was classified as either high, moderate, low, or very low (Guyatt et al., 2008). Risk of bias, consistency of results, directness (e.g. generalisability) and precision (e.g. sufficient data) were considered in assessing the overall quality (Armijo-Olivo et al., 2016).

3.2.6 Data extraction and statistical analysis

Descriptive data were extracted using an extraction table (Table 3.2). Authors were contacted if possible where there were difficulties extracting data from the published paper. Where feasible, study data that were comparable in terms of participant characteristics, outcome measures and follow-up periods, were pooled and a meta-analysis was performed.

For the meta-analysis, the standard mean difference (SMD) was calculated for the outcomes where the means and standard deviations were provided pre- and post-intervention. This conversion of the data to a common scale permitted comparison of studies that used different tools to measure the same outcome. This review followed the general practice of interpretation for small, medium, and large effect sizes (0·2=small effect, 0.5=medium effect, 0.8=large effect) (Cohen, 1988; Valentine & Cooper, 2013). The mean difference (MD) was calculated for studies using the same instrument for measurement. The results were reported in forest plots with 95% CI. The MCID was used to interpret the clinical meaningfulness of the findings. Inconsistency was quantified by calculating I² and interpreted as follows: 30% to 59% may represent moderate heterogeneity, 60% to 89% substantial heterogeneity, and 90% to 100% considerable heterogeneity between studies. If I² was greater than 30%, a random effects model was used to incorporate intertrial heterogeneity (Higgins et al., 2011).

In the instance of multiple comparison groups, the sham group was selected as the control condition. For the outcome of 'static balance', studies with eyes closed balance were selected to maintain the homogeneity of the analysis. Further, in studies with multiple time points, measurements taken at 2-3 weeks were selected for the meta-analysis (e.g., if effects were measured at the time points of 2 days, 3 weeks and at 2 months in a single study, data from measurements at 3 weeks were selected for the analysis). All statistical analyses were conducted using RevMan 5.3, Copenhagen (Revman, 2014).

Design	Sample	Intervention	Comparat	Measurement	Outcomes	Results
		and dosage	or	time points		
RCT	17	TCJ (antero-	sham	immediate	non weight-bearing	non weight-bearing
	(10M)	posterior)-			DFROM,	DFROM, significantly
	grade	mobilisation +			proprioception (joint	improved across time
	1/2	TCJ traction			position sense)	(p=0.04)
	chronic	30s				joint position sense
	lateral					significantly improved
	ankle					across time at target angle
	sprains					10° plantarflexion (p=0.03)
RCT	43	distal TFJ	no	immediate,	weight-bearing	weight-bearing DFROM
	chronic	manipulation +	interventio	short-term	DFROM,	not significant (p=0.82)
	ankle	HVLA thrust	n	(1, 2 and 3	static balance (single-	single-leg stance not
	sprains			wk)	leg stance), function	significant (n=0.42)
		1 repetition			(step down test, self-	Significant (p ^{-0.12})
					reported function,	function not significant-
					FAAM sports)	step down test (p=0.76),
						self-reported function
	Design RCT RCT	DesignSampleRCT17(10M)grade1/2chroniclateralanklesprainsunderRCT43chronicanklesprainssprains	DesignSampleInterventionRCT17TCJ (antero-(10M)posterior)-grademobilisation +1/2TCJ tractionchronic30slateral-sprains-RCT43distal TFJchronicmanipulation +ankle-sprains-It epetition-	DesignSampleInterventionComparatand dosageorRCT17TCJ (antero-sham(10M)posterior)-grademobilisation +J2TCJ tractionIchronice30sIlateraliateralIsprainssprainsIRCT43distal TFJnoankleinterventionanklesprainsIIIntervention1IankleIIIntervention1IAlteralIIRCT43distal TFJInterventionIIAlteralIIInterventionIAlteralIInterventionIAlteralIInterventionIAlteralIInterventionIInt	DesignSampleInterventionComparalMeasurementand dosageortime pointsRCT17TCJ (antero-shamimmediate(10M)posterior)-yadeshamimmediategrademobilisation +III1/2TCJ tractionIIIchronic30sIIIlateralIIIIankleIIIIgrainsinterventioshort-termankleIInterventioshort-termankleHVLA thrustn(1, 2 and 3)sprainsI repetitionistI	DesignSampleInterventionCompareMeasuremenOutcomesRCTind dosageorimediatenon weight-bearingRCT17TCJ (anteroshanimmediatenon weight-bearinggrademobilisation +FranceJFROM,porprioception (joint)1/2TCJ tractionFranceposition sense)lateralJosFranceFranceposition sense)ankleFranceinterventionshort-termpositon sense)RCT43distal TFJnoshort-termDFROM,ankleHVLA thrustno1,2 and 3staito balance (single-ankleINPACHna(1,2 and 3)staito balance (single-sprainsI repetitionFranceKath sports)interventionfatord staito balance (single-ankleI repetitionFranceKath sports)Kath sports)fatord staito balance (single-

Table 3.2 Description of the eligible studies

							(p=0.61),
							FAAM sports (p=0.83)
Collins et	randomised	16 (8M)	weight- bearing	placebo,	immediate	weight-bearing	weight-bearing DFROM
al, 2004	cross over	grade 2	MWM	no		DFROM,	significantly improved
		sub-	TCJ (posterior	interventio		pressure pain threshold,	across time
		acute	talar glide,	n		thermal pain threshold	(p=0.013) and no
		lateral	postero anterior				significant group
		ankle	tibial glide)				difference
		sprains	3 sets				(vs placebo p=0.202 vs
			of 10 repetitions				$(v_{3} \text{ praceod } p = 0.202, v_{3})$
							enageura noin threshold
							pressure pain unreshold
							and thermal pain threshold
							not significant (p<0.05)
Cruz-Díaz	RCT	81	weight-bearing	sham,	immediate,	weight-bearing	weight-bearing DFROM
et al, 2015		(47M)	MWM	no	short-term	DFROM,	significantly improved
		chronic	TCJ (posterior	interventio	(3 wk),	dynamic balance	p<0.0001 (at each time
		ankle	talar glide,	n	long-term	(SEBT)	point)
		sprains	postero-anterior		(6 mo)		dynamic balance
			tibial glide-)				significantly improved

			2 sets of 10 repetitions 2 sessions/wk for 3 wk				p<0.0001 (each direction of SEBT)
Gilbreath	prospective	11	weight -bearing	no control	short-term	weight -bearing	weight-bearing DFROM
et al, 2014	longitudinal	(5M)	MWM	group	(after 24-48 h)	DFROM,dynamic	not significant (p=0.69)
		chronic	TCJ (posterior			balance (SEBT),	dynamic balance not
		ankle	talar glide,			function	significant (SEBT-anterior
		sprains	postero anterior			(FAAM)	p=0.99; postero-medial -
			tibial glide)				p=0.15; postero-lateral
			2 sets of 4				p=0.24).
			repetitions				FAAM ADL not
			4m of MWM X				significant (p=0.19).
			3 sessions over				FAAM SPORTS
			a 1 wk				significantly improved
							across time (p=0.01)
Harkey et	RCT	30	Maitland	No	immediate	non weight-bearing	non weight- bearing
al, 2014		(14M)	mobilisation	interventio		DFROM, dynamic	DFROM significantly

		chronic	TCJ (antero-	n		balance (SEBT)	improved (p=0.049)
		ankle	posterior grade				dynamic balance no
		sprains	III)				improvement (p >0.05)
			3 sets of 60s				
Hoch &	randomised	20	Maitland	no	immediate	weight-bearing	weight-bearing DFROM
McKeon,	cross over	(9M)	mobilisation	interventio		DFROM, static balance	significantly improved
2011		chronic	TCJ-(anterior	n		dynamic balance	(p=0.01)
		ankle	posterior III)			(SEBT), talar stiffness	static balance significantly
		sprains	$50~^{\pm}5$ of 1s				improved time to boundary
			oscillations X2				antero-posterior minima
							significantly improved
							(p<0.0001)
							dynamic balance-not
							significant (p=0.98)
							(normalised reach
							distance)
							talar stiffness not
							significant (p=0.08)
Hoch et	prospective	12	Maitland	no control	short-term	Weight-bearing	weight-bearing DFROM
al, 2012	longitudinal	(6M)	mobilisation	group	(24–48 h and	DFROM,dynamic	significantly improved

		chronic	TCJ (antero-		1wk follow-	balance, function	across time (p<0.0001)
		ankle	posterior III)+		up)	(FAAM)	dynamic balance
		sprains	TCJ traction				significantly improved
			2 sets of 2min				across time (SEBT
			traction and 4				anterior- p<0.0001);
			sets of 2min				postero-medial- p=0.003;
			mobilisation				postero-lateral- p<0.0001)
							FAAM ADL and
							SPORTS significantly
							improved across time
							(p=0.001)
Hoch et	prospective	12 (6M)	Maitland	no control	short-term	static balance, talar	static balance not
al, 2014	longitudinal	chronic	Mobilisation	group	(24-48h, and	stiffness	significant.
		ankle	TCJ (antero		1 wk follow-		time to boundary antero-
		sprains	posterior III) +		up)		posterior and time to
			TCJ traction				boundary medio-lateral not
			2 sets of 2min				significant (p >0.05)
			traction and 4				talar stiffness not
			sets of 2min				significant (p>0.05)

Hopper et	randomised	20 (8M)	Mulligan ankle	injured	immediate	static balance, dynamic	static balance significantly
al., 2009	controlled	chronic	taping	taped,		balance (wandering,	improved in postural sway
	within-	ankle	not explicitly	injured un-		overshoot, reaction	recovery across time
	subjects	sprains	stated	taped,		time)	(p<0.001)
	design			uninjured taped, uninjured un-taped			single-leg stance not significant (0.792),
							dynamic tracking balance not significant ; wandering (p=0.559), overshoot- (p=0.547), reaction time- p=0.142.
Houstan	prospective	12 (6M)	Maitland	no control	immediate,	function	FAAM ADL some
et al, 2013	longitudinal	chronic	mobilisation	group	short-term	(FAAM sports)	components significantly
		ankle	TCJ (antero-		(1 wk follow-		improved across time;
		sprains	posterior III) +		up)		walking on even ground
			TCJ traction				(p=0.06); going down
			4min of traction				stairs (p=0.07); walking on

			and 8min of				uneven ground (p=0.03);
			mobilisation				light to moderate work
			6 sessions over				(p=0.06); heavy work
			2 wk.				(p=0.03); recreational
							activity (p=0.07)
							FAAM SPORTS some
							components significantly
							improved across time;
							landing (p=0.03); low
							impact activities (p=0.07);
							cutting p =0.02)
Iochon at	DCT	40	ontria avial	mugala	ah art tarm	DEPOM alertarflorion	DEPOM gionificantly
Josnep et	KC I	40	ankle axial	muscie	short-term	DFROM, planariexion	DFROM significantly
al., 2010		(19M)	elongation	energy	(1 mo)	ROM, static balance,	improved across time
		grade	TCJ (superior	technique		pain quality and	(p<0.001) and no
		1/2	inferior)-HVLA			intensity, function	significant group
		chronic	thrust			(functional evaluation	differences (p=0.713).
		lateral	6 sessions over			scale)	Plantarflexion ROM
		ankle	3 wk				significantly improved
		sprains					across time (p<0.001) and

no significant group
differences (p=0.300)
single-leg stance eyes
closed significantly
improved across time
(p<0.001) and no
significant group
differences (p=0.344)
single-leg stance eyes open
significantly improved
across time (p<0.001) and
no significant group
differences (p=0.413)
McGill significantly
improved across time
(p<0.001) and no
significant group
differences (p=0.077)
Functional evaluation scale
significantly improved

							across time (p<0.001) and
							no significant group
							differences (p=0.144)
Kohne, et	RCT	30	ankle axial	single	short-term	DFROM,	DFROM significantly
al 2007		(21M)	elongation	manipulati	(fifth wk	proprioception (joint	improved-p=0.028 (across
		grade	TCJ (superior	on	follow-up)	position sense),	time)
		1/2	inferior by a	treatment		pressure pain threshold,	joint position sense at 5°
		chronic	mortise			pain intensity	plantarflexion error
		recurrent	separation)-				significantly improved
		lateral	6				n=0.020 (screage time)
		ankle	manipulations				p=0.029 (across time)
		sprains	over 4 wk)				pressure pain threshold (p
							value not reported)
							pain intensity (p value not
							reported)
Lopez-	randomised	52	TCJ	placebo	immediate	proprioception	proprioception
Rodrıguez	controlled	(35M)	Manipulation				significantly improved;
et al, 2007	within-	grade 2	(caudal)				load support bilateral
	subject	chronic	HVLA thrust +				posterior load (p=0.016),
	repeated	lateral	posterior				anterior load (p=0.04),

	measures	ankle	gliding				posterior load (p=0.043),
		sprains	manipulation				posterior anterior load
			TCJ -HVLA				(p=0.016)
			thrust				
			1min				
Marron-	RCT	52	weight –	placebo	immediate,	weight-bearing	MWM- weight-bearing
Gomez,		(31M)	bearing MWM		short-term	DFROM	DFRFOM significantly
2015		chronic	TCJ (posterior		(24 and 48 h)		greater than placebo- p<
		ankle	talar glide,				0.05 (immediately and
		sprains	postero-anterior				short-term)
			tibial glide)				HVLA- weight-bearing
			1 set of 10				DFROM significantly
			repetitions				greater than placebo-
							p<0.001(immediately) and
			TCJ HVLA				p=0.001(short-term)
			distraction				
			thrust x 3				
Pellow et	RCT	30	ankle axial	detuned	short-term	non weight-bearing	non weight- bearing
al., 2001		(19M)	alongation	ultrasound	(1 mo follow-	DFROM, pain	DFROM significantly
		grade	(TCJ- superior	treatment	up)	threshold, pain quality	improved across time

1/2 sub-	inferior by a	and intensity, function	(p=0.001) and between
acute	mortise	(functional evaluation	groups-(p=0.001)
and	separation)	scale)	pain threshold significantly
chronic	8 manipulations		improved across time
lateral	over 4 wk		(p=0.002) and no
ankle			significant group
sprains			differences (p=0.395).
			McGill significantly
			improved across time
			(p=0.001) and between
			groups-(p=0.004)
			pain intensity significantly
			improved across time
			(p=0.002) and between
			groups-(p=0.004)
			functional evaluation scale
			significantly improved
			across time (p=0.001) and
			between groups-(p<0.001)

Plante,	RCT	20	TCJ (antero-	healthy	immediate	Weight-bearing	weight-bearing DFROM
2012		(12M)	posterior)	subjects		DFROM,	significantly improved
		chronic				static balance, function	across time (p<0.0001)
		ankle	10 oscillations			(dynamic functional	single-leg stance; centre of
		sprains				tasks)	pressure significantly
							improved (p<0.04)
							dynamic functional task
							(centre of pressure medial-
							lateral during jump task
							significantly improved
							[p<0.00]; centre of
							pressure medial- lateral
							during squat significantly
							improved [p< 0.022];
							centre of pressure medial -
							lateral during stance task
							significantly improved
							[p<0.0.039])
Reid et al,	randomised	23	weight-bearing	sham	immediate	weight-bearing	weight-bearing DFROM
2007	cross over	(8M)	MWM			DFROM	significantly improved

		chronic	(posterior talar				(p=0.02)
		lateral	glide, postero-				
		ankle	anterior tibial				
			glide)				
			10 repetitions X				
			2				
Someeh et	experimental	32	Mulligan ankle	healthy	immediate	dynamic balance	dynamic balance
al, 2015	study	(20M)	taping/ Fibular	subjects		(SEBT)	significantly improved
	design-	chronic	repositioning				across time- SEBT overall
	within	ankle	taping				reach (p=0.001)
	subjects	sprains	not explicitly				
			stated				

Someeh et	experimental	32	Mulligan ankle	healthy	immediate	function (dynamic	function significantly
al, 2015	study	(20M)	taping	subjects		functional tasks),	improved across time;
	design-	chronic	not explicitly			participants perceptions	single-leg hopping
	within	ankle	stated			of stability and	(p=0.014); figure of 8
	subjects	sprains				confidence	hopping (p=0.05); side
							hopping- (p=0.001)
							confidence in above
							mentioned functional tests
							significantly improved
							across time; p=0.023,
							0.048, and 0.038,
							respectively
X 7 [•] •	1 • 1	1.6	• • •		• • •	. 1 . 1 .	
Vicenzino	randomised	16	non weight-	no	immediate	weight-bearing	weight-bearing DFROM
et al,	cross over	(8M)	bearing MWM	interventio		DFROM,	significantly improved
2006		chronic	(antero	n		talar stiffness	(p=0.017)
		lateral	posterior talar				talar glide significantly
		ankle	glide for DF),				improved (p<0.001)
		sprains	4 glides of 10s				
			4 sets				

			weight-hearing				
			MWM				
			(posterior talar				
			glide, postero				
			anterior tibial				
			glide)				
			4 sets of 10				
			glides				
Wells,	RCT	17	Maitland	no	immediate	weight-bearing	weight bearing DFROM
2012		(7M)	mobilisation	interventio		DFROM,	not significant (p=0.95)
		chronic	(TCJ-antero-	n		non weight-bearing	non weight-bearing
		ankle	posterior IV)			BDFROM, dynamic	DFROM not significant
		sprains	3 repetitions,			balance, pain intensity,	(p=0.1)
			60s			static balance, stiffness,	dynamic balance not
						function	significant; SEBT
						(self-reported function)	composite (p=0.8);
							anterior (p=0.07); postero-
							medial (p=0.79); postero
							lateral (p=0.73)
							pain not significant

							(p=0.06).
							stiffness not significant
							(p=0.59)
							stability not significant
							(p=0.40)
							function (VAS) not
							significant (p=0.44)
Yeo et al,	randomised	13	Maitland	placebo,	immediate	weight-bearing	weight-bearing DFROM
2011	controlled	(10M)	mobilisation	no		DFROM,	significantly improved
	within-	grade 2	(distal TFJ	interventio		pressure pain threshold,	(p<0.0001)
	subject	sub-	antero-	n		pain intensity,	pressure pain threshold
	repeated	acute	posterior)			function	significantly improved
	measures	lateral	3 sets of 1min			(functional evaluation	(p<0.0001)
		sprain	mobilisation			scale)	pain intensity not
							significant (p=0.369)
							functional evaluation scale
							not significant (p=0.475)

Note: "immediate", measured immediately following the intervention. "short-term", measured up to 3 months following the intervention.

"long-term", measured at 3 or more months following the intervention.

ADL, activities of daily living; DFROM, dorsiflexion range of motion; FAAM, Foot and ankle ability measure; HVLA, high velocity low amplitude; M, Male; MWM, mobilisation with movement; RCT, randomised controlled trial; SEBT, star excursion balance test; TCJ, talocrural joint; TFJ, tibiofibular joint

3.3 Results

3.3.1 Selection and characteristics of included studies

The database search identified 1521 studies after duplicate removal and a further nine studies were identified through citation tracking and hand searching of reference lists (Figure 3.1). Following the first stage of screening (using study title and abstract), 56 studies (database search-n=47, hand search-n=9) were identified as eligible for inclusion from the original 1530 (database searchn=1521, hand search- n=9) studies. Common reasons for exclusion following title and abstract screening included; ineligible study design, joint mobilisation was not assessed in isolation, and the study aim was not relevant to the review research question. A further 33 studies were excluded in second stage (full text) screening, and reasons for exclusion included; study aim not relevant to research question (Ambarish et al., 2008; Baker, 2013; Cosby, 2012; Cosby et al., 2011; Delahunt et al., 2013; Eisenhart et al., 2003; Estrade, 2013; Fisher et al., 2009; Green et al., 2001; Hedlund et al., 2014; Kumari et al., 2014; Landrum et al., 2008; Lee et al., 2012; Lubbe et al., 2015; Nambi & Shah, 2012; Nilsson, 1983; Rashid et al., 2013; Silva et al., 2017; Teixeira et al., 2013) (n=19), conference proceedings, commentaries and research notes (Andersen et al., 1987; Ayad et al., 2015; Clinch, 1986; Fay & Egerod, 1984; Hart & Macintyre, 2002; Rundle, 1988; Vicenzino et al., 2005) (n=7), not peer reviewed (Andersen et al., 2003; Chen et al., 2012; Yu, 2007) (n=3), full text not available (Diebschlag, 1987; Lee & Kim, 2005) (n=2), study protocol only (Davenport et al., 2010) (n=1), and thesis removed as the relevant published paper was included (Hoch, 2011) (n=1). Twenty-three studies (including three theses) were therefore included in the current review. The inter-reviewer agreement for the title/abstract and full text screenings was considered to be very good (k=0.80, 95% CI 0.72-0.89) and good (k=0.71, 95% CI 0.52-0.90) respectively. All disagreements were resolved by consensus. The data from 11 studies (including two theses (Plante,

2012; Wells, 2012)) were available and deemed appropriate for inclusion in the meta-analysis (Figure 3.1). Publication bias was visually observed using funnel plots (Figure 3.2).

The included studies were conducted in seven countries (Australia, Canada, Iran, New Zealand, South Africa, Spain, and United States) and involved a total of 585 participants. Twenty- one studies evaluated chronic ankle sprains and three studies investigated sub-acute sprains. Outcomes measured varied widely and included DFROM, proprioception, stability/balance, pain threshold (pressure and thermal), pain intensity and quality, function, talar stiffness, postural sway, and patient confidence. A range of joint mobilisation techniques were used and these included MWM in both weight-bearing or/and non weight-bearing (n=6) (Collins et al., 2004; Cruz-Diaz et al., 2015; Gilbreath et al., 2014; Marron-Gomez et al., 2015; Reid et al., 2007; Vicenzino et al., 2006), antero-posterior talocrural mobilisation (Maitland grades III and IV (Maitland, 2005), (n=4) (Harkey et al., 2014; Hoch & Mckeon, 2011; Plante, 2012; Wells, 2012), high velocity low amplitude (HVLA) ankle axial elongation manipulation and manipulation of the talocrural joint (n=6) (Alanson, 2012; Joseph et al., 2010; Kohne et al., 2007; Lopez-Rodriguez et al., 2007; Marron-Gomez et al., 2015; Pellow & Brantingham, 2001), Mulligan ankle taping (MAT) (n=3) (Hopper et al., 2009; Someeh et al., 2015a, 2015b), distal tibiofibular joint manipulation or mobilisation (n=2) (Beazell et al., 2012; Yeo & Wright, 2011), and combined mobilisation and traction of the talocrural joint (n=4) (Alanson, 2012; Hoch, Andreatta, et al., 2012; Hoch et al., 2014; Houston et al., 2013). MAT was included because it aims to mimic a MWM by sustaining the fibula glide during daily activities (Hing et al., 2015). These techniques were variously applied by physiotherapists, medical practitioners, chiropractors and athletic trainers. Table 3.2 describes the participants, interventions, comparators, outcome measures and results of the included studies.



Figure 3.1 Flow chart of study selection

Immediate effect of mobilisation on weight-bearing dorsiflexion, pain, static balance and dynamic balance





Short-term effect of mobilisation on weight-bearing dorsiflexion

a) Weight-bearing dorsiflexion

SE=Standard Error; SMD=standard mean difference; MD=mean difference, PM=postero-medial; PL=postero-lateral; SEBT star excursion balance test

Figure 3.2 Funnel plots

The immediate effects of joint mobilisation were evaluated in 17 studies, short-term effects in 10 studies, and the long-term effects were assessed in only one study (Table 3.3). No studies evaluating effects on gait parameters, QOL, re-injury rate or strength were located in our search. In this systematic review, participants with chronic ankle sprains were included in 21 studies and three studies included participants with sub-acute sprains. No studies measuring the effectiveness of mobilisation in isolation for acute ankle sprains were able to be found. A meta-analysis was conducted using 11 studies, all involving participants with chronic ankle sprains.

3.3.2 Common mobilisation techniques used in rehabilitation of ankle sprainsFive combinations of mobilisation techniques were used in the 23 studies, including MulliganMWM and taping techniques, Maitland mobilisation with and without traction, and manipulation.

The number of studies with positive effects on any clinically relevant outcome are contrasted against the number of studies with no positive effects, for each mobilisation technique (Figure 3.3). The findings also suggest that the combination of Mulligan MWM and taping is more likely to produce a clinical benefit than the other three mobilisation combinations, as more (17) of the studies using MWM techniques found positive outcomes compared to other techniques (Maitland mobilisation 12, manipulation 14). Further, studies reporting no positive outcomes with MWM techniques are fewer in number (6) compared with the other techniques (Maitland mobilisation 13).



Percentage of outcome evaluations with positive findings

Percentage of outcome evaluations without positive findings

Figure 3.3 Percentage and number of outcome evaluations with and without positive findings following each technique combination of mobilisation for any clinically relevant outcome at any time point
3.3.3 Quality of studies

Due to differences in study design, two tools were used to assess the methodological quality of the included studies. PEDro was used for the assessment of RCTs (n=19) and the quality assessment tool for observational cohort and cross-sectional studies was used for all other study types (n=4). The level of agreement between reviewers for the quality assessment was considered to be high (k=0.63, 95% CI=0.53-0.73) and all disagreements were resolved by consensus.

Most studies scored well on random allocation, adequate follow-up, and for providing both point measures and measures of variability for at least one key outcome. In studies assessed using the PEDro scale (Figure 3.4), the most common risk of bias was for therapist and subject blinding. For the Quality assessment tool for observational cohort and cross-sectional studies, all four studies demonstrated bias in terms of insufficient timeframe, different levels of exposures as related to the outcome examined, and clearly defined valid and reliable exposure measures (Figure 3.5). All studies scored at least moderate in terms of the overall quality of the methodology for both the scales utilised (Figure 3.6 and 3.7).



PEDro Scores per Criteria

Figure 3.4 PEDro scores for assessment of quality of individual criteria (Pedro-Scale, 1999). Details about criteria: 1, Eligibility criteria were specified (Explanation: This criterion influences external validity, but not the internal or statistical validity of the trial. It has been included in the PEDro Scale so that all items of the Delphi Scale are represented on the PEDro Scale. This item is not used to calculate the PEDro score) (PEDro Scale); 2, Participants were randomly allocated to groups; 3, Allocation was concealed; 4, Groups were similar at baseline regarding most important prognostic indicators; 5, Blinding of all participants; 6, Blinding of therapists who administered the therapy; 7, Blinding of all assessors who measured at least 1 key outcome; 8, Measures of at least 1 key outcome were obtained from more than 85% of the participants; 9, All participants for whom outcome measures were available received the treatment or control condition as allocated; 10, Results of between-group statistical comparisons are reported for at least 1 key outcome; 11, Study provides both point measures and measures of variability for at least 1 key outcome.

Quality Assessment Tool for Observational Cohort and Cross-Sectional Studies Score per Criteria



Figure 3.5 Quality assessment tool for observational cohort and cross-sectional studies (National

Institutes of Health, 2014)

PEDro scale												
Study	1) Eligibility criteria	2) Random allocation	3) Concealed allocation	4) Baseline comparability	5) Blinding subjects	6) Blinding therapists	7) Blinding assessors	8) Adequate follow-up (than 85% of subjects)	9) Intention to treat analysis	10) Between-group comparisons	11) Point measures and variability	Total Score out of 10
Alanson 2012	+	+	+	-	-	-	+	+	+	+	+	7
Beazell, Grindstaff et al. 2012	+	+	-	+	-	-	+	+	+	+	+	7
Collins, Teys et al. 2004	+	+	-	+	+	-	+	+	-	+	+	7
Cruz-Diaz, Lomas Vega et al. 2015	+	+	+	+	-	-	+	+	-	+	+	7
Harkey, McLeod et al. 2014	+	+	+	+	-	-	+	+	+	+	+	8
Hoch and McKeon 2011	+	+	+	+	-	-	+	+	+	+	+	8
Hopper, Samsson et al. 2009	+	+	-	+	-	-	-	+	+	+	+	6
Joseph, de Busser et al. 2010	+	+	+	+	-	-	-	+	+	+	+	7
Kohne, Jones et al. 2007	+	+	+	-	-	-	-	+	+	+	+	6
Lopez-Rodriguez, de-Las- Penas et al. 2007	+	-	-	+	-	-	-	+	+	+	+	5
Marron-Gomez, Rodriguez- Fernandez et al. 2015	+	+	-	+	+	-	+	+	+	+	+	8
Pellow and Brantingham 2001	+	+	-	+	-	-	-	+	-	+	+	5
Plante 2012	+	+	-	+	-	-	-	+	+	+	+	6
Reid, Birmingham et al. 2007	+	+	-	+	-	-	+	+	-	+	+	6
Someeh, Norasteh et al. 2015	+	+	-	+	-	-	-	+	+	+	+	6
Someeh, Norasteh et al. 2015	+	+	-	+	-	-	-	+	+	+	+	6

Vicenzino, Branjerdporn et al. 2006	+	+	-	+	+	-	+	+	+	+	+	8
Wells 2012	+	+	+	+	-	-	+	+	+	+	+	8
Yeo and Wright 2011	+	+	-	+	-	-	+	+	+	+	+	7

Figure 3.6 PEDro scores for assessment of quality of individual intervention studies

+ meet criteria, - do not meet criteria

1, eligibility criteria were specified (*Explanation: This criterion influences external validity, but not the internal or statistical validity of the trial. It has been included in the PEDro scale so that all items of the Delphi scale are represented on the PEDro scale. This item is not used to calculate the PEDro score*) (PEDro Scale); 2, participants were randomly allocated to groups; 3, allocation was concealed; 4, groups were similar at baseline regarding most important prognostic indicators; 5,blinding of all participants; 6, blinding of therapists who administered the therapy; 7, blinding of all assessors who measured at least one key outcome; 8, measures of at least one key outcome were obtained from more than 85% of the participants; 9, all participants for whom outcome measures were available received the treatment or control condition as allocated; 10, results of between-group statistical comparisons are reported for at least one key outcome; 11, study provides both point measures and measures of variability for at least one key outcome (Pedro-Scale, 1999).

	Qu sca	Quality assessment tool for observational cohort and cross-sectional studies scale													
Study	1	2	3	4	5	6	7	8	9	10	11	12	13	14	Score out of 14
(Gilbreath, Gaven et al. 2014)	+	+	+	+	+	-	-	-	+	+	+	-	+	+	10
(Hoch, Andreatta et al. 2012)	+	+	+	+	+	-	-	-	+	+	+	+	+	+	11
(Hoch, Mullineaux et al. 2014)	+	+	+	+	+	-	-	-	+	+	+	-	+	+	10
(Houston, McKeon et al. 2013)	+	+	+	+	-	-	-	-	+	-	+	-	+	-	7

Figure 3.7 Quality assessment tool for observational cohort and cross-sectional studies scores for assessment of quality of individual cohort studies

+ meet criteria, - do not meet criteria

1, Research question or objective clearly stated; 2, Study population clearly specified and defined; 3, Participation rate of eligible persons \geq 50%; 4, Subjects selected from same or similar population; 5, Sample size justification; 6, Exposure(s) of interest measured prior to outcome(s); 7,Timeframe sufficient; 8, Different levels of exposures as related to the outcome are examined; 9, Exposure measures clearly defined, valid, and reliable;10, Exposure(s) assessed more than once over time; 11,Outcome measures clearly defined, valid, and reliable; 12, Outcome assessors blinded to the exposure status; 13, Follow-up after baseline \leq 20%; 14, Adjusted for potential confounding variables Total (0 to 14) (Higgins et al., 2011; National Institutes of Health, 2014) 3.3.4 Effects of mobilisation on sub-acute/chronic ankle sprains

The outcome measures of DFROM, proprioception, stability/balance, pain threshold, pain intensity and quality, function, talar stiffness, postural sway, and patient's confidence towards stability were assessed at varying time points across the studies after application of joint mobilisation. Table 3.3 lists each outcome evaluation, indicating positive effects of mobilisation at each of the three time points of interest.

Eleven studies on chronic sprains reported quantitative data on five different outcomes, including weight-bearing DFROM, static balance, dynamic balance, pain intensity and pain threshold. However, due to study heterogeneity and a lack of useable data for some outcomes, data could only be pooled for weight-bearing DFROM, static balance, dynamic balance and pain intensity in order to evaluate immediate effects, and weight-bearing DFROM was the only outcome measure available to assess the short-term effects of ankle mobilisation.

Positive findings											
Quitcome	Immed	liate	Short-te	erm	Long-term						
	Yes	No	Yes	No	Yes	No					
1. DFROM	11	3	4	4	1	0					
weight bearing DFROM	9	2	3	2	1	0					
non weight bearing DFROM	2	1	0	1	0	0					
unspecified	0	0	1	1	0	0					
2. Proprioception	2	0	1	0	0	0					
3. Stability/balance	3	7	3	3	1	0					
static balance	1	3	1	3	0	0					
dynamic balance	2	4	2	0	1	0					
4. Pain threshold	1	1	1	1	0	0					
5. Pain intensity	0	2	2	1	0	0					
6. Functional outcomes	2	4	4	2	0	0					
7. Talar stiffness	1	2	0	1	0	0					
8. Recovery from postural sway	1	0	0	0	0	0					
9. Patient's confidence towards stability	1	0	0	0	0	0					

Table 3.3 Number of outcome evaluations investigating at each time point of interest, listed by the reported positive effects

DFROM, dorsiflexion range of motion

Note: "Immediate", measured immediately following the intervention, "Short-term", measured up to 3 months following the intervention, "Long-term", measured at 3 or more months following the intervention

3.3.5 Immediate effects of mobilisation on ankle sprains

The immediate effects on DFROM were assessed in 14 outcome evaluations, of which 11 reported improvement with mobilisation techniques (Table 3.3). The findings for other outcomes were less notable. Of the 10 studies which investigated the immediate effects of mobilisation on stability/balance, three had demonstrable improvement (Cruz-Diaz et al., 2015; Hoch & Mckeon, 2011; Someeh et al., 2015a). Similarly, studies which assessed pain, talar stiffness and function revealed inconsistent results. When considering the immediate effects of mobilisation on functional outcomes, two outcome evaluations out of six demonstrated that it was effective (Houston et al., 2013; Someeh et al., 2015b). A summary of the reported immediate effects is provided in Table 3.3.

Pooled data from five studies with a total of 180 participants were grouped for analysis of the effects of mobilisation on each direction of the SEBT; anterior, postero-medial, and postero-lateral. This analysis provided significant findings for the postero-medial direction of the SEBT (MD=3.22, CI=1.43-5.01, p=0.0004), however the postero-lateral direction (MD=3.55, CI=-0.18-7.28, p=0.06) and the anterior direction (MD=4.10, CI=-0.35-8.54, p=0.07) results of the SEBT, were not significant (Figure 3.8). Pooled data for static balance from three studies with a total of 100 participants indicated there were no significant immediate benefits following mobilisation of individuals with chronic sprains, when compared to control participants (SMD=0.01, CI=-0.38-0.40, p=0.96) (Figure 3.9).

	Expe	eriment	iental Control					Mean Difference Mean Difference		
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% Cl	IV, Random, 95% Cl	
1.2.1 Postero-medial SEBT										
Cruz-diaz et al, 2015(14)	88.51	5.43	30	85.52	2.86	31	14.7%	2.99 [0.80, 5.18]		
Harkey et al, 2014(73)	83	11	15	78	9.7	15	4.2%	5.00 [-2.42, 12.42]	+	
Hoch & McKeon, 2011(74)	94.06	8.15	20	93.3	8.48	20	7.1%	0.76 [-4.39, 5.91]		
Someeh et al, 2015(81)	101.8	8.3	16	96.4	7.2	16	6.7%	5.40 [0.02, 10.78]		
Wells, 2012(69)	85.04	7.55	9	79.12	11	8	3.1%	5.92 [-3.16, 15.00]		
Subtotal (95% CI)			90			90	35.8%	3.22 [1.43, 5.01]	◆	
Heterogeneity: Tau ² = 0.00; (Chi ř = 2.1	11, df=	4 (P = 0	0.72); I ^z a	= 0%					
Test for overall effect: Z = 3.5	i2 (P = 0.	.0004)								
1.2.2 Anterior SEBT										
Cruz-diaz et al, 2015(14)	84.72	5.07	30	77.09	6.01	31	12.8%	7.63 [4.84, 10.42]		
Harkey et al, 2014(73)	72.5	7.5	15	64.6	6.8	15	7.1%	7.90 [2.78, 13.02]		
Hoch & McKeon, 2011(74)	79.44	4.73	20	78.91	5.51	20	11.6%	0.53 [-2.65, 3.71]		
Wells, 2012(69)	64.82	6.12	_9	65.07	6.42	- 8	5.8%	-0.25 [-6.23, 5.73]		
Subtotal (95% CI)			74			74	37.4%	4.10 [-0.35, 8.54]	-	
Heterogeneity: Tau ² = 15.83;	Chi ² = 1	4.94, df	'= 3 (P	= 0.002); I ² = 80	0%				
Test for overall effect: Z = 1.8	(0 (P = 0.	.07)								
1.2.3 Postero-lateral SEBT										
Cruz-diaz et al, 2015(14)	89.28	3.04	30	87.11	3.25	31	16.5%	2.17 [0.59, 3.75]		
Harkey et al, 2014(73)	77.6	13.9	15	70.4	12.2	15	2.9%	7.20 [-2.16, 16.56]		
Hoch & McKeon, 2011(74)	87.48	10.55	20	86.89	11.02	20	5.0%	0.59 [-6.10, 7.28]		
Wells, 2012(69)	82.65	10.04	_9	70.47	11.57	8	2.4%	12.18 [1.82, 22.54]		
Subtotal (95% CI)			74			74	26.8%	3.55 [-0.18, 7.28]	-	
Heterogeneity: Tau² = 5.81; (Chi² = 4.8	30, df=	3 (P = (0.19); I²∘	= 37%					
Test for overall effect: Z = 1.8	17 (P = 0)	.06)								
Tatal (DEV/ CI)			220			220	100.0%	2 72 12 00 5 461		
Total (95% CI)		~~ ~	238			238	100.0%	3.73 [Z.UU, 5.46]		
Heterogeneity: Tau ² = 4.06; (⊃ni* = 24	.83, df =	= 12 (P	= 0.02);	If = 529	%			-20 -10 0 10 20	
lest for overall effect: Z = 4.2	:3 (P < 0.	.0001)							Favours (control) Favours (experimental)	
Test for subgroup difference	s: Chi ² =	0.14, d	t= 2 (P	= 0.93)	. I* = 0%	b				

Figure 3.8 MD (95% CI) of the immediate effect of joint mobilisation on dynamic balance by

pooling data from 5 studies (n=180). Abbreviations: df, degrees of freedom; IV, inverse variance

	Exp	eriment	al	Control				Std. Mean Difference	Std. Mean Difference			e	
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Fixed, 95% CI		IV, Fixe	d, 95% Cl		
Hoch & McKeon, 2011(74)	1.96	0.6	20	1.95	0.45	20	40.0%	0.02 [-0.60, 0.64]			+		
Hopper et al, 2009(82)	3.23	1.06	20	3.31	1.26	20	40.0%	-0.07 [-0.69, 0.55]			•		
Plante, 2012(70)	1.028	0.029	10	1.024	0.024	10	20.0%	0.14 [-0.73, 1.02]			†		
Total (95% CI)			50			50	100.0%	0.01 [-0.38, 0.40]			•		
Heterogeneity: Chi ² = 0.15, c	-20	-10	0	10	20								
Test for overall effect. $Z = 0.0$	15 (P = 0	.96)								Favours (control	Favours	[experime	intal]

Figure 3.9 SMD (95% CI) of the immediate effect of joint mobilisation on static balance by pooling data from 3 studies (n=100). Abbreviations: df, degrees of freedom; IV, inverse variance; Std., standardised

Similarly, data from seven studies with a total of 249 participants indicated there were no significant immediate effects of mobilisation on the weight-bearing DF-ROM of individuals with chronic sprains (SMD=0.66, CI=-0.25-1.58, p=0.16) (Figure 3.10). For pain intensity, pooled data from two studies with a total 47 participants indicated mobilisation had no immediate effect on individuals with chronic sprains (SMD=-0.21, CI=-0.78-0.37, p=0.48) (Figure 3.11). There were insufficient data to analyse the immediate benefits of mobilisation on pain threshold.

	Expe	rimen	tal	Control				Std. Mean Difference	Std. Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% Cl	IV, Random, 95% Cl
Cruz-diaz et al, 2015(14)	16.8	0.47	30	15.07	0.48	31	14.0%	3.59 [2.77, 4.42]	_ _
Hoch & McKeon, 2011(74)	12.62	2.79	20	12.2	3.01	20	14.8%	0.14 [-0.48, 0.76]	
Marron-Gomez, 2015(15)	11.5	3.8	18	8.3	1.5	15	14.4%	1.04 [0.31, 1.78]	_
Plante, 2012(70)	13	3	10	12.4	4	10	13.9%	0.16 [-0.72, 1.04]	
Reid et al, 2007(71)	10.55	3.79	23	10.32	3.89	23	14.9%	0.06 [-0.52, 0.64]	
Vicenzino et al, 2006(16)	4.8	1.5	16	4.4	1.6	16	14.5%	0.25 [-0.44, 0.95]	
Wells, 2012(69)	9.5	2.73	9	11.04	1.93	8	13.4%	-0.61 [-1.59, 0.37]	
Total (95% CI)			126			123	100.0%	0.66 [-0.25, 1.58]	-
Heterogeneity: Tau ² = 1.37; 0	Chi² = 64	.78, df							
Test for overall effect: Z = 1.4	12 (P = 0.	16)	Favours (control) Favours (experimental)						

Figure 3.10 SMD (95% CI) of the immediate effect of joint mobilisation on weight-bearing DFROM by pooling data from 7 studies (n=249). Abbreviations: df, degrees of freedom; IV, inverse variance: Std., standardised

inverse variance; Std., standardised



Figure 3.11 SMD (95% CI) of the immediate effect of joint mobilisation on pain intensity by pooling data from 2 studies (n=47). Abbreviations: df, degrees of freedom; IV, inverse variance; Std., standardised

3.3.6 Short-term effects of mobilisation on ankle sprains

Half of the outcome evaluations reported that mobilisation improved DFROM, stability/balance and pain threshold in the short-term (Table 3.3). Demonstrable improvement was also observed in pain intensity and function (Table 3.3), and two studies (Hoch et al., 2014; Kohne et al., 2007) which evaluated short-term outcomes on talar stiffness and proprioception reported improvements. No studies reported short-term findings on postural sway or patient's balance confidence.

Pooled data from two studies with 94 participants with chronic sprains indicated joint mobilisation was effective in the short-term for improving weight-bearing DFROM (MD=2.56, CI=0.89-4.23, p=0.003) (Figure 3.12). There were insufficient data evaluating static balance, dynamic balance, pain threshold and pain intensity to permit analysis of the short-term benefits of mobilisation on these outcomes.

	Expe	erimen	tal	Control Std. Mean Difference				Std. Mean Difference	Std. Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% Cl	IV, Random, 95% Cl
Cruz-diaz et al, 2015(14)	17.04	0.48	30	15.12	0.52	31	49.7%	3.79 [2.93, 4.64]	
Marron-Gomez, 2015(15)	12	3.7	18	8.3	1.5	15	50.3%	1.24 [0.48, 1.99]	•
Total (95% CI)			48			46	100.0%	2.50 [0.00, 5.00]	◆
Heterogeneity: Tau ² = 3.08; Test for overall effect: Z = 1.	Chi² = 1 .96 (P = I	9.13, d 0.05)	-20 -10 0 10 20 Favours [control] Favours [experimental]						

Figure 3.12 SMD (95% CI) of the short-term effect of joint mobilisation on weight-bearing DFROM by pooling data from 2 studies (n=94). Abbreviations: df, degrees of freedom; IV, inverse variance; Std., standardised

3.3.7 Long-term effects of mobilisation on ankle sprains

Only one study evaluated the long-term effects of mobilisation on ankle sprains. Long-term improvement in DFROM and stability/balance were reported in the single included study (Cruz-Diaz et al., 2015).

3.3.8 Quality of evidence

According to the GRADE assessment (Table 3.4), the evidence for DFROM (immediate and short term), static balance and dynamic balance can be considered to be of moderate quality. The evidence for pain was considered to be of low quality due to lack of generalisability of one of the included studies. Overall, the evidence included in this meta-analysis was considered to be of moderate quality, with the risk of bias and the level of heterogeneity the main factors influencing the quality of the evidence.

Table 3.4 Assessment	of the	quality	of evidence
----------------------	--------	---------	-------------

Number of studies (sample size, n)	Risk of bias	Inconsistency	Indirectness	Imprecision	Quality of evidence
Immediate effects					
Outcome: DFROM					
7 studies (n; experimental=126 : control=123)	low risk of bias (Pedro scores: 6,6,7,8,8,8 and 8)	p value on test for heterogeneity p<0.00001, I ² =91% high inconsistency	low indirectness	low imprecision	moderate quality (low risk of bias and high inconsistency)

Outcome: dynamic	balance				
5 studies (n; experimental=90: control=90)	low risk of bias (Pedro scores: 6,7,8,8 and 8)	p value on test for heterogeneity p=0.02, I ² =52% moderate inconsistency	low indirectness	low imprecision	moderate quality (low risk of bias and moderate inconsistency)
Outcome: static bal	ance				
3 studies (n; experimental=50: control=50)	moderate risk of bias (Pedro scores: 6,6 and 8)	p value on test for heterogeneity p=0.93, I ² =0% low inconsistency	low indirectness	low imprecision	moderate quality (moderate risk of bias and low inconsistency)
Outcome: pain inte	nsity				
2 studies (n; experimental=24: control=23)	moderate risk of bias (Pedro scores: 5 and 8)	p value on test for heterogeneity p=0.73, I ² =0% low inconsistency	moderate indirectness (less generalisable)	low imprecision	low quality (moderate risk of bias, moderate inconsistency and low indirectness)
Short-term effects					
Outcome: DFROM					
2 studies (n; experimental=48: control=46)	low risk of bias (Pedro scores: 7 and 8)	p value on test for heterogeneity p<0.0001, I ² =95% high inconsistency	low indirectness	low imprecision	moderate quality (low risk of bias and high inconsistency)

DFROM, Dorsiflexion range of movement

3.4 Discussion

This is the first systematic review to assess the clinical benefits of joint mobilisation in the management of either lateral or medial ankle ligament sprains at all stages of recovery. Importantly, this is the first review to only include studies in which joint mobilisation is the sole intervention. The current review did not identify any studies evaluating the clinical benefits of joint mobilisation on acute ankle sprains, perhaps because mobilisation is not typically the preferred choice of management in the acute stage of ankle sprains (Van Den Bekerom et al., 2013). Findings about the clinical benefits of mobilisation on the majority of outcome measures were inconsistent across studies, and a lack of reported quantitative data, heterogeneity of subjects and the differing types of joint mobilisation applied made direct comparisons difficult. Despite this, meta-analysis indicated there are immediate benefits of mobilisation for improving dynamic balance, and a short-term benefit in improving weight-bearing DFROM in chronic ankle sprains. These results provide compelling evidence that joint mobilisation may be effective in improving balance immediately and in increasing DFROM in the short-term in chronic ankle sprains.

Dynamic balance and weight-bearing DFROM improvements following joint mobilisation were both associated with clinically meaningful changes. The modified SEBT test assesses performance during single-leg balance with reaching in three directions (anterior, postero-medial, posterolateral) (Chimera et al., 2015; Rehabilitation-Measures-Database, 2010). The MCID for this test is reported as being 3.5%, and therefore the immediate effect on dynamic balance found in the metaanalysis (MD=3.73) can be considered as clinically meaningful (Chimera et al., 2015; Rehabilitation-Measures-Database, 2010). It is plausible that the immediate improvements in dynamic balance following joint mobilisation may increase the individual's balance confidence and perhaps reduce the risk of re-injury. Clinically, this may assist the individual with an ankle sprain to more safely proceed to the next level of functional exercise in the rehabilitation process.

There were no immediate improvements in either anterior SEBT performance or DFROM. Interestingly, previous research supports the existence of a correlation between anterior SEBT performance and the weight-bearing lunge test (Hoch, Staton, et al., 2012). This correlation could help explain the current review's findings on immediate anterior SEBT performance and DF-ROM. Notably, the MCID for ankle DFROM has not been established (Young et al., 2013). However, approximately 3.6° of DFROM is associated with 1cm in distance from the wall in the lunge test (Hoch & Mckeon, 2011). The MD in the short-term measurement of weight-bearing DFROM from the current meta-analysis was 2.56cm and this equates to 9.2° of dorsiflexion, which can be considered as clinically meaningful given that the normal total range is only 15- 20° (Hallaceli et al., 2014; Roaas & Andersson, 1982).

Joint mobilisation techniques are aimed at restoring the normal joint ROM (Hertling & Kessler, 2006; Oatis, 2004), and indeed this review found DFROM improved following mobilisation. However, the mechanisms by which restoring ankle ROM may assist other impairments is unclear, as are the underlying mechanisms by which mobilisation may actually work (Marron-Gomez et al., 2015; Vicenzino et al., 2006). It has been proposed that increased ankle ROM is due to the correction of a bony positional fault (Vicenzino et al., 2011). It is further postulated that the correct alignment of the articular surfaces may help to restore normal biomechanics, as well as sensorimotor function (Vicenzino et al., 2011). However, it may be that mobilisation produces less impact on pain, as evidenced by the lack of improvement in ankle pain outcome measures in this review. Potential underlying central nervous system changes related to persistent pain in chronic sprains remain unclear, but central sensitisation may be a possible factor for persistence of chronic pain. If central sensitisation is actually a key factor contributing to chronic ankle sprain pain, then changing the bony alignment would be unlikely to improve pain in chronic sprains as it is not the usual localised pressure pain hypersensitivity (Ramiro-Gonzalez et al., 2012) experienced immediately after a sprain.

According to the Clinical Practice Guidelines Linked to the ICF from the APTA, clinicians should use joint mobilisation to improve ankle dorsiflexion, proprioception, and weight-bearing tolerance in patients recovering from a lateral sprain (Martin et al., 2013). Of these three outcomes, the findings of the current review only support the benefit of mobilisation for dorsiflexion. There was insufficient research available to conclude whether mobilisation is effective for improving proprioception or weight-bearing tolerance. However, the current review found clinically meaningful evidence for the effect of mobilisation on dynamic balance, an outcome not mentioned in the Clinical Practice Guidelines from the APTA. One explanation for this difference may be that the Guidelines only included literature published prior to April 2012, while the current review has included seven more recently published studies.

The inclusion and exclusion criteria of the current review differ in important ways from previous systematic reviews on this topic. In contrast to these prior reviews, our search criteria included both lateral and medial ligament sprains, covered all stages of recovery from acute to chronic, and encompassed all clinically relevant outcomes used to assess the effects of mobilisation. Importantly, of the six prior reviews which have evaluated the efficacy of mobilisation techniques

on ankle sprains, all included studies which did not evaluate joint mobilisation as a unique intervention, but rather as an adjunct to other interventions (such as home exercise programs, RICE protocol and external supports included in their review (Bleakley et al., 2008; Loudon et al., 2014; Southerst et al., 2015; Terada et al., 2013; Van Der Wees et al., 2006; Wikstrom & Mckeon, 2011). The current review excluded these multi-modal studies to ensure the homogeneity of the included studies, and to increase the precision of the results in relation to the effects of joint mobilisation. Compared to the recent review by Loudon et al (Loudon et al., 2014), the present review included almost three times more studies (23), with all of these only investigating the clinical effects of joint mobilisation techniques in isolation. In the review by Loudon et al (Loudon et al., 2014), only eight studies were included, and of those mobilisation was used as the sole intervention in only five (Loudon et al., 2014). This disparity in the number of included studies may be due to our searching a greater number of databases (11), including medial ankle sprains in the search criteria, by reviewing dissertations and theses, and by not limiting clinical outcomes.

This review includes the first meta-analysis undertaken to assess the clinical benefits of joint mobilisation for ankle sprains. When comparing the findings of the current review to previous systematic reviews, there were some agreements and some inconsistent results. When considering the immediate effects of mobilisation, the review by van der Wees et al (Van Der Wees et al., 2006) reported an improvement in DFROM (Van Der Wees et al., 2006). However, the current review did not support an immediate effect on weight-bearing DFROM, with mobilisation providing only a short-term effect. Pain and function are concluded to improve immediately in the review by Southerst et al (Southerst et al., 2015), but in our review immediate pain relief was not evident and inconclusive results were found for immediate function. When considering the short-term effects, the effectiveness of mobilisation in increasing ankle ROM was supported in the

review of Bleakely et al (Bleakley et al., 2008), and this was consistent with the findings of the current review (Bleakley et al., 2008). The review by van Ochten et al (Van Ochten et al., 2014) reported positive changes in short-term pain and function in chronic sprains, however the findings of the present review were inconclusive for both of these outcomes (Van Ochten et al., 2014). When considering the long-term effects of mobilisation, pain and function are improved according to the review by Southerst et al (Southerst et al., 2015). The findings of the current review on these outcomes were inconclusive due to lack of data. Different definitions of inclusion criteria for mobilisation techniques included within reviews (e.g., including other therapies such as home exercise or RICE treatment along with mobilisation), as well as differences in the databases searched and the periods of the data searches, are all factors contributing to these differing findings.

3.4.1 Study limitations

Limitations of this review include the wide variation in follow-up time points that we defined as short-term (from one day to less than three months). Additionally, the included studies have used a range of different mobilisation techniques and comparators. It was beyond the scope of this review to attempt to determine the independent merits of individual techniques. In particular, there may be value in analysing joint mobilisation and high velocity thrust manipulation techniques separately rather than together, but given the lack of available research at this time directly comparing these two manual therapy approaches this level of scrutiny is not possible. In addition, it was not possible to pool data to analyse the effectiveness of mobilisation for some important outcomes that were reported in single studies. Despite attempts to contact authors of included studies, data were insufficient to analyse immediate effects on PPT and short-term effects on PPT and pain intensity. Finally, no high quality evidence was found, to provide robust evidence for the effectiveness of joint mobilisation for ankle sprains.

Further research is required to determine the mechanisms by which mobilisation improves dynamic balance and weight-bearing DFROM. Also, the long-term effects of mobilisation on ankle sprains should be further investigated using clinically relevant outcomes.

3.5 Conclusion

Joint mobilisation appears to clinically benefit individuals with chronic ankle sprains, improving dynamic balance immediately and weight-bearing DFROM in the short-term. It is unlikely to have an immediate effect on static balance, pain intensity, and weight-bearing DFROM. Other clinical outcomes that have been reported following mobilisation demonstrate an inconsistent response to mobilisation, and this may be a reflection of previous study designs or of the intervention itself.

Chapter 4 Is the fibula positioned anteriorly in weight-bearing in individuals with chronic ankle instability?

This chapter has been has been submitted (10 July 2019) for publication in a peer-reviewed journal as:

Weerasekara, I., Osmotherly, P., Snodgrass, S., Tesseir, J., & Rivett, D. A. (2019). Is the fibula positioned anteriorly in weight-bearing in individuals with chronic ankle instability? (under review).

The work presented in this manuscript was completed in collaboration with the co-authors (Appendix 1). The ethics approval and supporting documents for the study reported in this chapter appear in Appendices 2, 4, 5, 9, 10, 13 and 15.

Overview

The findings of the systematic review described in Chapter 3 did not identify any studies evaluating the clinical benefits of joint mobilisation on acute ankle sprains, and all studies included in the meta-analysis were on chronic recurrent sprains or on CAI. Other than development of residual symptoms such as a subjective feeling of 'giving way' and a history of recurrent sprains, various anatomical and clinical deficits have been purported to be associated with CAI. However, these deficits have not been consistently reported in the literature and there is no agreement as to the clinical features of CAI. Chapters 4 and 5 together aim to further investigate the identifying characteristics of CAI.

The presence and direction of a fibular positional abnormality in CAI remains controversial, although this could be one of the key potential anatomical characteristics of CAI. All previous studies examining fibular position in participants with ankle injuries have used non-functional (non weight-bearing) radiographs. Therefore, the purpose of the case-control study described in this chapter was to determine any differences in normalised fibular position in a weight-bearing position between participants with CAI and healthy volunteers, and to establish diagnostic utility measures.

This chapter presents the findings of the case-control study with respect to a comparison of the position of the fibula of individuals with CAI, to those with healthy ankles. Importantly, this is the first study to assess fibular position using weight-bearing radiographs, arguably a more functional position. Utility data for normalised fibular position including a discrimination score, reliability values, specificity and sensitivity are also discussed in this chapter.

4.1 Introduction

The 'positional fault hypothesis' was proposed in 1993 by Brian Mulligan (Mulligan, 1993) to explain the benefits of MWM in the treatment of joint injuries. According to this hypothesis, joint injuries or sprains might result in a minor bony incongruence. In relation to ankle joint inversion injuries, the distal fibula may become anteriorly positioned in relation to the tibia following injury (Vicenzino et al., 2011), causing painful restrictions in physiological movement (Mulligan, 1995). The presence of such a fibular positional abnormality remains controversial. Whether an abnormally positioned fibula predisposes to injury, or whether inversion injury may result in an anteriorly positioned fibula, is yet to be examined (Hubbard et al., 2006).

The presence of an abnormally positioned fibula has been explored in acute to chronic ankle sprains and also in CAI (Berkowitz & Kim, 2004; Eren et al., 2003; Fukuhara et al., 2012; Hubbard & Hertel, 2008; Hubbard et al., 2006; Kavanagh, 1999; Kobayashi et al., 2014; Li et al., 2017; Mavi et al., 2002; Merlin et al., 2005; Scranton et al., 2000), using fibular movement measured by potentiometer, 3D CT-based bone models, and radiological investigations involving MRI, CT, fluoroscopy and X-rays. These studies utilised different indices to determine fibular position including axial malleolar index (AMI) and intra–malleolar index (IMI) (Berkowitz & Kim, 2004; Eren et al., 2003; Mavi et al., 2002; Scranton et al., 2000). Fibular position has also been measured radiographically as the distance between the most anterior margin of the tibia and the most anterior margin of the fibula on a lateral view (Hubbard et al., 2006; Vicenzino et al., 2011). Because of possible superimposition of anatomy (fibular position by the size of the tibia) on the lateral projection, a normalisation technique has been suggested which reports fibular position as a percentage of tibial width (Wikstrom, Tillman, et al., 2010).

While some studies support the presence of an alteration in fibular position in CAI, others indicate no such findings (Li et al., 2017; Wikstrom et al., 2012). Where a fibular positional anomaly has been detected, there are mixed findings regarding the direction of the displacement, including anterior (Hubbard et al., 2006), posterior (Berkowitz & Kim, 2004), lateral (Kobayashi et al., 2014) and antero-inferior (Merlin et al., 2005). These inconsistencies may possibly be due to different radiological methods or methods of measurement (Vicenzino et al., 2011) used across the various studies. It is also possible that alterations in fibular position may be better observed in functional positions (such as standing). However, all studies undertaken to date have utilised non weight-bearing positions.

It may be important to assess the position of the fibula in a weight-bearing position as this is a more functional position and likely more clinically relevant. A detectable difference in fibular position in a weight-bearing position in patients with CAI, as compared to healthy individuals could well be important in their clinical management. Demonstration of the existence of an altered fibular position could provide some support for physical interventions aimed at 'correcting a fibular positional fault', and provide a possible explanation for persistent ankle pain and dysfunction following injury in some cases (Vicenzino et al., 2011).

To date, researchers have not investigated differences in fibular position between injured and healthy individuals in a weight-bearing position, or adequately demonstrated the diagnostic utility of any imaging method of fibular position. Quantifying values for reliability, specificity and sensitivity measures, and cut-off scores for fibular positional changes that may be clinically relevant, could be of assistance in the therapeutic management of CAI.

The purpose of this study was to determine any differences in normalised fibular position in a weight-bearing position between participants with CAI and healthy volunteers, and to establish diagnostic utility measures including inter-rater and intra-rater reliability, specificity and sensitivity, and cut-off scores for normalized fibular position.

4.2 Materials and methods

Participants with CAI and healthy ankles aged 18 years and over were recruited through posted flyers, social media and using media releases (from October, 2017 to April, 2018). The volunteers with CAI were considered eligible if they satisfied the inclusion and exclusion criteria as endorsed by the IAC (Gribble et al., 2013), with the exception that the duration for undergoing at least two episodes of giving way of the ankle was changed from six to 12 months considering the seasonal nature of some sports. Individuals with a history of at least one significant ankle sprain and of the ankle giving way, and/or a recurrent sprain, and/or a feeling of instability, were included in the CAI group. Individuals were excluded from the CAI group if they reported a history of previous surgery or fractures in the lower extremity, current or previous injury to the ankle, neuromuscular disorders causing problems in the lower limb, conditions contraindicating radiological imaging, and an inability to read English. A significant ankle sprain was defined as an initial sprain occurring at least 12 months prior to study enrolment associated with inflammatory symptoms and which created at least one day of interrupted desired physical activity, with the most recent injury occurring more than three months prior to study enrolment (Gribble et al., 2013). Volunteers with healthy ankles (age and gender matched) were accepted into the study if they had no prior history of ankle problems, lower limb surgery or other ankle treatment, and reported no current pain or problems in or around the ankle while performing daily activities. The same exclusion criteria were applied for volunteers with both CAI and healthy ankles. The University of Newcastle Human Research Ethics Committee granted ethical approval for the study (H-2017-0217). Informed consent was obtained from all participants and the rights of them were protected.

An X-ray (55k Vp and 2.1 mAs) was taken of the most affected ankle of the individuals with CAI to measure the fibular position with respect to the tibia in weight-bearing (neutral ankle in standing position). The most affected side was determined using the CAIT (Hiller et al., 2006). CAIT is a self-reported questionnaire used to determine the presence of ankle instability, with a cut-off score \leq 24 indicating an individual's ankle is unstable (Gribble et al., 2013). If participants presented with a similar score for both ankles, they were asked to verbally nominate the most problematic ankle. If they were unable to distinguish between ankles, the dominant side was imaged. Volunteers with healthy ankles were age and gender matched to the participants with CAI, with the same ankle (left, right) imaged.

Each participant was instructed to stand on the affected foot with the knee slightly flexed representing mid-stance of the gait cycle, with the foot of the other leg hanging relaxed. Approximately 2cm distance was maintained between the imaged foot and the parallel image receptor. All participants were provided with similar instructions and the position of the leg was monitored throughout the procedure. The X-ray was repeated once if any leg rotation was observed on imaging. The central ray was directed to the base of the metatarsals and perpendicular to the image receptor. The focal-film distance was set to 110cm. To ensure the lateral X-ray was acceptable, each individual radiograph was viewed immediately after exposure, while the participant maintained the same position of the imaged foot but could take some weight back on the other foot if desired. As the image was visible after approximately 10 seconds, this allowed for adjustments to the participant's position and a new image to be taken if required. As the radiograph was a lateral view, it was considered acceptable if superimposition of the talar domes was present, thus permitting a clear view of an open tibio-talar joint and allowing planned measurements to be undertaken. The participant was permitted to lightly hold the body of the X-ray machine for balance, if necessary (Weerasekara, Osmotherly, Snodgrass, Tesseir, et al., 2019).

Merge PACS software (Merge Health Care, 2012) was used to digitally record all radiographic images. The distance between the anterior edges of the distal fibula and the distal tibia was recorded as the fibular position (Figure 4.1) (Hubbard et al., 2006). Measures of fibular position were normalised to tibial width, and the tibial width defined as the maximum distance between the anterior and posterior tibial processes within the distal epiphysis, in a lateral view X-ray image. Normalisation of measurements was undertaken to minimise the potential error that could be introduced due to anatomical variation between individuals. All the radiographs were performed by a registered diagnostic radiographer (JT) with over 30 years of clinical experience.

A random selection of 24 ankle X-rays (CAI=12, healthy=12) were used to determine the reliability measures. These radiographs were independently evaluated by two assessors (a registered physical therapist [SW], and a registered radiographer [JT]) for reliability measures. Both the assessors were experienced in the use of the Merge PACS software (Merge Health Care, 2012). Each tester

individually completed measurements on one occasion to determine inter-rater reliability measures, and one tester (IW) undertook further blinded measurements on a second occasion (two weeks later) for intra-rater reliability measures. The standard error of measurement (SEM) was calculated for each reliability measure.



Figure 4.1 Measurement of normalised fibular position (normalised to tibial width; normalised fibular position, 17.39% =(fibular position [A], 6.8mm / maximum tibial width [B], 39.1mm) x 100. (A)=distance between the anterior edge of the distal fibula and the anterior edge of the distal tibia. (B)=maximum distance between the anterior tibial process and the posterior tibial process within the distal epiphysis

4.2.1 Analysis

The primary outcome measures of fibular position (Hubbard et al., 2006) and self-reported function (FAAM subscales) (Croy et al., 2012) were used in sample size calculations. The largest sample size estimation (MD=2.5, SD=3.4) (Hubbard et al., 2006) resulted in a minimum sample size of 33 participants per group allowing for a 10% for data loss, alpha of 0.05, and achieving power of 0.80.

Baseline data were assessed for normality using the Shapiro-Wilk normality test. Descriptive statistics were calculated for all variables. Comparison of means between the normal and CAI groups were analysed using independent t-tests. The effect size was calculated using Cohen's d for normalised fibular position.

Relative reliability of the measures was assessed using ICC _{2,1} and 95% CIs. The SEM was calculated to assess measurement precision using the formula, SEM=SD × $\sqrt{(1 - ICC)}$, with SD representing the standard deviation of the measure (Powden et al., 2015).

The ability of the measures of normalised fibular position to identify individuals with CAI was calculated using the area under the curve (AUC) for receiver operating characteristic (ROC) curves and the 95% CIs of the AUC. A traditional academic point scale was utilised to determine the accuracy of the AUC and the 95% CIs of the AUC for discriminating between healthy participants and those with CAI (0.90 - 1.00, excellent; 0.80 - 0.89, good; 0.70 - 0.79, acceptable; 0.60 - 0.69, poor; and 0.00 - 0.59, failure) (Wikstrom et al., 2012). In addition, a cut-off score (for discriminating individuals with CAI from healthy individuals) with resulting likelihood ratios and 95% CIs were quantified for normalised fibular position. The cut-off score was determined by calculating the Youden index (J) for fibular position along the ROC curve, with the largest J value representing the cut-off score (Beninato et al., 2014; Wikstrom et al., 2012). Further, positive (LR+) and negative (LR–) likelihood ratios were produced (LR+=(sensitivity /(1 – specificity), LR-=(1–sensitivity)/specificity) (Beninato et al., 2014; Powers et al., 2017). All statistical analyses were performed using IMB SPSS (Version 23.0, Armonk, NY, IBM Corp).

4.3 Results

Sixty-six participants were included in the study after assessing eligibility. An outline of the recruitment process is provided in Figure 4.2. Common reasons for exclusion included previous ankle injury (n=13 in CAI group) and unable to match with CAI group (n=6 in healthy group).



Figure 4.2 Overview of the recruitment process

4.3.1 Participant characteristics

Data from 33 individuals with chronic unstable ankles (11 males; mean (SD) age 30.24 ± 8.70 years; mean (SD) BMI 25.30 \pm 4.46 kgm⁻²) and 33 age and gender matched healthy controls (11 males; mean (SD) age 30.45 ± 8.71 years; mean (SD) BMI 23.47 \pm 3.51 kgm⁻²) were included in the study. Participant characteristics are presented in Table 4.1.

The height, weight and BMI measures were similar in both groups. The CAI group was functionally impaired with a significantly different mean FAAM score (Table 4.1). Further, in the CAI group, the dominant leg was unstable in 20 participants and 26 of them had sprained their other ankle even though it was not reported by them as unstable.

4.3.2 Comparison of fibular position

No significant difference between the two groups was noted for non-normalised fibular position (MD=-1.24mm [95% CI=-2.85 - 0.37], p=0.13). However, there was a significant difference in normalised fibular position (MD=-3.01% [95% CI=-5.83--0.19], p=0.04) (Table 4.2). The effect size was d=0.53 for normalised fibular position.

4.3.3 ROC analysis

The AUC for the ROC curve (Figure 4.3) was not significant indicating that the fibular position values cannot independently predict having CAI. The largest Youden index value indicated that a normalised fibular position value of greater than 27% was the cut-off score to distinguish the CAI group (Table 4.3). Moderate sensitivity (70%) and fair specificity (55%) were calculated at this cut-

off for normalised fibular position. Resulting positive and negative likelihood ratios were 1.53 and 0.56 respectively.

Baseline data	Healthy	95%	CAI (n=33)	95%	p value
	(n=33)	CI	mean (SD)	CI	
	mean (SD)				
age	30.5	27.4-33.5	30.2	27.2-33.3	0.92
	(8.7)		(8.7)		
height	169.5	1663-172.8	170.6	167.9-173.3	0.61
	(9.2)		(7.6)		
weight	67.8	63.0-72.6	73.7	68.7-78.8	0.09
	(13.6)		(14.3)		
BMI	23.5	22.2-24.7	25.3	23.7-26.9	0.07
	(3.5)		(4.5)		
CAIT score	29.0	28.5-29.6	13.8	12.3-15.3	< 0.01
affected/ matched	(1.5)		(4.3)		
side					
CAIT score	29.0	28.5-29.5	21.1	19.7-23.4	< 0.01
other side	(1.5)		(6.7)		
FAAM ADL	99.9	99.8-100.0	89.1	86.1-92.1	< 0.01
score	(0.4)		(8.5)		
FAAM	99.4	98.6-100.1	70.1	65.7-74.4	< 0.01

Table 4.1 Characteristics of the participants

Sports score	(2.2)	(2.3)
--------------	-------	-------

ADL, activities of daily living; BMI, body mass index; CAI, chronic ankle instability; CAIT, Cumberland Ankle Instability Tool; FAAM, Foot and Ankle Ability Measure; SD, standard

deviation

Table 4.2 Comparison of fibular position between individuals with unstable ankles and healthy ankles

Fibular position	Healthy	CAI (n=33)	p value
(n=:	(11-33)	mean (SD)	(MD, 95% CI)
	mean (SD)		
non-normalised	11.54	12.78 (3.63)	0.13
fibular position (mm)	(2.89)		(1.24, -0.37 to 2.85)
normalised fibular	26.69	29.7 (6.55)	0.04
position (%)	(4.78)		(3.01, 0.19 to 5.8)

CAI, chronic ankle instability; MD, mean difference; SD, standard deviation



Figure 4.3 Receiver operating characteristic (ROC) curve used to calculate the area under the curve and cut-off score for normalised fibular position

Fibular positon	Non-normalised fibular	Normalised fibular
	position	position
Area under the curve (95% CI)	0.60 (0.46-0.73)	0.63 (0.50-0.77)
p value	0.18	0.07
Sensitivity (95% CI)	0.49	0.70
Specificity (95% CI)	0.73	0.55

Table 4.3 Characteristics of receiver operating characteristic (ROC) curve for fibular position

Positive likelihood ratio (LR ⁺)	1.78	1.53
Negative likelihood ratio (LR ⁻)	0.71	0.56
Cut–off value	12.70	26.81

CI, confidence interval; LR, likelihood ratio

4.3.4 Reliability of the weight-bearing X-Ray measures of fibular position

Inter-rater reliability and intra-rater reliability were excellent for all the fibular position measures with high ICC _(2,1) values, and low SEM values (Table 4.4).

Table 4.4 Intra-class correlation coefficient (ICC_{2,1}) and standard error of measurement (SEM) with 95% CI for inter-rater and intra-rater reliability of measurements of fibular position

Measurement	Intra-rater reliability	Inter-rater reliability
Non-normalised fibular position		
ICC (95% CI)	1.00 (0.99-1.00)	0.98(0.96-0.99)
SEM	0.56	0.97
95% CI (1.96 x SEM)	1.10	1.91
Tibial width		
ICC (95% CI)	1.00 (1.00)	0.98(0.96-0.99)

SEM	0.57	
95% CI (1.96 x SEM)	1.12	1.12
		2.19
Normalised fibular position		
ICC (95% CI)	0.99(0.98-1.00)	0.98(0.96-0.99)
SEM	1.33	1.75
95% CI (1.96 x SEM)	2.60	3.43

ICC, intra-class correlation coefficient; SEM, standard error of measurement; CI, confidence interval

4.5 Discussion

The present study of fibular position in individuals with CAI is the first to be conducted with participants weight-bearing in standing. A significant difference in fibular position in individuals with CAI was found compared to individuals with healthy ankles when normalised for tibial width, which may suggest CAI is associated with a slight anterior position of the fibula in a weight-bearing position. Although the magnitude of detected difference in the current study were small, the effect size was moderate (d=0.53). This indicates that individuals with CAI have a more anteriorly displaced fibula (0.53 standard deviations higher) than individuals with healthy ankles (Cohen, 1988). Clinically, this may be a factor contributing to the persistence of pain, ROM restriction and other symptoms and signs of CAI. The finding of an anteriorly positioned fibula in the present study lends support to Mulligan's hypothesis of a fibular 'positional fault' in chronic cases of injury or sprain of the ankle (Vicenzino et al., 2011). It also supports the biological rationale for MWM which attempts to correct the 'positional fault' through the pain-free application of a manual glide (usually a posteriorly directed mobilisation) of the fibula while the
patient performs the impaired active or functional movements. While the mechanism leading to the observed anterior fibular positional change in CAI is unclear, it may be associated with mechanical instability in some cases. The incidence of isolated distal tibiofibular syndesmotic sprains is reported as being between 1% and 11% (Hopkinson et al., 1990). Any change in fibular position could potentially be due to an undiagnosed syndesmotic sprain or instability of the anterior syndesmosis of the ankle in certain individuals.

In our study, the distance between the anterior margin of the tibia and the anterior margin of the fibula was measured to determine the fibular position, using lateral X-rays taken in a weightbearing position. To allow for individual morphological differences, this measurement was also normalised by calculating it as a percentage of the tibial width. No previous studies have investigated fibular position in a weight-bearing position to facilitate comparison of our findings. However, there are a few studies investigating fibular positional changes conducted in non weight-bearing positions using radiological methods, such as fluoroscopy (Hubbard et al., 2006), X- ray generators (Wikstrom, Tillman, et al., 2010), MRI (Berkowitz & Kim, 2004; Mavi et al., 2002; Merlin et al., 2005) and CT (Berkowitz & Kim, 2004; Eren et al., 2003; Kobayashi et al., 2014; Scranton et al., 2000), and non-radiological methods, such as a potentiometer (Kavanagh, 1999). In these studies, participants were positioned in either side-lying or supine lying. In addition, the ankle was held in a variety of positions, including maximal dorsiflexion in one study (Hubbard et al., 2014; Mavi et al., 2006), neutral in the majority of studies (Eren et al., 2003; Kavanagh, 1999; Kobayashi et al., 2014; Mavi et al., 2002; Merlin et al., 2005) and was not mentioned in one study (Wikstrom, Tillman, et al., 2010).

Previous research on the existence of an altered fibular position in CAI is somewhat conflicting. Two studies found no fibular positional differences between healthy and CAI participants (Kavanagh, 1999; Wikstrom, Tillman, et al., 2010), however the others all support the existence of a fibular positional change. However the direction of this change varies between these studies, with most finding either an anteriorly positioned (Hubbard et al., 2006; Kavanagh, 1999; Mavi et al., 2002) (consistent with the present study) or a posteriorly positioned fibula (Berkowitz & Kim, 2004; Eren et al., 2003; Scranton et al., 2000). Further, an antero-inferior displacement (Merlin et al., 2005) and a laterally positioned fibula have also been observed (Kobayashi et al., 2014). Some of the differences in findings across studies may be due to the use of different measures. The AMI (Berkowitz & Kim, 2004; Eren et al., 2003; Scranton et al., 2000) is most commonly utilised in MRI studies where it shows the position of the fibula in relation to the tibia at the ankle mortise (Berkowitz & Kim, 2004). In radiographic studies, the 'distance between anterior margin of the tibia and anterior margin of the fibula' is commonly used (Hubbard et al., 2006; Mavi et al., 2002; Wikstrom, Tillman, et al., 2010). The studies of Kobayashi et al (Kobayashi et al., 2014) and Kavanagah et al (Kavanagh, 1999) used other customised measures. Moreover, normalisation of fibular position was used in only one previous study because of possible superimposition of fibular position by the size of the tibia on the lateral projection (Wikstrom, Tillman, et al., 2010), making this just the second study to do so.

Consistent with previous research using fluoroscopic non weight-bearing images (Hubbard et al., 2006), weight-bearing lateral X-ray measures in the current study demonstrated excellent reliability for both normalised and non-normalised fibular positon. A weight-bearing lateral X-ray is arguably more functional and more clinically relevant than other available methods for measuring fibular position. Further, the low SEM values (Table 4.4) indicate good precision for estimation of both

normalised and non-normalised fibular position. On the other hand, only moderate sensitivity (70%) and fair specificity (55%) were observed at the cut-off value for identifying the individuals with CAI using normalised fibular position ($\geq 27\%$) in the current findings. These sensitivity and specificity values demonstrate a moderate ability of the normalised fibular position to predict the presence of CAI when the test is positive ($\geq 27\%$), and a minimal ability to exclude the presence of CAI when the test is negative (Lalkhen & Mccluskey, 2008).

Given that the mechanism of CAI may sometimes be multifactorial (mechanical and/or functional), it is perhaps not surprising that the normalised fibular position measure discriminated between the two groups (CAI and healthy) with a point estimate accuracy of 0.63 for the AUC (95% CI 0.50-0.77, p=0.07) (Table 4.3). Thus, only 63% of participants were correctly classified according to CAI status using 'normalised fibular position' as the predictor alone. Moreover, the lack of statistical significance of the AUC could indicate that this finding is simply due to chance. The positive and negative likelihood ratios of 1.53 and 0.56 (respectively) do not provide a strong indication for ruling CAI in or out in these individuals, further suggesting that normalised fibular position alone is not an appropriate sole predictor of CAI (Jaeschke et al., 1994; Wright et al., 2014). Therefore, utilisation of normalised fibular position measures combined with other clinical findings, might be helpful in identifying individuals with CAI.

Future research should investigate whether lateral X-ray findings of fibular position may effectively be used in determining the appropriate application of MWM techniques in rehabilitating CAI. Potentially the practitioner could choose whether or not to apply MWM as a treatment for the client based on the amount of displacement. Further, they could also potentially choose the direction of the MWM glide so as to reverse the evident direction of positional displacement. Whether MWM treatment actually reverses fibular displacement on imaging may also merit investigation using the method employed in this study. However, the clinical utility of this method should consider the ease of obtaining an accurate X-ray image and the time spent on taking the fibular position measures. Finally, it would be interesting to measure fibular displacement of the other ankle of individuals with CAI to explore whether the fibular position was present before the injury or if it resulted from the injury.

First, although every effort was made to minimise axial rotation of the lower limb during radiographs, possible rotation of the tibia and fibula may have introduced minor variability in the fibular positional measurements. Second, a priori sample size estimation was not conducted for the ROC analysis because it was a planned but secondary objective. Third, the study was not powered to explore weight-bearing fibular position in subgroups (mechanical and functional instability) of CAI, and thus the potential for heterogeneity of the study sample may have influenced the results. Fourth, the assessor was not blinded to group allocation during the measurements, however fibular position measures were obtained using a computer program. Fifth, in positioning the knee of the non-imaged limb (slightly flexed, representing mid-stance of the gait cycle) some variability may have been introduced, and in future researchers should consider standardising this. Finally, the case-control design of the study precludes any cause and effect relationship being ascribed to the fibular positional difference.

As a conclusion, an anteriorly positioned distal fibula in relation to the tibia was observed in participants with CAI compared with healthy controls. This fibular positional difference may

contribute to the persistence and recurrence of pain and dysfunction in some cases of CAI. Weightbearing lateral radiographic measurements of fibular position can be performed reliably and reproducibly. However, the low specificity and sensitivity utility scores for normalised fibular position indicate that it has very little ability to predict CAI alone.

Chapter 5 Clinical characteristics of chronic ankle instability: a casecontrol study

This chapter has been submitted (03 June 2018) for publication in a peer-reviewed journal as:

Weerasekara, I., Snodgrass, S., Osmotherly, P., Tessier, J., & Rivett, D. A. (2019). Clinical characteristics of chronic ankle instability: a case-control study (under review).

The work presented in this manuscript was completed in collaboration with the co-authors (Appendix 1). The ethics approval and supporting documents for the study reported in this chapter appear in Appendices 2, 4, 5, 9, 10, 13 and 15.

Overview

The case-control study detailed in Chapter 5 is intended to address the inconsistency of evidence for common clinical characteristics of CAI, by evaluating differences in ankle dorsiflexion range, balance, self-reported function, pain and PPT between individuals with CAI and those with healthy ankles. In this regard, it is an extension of and complementary to the study reported in Chapter 4 investigating displacement of the fibula as an anatomical characteristic of CAI.

This chapter therefore presents the findings of the case-control study comparing the aforementioned proposed clinical characteristics of individuals with CAI to those with healthy ankles. In addition, the association of each of these clinical features with CAI are described in this chapter.

5.1 Introduction

The ankle is the second most commonly injured body area in sport with ankle sprain being the most common type of ankle injury (Fong et al., 2007). Up to 40% of individuals who sprain their ankles may develop CAI (Miklovic et al., 2018). Other than development of residual symptoms such as a subjective feeling of 'giving way' and recurrent sprains, other deficits have been purported to be associated with CAI, including reduced ankle dorsiflexion range, impaired balance and compromised function (Hertel, 2002). However, these deficits have not been consistently reported in the literature and clinical relationships are yet to be investigated between these deficits. Moreover, little research has been conducted to identify pain (Adal et al., 2019) and PPT characteristics in CAI. Better understanding of the major clinical differences between individuals with CAI as compared to healthy ankles, may assist in the design of more effective treatment and prevention programs for this condition.

Assessing pain, functional assessment tool scores (such as weight-bearing dorsiflexion range and balance tests), and self-reported functional assessment tool scores, as well as exploring the existing associations between these measures may be clinically relevant in interpreting the nature and presentation of this condition. Reduced ankle dorsiflexion range has been demonstrated to impact on dynamic balance in an anterior direction as measured by the SEBT in individuals with CAI compared with healthy controls (Hoch, Staton, et al., 2012). The SEBT has been recommended as a highly representative non-instrumental measure of dynamic balance for physically active people (Gribble et al., 2012). However, anterior direction balance deficits identified by this test can be due to ankle DFROM restriction rather than any balance impairment. Previously reported positive correlations between dorsiflexion range restriction and anterior direction balance deficit measured using the SEBT (Basnett et al., 2013; Terada et al., 2014), has raised questions regarding the

appropriateness of SEBT as a representative test for dynamic balance alone. Notably, CoP during performance of the on the SEBT has been quantified in individuals with acute first-time lateral ankle sprains (Doherty et al., 2015) and individuals with recurrent sprains in previous studies (Nakagawa & Hoffman, 2004). Thus exploring the performance of individuals with CAI on the SEBT combined with CoP measures, may be of greater value in quantifying dynamic balance impairments than with the SEBT alone. Further, static balance deficits associated with CAI have not been consistently reported in the previous literature (De Vries et al., 2010; Nakagawa & Hoffman, 2004).

Both self-report and objective measures of pain characteristics are similarly important in understanding the clinical presentation of CAI. The importance of investigating the impact of pain on individuals with CAI was highlighted in a recent study with 1147 participants, in which 60% reported ankle pain (Adal et al., 2019). Self-perceived pain in CAI has been previously reported to not correlate with SEBT performance (Terada et al., 2014), and it would be clinically interesting to further explore whether pain actually limits the functional ability of individuals with CAI. Further, no previous study has investigated peripheral and central sensitisation pain mechanisms in CAI despite it being documented in acute ankle sprains (Ramiro-Gonzalez et al., 2012). Clinical recognition of central sensitisation in patients with persistent musculoskeletal pain is essential in appropriately applying the growing knowledge of pain neurophysiology in clinical practice (Nijs et al., 2010).

This study aimed to explore whether people with CAI have specific deficits in terms of ankle dorsiflexion range, balance, self-reported function, pain and PPT compared to healthy matched

controls. Further, the study aimed to investigate whether any of these measures may be associated with the presence of CAI.

5.2 Methods

5.2.1 Participants

From October 2017 to April 2018, participants with CAI (n=33) and age, gender and side matched individuals with healthy ankles (n=33) were recruited into this case-control study. Recruitment was from a university and the general community in the Newcastle area of New South Wales, Australia through flyers, social media, media release and web posts. All volunteer participants over 18 years of age with CAI were considered eligible if they satisfied the inclusion and exclusion criteria as endorsed by the IAC, one with modification; the period for experiencing at least two episodes of giving way was changed to 12 months instead of six to account for the seasonal nature of some sports (Figure 5.1) (Gribble et al., 2013; Weerasekara, Osmotherly, Snodgrass, et al., 2019b). Volunteers with healthy ankles were eligible for the study provided they had not experienced previous ankle problems, lower limb surgeries, or other treatment for an ankle problem, and had no current pain or problems in or around the ankle while performing daily activities. The same exclusion criteria applied to CAI participants were also applied to the volunteers with healthy ankles (Weerasekara, Osmotherly, Snodgrass, et al., 2019b). Written informed consent was obtained from all participants and ethics approval (H-2017-0217) was obtained from the Human Research Ethics Committee of The University of Newcastle, Australia.

Inclusion criteria A history of at least one significant ankle sprain; Initial sprain must have occurred at least 12 months prior to study enrolment Was associated with inflammatory symptoms _ Created at least one interrupted day of desired physical activity The most recent injury must have occurred more than three months prior to study enrolment A history of the previously injured ankle joint 'giving way' and/or recurrent sprain and/or 'feelings of instability' Participants should report at least two episodes of giving way in the 12 months prior to study enrolment Self-reported ankle instability should be confirmed with the Cumberland Ankle Instability Tool (CAIT) (≤ 24) **Exclusion criteria** 1. A history of previous surgeries to the musculoskeletal structures (i.e., bones, joint structures, nerves) in either lower extremity 2. A history of a fracture in either lower extremity requiring realignment 3. Acute injury to musculoskeletal structures of other joints of the lower extremity in the previous three months that impacted joint integrity and function (i.e., sprains, fractures), resulting in at least one interrupted day of desired physical activity 4. Any previous injuries still causing problems to lower limb (i.e., balance issues) 5. Any neuromuscular disorder that may affect lower limb (i.e., muscle weakness, balance issues) 6. Have conditions for which radiological imaging is contraindicated (e.g., pregnancy) 7. Inability to read English Figure 5.1: Inclusion and exclusion criteria for individuals with CAI (Gribble et al., 2013)

In cases of bilateral CAI, the CAIT was utilised to screen for the most affected side. Greater instability is represented by a lesser score in this questionnaire (Hiller et al., 2006). The cut-off score of ≤ 24 was taken to indicate a self-reported unstable ankle (Gribble et al., 2013). If both ankles had a similar score, the participant was asked to select their most problematic ankle. If they were unable to distinguish between their ankles, the dominant side was selected for the measurement. Individuals with healthy ankles were age and gender matched to individuals with CAI. Clinical characteristics including ankle DFROM, balance (static and dynamic), and PPT were measured by the same researcher throughout the study. The sequence of testing was randomised and the sound side (or less unstable) was measured first in every measure in individuals with CAI.

5.2.3 Procedures

Static balance:

The participant was instructed to stand barefoot on the marked centre of the force plate (KISTLER 9260AA6, Winterthur, Switzerland), assuming a single-leg stance position. Each participant was then asked to bend the non-stance leg slightly at the hip, with the knee bent to approximately 90 degrees, representing mid-stance of the gait cycle. Their arms were crossed on their chest during both eyes close and open balance tests. The participant was asked to remain as still as possible for 10 seconds in both static balance tests (eyes open, eyes close), and maintaining a fixed gaze on a cross marked on the wall six metres in front of them during the eyes open balance test. If the participant was unable to stand for 10 seconds, the total standing time completed was recorded. Averaged CoP data for sway velocity (absolute mean value of the instantaneous velocity of the CoP in a given direction during a given period) (Ross et al., 2009) and sway area (rectangular area defined by the maximum anterior, posterior, medial, and lateral sways during a given time) (Ross et

al., 2009) per second were recorded and used in the analysis. Prior to each data collection session the force plate was calibrated automatically, and CoP data were acquired at 100Hz.

Dynamic balance (SEBT):

The participant was instructed to stand on a stable base of support on the stance limb in the middle of the testing grid on a force plate. The force plate was calibrated prior to each data collection session. Then the participant was asked to reach as far as possible with the reaching limb along each of the lines marked on the platform in anterior, postero-medial and postero-lateral directions. They were then asked to tap the line with the great toe of the reaching foot, and to return the reaching limb to the starting position in the centre of the grid. The great toe of the reaching foot was painted with a washable paint, so the researcher could easily measure how far the limb had reached. The shortest distance from the centre to the paint print was taken as the measure. The balance time was the time spent from the moment the participant lifted their limb until they returned to the starting position (Jaber et al., 2018). If the participant lifted or shifted any part of the foot of the stance limb during the trial, the trial was not considered complete. After performing a maximum of four non-recorded trials for familiarisation, one trial per direction was recorded (Pionnier et al., 2016). Normalised SEBT values were taken as the ratio between excursion distance and the participant's leg length (the distance between the anterior superior iliac spine and the ipsilateral medial malleolus) (Pionnier et al., 2016). Sway velocity (Ross et al., 2009) measures to quantify spatio-temporal parameters were acquired at 100Hz, under the foot during single-leg stance (Pionnier et al., 2016).

Self-reported function:

The FAAM (Martin et al., 2005) and FAOS questionnaires (Roos et al., 2001) for self-reported foot and ankle function were used to describe the level of disability of the cohort, as endorsed by the IAC (Gribble et al., 2013). The participants were asked to answer the questions with responses that most closely described their condition during the last week. The FAAM consists of two sub-scales to assess function during usual ADL and during sports-related activities. The FAOS consists of five sub-scales to assess symptoms, pain, function related to ADL, function related to sports and recreational activities, and QOL.

Pain:

Pain intensity during rest was assessed using a VAS which consisted of a 100mm horizontal line, with 'no pain' anchoring the left side of the line and 'worst possible pain' anchoring the right. The validity of the VAS for detecting changes in pain intensity has been supported by numerous studies (Ferreira-Valente et al., 2011; Price et al., 1983).

Pressure pain threshold (PPT):

PPT measurements were obtained in each leg from two points around the ankle (anterior to the lateral malleolus over the ATFL, inferior to the medial malleolus over the deltoid ligament) and one remote point over the proximal third of the tibialis anterior muscle belly (Ramiro-Gonzalez et al., 2012). A Freedom Tracker hand-held algometer (JTECH Medical, Salt Lake City, UT, USA) was used during measurement, and a probe with a contact surface of 1cm² was placed perpendicular to the skin and pressure applied at 40kPa/s. The least affected/not affected leg was

measured first, and the order of the measurement sites was randomised. The tested leg was rested on the bed, and calibration was carried out prior to each data collection session. The participant was instructed to indicate when the feeling of the stimulus changed from 'pressure only' to 'pressure and discomfort' by pressing an indicator switch. This process was repeated three consecutive times at each measurement site and a 10s rest time was allowed between each measure. Pressure algometry is considered a stable and reliable measure of PPT (Frank et al., 2013).

Weight-bearing dorsiflexion range of motion (DFROM):

The participant was instructed to perform a weight-bearing lunge by bringing their knee towards the wall until they lightly touched the wall, whilst maintaining their heel in contact with the floor. Then the participant was asked to move their foot away from the wall in 1cm increments until the heel no longer maintained contact with the floor or the knee was no longer in contact with the wall. Participants were instructed to keep their hands on their hips and to avoid twisting their trunk during the test. Maximal dorsiflexion was measured by the greatest distance between the great toe and wall with the participant's knee maintaining contact with the wall (Gilbreath et al., 2014). The same procedure was followed for the opposite leg. The weight-bearing lunge test is reported as a reliable measure (Powden et al., 2015).

5.2.3 Analysis

The sample size was calculated using an alpha significance level of 0.05 and a power of 0.80, in relation to the largest estimation resulted for primary outcome measures (MD=2.5, SD=3.4) of self-reported function and fibular position (Hubbard et al., 2006). This resulted in a minimum sample size of 33 per group, while allowing for a 10% dropout rate. Statistical analyses were

performed using SPSS Statistics for Windows (Version 23.0, Armonk, NY, IBM Corp). Descriptive statistics were calculated for all variables.

Mean comparison

Data were assessed for normality both visually and using the Shapiro-Wilk test. The comparison of means between normal and CAI groups were analysed using independent t-tests, and comparison of means between the CAI affected side and the other side were analysed using paired t-tests. When the condition of normality was not met, non parametric equivalent tests were applied. The level of significance was set at $\alpha = 0.05$.

Regression analysis

The variables were selected for the regression model according to the assumptions of regression analysis, and highly correlated (over 0.7-1) variables were removed from the model. Univariate regression was carried out between each variable (BMI, weight-bearing DFROM, normalised anterior SEBT scores, antero-posterior sway velocity recorded during both postero-lateral and postero-medial SEBT testing, and both antero-posterior and medio-lateral eyes open balance scores, and fibular position normalised to tibial width), and those with p < 0.2 were retained for inclusion in the multivariate model. Logistic regression was carried out using a backwards-stepwise Wald selection method. The level of significance was set at $\alpha = 0.05$.

5.3 Results

A total of 66 participants (CAI=33, healthy=33) who met the eligibility criteria were included in the study (Figure 5.2).



Figure 5.2 Flow chart of participants in the selection process

5.3.1 Participant characteristics

Results of the comparison between group characteristics are presented in Table 5.1. There were 22 females and 11 males in each group. In 18 individuals with CAI, the most affected side was the right. The right side was dominant in 27 participants in the CAI group, and in 31 with healthy

ankles. Age, height, weight and BMI were normally distributed in the two groups, and no statistical differences were found in participant characteristics between the groups except for CAIT score (p < 0.01). All of the self-reported outcomes in relation to the FAAM and FAOS sub-scales were significantly lower in the CAI group compared to the healthy group (p > 0.01) (Table 5.1).

	(Groups	Difference between	
	CAI (n=33)	Healthy (n=33)	— groups	
	mean (SD)	man (SD)	MD (95% CI)	
age (years)	30.2 (8.7)	30.5 (8.7)	0.2 (4.1-4.5)	
height (cm)	170.6 (7.6)	169.5 (9.2)	-1.1 (-5.2-3.1)	
weight (kg)	73.7 (14.3)	67.8 (13.6)	-5.9 (-12.8-0.9)	
BMI (kgm ⁻²)	25.3 (4.5)	23.5 (3.5)	-1.8 (-3.8-0.1)	
CAIT score	13.8 (4.30)	29.0 (1.5)	15.2 (13.6-16.8)*	
FAOS symptoms	66.2 (15.8)	96.8 (4.2)	30.6 (24.9-36.4)*	
FAOS pain	77.3 (14.1)	99.5 (0.8)	22.4 (17.4-27.4)*	
FAOS sports	59.2 (22.0)	99.6 (2.0)	37.2 (29.9-44.5)*	
FAOS ADL	88.7 (11.5)	99.9 (0.3)	11.3 (7.3-15.3)*	
FAOS QOL	56.4 (14.9)	99.4 (1.8)	43.0 (37.8-48.3)*	
FAAM ADL	89.1 (8.5)	99.9 (0.4)	10.9 (7.9-13.9)*	
FAAM sports	70.1 (12.3)	99.4 (2.2)	29.3 (24.9-33.6)*	

Table 5.1 Characteristics of the participants in each group

ADL, activities of daily living; BMI, body mass index; CAI, chronic ankle instability; CAIT, Cumberland ankle instability tool; FAAM, foot and ankle ability measure; FAOS, foot and ankle outcome score; QOL, quality of life

*Significant difference (p < 0.05)

5.3.2 Comparison of the characteristics

A significantly lower normalised anterior SEBT score was observed in the affected ankle of the CAI group compared to the healthy group (MD=-5.8, 95% CI -9.5 to -2.1) and also compared to the other ankle of the CAI group (MD=-2.5, 95% CI -4.5 to -0.5) (Table 5.2). The normalised postero-medial SEBT score was significantly less in the affected ankle of participants with CAI, compared to their other ankle (MD=-4.0, 95% CI -6.8 to -1.2). All the other SEBT measures were found to be non-significant (Table 5.2). A significantly lower weight-bearing DFROM was found between the ankles of individuals with CAI and between the two groups (MD=-2.6, 95% CI -4.2 to -1.0; MD=-1.0, 95% CI -1.7 to -0.4, respectively) (Table 5.2). Pain during rest was significantly higher in the affected ankle of CAI participants compared to healthy individuals (p<0.01), as well as compared to the other ankle of the CAI participants (p=0.06) (Table 5.2). None of the PPT values were significantly different, except for a lower threshold over the lateral ligament on the affected side compared to the other ankle of CAI participants (MD=-2.5, 95% CI -4.8 to -0.2) (Table 5.2).

In the SEBT tests, greater postero-medial sway velocity in the postero-medial direction were observed in the CAI group (MD 7.8, 95% CI 0.2 to 15.5), with all the other sway velocity measures (in any group comparison) non-significant (Table 5.2). When comparing CoP data for the eyes

open standing balance tests of the healthy and CAI groups, all were significantly higher in both the antero-posterior and medio-lateral directions for sway velocity (p=0.02 and p<0.01, respectively), and medio-lateral sway area per second (p<0.01) in individuals with CAI (Table 5.2). However, no differences were found between the affected and the other ankle for CAI participants (Table 5.2). In eyes close balance, only antero-posterior sway area per second was significantly higher in the CAI group (Table 5.2). Interestingly, antero-posterior and medio-lateral sway velocity were significantly greater in the other ankle of CAI participants compared to the affected ankle (p=0.03 and p=0.02, respectively) (Table 5.2).

Table 5.2 Comparison of characteristics between individuals with CAI and healthy individuals, and between the affected ankle and other ankle of CAI participants

Clinical characteristic	CAI group (affected ankle) (n=33) mean (SD)	CAI group (other ankle) (n=33) mean (SD)	Healthy group (n=33)	MD (CAI affected ankle - healthy group) (p value [95% CI])	MD (CAI affected ankle - CAI other ankle) (p value [95% CI])
			mean (SD)		
dynamic balance (n	on-normalised) (ci	n)			
SEBT anterior	54.3	56.5	58.0	-3.7 (0.02,	-2.1 (0.02,
	(7.0)	(6.5)	(5.6)	-6.8 to -0.6)*	-3.9 to -0.3)*
SEBT postero-	73.8	75.8	75.5	-1.7 (0.43,	-1.9 (0.12,
lateral	(9.2)	(9.2)	(8.0)	-5.9 to 2.5)	-4.4 to 0.5)
SEBT postero-	59.4	62.9	60.8	-1.4 (0.66,	-3.4 (0.01,
medial	(13.1)	(12.4)	(13.3)	-7.9 to 5.1)	-6.0 to -0.9)*
dynamic balance (r	ormalised)				
SEBT anterior	61.1	63.7	66.9	-5.8 (> 0.01,	-2.5 (0.02,

	(8.0)	(8.5)	(6.9)	-9.5 to -2.1)*	-4.5 to -0.5)*
SEBT postero-	83.0	85.3 (11.2)	87.2	-4.1 (0.11,	-2.3 (0.09,
lateral	(10.5)		(9.9)	-9.1 to 0.9)	-5.0 to 0.4)
SEBT postero-	66.9	70.9 (15.1)	70.1	-3.3 (0.39,	-4.0 (0.01,
medial	(15.2)		(15.4)	10.8 to 4.2)	-6.8 to -1.2)*
weight-bearing	9.4	10.4	12.0	-2.6 (> 0.01,	-1.0 (> 0.01,
DFROM (cm)	(3.9)	(3.4)	(2.4)	-4.2 to -1.0)*	-1.7 to -0.4)*
PPT (kPa)					
Tibialis anterior	27.4	27.2	26.9	p=0.33#	0.2 (0.85,
	(13.8)	(10.1)	(18.5)		-2.1 to 2.6)
medial ligament	27.9	30.4	26.5	p=0.21#	-1.7 (0.38,
	(9.4)	(11.4)	(12.6)		-5.6 to 2.2)
lateral ligament	24.9	26.58	24.0	p=0.43#	-2.5 (0.04,
	(10.0)	(10.6)	(12.4)		-4.8 to -0.2)*
pain at rest (mm)	7.8	4.8	0	p<0.01#*	p=0.06#
	(11.1)	(10.4)	(0)		

SEBT tests: sway velocity measures (mm/s)						
anterior direction-	78.7	74.4	80.6	-1.9 (0.74,	4.4 (0.18,	
sway velocity AP	(23.9)	(20.8)	(22.0)	-13.1 to 9.4)	-2.1 to 10.8)	
anterior direction-	54.2	55.4	53.0	1.2 (0.79,	-1.2 (0.77,	
sway velocity ML	(18.7)	(17.2)	(54.2)	-7.2 to 9.5)	-9.3 to 6.9)	
postero-medial-	73.7	77.1	69.4	4.2 (0.39,	-3.4 (0.44,	
sway velocity AP	(20.6)	(24.7)	(18.7)	-5.4 to 13.9)	-12.3 to 5.4)	
postero-medial-	62.4	58.6	54.5	7.8 (<0.05,	3.8 (0.25,	
sway velocity ML	(16.9)	(14.5)	(14.3)	0.2 to 15.5)*	-2.7 to 10.3)	
postero-lateral-	73.1	75.6	70.1	3.0 (0.55,	-2.5 (0.40,	
sway velocity AP	(23.6)	(20.1)	(16.2)	-7.0 to 13.0)	-8.4 to 3.4)	
postero-lateral-	60.1	59.2	53.6	6.4 (0.18,	0.8 (0.86,	
sway velocity ML	(21.7)	(21.4)	(16.4)	-3.0 to 15.9)	-8.6 to 0.3)	
eyes open balance -	sway velocity (mn	1/s) and sway are	ea (mm) measures			
sway velocity AP	31.8	33.0	24.8	p=0.02#*	p=0.64#	
	(16.0)	(24.0)	(6.3)			

sway velocity ML	35.5	35.9	29.6	p<0.01#*	p=0.78 [#]
	(7.7)	(11.0)	(6.5)		
sway area per	8.1	9.6	6.1	p=0.10 [#]	p=0.09 [#]
second AP	(5.3)	(7.3)	(2.2)		
sway area per	7.7	8.6	5.5	p<0.01 ^{#*}	p=0.81 [#]
second ML	(3.5)	(5.7)	(1.8)		
eyes close balance -	sway velocity (mm	/s) and sway are	ea (mm) measures		
sway velocity AP	62.5	74.1	57.0	p=0.18 [#]	p=0.03 ^{#*}
	(20.0)	(33.4)	(20.0)		
sway velocity ML	63.4	70.8	60.5	p=0.73#	p=0.02 ^{#*}
	(17.4)	(14.1)	(17.8)		
sway area per	13.0	14.1	11.2	p=0.04#*	p=0.69 [#]
second AP	(4.0)	(7.7)	(4.3)		
sway area per	10.77	13.6	10.3	p=0.73#	p=0.07#
second ML	(4.22)	(8.9)	(3.3)		

[#]non-parametric tests were used (Wilcoxon Signed Ranks Test for paired comparison, Mann Whitney Test for independent group comparison)

*Significant difference (p < 0.05)

AP, antero-posterior; CAI, chronic ankle instability; ML, medio-lateral; PPT, pressure pain threshold; SEBT, star excursion balance test; DFROM, dorsiflexion range of motion

5.3.3 Regression analysis

Results of correlation analyses between each variable are shown in Supplemental Digital Content 2 (Table 5.3). Significantly high correlations (r > 0.70) were found between anterior SEBT and postero-lateral SEBT, and between postero-medial SEBT and postero-lateral SEBT variables. Significantly moderate correlations (0.50 < r < 0.70) were found between antero-posterior sway velocity of postero-medial SEBT and antero-posterior sway velocity of postero-lateral SEBT, and antero-posterior sway velocity of postero-lateral SEBT, and even posterior sway velocity of postero-lateral SEBT and antero-posterior sway velocity of postero-lateral SEBT, antero posterior sway velocity of eyes open standing balance and medio-lateral sway velocity of eyes open standing balance; weight-bearing DFROM and anterior SEBT; and anterior SEBT and postero-medial SEBT variables (Table 5.3).

Variable 1	Variable 2	r	p value
BMI	 weight-bearing DFROM 	-0.26	0.03
	- PL SEBT: ML sway velocity	0.35	< 0.01
	- eyes open: AP sway velocity	0.35	< 0.01
	- eyes open: ML sway velocity	0.29	0.02
weight-bearing	 normalised SEBT anterior 	0.55	< 0.01
DFROM	 normalised SEBT PL 	0.37	< 0.01
	 normalised SEBT PM 	0.33	0.01
normalised SEBT	 normalised SEBT PL 	0.76	< 0.01
anterior	 normalised SEBT PM 	0.58	< 0.01
	- PM SEBT: AP sway velocity	0.25	< 0.05
	- eyes open: AP sway velocity	-0.31	0.01
	- eyes open: ML sway velocity	-0.32	0.01

Table 5.3: Bivariate correlations identified as being statistically significant (p<.05) for the CAI affected ankle

normalised SEBT PL	-	normalised SEBT PM	0.76	< 0.01
	-	eyes open: AP sway velocity	-0.29	0.02
	-	eyes open: ML sway velocity	-0.31	0.01
normalised SEBT PM	-	eyes open: AP sway velocity	-0.32	0.01
anterior SEBT: AP	-	anterior SEBT: ML sway velocity	0.31	0.01
sway velocity	-	PL SEBT: AP sway velocity	0.48	< 0.01
	-	PL SEBT: ML sway velocity	0.31	0.01
	-	PM SEBT: AP sway velocity	0.42	< 0.01
	-	PM SEBT: ML sway velocity	0.25	< 0.05
	-	eyes open: AP sway velocity	0.41	< 0.01
	-	eyes close: AP sway velocity	0.25	<0.05
antonion SEDT: MI		avag alagat ML away valagity	0.25	0.02
	-	fibular position	0.28	0.02
sway velocity	-	noular position	0.44	<0.01
PM SEBT: AP sway	-	PL SEBT: AP sway velocity	0.60	< 0.01
velocity	-	PM SEBT: ML sway velocity	0.38	< 0.01
	-	eyes open: AP sway velocity	0.28	0.02
	-	eyes close: AP sway velocity	0.32	0.01
PM SEBT: ML sway	-	PL SEBT: AP sway velocity	0.29	0.02
velocity	-	PL SEBT: ML sway velocity	0.43	< 0.01
	-	eyes open: AP sway velocity	0.29	0.02
PL SEBT: AP sway	-	PL SEBT: ML sway velocity	0.41	< 0.01
velocity	-	eyes close: AP sway velocity	0.36	< 0.01
PL SEBT: ML sway	-	eyes open: AP sway velocity	0.27	0.03
velocity				
eyes open: AP sway	-	eyes close: AP sway velocity	0.41	< 0.01
velocity	-	eyes open: ML sway velocity	0.56	< 0.01
eyes open: ML sway	-	eyes close : ML sway velocity	0.27	0.03
velocity	_	fibular position	0.28	0.03

eyes close: AP sway	-	eyes open: ML sway velocity	0.43	< 0.01
velocity	-	eyes close: ML sway velocity	0.58	< 0.01
eyes close: ML sway	-	fibular position	0.33	0.01
velocity				

AP, antero-posterior; BMI, body mass index; ML, medio-lateral; PL, postero-lateral; SEBT, star excursion balance test; DFROM, dorsiflexion range of motion

The logistic regression model used the predictor variables of BMI, normalised fibular position, weight-bearing DFROM, anterior excursion of the SEBT, medio-lateral sway velocity of postero-lateral and postero-medial SEBT excursions, and antero-posterior and medio-lateral sway velocity of eyes open standing balance (Table 5.4). The final regression model demonstrated that reduced weight-bearing DFROM is associated with having CAI (OR 0.72, 95% CI 0.58 to 0.88, p<.001), and greater medio-lateral sway velocity of eyes open balance is also associated with having CAI (OR 1.17, 95% CI 1.06 to 1.29, p<.001).

Table 5.4 Results of final multivariate regression model (backwards Wald method, R^2 =.413) indicating variables predicting group membership of CAI (n=66)

Variable	Group	OR (95% CI)	p-value
BMI	Healthy	1	0.60
	CAI	1.04 (0.88-1.25)	-
weight-bearing DFROM	Healthy	1	0.03
	CAI	0.74 (0.57-0.97)	_
normalised SEBT	Healthy	1	0.80
anterior excursion	CAI	0.99 (0.89-1.09)	-
medio-lateral sway	Healthy	1	0.74
velocity of postero-	CAI	0.99 (0.95-1.03)	_
lateral excursion of			
SEBT			
medio-lateral sway	Healthy	1	0.34
velocity of postero-	CAI	1.02 (0.98-1.07)	-
medial excursion of			
SEBT			
antero-posterior sway	Healthy	1	0.62
velocity of eyes open	CAI	0.98 (0.91-1.06)	_
standing balance			
medio-lateral sway	Healthy	1	0.03
velocity of eyes open	CAI	1.17 (1.10-1.35)	-
standing balance			
normalised fibular	Healthy	1	0.63
position	CAI	1.03 (0.91-1.17)	-

BMI, body mass index; CAI, chronic ankle instability; OR, odds ratio; SEBT, star

excursion balance test; weight-bearing DFROM, weight-bearing dorsiflexion range of motion

5.4 Discussion

There are several key findings from this study. First, individuals with CAI achieved a smaller excursion in the anterior direction during the SEBT, demonstrated restricted weight-bearing DFROM, reported greater functional impairment with both self-reported functional outcome questionnaires, and indicated greater resting pain, both compared to healthy individuals and to their other leg. Second, there were no differences in PPT in any of the tested locations, other than locally over the lateral ligament in individuals with CAI as compared to their other ankle. Third, there was a clearly observed pattern in all force plate parameters of CoP during both eyes open and close balance tests, possibly due to some individuals with bilateral ankle injuries in the CAI group. Alternatively, the pattern may possibly be due to changes in the central nervous system and motor control affecting both limbs. However, analysis of balance in participants with unilateral CAI (excluding bilateral CAI participants) was not undertaken due to the small sample size for a perprotocol post hoc analysis (unilateral CAI, n = 10). Generally, this was lower CoP measures in healthy individuals compared to individuals with CAI, but not on the affected side of individuals with CAI compared to their other side. Finally, weight-bearing DFROM and eyes open standing balance medio-lateral sway velocity parameters were associated with ankle instability in this sample. Overall, these findings suggest that there are multiple differences in CAI compared to healthy ankles, including restriction in dorsiflexion, a balance deficit in anterior directions, greater resting pain, higher functional impairment, and higher sway velocity and sway area measures in eyes open balance.

Some of the SEBT reach performance differences between groups are consistent with previous research, including a reduction in anterior SEBT reach distance (Hoch, Staton, et al., 2012; Jaber et al., 2018) and postero-medial reach distance during the SEBT in individuals with CAI, as

previously reported by Plante et al (Plante & Wikstrom, 2013). Similarly, a reduced DFROM of the affected ankle in individuals with CAI has been reported in previous studies (Hoch, Staton, et al., 2012; Plante & Wikstrom, 2013; Terada et al., 2014). It is plausible that the observed reduction in SEBT reach distance may be related to the reduced available weight-bearing DFROM. Some possible reasons for the limitation in dorsiflexion may include soft tissue dysfunction or tissue adhesions around the ankle (Nyska & Mann, 2002), incongruences of the bones in and around the foot and ankle (Del Buono et al., 2013), foot or ankle muscle strength deficits (Guillén-Rogel et al., 2017), and various other biomechanical influences (Baumbach et al., 2014; Brockett & Chapman, 2016). Clinically, this range limitation may contribute to an increased risk of injury recurrence in CAI, particularly when it is associated with a balance deficit (Basnett et al., 2013).

Other studies have also found a correlation between SEBT performance and DFROM. In our study, a significant positive correlation was observed between every direction of the SEBT and weightbearing DFROM (anterior, r=0.55, p<0.01; postero-lateral, r=0.37, p<0.0; postero-medial, r=37, p=0.01) (Table 5.3). Very similar findings were noted in the study of Basnett et al (Basnett et al., 2013), and in some other studies weight-bearing DFROM was correlated with the anterior direction of the SEBT, but not with the postero-lateral or postero-medial directions (Hoch, Staton, et al., 2012; Terada et al., 2014). The correlation of weight-bearing DFROM with anterior SEBT reach distance observed in our study may therefore provide one possible reason for the significant reduction in anterior SEBT reach distance in the CAI group. Although SEBT is recommended as a non-instrumented dynamic balance test for physically active people (Gribble et al., 2012), it appears a restriction in weight-bearing DFROM could affect excursions in some or all the directions of the SEBT (Hoch, Staton, et al., 2012; Terada et al., 2014). Therefore, the SEBT may not purely assess balance because a weight-bearing DFROM restriction may impact the results. Within this context, introducing CoP measures during the SEBT may enhance the accuracy of assessing balance.

A combination of CoP measures (using force plate platforms) with the SEBT has been previously utilised in two studies investigating CAI (Jaber et al., 2018; Pionnier et al., 2016). Both of those studies reported significant differences in all sway velocity measures during the SEBT (Jaber et al., 2018; Pionnier et al., 2016), which contrasts with the results of the present study. However, there were several differences in the measures in these studies compared to the present study. In one study, balance time for CoP velocity was the time interval between toe-off and touchpoint, whereas in our study it was toe-off to the touchpoint and then back to mid-stance. In that study a significant reduction of CoP velocity of the medio-lateral component was reported for CAI participants, but no changes were found in the antero-posterior component (Pionnier et al., 2016). In the other study, single-leg stance balance was measured for about 10 seconds before resting after each excursion of the SEBT, and the sway velocity values were significantly higher in each of the three SEBT directions in individuals with CAI (Jaber et al., 2018). The non-significant findings in our study may have been because CAI participants felt less confident in moving (or perhaps experienced kinesiophobia due to fear of a recurrent sprain) when performing functional or sports tasks, as compared to healthy individuals. Thus, future research could consider assessing individuals with CAI for kinesiophobia during functional or sports tasks.

In sports medicine settings, the most frequent measures of postural-control assessment using CoP based variables have shown contradictory findings on altered postural stability in CAI (Ross et al., 2009). The most commonly assessed CoP variable is sway velocity (the absolute mean value of the

instantaneous velocity of the CoP in a given direction during a given period) (Ross et al., 2009) which indicates how quickly a person shifts and is able to control the CoP (Childs, 2016). In our study, both sway velocity parameters and ML sway area were significantly greater in CAI participants in eyes open balance when compared between groups, but not when compared between ankles in CAI participants. In eyes close balance, sway velocity was not significantly different between groups, but was significantly different between ankles within the CAI participants. The unaffected side exhibited greater sway velocity than the affected side. Some previous studies similarly report a significantly greater sway velocity measures in antero-posterior (Wikstrom, Fournier, et al., 2010) and medio-lateral directions (Chen et al., 2014; Wikstrom, Fournier, et al., 2010), whereas in contrast to our findings, no significant difference in postural sway has also been reported (Bernier et al., 1997; Chen et al., 2014). Small sample sizes (Bernier et al., 1997) and the use of CAI subgroups (functionally (Bernier et al., 1997) or mechanically unstable ankle (Chen et al., 2014)) may account for some of the differences in findings between studies.

The significant difference within groups in sway velocity measures for eyes close balance suggests that proprioception is impaired as a part of compensation mechanism where contralateral ankle counteract the balance insufficiency. Impaired anterior-posterior measures may indicating the increased antero-posterior movements due to lack of movement control by injured lateral ligament (Chen et al., 2014). In addition, having a more antero-lateral sway measures likely explains, the closed-packed ankle joints assist the unstable ankle to maintain a more stable position (Pope et al., 2011). Also, the greater sway velocity of the unaffected side suggests that local impairments in proprioception may be related to the original injury to local tissues that contain proprioceptive organs. This bilaterally poor proprioception may be centrally driven, however due to the small sample size was unable to be tested. Notably, poor postural control has previously been found to be

related to an increased risk of ankle sprain (Chen et al., 2014; McKeon & Hertel, 2008), and this may therefore be one of the possible causes for the common occurrence of bilateral instability.

The presence of ankle pain has been found to be common (>60%, n=689) in patients with CAI (Adal et al., 2019). In our sample, 60.1% (n=20) of individuals with CAI reported a mean (SD) resting ankle pain of 7.8 (11.1) mm on the VAS scale. Therefore, pain-relieving interventions may sometimes be clinically indicated, along with other interventions to address impairments associated with CAI. Persistent or chronic pain is one common complaint of CAI, (Konradsen et al., 2002) and it was expected that the presence of central sensitisation would be indicated by generalised lower PPT (Ramiro-Gonzalez et al., 2012). So far as we are aware this is the first study to assess the role of central sensitisation in CAI using PPT, and our study found no evidence suggestive of central sensitisation, other than a local pressure pain hypersensitivity over the lateral ligament. This is consistent with some peripheral musculoskeletal conditions such as acute ankle sprain ((Ramiro-Gonzalez et al., 2012) and shoulder pain (Coronado et al., 2014), however differs to other musculoskeletal conditions such as knee osteoarthritis (Fingleton et al., 2015) and chronic low back pain (Corrêa et al., 2015). Possibly the occurrence of recurrent acute sprains within the course of CAI is key. Potentially, activated nociceptors due to recurrent injuries, ongoing inflammation and mechanical irritancy in CAI can prolong the pain experience and may perpetuate peripheral sensitisation, without any central sensitisation (Chimenti, Frey-Law, & Sluka, 2018). The conflicting findings of bilateral changes in CoP measures and reduced PPT measures over the affected area suggest the need for assessing more than one aspect of sensorimotor function, and also for investigating specific subgroups of CAI.

Patient-reported outcome tools could provide clinicians with some insight into the disability experienced by patients with CAI. The two self-reported functional questionnaires endorsed by the IAC and used in the present study (Gribble et al., 2013), demonstrated significantly lower selfreported function in participants with CAI, consistent with the findings of the systematic review by Houstan et al (Houston et al., 2015). Several clinical factors may have contributed to the decreased function in the CAI group, including impaired weight-bearing DFROM, impaired balance measures and pain. This suggests the need for a comprehensive treatment plan to address the various clinical characteristics of this multifactorial problem.

One limitation of this study is that for the eyes close balance test, the participant was asked to remain as still as possible for 10 seconds. However, some participants were unable to stand for this long, therefore averaged CoP data were used in analysis to maintain consistency. A further limitation is that some participants with CAI reported bilateral ankle injuries, and this may have affected comparisons undertaken within the CAI group. Additionally, differences between subgroups of CAI were not able to be investigated due to an insufficient sample size required for subgroup analysis. Finally, the assessor was not blinded to the group allocation, however all the measures except for DFROM were obtained using a computer program.

5.5 Conclusion

The findings of this study support the multi-faceted nature of CAI, including reduced weightbearing DFROM, impaired static and dynamic balance, decreased self-reported function, and increased resting pain as compared to individuals with healthy ankles. However, there was no difference in PPT in individuals with CAI, suggesting central sensitisation may not be a component in their chronic pain. Further, impaired weight-bearing DFROM and eyes open medio-lateral sway velocity are the variables most associated with having CAI.

Chapter 6 Effects of mobilisation with movement (MWM) on anatomical and clinical characteristics of chronic ankle instability: a randomised controlled trial protocol.

This chapter has been published as:

Weerasekara, I., Osmotherly, P. G., Snodgrass, S. J., Tessier, J., & Rivett, D. A. (2019). Effects of mobilisation with movement (MWM) on anatomical and clinical characteristics of chronic ankle instability: a randomised controlled trial protocol. *BMC Musculoskeletal Disorders, 20*(1), 75. doi:10.1186/s12891-019-2447-x

The work presented in this manuscript was completed in collaboration with the co-authors (Appendix 1). The ethics approval, clinical trial registration and supporting documents for the study reported in this chapter appear in Appendices 3, 6, 8, 11, 12 and 14-16.

Overview

The systematic review presented in Chapter 3 suggested that MWM and associated fibular repositioning taping may produce greater clinical benefits in chronic ankle sprains than other common joint mobilisation techniques. Further, there is a notable paucity of RCTs investigating the efficacy of manual therapies for CAI, especially in the long-term. The RCT protocol described in this chapter is designed to assess the immediate to long-term effects of MWM (with fibular repositioning taping) on the anatomical and clinical characteristics of CAI, as identified within the studies described in Chapters 4 and 5.
This chapter outlines the study design for the RCT that is the basis for the final study of the thesis. This RCT is designed to compare the effects of MWM (with fibular repositioning taping) to a placebo intervention on fibular position and clinical outcomes including ankle dorsiflexion range, balance, self-reported function, pain and PPT immediately post-intervention, at 12 weeks and at 12 months.

6.1 Introduction

Up to 40% of patients with an initial ankle sprain develop CAI, which is frequently associated with recurrent sprains and persistent pain (Doherty et al., 2014; Hershkovich et al., 2015). A recurrent subjective perception of the ankle joint 'giving way' is clinically indicative of CAI (Gigi et al., 2015), which is defined as "repetitive bouts of lateral ankle instability resulting in numerous ankle sprains" (Hertel, 2002). Clinical management of CAI often involves balance and sport-related activity training (Martin et al., 2013). In a recent meta-analysis, preliminary evidence was also found supporting joint mobilisation as a clinically effective intervention in improving dynamic balance and DFROM in CAI (Weerasekara et al., 2018).

Several ankle joint mobilisation procedures have been developed and described by renowned manual therapists such as Geoffrey Maitland, Freddy Kaltenborn and Brian Mulligan, and are commonly used in rehabilitation (Pettman, 2007). These procedures are applied to a joint, either in the form of non-thrust passive joint mobilisations, high velocity thrust manipulation, or MWM. MWM is defined as the application of a sustained passive accessory movement to a joint while the patient actively performs a task/movement that was previously identified as being painful or limited (Vicenzino et al., 2011). After manual application of MWM, tape is applied to help maintain the glide and corrected fibular position (Mau & Baker, 2014). Biomechanically and neurophysiologically mediated mechanisms have been proposed to explain how these joint mobilisation procedures may work (Mccarthy et al., 2015; Vicenzino et al., 2011). The proposed neurophysiological mechanisms are based on animal (Chien et al., 2009) and human experiments (Sterling et al., 2001) related to pain science and motor systems (Vicenzino et al., 2011). These have shown that joint manual therapy techniques including MWM, activate a descending pain inhibitory pathway which is non-opioid mediated (Vicenzino et al., 2011). One proposed

biomechanical mechanism relates to a reduction of an entrapped meniscoid or synovial fringe by a specifically directed MWM glide particularly in those instances where only one repetition is required to bring about a substantial and long lasting effect (Vicenzino et al., 2011).

Our recent systematic review and meta-analysis identified greater effects for MWM and Mulligan taping compared to Maitland joint mobilisation (with and without traction) and joint thrust manipulation (Beazell et al., 2012; Weerasekara et al., 2018). DFROM and self-reported instability were some of the outcomes for which there was evidence of improvement from MWM, although the long-term benefits were unclear (Cruz-Diaz et al., 2015; Reid et al., 2007; Vicenzino et al., 2006). Most of the previous studies on chronic ankle sprains have applied MWM to the talocrural joint (Cruz-Diaz et al., 2015; Gilbreath et al., 2014; Marron-Gomez et al., 2015; Reid et al., 2007; Vicenzino et al., 2006), and few studies have applied MWM taping (Hopper et al., 2009; Someeh et al., 2015a, 2015b). However Mulligan proposes that an anterior fibular positional fault commonly results from ankle inversion sprains, and that a MWM using a posterior glide of the fibula to correct this should be trialled after 48 hours following such an injury (Hing et al., 2015). Patients with recurrent ankle sprains may also benefit from this MWM treatment combined with taping aimed at maintaining the posterior fibula glide, with reportedly less 'giving way' and greater confidence in using the ankle in patients with functional instability and pain (Hing et al., 2015). Therefore this study is designed to evaluate the clinical benefits of the fibular posterior glide MWM with Mulligan taping, and whether it corrects any demonstrable positional fault which may exist. The prevalence of pain in people with CAI is high (60.1%) (Adal et al., 2019) and to our knowledge no studies have assessed the effect of MWM on pain. In addition, the present study will assess the effects of MWM on PPT in CAI. The presence of localised peripheral sensitisation has been previously identified in acute inversion ankle sprains (Ramiro-Gonzalez et al., 2012) and in

sub-acute ankle sprains (Collins et al., 2004). Balance impairments in CAI are frequently reported in the literature and MWM has been found to be effective immediately after application, but there is presently insufficient research to determine the short-term benefits of MWM for balance impairments (Weerasekara et al., 2018). The present study plans to address this deficiency in the literature as well.

A positional fault at the inferior tibio-fibular joint, is one arthrokinematic abnormality proposed to be related to persistent/recurrent symptoms and repetitive ankle sprains in CAI (Hertel, 2002). In the case of an ankle joint sprain, Mulligan suggests that the distal fibula is 'mal-positioned' anteriorly (anterior positional fault) following an inversion injury and that chronicity may result if this remains uncorrected (Hing et al., 2015; Mulligan, 1995). Preliminary evidence for such an anterior fibular positional fault was identified in Hubbard et al's study of individuals with CAI (Hubbard et al., 2006). However it is unclear whether ankle instability caused the anterior fibular position or whether the fault itself was actually the predisposing factor to re-injury. Also, the clinical importance of an anterior fibular positional fault in relation to other potential contributors to be further investigated (Cruz-Diaz et al., 2015; Hoch et al., 2014). It has been proposed by Mulligan in his positional fault hypothesis, that MWM effects an immediate and lasting improvement by correcting a minor bony incongruity which is the source of the patient's presenting problem (Hing et al., 2015). However, the capacity of MWM to reverse any positional fault remains unclear and further studies are required to assess the effectiveness of this technique.

The objective of this study is to determine the effect of MWM on anatomical and clinical characteristics of CAI, and to determine the long-term effectiveness of this treatment.

The specific aims of the study are therefore:

To evaluate the effectiveness of MWM on clinically relevant outcomes, including patient-reported outcomes (dorsiflexion range, pain intensity, self-reported function, PPT, static and dynamic balance), including long lasting benefits assessed at 12 months post treatment.

To assess the effect of MWM on changing the fibular position relative to the position of the tibia in CAI.

6.2 Methods

6.2.1 Design

This randomised controlled study has been registered in the Australian New Zealand Clinical Trial Registry (ANZCTR) and ethical approval has been granted by the Human Research Ethics Committee of The University of Newcastle, Australia (H-2017-0354). Informed consent will be obtained in writing from all participants.

6.2.2 Participants

Participants aged 18 years or over will be recruited from the general community in the Newcastle area of New South Wales, Australia through flyers posted on noticeboards in shopping centres, the University of Newcastle main campus, and various other public places. Recruitment advertising will also be via University of Newcastle social media channels. Volunteers with CAI will be accepted into the study if they satisfy the inclusion and exclusion criteria as endorsed by the IAC (Gribble et al., 2013), except the time period for experiencing at least two episodes of giving way is changed from six months to 12 months to account for the seasonal nature of some sports (Table 6.1).

Data collection will be carried out at the physiotherapy and radiography research laboratories of the School of Health Sciences, The University of Newcastle, Australia.

6.2.3 Procedure

This trial will adopt a pragmatic RCT design to allow for real world application of MWM in a randomised setting (Alsop et al., 2016). This design has been used by previously published trials of manual therapy to better reflect routine clinical practice (Deyle et al., 2016; Groeneweg et al., 2017; Reid et al., 2015). It enhances the external validity, but still controls for threats to internal validity.

The initial screening will be performed over the telephone after the potential participant contacts the research team. The screening questions are to determine if the potential participant meets some of the inclusion/exclusion criteria (Table 6.1). If a potential participant appears eligible following the telephone interview, further screening will be carried out using two standardised questionnaires: the FAAM (Martin et al., 2005), which measures function, and the CAIT (Hiller et al., 2006), which measures ankle instability. A link to access these questionnaires on the Qualtrics online survey platform (Qualtrics, Provo, Utah, USA) will be sent to the potential participant, along with the participant information statement and the consent form, through an email. Once the potential participant returns their completed forms, their final eligibility will be determined

according to their scores (FAAM: ADL subscale <90%, sport subscale <80%; CAIT \leq 24) on the two screening questionnaires. The participant will then be contacted to schedule an appointment for data collection.

Inclusion criteria	Exclusion criteria
 A history of at least one significant ankle sprain; Initial sprain must have occurred at least 12 months prior to study enrolment Was associated with inflammatory symptoms 	 A history of previous surgeries to the musculoskeletal structures (i.e., bones, joint structures, nerves) in either lower extremity
 Created at least one interrupted day of desired physical activity The most recent injury must have occurred more than three months prior to study enrolment 	• A history of a fracture in either lower extremity requiring realignment
	• Acute injury to musculoskeletal structures of other joints of the lower extremity in the previous three months that impacted joint integrity and function (i.e., sprains, fractures), resulting in at least one interrupted day of desired physical activity

Table 6.1 Inclusion and exclusion criteria

	• Have conditions for which manual therapy is generally contraindicated (such as the presence of a tumour, fracture, rheumatoid arthritis, osteoporosis, prolonged history of steroid use, or severe vascular disease)
 A history of the previously injured ankle joint 'giving way' and/or recurrent sprain and/or 'feelings of instability' Participants should report at least two episodes of giving way in the 12 months prior to study enrolment 	 Have conditions for which radiological imaging is contraindicated (e.g., pregnancy)
• Self-reported ankle instability should be confirmed with the Cumberland ankle instability tool (CAIT) (≤ 24)	• Have conditions for which taping is contraindicated (e.g., allergy to strapping tape)
 General self-reported foot and ankle function questionnaire minimum score (foot and ankle ability measure (FAAM): activities of daily living (ADL) subscale <90%, sport subscale <80%) 	 Receiving concurrent treatment The most recent treatment for the ankle condition should have been received at least a week prior to study enrolment Inability to read English

Consenting participants will be randomised into two groups: an experimental group who will receive MWM, and a control group who will receive a placebo (detuned laser). All of the participants will be assessed for general joint hypermobility using the Beighton score (Smits-Engelsman et al., 2011). Mechanical ankle instability will be tested separately for each ankle using an X-ray while undergoing an anterior drawer stress test (Hubbard & Cordova, 2009). The clinically important outcome measures will include; radiological imaging of fibular position with

respect to the tibia (positional fault), DFROM, PPT, pain intensity, function, static balance and dynamic balance. These procedures and outcome measures are further explained below. The researcher who collects the clinical measurements, and the radiographer taking the X-rays, will be blinded to the participant's group (intervention) allocation. This researcher will remain blinded to the group allocation until the three month follow-up. The 12 month follow-up data will be collected using online questionnaires.

Each participant will be randomly allocated to a group to receive either MWM (active) treatment or detuned laser treatment (placebo) (these interventions are fully explained below). The participant will be blinded as to whether they are receiving an active or placebo intervention, however due to the nature of the interventions, the treating practitioner cannot be blinded. Participants will be randomly allocated to groups according to a computer generated (GraphPad Software, Inc., CA, USA) randomisation schedule by another researcher not involved in data collection using sealed opaque envelopes. Each envelope will contain a piece of paper printed with either '1' or '0', for which '1' denotes 'MWM' group and '0' denotes 'placebo' group. The treating practitioner will open the envelope and allocate the participant to a group according to the number in the envelope, and deliver the designated treatment accordingly.

Participants of both groups will attend for 2-8 treatment sessions over 4 weeks. The exact number of treatments needed to achieve an optimal change is not presently known, so a range allows the practitioner to exercise their clinical judgement. We have chosen two as the minimum number of treatments, because usual clinical practice would involve a minimum of two visits to enable re-assessment following the initial treatment (Maitland, 2005). The actual number of treatment

sessions delivered to participants in each group will be determined according to the clinical judgement of the treating practitioner, who is a registered physiotherapist with a post-professional tertiary qualification in the field of manual therapy and more than 20 years of clinical experience in treating musculoskeletal conditions. The physiotherapist will also be individually instructed in the MWM intervention by an accredited member of the Mulligan Concept Teachers Association. The physiotherapist will conclude the course of intervention if the patient reports they have fully recovered or if no further improvement is possible up to a limit of eight sessions over a four week period. The number of sessions and the duration of each session will be recorded. The same measures taken at baseline will be repeated at the conclusion of the course of intervention, within a maximum of four days after the participant's last intervention session. Further measurements will be repeated at the twelfth week with the exception of the imaging, and only self-report outcomes will be assessed at 12 months. Participants will be contacted by telephone every four weeks after finishing treatment for up to one year to record any new ankle injuries, any treatments undertaken, and their level of engagement in sport and other activities. Figure 6.1 describes the flow of the study.



Figure 6.1 Flow of the study

6.2.4 Outcome measures

6.2.4.1 Measurement of fibular positional fault from radiograph

A weight-bearing (neutral ankle in standing position) X-ray (55k Vp and 2.1 mAs) will be taken of the affected ankle of the participant. The participant will be asked to stand on the foot to be imaged on a wooden box with the knee slightly flexed to simulate mid-stance of the gait cycle, with the foot of the non-stance leg hanging in a relaxed manner. The imaged leg will be maintained ~2cm away and parallel to the image receptor. The same instructions will be given to all participants and the participant's leg position will be monitored throughout the procedure. If any leg rotation is noted on imaging, the X-ray will be redone. The central ray will be centred at the base of the metatarsals and perpendicular to the image receptor, and the focal-film distance will be set to 110cm. The participant will be allowed to hold on to body of the X-ray machine for balance if required.

Radiographic images will be digitally obtained using Merge PACSTM software (Merge Health Care, 2012). The fibular position will be measured as the distance between the anterior edge of the distal fibula and the anterior edge of the distal tibia (Hubbard et al., 2006) (Figure 6.2). The test-retest reliability ICC_{3,1} has been estimated as 0.98, with a SEM of 0.64mm for this measurement, and for intra-tester reliability, the ICC_{3,1} is 0.92 and SEM is 0.72mm (Hubbard et al., 2006).



Figure 6.2 Fibular position measurement; the distance between the anterior edge of the distal fibula and the anterior edge of the distal tibia (right ankle, 4.2mm in this image)

6.2.4.2 Weight-bearing dorsiflexion range of movement

Weight-bearing DFROM will be measured using the weight-bearing lunge test. The participant will be instructed to lunge towards the wall, touch their knee to the wall, and keep their heel in contact with the floor. Then the participant will be asked to move their foot away from the wall in 1cm increments until the heel no longer maintains contact with the floor or the knee is no longer in contact with the wall. Maximal dorsiflexion will be considered to be the greatest distance between the great toe and wall with the participant's knee maintaining contact with the wall (Gilbreath et al., 2014; Hoch, Andreatta, et al., 2012). Both inter-rater reliability (ICC=0.80-0.99) and intra-rater reliability (ICC=0.65-0.99) have been reported as high for this test (Powden et al., 2015). The same procedure will be followed for the opposite leg. Each centimetre away from the wall in the lunge test represents approximately 3.6 degrees of dorsiflexion (Bennell et al., 1998). Three test attempts will be performed and the average value will be used for analysis.

6.2.4.3 Pressure pain threshold (PPT)

PPT measurements will be obtained in each leg from two local points (to assess local hypersensitivity) and one remote body area (to assess central sensitisation), in accordance with the method used in a previous study on acute ankle sprain (Ramiro-Gonzalez et al., 2012). The points include anterior to the lateral malleolus over the ATFL, inferior to the medial malleolus over the deltoid ligament, and over the proximal third of the tibialis anterior muscle belly.

A Freedom Tracker hand-held algometer (JTECH Medical, Salt Lake City, UT, USA) will be used for measuring PPT. A probe (contact surface of 1cm²) will be placed perpendicular to the skin and pressure will be applied (40kPa/s). The participant will be asked to indicate when the feeling of the stimulus changes from 'pressure only' to 'discomfort' by pressing an indicator switch (Arendt-Nielsen et al., 2010; Rebbeck et al., 2015). This process will be performed three consecutive times and a 10 second rest period will be allowed between each set of measurements. Pressure algometry is considered a stable and reliable measure of PPT (Frank et al., 2013). The inter-rater reliability of pressure algometry has been reported to be high when the algometer pressure is applied at a consistent rate (ICC 0.91, 95% CI 0.82–0.97) (Chesterton et al., 2007).

6.2.4.4 Pain intensity

Current pain intensity will be assessed using the VAS which consists of a 100mm horizontal line, with 'no pain' anchored on the left of the line and 'worst possible pain' anchored on the right. The validity of the VAS for detecting changes in pain intensity has been supported by several studies (Ferreira-Valente et al., 2011; Price et al., 1983).

The participant will also be asked to indicate all areas in which they currently feel symptoms on a body chart. The areas in which they feel 'pain' will be shaded; the areas in which they feel 'tingling, pricking, or burning' will be circled; and the areas where they feel 'numbness, heaviness or other sensations' will be indicated on the chart by an 'N'.

6.2.4.5 Function

Self-reported physical function of the participant will be evaluated using the FAAM which consists of a 21-item ADL subscale and an 8-item sport subscale (Martin et al., 2005). This tool has been documented as a reliable, responsive and valid measure of physical function for individuals with a broad range of musculoskeletal disorders of the lower leg, foot and ankle (Martin et al., 2005). The FAOS questionnaire comprising 42 items will also be used, and has been reported as also being a reliable and valid measure (ICCs reported as 0.78, 0.86, 0.70, 0.85, 0.92 for the five subscales of pain, symptoms, ADL, sport and recreation function, and foot and ankle related QOL, respectively) (Roos et al., 2001). Further, the participant will be asked to identify up to three important activities that they are unable to perform or are having moderate to extreme difficulty performing due to pain. For each activity, the participant will be asked to rate between 0 and 10 the level of difficulty they experience performing that activity using the patient-specific functional scale (PSFS) (Stratford et al., 1995). The construct validity of the PSFS is well supported, and the test-retest reliability has been assessed as moderate to good (ICC_{2,1}=0.713) (Hefford et al., 2012).

6.2.4.6 Static balance

For static balance, the participant will stand barefoot on the centre of a force plate (KISTLER 9260AA6, Winterthur, Switzerland), assuming a standardised single-leg stance position. The participant will then be instructed to flex the other leg slightly at the hip, with the knee flexed to 90 degrees. Their arms will be crossed at their chest with each hand resting on the opposite shoulder. Measurements will be recorded with both 'eyes open' and 'eyes close'. For 'eyes open', the participant will be asked to maintain a fixed gaze on a cross marked on the wall three metres in front of them and remain as still as possible for 10 seconds (Trojian & Mckeag, 2006). For 'eyes close', the participant will be asked to close their eyes and remain as still as possible for 10 seconds (Trojian & Mckeag, 2006). If the participant is unable to stand for 10 seconds, the standing time achieved will be recorded. Only averaged CoP data including sway velocity, sway area per second, sway average amplitude and sway maximal amplitude will be used in the analysis to maintain consistency. CoP data obtained through the force platform will be acquired at 100Hz (Hopper et al., 2009).

6.2.4.7 Dynamic balance

Dynamic balance will be assessed using the SEBT which has been shown to be a reliable measure to identify dynamic balance deficits in patients with a variety of lower extremity conditions (Gribble et al., 2012). The participant will be asked to establish a stable base of support on the stance limb in the middle of the testing grid on a force plate (KISTLER 9260AA6, Winterthur, Switzerland). While standing on a single limb, the participant will be asked to reach as far as possible with the reaching limb along each line (anterior, postero-medial and postero-lateral directions), lightly touching the line with the most distal portion of the reaching foot without shifting weight or coming to rest on the foot of the reaching limb. The participant will then be asked to return the reaching limb to the starting position in the centre of the grid. If the individual lifts or shifts any part of the foot of the stance limb during the trial, the trial will be not considered as complete (Gribble et al., 2012).

After performing a maximum of four non-recorded trials for familiarisation, the next trial for each direction will be recorded for the purpose of analysis (Pionnier et al., 2016; Robinson & Gribble, 2008). Normalised SEBT values will be obtained by dividing the excursion distance by the participant's leg length (the distance between the anterior superior iliac spine and the ipsilateral medial malleolus), and then multiplying by 100 (Gribble & Hertel, 2003; Pionnier et al., 2016). Data for CoP velocity (V) to quantify spatio-temporal parameters (VCoP-total, VCoP-mediolateral, VCoP-anteroposterior) will be acquired at 100Hz, under the foot during unipodal stance (Pionnier et al., 2016).

6.2.4.8 Perceptions of the credibility of the placebo

At the data collection session at the conclusion of course of intervention, the participant will be asked to indicate which intervention (active or placebo) they thought they had received during the last four weeks and to give a confidence rating on a scale of 0–10 (with 0='not at all confident' and 10='extremely confident' (Owens & Menard, 2011)). Global perceived effect will also be measured using a self-assessment of improvement on a seven point rating scale (1=completely recovered, 2=much improved, 3=slightly improved, 4=not changed, 5=slightly worsened, 6=much worsened, 7=worse than ever) in response to the question 'How would you rate the course of your ankle complaints since the start of this study?' (Kamper et al., 2010; Van Der Windt et al., 1998).

6.2.4.9 Other measures

Telephone interviews will be conducted monthly after enrolment up to one year to record any new injuries, any treatments undertaken, and the level of engagement in sports and other activities. These variables will be used as covariates in the analysis of the 12 month follow-up data as they are possible confounders. Further, the Beighton score for hypermobility and radiographic measurement of the anterior drawer stress test will be recorded.

6.2.4.10 Beighton score

Scoring for joint hypermobility will be undertaken according to previously published methods(Smits-Engelsman et al., 2011). Each participant will be assessed in five test positions, as follows:

 Passive extension of the fifth metacarpophalangeal (MCP) joint to ≥ 90 degrees. The participant sits on a chair at the short side of the table with the shoulder in 80 degrees abduction, elbow flexed at 90 degrees, and the forearm resting on the table in a pronated position. The fifth MCP joint is passively extended by the researcher and a goniometer is used to measure the angle.

- Passive hyperextension of the elbow ≥ 10 degrees. The participant sits on a chair with the shoulder at 90 degrees of flexion and the forearm supinated. A goniometer is placed at the lateral epicondyle and the measurement is taken at maximum elbow extension.
- Passive hyperextension of the knee ≥ 10 degrees. The participant lies supine with their legs in the horizontal plane. The goniometer is placed at the lateral femoral condyle and the measurement taken at maximum knee extension.
- 4. Passive apposition of the thumb to the flexor side of the forearm. The score is positive if the entire thumb touches the flexor side of the forearm while the shoulder is flexed at 90 degrees, the elbow extended, and the forearm pronated.
- 5. Forward flexion of the trunk with the knees straight. The score is positive if the participant's hand palms rest easily on the floor.

6.2.4.11 Anterior drawer stress test with radiographic measurement

Ankle joint mechanical instability will be assessed using a lateral X-ray to measure the amount of anterior translation of the talus during a ligament stress test for each ankle. The radiograph will be taken while the ankle is undergoing a simulated anterior drawer test using 125N (Maitland, 2005). The stress radiograph will be taken with the participant in a supine lying position with the foot relaxed in a resting position and the lower leg resting on a support, with the hip and knee each flexed approximately 45 degrees. The heel will be supported on a dynamometer (Lafayette Manual Muscle Tester, Model 01165, Lafayette, IN, USA) attached to a customised device which produces the anteriorly directed force. The distal tibia will be fixed on the support using a stabilising belt placed over the distal aspect of the tibia (Seebauer et al., 2013). The central ray will be centred just

above the tip of the lateral malleolus and perpendicular to the image receptor (Johannsen, 1978). Then an anterior force of 125N will be applied (Hubbard & Cordova, 2009) to the heel of the participant at an angle of 20 degrees to the vertical plane as per recommended clinical practice (Dutton, 2017), using the customised device. The force will be monitored using the digital display of the dynamometer attached to the customised device, and the radiograph will be taken at 125N. The ankle radiograph will be taken at the focal-film distance of 110cm (Lee et al., 2013) and will set to 55 kVp and 2.1 mAs. The same procedure will be applied to the other ankle. These images will be taken at the baseline data collection session to assess mechanical instability for use in subgroup analysis.

Radiographic images will be digitally obtained using Merge PACS[™] software (Merge Health Care, 2012). Anterior translation of the talus will be measured between the posterior lip of the tibial articular surface and the nearest point of the talar dome (Figure 6.3) (Beynnon et al., 2005; Ellis et al., 2011; Lee et al., 2013; Prado et al., 2013) to identify any mechanical instability. Anterior drawer stress radiographs have been found to have moderate sensitivity, high specificity and a high positive predictive value for the evaluation of lateral ankle instability (Jolman et al., 2017). A between-limb difference of 3mm in anterior translation of the talus or an absolute value of 10mm is considered clinically significant (Jolman et al., 2017).



Figure 6.3 Anterior translation of the talus during the anterior drawer stress test is measured as the distance on X-ray from the posterior lip of the tibial joint surface to the nearest point of the talar dome (left ankle, 13.2mm in this image)

6.2.5 Application of the intervention

Participants in the experimental group will be treated with a manual MWM technique to the ankle and will be taped after the intervention using the Mulligan approach (Hing et al., 2015) to attempt to maintain the effects of the MWM. The control group will receive a detuned (inactive) therapeutic laser treatment to the lateral region of the ankle. The number of treatment sessions delivered for each participant will be based on their symptomatic response to treatment, as determined by the clinical judgement of the treating practitioner. Each participant will be asked to avoid concurrent interventions during their participation in the study.

6.2.5.1 MWM intervention

The participant's inferior tibio-fibular joint will be mobilised using Mulligan's fibula MWM for dorsiflexion and/or inversion (Hing et al., 2015). Initially, the technique will be performed in supine lying with the tibia resting on the treatment table and the foot unsupported off the table's edge. The practitioner applies a sustained pain-free anteroposterior glide with a slight cephalad and lateral inclination to the distal fibula (lateral malleolus). This glide is maintained while the participant performs active inversion or dorsiflexion (depending on which is more limited in range) to end of range. There should be no pain with the active movement. At the end of range, the practitioner will apply and sustain overpressure to the active movement for a few seconds (or the participant will do so after appropriate instruction) (Hing et al., 2015; Vicenzino et al., 2006). If dorsiflexion remains restricted, this technique can be progressed and performed in partial and/or full weight-bearing. One treatment session will consist of three to five sets, with six to ten repetitions of the active movement in each set, with the actual dosage depending on the individual response of the participant (Hing et al., 2015). Participants will receive between two to eight sessions according to the clinical reasoning of the practitioner, over a period of four weeks. After each session, Mulligan MWM taping will be applied in an attempt to replicate the sustained fibula glide (Vicenzino et al., 2011). Non-elastic tape will be applied to the ankle starting 2cm anterior to the fibula and 1cm proximal to the tip of the lateral malleolus. The tape will be spiralled obliquely around the lower limb while the fibula glide is sustained, finishing on the anterior aspect of the leg (Hing et al., 2015). The participant will be instructed to keep the tape on for 24 hours. In the case of an adverse reaction, they will be advised to remove the tape immediately and note the length of time the tape was in place.

6.2.5.2 Detuned laser intervention

The placebo intervention will be applied using a detuned therapeutic laser device (Meyer Medical Electronics, Mordialloc, Australia) for five minutes to the lateral region of the ankle, maintaining the probe 0.5-1cm away from the skin (De Bie et al., 1998; Kingsley et al., 2014; Reid et al., 2015; Teo et al., 2010). The detuned laser device will appear to function normally (both audibly and visually) to participants, but no effective emission will be produced. Both the participant and the practitioner will be required to wear protective glasses as per normal clinical practice (Cotler et al., 2015). Participants will receive two to eight treatments over four weeks, according to the clinical judgement of the treating practitioner. Detuned laser has been used in several other studies assessing manual therapy including for chronic ankle sprains. It avoids any possible direct mechanical effects to the ankle being treated and also does not activate somatosensory receptors (Irnich et al., 2001; Pellow & Brantingham, 2001; Reid et al., 2008). Further, it has been shown to

have a strong placebo effect (Reid et al., 2008). Scheduling of participant appointments will be arranged to avoid interaction between participants.

6.2.6 Sample size and data analysis

Previously published data related to the primary outcome measure of function (FAAM subscales, ADL and sports) (Gilbreath et al., 2014; Martin et al., 2005) (MCID=8.0, SD=5.68; MCID=9.0, SD=7.42 respectively) were used in sample size calculations (Gilbreath et al., 2014; Martin et al., 2005; Merlin et al., 2005). A sample size of 16 per group allowing for a 30% drop-out rate was estimated, for a minimal statistical power of 0.80 and an alpha significance level of 0.05. Secondary analysis based on the subgroups of ankle instability (mechanical, functional) will be preliminary in nature as the study is not powered for this aim. Data will be analysed using SPSS Statistics for Windows (Version 23.0, Armonk, NY, IBM Corp). Continuous data will be assessed for normality using the Shapiro-Wilk test.

Baseline comparability between groups will be analysed using the independent t-test or nonparametric equivalent, as appropriate. Linear mixed models will be used to analyse the outcome measures. For the primary outcome measure, 'function' will be the outcome variable and time, group and an interaction term for time by group will be the predictors. Any statistically significant difference in change in the outcome variable over time between the groups will be indicated by the p value for the interaction term. Pairwise Bonferroni comparisons will be performed to explore the differences between time points and between groups if a significant interaction is identified. Independent t-tests will be used to compare outcome measures between groups at each time point and the changes of the scores will be used to detect any changes in the outcomes of interest. Intention to treat (ITT) analysis will be performed with all participants allocated to each group condition to evaluate the effect of the independent variable. For missing data in ITT analysis, a participant's last observation for each outcome measure will be carried forward. The average number and the average duration of intervention sessions between groups will be compared. If any significant difference observed, secondary analysis will be taken to find any correlation between the treatment volume and the outcomes.

Additional variables recorded during monthly phone interviews (new injuries, changes in activity level, and occurrence of other treatments) will be used as covariates in the analysis of the 12 month follow-up data as they are possible confounders. Further, the Beighton score for hypermobility will also be included in regression analysis as a covariate. Radiographic measurement of the anterior drawer stress test will be used to differentiate subgroups of CAI in potential sub group analysis.

6.3 Discussion

One proposed anatomical mechanism underpinning MWM is theorised to be a correction of a minor bony incongruity (positional fault) which is at the source of the patient's presenting problem (Hing et al., 2015; Vicenzino et al., 2007). The existence of an anterior fibular 'positional fault' in individuals with CAI has some preliminary radiological support (Hubbard et al., 2006). There are also limited MRI data supporting Mulligan's positional fault hypothesis in cases of lateral ankle pain (Merlin et al., 2005), however there is no evidence to date that MWM reverses any positional anomaly. Further, should any fibular positional anomaly be reversed immediately after the application of MWM, the length of time this reversal or correction is maintained is unknown. The proposed study protocol is designed to determine the presence of any positional fault of the fibula

in CAI, and whether MWM can reverse this, and if so, whether this reversal is evident four weeks after treatment commences. Moreover, this study protocol will explore the correlation between an anatomical measure (fibular position) and other clinical outcomes (pain, function, PPT, DFROM, static and dynamic balance). Potential relationships between these measures may help explain how changing an anatomical measure may effect a clinically meaningful outcome. The effect of MWM in CAI will also be explored in relation to the presence or not of radiologically measurable mechanical instability.

There are very few clinical trials with long-term follow-ups which have assessed MWM for any musculoskeletal condition, and only one for CAI which had a six month follow-up (Cruz-Diaz et al., 2015; Weerasekara et al., 2018). The proposed study protocol is therefore the first designed to evaluate the long-term effectiveness of MWM on CAI. Moreover, the treatment effect may depend on the type of instability present (mechanical or functional), and this study protocol may evaluate the efficacy of MWM on these two subgroups of CAI. However, the subgroup analysis will be exploratory as the study was only powered to detect the main effect being the intervention on the functional outcome.

Chapter 7 Mobilisation with movement with taping is not effective in changing fibular position or improving clinical outcomes in chronic ankle instability: a randomised controlled trial

This chapter has been prepared in accordance with the requirements for manuscript submission to a peer-reviewed journal. For the purpose of this thesis, only immediate post-intervention and 12 week follow-up data are presented in this chapter as 12 month data were still being collected at the time of thesis submission. The manuscript to be submitted for publication will be finalised after the completion of the 12 month follow-up data collection and analysis.

The ethics approval, clinical trial registration and supporting documents for the study reported in this chapter appear in Appendices 3, 6, 8, 11, 12, 14 and 15.

Overview

This chapter reports some of the findings of the RCT study for which the protocol is described in Chapter 6, including at baseline, immediately post-intervention and at 12 week follow-up. Data for 12 month follow-up were incomplete as they were still being collected at the time of submission of the thesis, and therefore are not included in this chapter.

An average of six sessions of 'MWM with fibular repositioning taping', which included four sets of eight repetitions, was compared to an average of five sessions of '6 minutes of detuned placebo laser' in assessing their effects on fibular displacement and common clinical characteristics of CAI. The clinical characteristics of self-reported function, weight-bearing dorsiflexion range, static and

dynamic balance, PPT and pain intensity were assessed in both groups up to 12 weeks, and the results are described in this chapter.

7.1 Introduction

Chronic ankle instability is typically a result of recurrent ankle sprains and long-term weakness in surrounding supporting structures, particularly the lateral ligament complex (Hertel, 2002). It can present bilaterally or unilaterally and is often accompanied by weakness, chronic ankle pain, and a feeling of giving-away (Fong et al., 2007; Hiller et al., 2011). CAI may also disrupt regular daily activities, including potential lost time from sporting activities, and may result in both direct and indirect costs of management (Childs, 2016; Gribble et al., 2016b). There are two widely accepted subgroups of CAI, mechanical ankle instability and functional ankle instability (Hiller et al., 2011).

Conservative interventions are applied alone or as a combination of strategies to manage ankle instability (Ajis & Maffulli, 2006), and often include manual therapy (Weerasekara et al., 2018). Brian Mulligan's MWM and fibular repositioning taping is a frequently used manual therapy combination for managing CAI (Marron-Gomez et al., 2015; Vicenzino et al., 2006), and has been suggested to be more beneficial when compared to other forms of manual therapy, a placebo or no intervention controls (Weerasekara et al., 2018; Westad et al., 2019). The fibula has been reported to be more anteriorly positioned compared to the tibia in CAI in the weight-bearing position (Weerasekara, Osmotherly, Snodgrass, et al., 2019a). To date, there have been no studies investigating the effectiveness of Mulligan's MWM with taping on correcting the displaced fibular position in CAI.

The effects of MWM have been examined previously in individuals with CAI using DFROM (Cruz-Diaz et al., 2015; Gilbreath et al., 2014; Marron-Gomez et al., 2015; Reid et al., 2007; Vicenzino et al., 2006), dynamic balance (Cruz-Diaz et al., 2015; Gilbreath et al., 2014) and self-

reported function (Gilbreath et al., 2014). Mulligan's fibular repositioning taping has also been examined for its effects on static balance (Hopper et al., 2009), dynamic balance (Hopper et al., 2009; Someeh et al., 2015a) and function (Someeh et al., 2015b) in participants with CAI. These studies have reported inconsistent conclusions on improvement of measured outcomes. The potential to alter fibular position using MWM with fibular repositioning taping, could impact on other clinical outcomes for CAI. While there is a growing evidence for the use of MWM alone for CAI (Weerasekara et al., 2018), the evidence for the efficacy of a combination of Mulligan's MWM with taping to improve clinical outcomes is currently lacking. Further, it may be insightful to explore the effect of MWM on a group of individuals with CAI that present with a displaced fibula at the distal tibio-fibular joint.

Therefore, the research questions for this study were:

- 1. Is there a significant difference in the immediate effect of Mulligan's MWM with fibular repositioning taping and a placebo intervention on fibular position in individuals with CAI?
- 2. Is there a significant difference in the immediate or short-term effects of Mulligan's MWM with fibular repositioning taping and a placebo intervention on measures of weight-bearing DFROM, static balance, dynamic balance, PPT, pain intensity and self-reported function in individuals with CAI?
- 3. Is there any effect of Mulligan's MWM with fibular repositioning taping on a subgroup of CAI (mechanically unstable vs mechanically stable, or displaced fibula vs non-displaced fibula), using primary outcome measures (fibular position, self-reported function)?

7.2 Materials and Methods

This study has been registered in the Australian New Zealand Clinical Trials Registry (ACTRN12617001467325) and granted ethical approval by the Human Research Ethics Committee of The University of Newcastle, Australia (approval no. H-2017-0354). Written consent was obtained from all participants.

7.2.1 Design

A detailed protocol of this study has been published together with prospective registration (Weerasekara, Osmotherly, Snodgrass, et al., 2019b). The current study used a pragmatic RCT, assessor-blinded design. The participants were blinded as to whether they were receiving an active or placebo intervention, however, in keeping with any trial using manual therapy, the therapist was not able to be blinded. After completion of baseline measurements, the participants were randomly allocated to groups according to a computer-generated random number schedule (GraphPad Software, Inc., CA, USA) by a researcher not involved in data collection, using sealed opaque envelopes. The therapist allocated the participant to the group and delivered the designated intervention according to the number in the envelope.

Once the therapist concluded the course of intervention up to a maximum of eight sessions over a four week period, the baseline assessment was repeated. The endpoint of treatment was decided by the therapist based on their clinical judgement, using a pragmatic approach within this time frame (Weerasekara, Osmotherly, Snodgrass, et al., 2019b). The same assessment was repeated at the twelfth week with the exception of the radiographic measures (Figure 7.1). Further, the participants were telephoned every four weeks following the completion of the intervention to determine any

new ankle or leg injuries, any treatment undertaken, and their level of engagement in activities.



Figure 7.1 Design and flow of participants through the trial

CAI, chronic ankle instability; CAIT, Cumberland Ankle Instability Tool; DFROM, dorsiflexion range of motion; FAAM, foot and ankle ability measure; MWM, Mobilisation with movement; PPT, pressure pain threshold

7.2.2 Participants, therapists, centres

Participants were recruited via flyers and social media channels. Inclusion criteria included aged 18 years or over, a history of at least one significant ankle sprain, a history of ankle joint 'giving way'

and/or recurrent sprain and/or 'feelings of instability' confirmed with a CAIT score of ≤ 24 (Gribble et al., 2013). and cut-off scores of ADL scale < 90% and sport scale < 80% in the FAAM questionnaire (Gribble et al., 2013; Weerasekara, Osmotherly, Snodgrass, et al., 2019b). A history of previous surgeries or fractures requiring realignment in the lower extremity, an acute injury to the lower extremity in the previous three months, any condition for which manual therapy, taping or radiological imaging is generally contraindicated, receiving concurrent treatment, and an inability to read English were the exclusion criteria for the study (Gribble et al., 2013).

Questionnaires were completed through the Qualtrics online survey platform (Qualtrics, Provo, Utah, USA) and other data were collected at the physiotherapy and radiography research laboratories of The University of Newcastle, Australia.

7.2.3 Intervention

Participants in both groups received 2-8 treatments over four weeks, according to the clinical judgement of the therapist based on their symptomatic response to treatment. Both interventions involved approximately the same amount of treatment time. The participant appointments were arranged in a way to avoid interactions between the participants of the two groups. The interventions for both groups were delivered by a registered physiotherapist with a post-professional qualification in manual therapy and more than 20 years of musculoskeletal clinical experience. An honorary member of the Mulligan Concept Teachers Association instructed the physiotherapist in the MWM intervention. Including an adjunct intervention such as home exercise was avoided in the current study to increase the precision of the results in relation to the effects of joint mobilization as a unique intervention, and also to ensure the homogeneity of the sample.

7.2.3.1 Experimental group

Participants in this group were treated with a manual MWM technique to the ankle region and taped after the intervention, as described by Mulligan (Hing et al., 2015; Weerasekara, Osmotherly, Snodgrass, et al., 2019b). The inferior tibio-fibular joint of the supine lying participant was mobilised using Mulligan's fibula MWM for dorsiflexion and/or inversion (Hing et al., 2015), with the tibia resting on the treatment table and the foot unsupported off the edge of the table. A sustained pain-free anteroposterior glide with a slight cephalad and lateral inclination was then applied to the distal fibula. The participant was then asked to perform active inversion or dorsiflexion (whichever was most limited) to end of range, while the physiotherapist maintained the glide. An overpressure was applied at the end of the range and maintained by the physiotherapist for a few seconds (Hing et al., 2015; Vicenzino et al., 2006). Consistent with MWM treatment principles, this technique was performed pain free. The technique was progressed to partial and/or full weight-bearing if dorsiflexion remained restricted. Three to five sets of eight repetitions were included in an intervention session, depending on the individual response of the participant (Hing et al., 2015; Weerasekara, Osmotherly, Snodgrass, et al., 2019b).

After each session, taping was applied to maintain the pain-free fibula glide (Vicenzino et al., 2011). Non-elastic tape was spiralled obliquely starting 2cm anterior to the fibula and 1cm proximal to the tip of the lateral malleolus, while the fibula glide was manually sustained by the therapist (Hing et al., 2015; Weerasekara, Osmotherly, Snodgrass, et al., 2019b). The participant was instructed to keep the tape on for 24 hours. In the case of an adverse reaction to the tape, the

participant was asked to remove the tape immediately and note the length of time the tape had been in place.

7.2.3.2 Control group

Participants in the control group received a detuned (inactive) therapeutic laser (Meyer Medical Electronics, Mordialloc, Australia) treatment to the lateral region of the ankle. The detuned laser device functioned with visual and audible parameters, but did not produce any emission. As per normal clinical practice, both the participant and the practitioner wore protective glasses (Cotler et al., 2015; De Bie et al., 1998).

7.2.4 Outcome measures

The primary outcomes of this study were fibular position and self-reported function. In addition, there were several secondary outcomes including weight-bearing DFROM, balance (static and dynamic), PPT and pain intensity. Further, radiographic assessment of the anterior drawer stress test of the ankle and an assessment of overall joint hypermobility using the Beighton score (hypermobility defined as a Beighton scale score of ≥ 4) (Clinch et al., 2011) were undertaken. Due to ethical considerations of the radiation exposure from repeated X-rays and the related cost, we chose not to follow up fibular position measures beyond the 12 week follow-up. All other measures were followed up for 12 months. All these measurements have been explained in detail in the protocol paper of this RCT (Weerasekara, Osmotherly, Snodgrass, et al., 2019b), but are briefly outlined below. This paper reports the results obtained up to the 12 week follow-up.

7.2.4.1 Primary measures

The position of the fibula was measured with the participant standing on the affected foot, representing mid-stance of the gait cycle. A digital lateral radiographic image was obtained using Merge PACS software (Merge Health Care, 2012). The fibular position was recorded as the distance between the anterior edges of the distal fibula and the distal tibia (Hubbard et al., 2006). These measurements were then normalised for tibial width, defined as the maximum distance between the anterior and posterior tibial processes within the distal epiphysis (Weerasekara, Osmotherly, Snodgrass, et al., 2019b). A cut-off for normalised fibular position of \geq 27 was used to distinguish a displaced fibula (Weerasekara, Osmotherly, Snodgrass, et al., 2019a). All radiographic assessments were performed by the same radiographer.

Self-reported function was measured using the FAAM (Martin et al., 2005) and the FAOS (Roos et al., 2001; Weerasekara, Osmotherly, Snodgrass, et al., 2019b).

7.2.4.2 Secondary measures

Weight-bearing DFROM was measured using the weight-bearing lunge test. Maximal dorsiflexion was recorded as the greatest distance between the great toe and the wall (Gilbreath et al., 2014; Hoch, Andreatta, et al., 2012; Weerasekara, Osmotherly, Snodgrass, et al., 2019b).

Static balance measurements were recorded in 'eyes open' and 'eyes close' conditions for 10 seconds. The participant was asked to stand barefoot assuming a single-leg stance position on the centre of a force plate (KISTLER 9260AA6, Winterthur, Switzerland). Averaged CoP data during these measures, including sway velocity and sway area per second acquired at 100Hz, were used in
the analysis (Hopper et al., 2009; Weerasekara, Osmotherly, Snodgrass, et al., 2019b).

The SEBT was performed on a force plate to measure dynamic balance. The SEBT reach distances were normalised by dividing the excursion distance by the participant's leg length, and then multiplying by 100 (Gribble & Hertel, 2003; Pionnier et al., 2016; Weerasekara, Osmotherly, Snodgrass, et al., 2019b). CoP data were recorded during these excursions (Weerasekara, Osmotherly, Snodgrass, et al., 2019b).

PPT was measured using a Freedom Tracker hand-held algometer (JTECH Medical, Salt Lake City, UT, USA). Measurements were obtained at local points anterior to the lateral malleolus and inferior to the medial malleolus, and from one point remote to the ankle (over the proximal third of the tibialis anterior muscle belly) (Ramiro-Gonzalez et al., 2012; Weerasekara, Osmotherly, Snodgrass, et al., 2019b).

Each participant was asked to indicate their pain intensity at rest using a VAS, by marking a 100mm line anchored with 'no pain' at one end and 'worst possible pain' at the other (Weerasekara, Osmotherly, Snodgrass, et al., 2019b).

Ankle joint mechanical instability was measured during the anterior drawer test using a lateral Xray for each ankle. The amount of anterior translation of the talus for each ankle was measured while a 125N force was applied (Hubbard & Cordova, 2009).Where the participant could not tolerate 125N of force, the maximum force they could tolerate was applied to both ankles. Anterior translation of the talus was recorded as the distance between the posterior lip of the tibial articular surface and the nearest point of the talar-dome (Beynnon et al., 2005; Ellis et al., 2011; Lee et al., 2013; Prado et al., 2013; Weerasekara, Osmotherly, Snodgrass, et al., 2019b). A difference between the limbs of 3mm or more in anterior translation of the talus was considered indicative of a mechanically unstable ankle (Jolman et al., 2017).

7.2.5 Data analysis

The sample size calculation used both self-reported function (FAAM) and fibular position as the primary outcome measures (Gilbreath et al., 2014; Merlin et al., 2005). The largest estimation was 17 participants per group providing 80% power to detect a 3.5mm between-group difference for fibular position, assuming a standard deviation of 3.1mm (Merlin et al., 2005), a dropout rate of 30% and an alpha of 0.05.

The normal distribution of quantitative data was assessed graphically and by normality tests. Baseline comparability between groups was analysed using the independent t-test or nonparametric equivalent, as appropriate. A general linear model repeated measures with time (preintervention, immediately post-intervention, 12 week follow-up) as the within-subject factor, and group (MWM or placebo) as the between-subject factor (independent variable), was applied. An ITT analysis was utilised, with imputation using last observation carried forward (Salkind, 2010). Effect sizes for within-group and between-group were calculated. Further, changes in outcome measures were compared between groups using independent t-tests, and between time points using paired t tests. Non-parametric tests were used appropriately when assumptions were violated for parametric tests. Further, subgroups (displaced vs non-displaced fibula; mechanically unstable vs mechanically stable ankle) were compared post hoc.

Data were analysed using IMB SPSS (Version 24.0, Armonk, NY, IBM Corp) software. The statistical analysis was conducted using a 95% CI, with a p value of less than 0.05 considered statistically significant.

7.3 Results

7.3.1 Flow of participants, therapists, centres through the study

Recruitment occurred between February-July 2018. Twelve week follow-up assessments were completed in November 2018. A total of 166 participants were screened over the phone and 89 were excluded. Common reasons for exclusion were non-response, lower limb fractures and surgeries. The online screening excluded a further 43 participants, with the most common reason for exclusion being below the FAAM cut-off score required to be eligible for the study. A total of 34 participants were randomly assigned to the experimental group (n=16) or control group (n=18). However, four participants (experimental=3, control=1) did not continue the intervention and were considered as lost to follow-up (Figure 7.1). All participants received the correct intervention as per the randomisation schedule. Table 7.1 presents the baseline characteristics of the participants. No differences were found in baseline characteristics between the two groups.

Characteristic	Randomis	sed (n=34)	Lost to follow up (n=4)					
	Exp (n=16)	Con (n=18)	Exp (n=3)	Con (n=1)				
age (years), mean (SD)	33.1 (8.1)	31.9 (11.8)	29.7 (8.1)	20.0 (0)				
female, n (%)	8 (50)	13 (72.2)	2 (66.7)	0 (0)				
height (cm), mean (SD)	172.5 (8.5)	168.7 (5.2)	172.5 (1.8)	166.0 (0)				
weight (Kg), mean (SD)	75.5 (16.1)	71.5 (17.7)	72.7 (19.8)	53.0 (0)				
BMI (Kg.m ⁻²), mean (SD)	26.8 (5.2)	25.1 (5.9)	24.3 (6.2)	19.2 (0)				
other ankle sprained, n (%)	13 (81.3)	16 (88.9)	3 (75)	1 (0)				
hypermobile, n (%)	3 (18.8)	6 (33.3)	2 (50)	1 (0)				
CAIT score (affected) (SD)	8.1 (3.5)	8.8 (5.1)	9 (6.1)	7 (0)				
CAIT score (other) (SD)	19.6 (7.8)	15.3 (5.8)	27.0 (5.2)	18.0 (0)				
FAAM score (ADL) (SD)	72.9 (8.7)	73.7 (15.6)	74.3 (10.3)	62.0 (0)				
FAAM score (sports) (SD)	54.5 (12.7)	54.8 (15.8)	66.7 (8.0)	56.0 (0)				

Table 7.1 Participants characteristics

ADL, activities of daily living; BMI, body mass index; CAIT, Cumberland ankle instability tool; Con, control group (placebo, detuned laser); Exp, experimental group (MWM with taping); FAAM, foot and ankle ability measure

7.3.2 Compliance with trial method

The 13 participants who were randomized into the experimental group (MWM) each received an average of 6 (1.5 SD) sessions, consisting of an average of 4 (0.5 SD) sets of eight repetitions. The 17 participants in the placebo (detuned laser) group underwent an average of 5 (1.9 SD) sessions per participant. Both groups received the intervention for approximately the same duration (six minutes). No adverse effects were reported. The same physiotherapist was responsible for all intervention sessions for both groups.

During the 12 week follow-up, eight new ankle injuries (MWM, n=5; laser, n=3) and eight other leg injuries (MWM, n=6; laser, n=2) were reported. Compared to the levels reported during the intervention period, 14 participants (MWM, n=5; laser, n=7) increased their activity levels and eight (MWM, n=6; laser, n=2) decreased their activity levels. Eleven participants (MWM, n=3; laser, n=8) underwent other treatments for their ankle (massage therapy, n=2; non-steroidal anti-inflammatory medication, n=5; orthostatic support, n=1); physiotherapy, n=2; surgery, n=1).

7.3.3 Effect of 'Mulligan's MWM with fibular repositioning taping' intervention No differences were found between groups in all variables at baseline, except for higher weightbearing DFROM of the ankle in the placebo group (p=0.01, MD=-2.7cm, 95% CI=-4.6 to -0.8). In the group-by-time interaction, no significant difference was found in any of the outcomes (Table 7.2).

7.3.4 Effect on fibular position in individuals with CAI

No significant changes between the two groups were identified in fibular position or normalised fibular position following completion of the intervention (p=0.61, effect size=0.01; p=0.81, effect size=<0.01, respectively). No significant interaction effect and no significant effect for time was detected (Table 7.2). No difference was found in the relative change of fibular position between or within groups at any of the time points (Table 7.3).

7.3.5 Effect on common clinical measures in individuals with CAI

In the comparison between the two interventions, significant improvements were observed in the FAOS pain (p=0.02, effect size=0.18) and sports (p< 0.01, effect size=0.23) subscales in the placebo group (Table 7.2). Significant improvements were reported across time in the placebo group; for pain intensity (p=0.03, effect size=0.23), SEBT normalised postero-medial reach distance (p< 0.01, effect size=0.33), FAAM sports subscale (p=0.02, effect size=0.24), FAOS sports subscale (p=0.03, effect size=0.22), FAOS quality of life (QOL) subscale (p=0.03, effect size=0.24) and medio-lateral sway velocity of eyes close balance (p=0.03, effect size=0.23) (Table 7.2). In the group-by-time interaction, no significant difference was found in any of the outcomes (Table 7.2).

In the comparison between the two groups, a significant increase of the relative change of the antero-posterior sway velocity of anterior excursion of the SEBT was observed in the placebo group immediately after application (MD=-28.5; 95CI% 53.6 to -3.4). No difference was found in the relative change of any of the outcomes after immediate measurement (immediately post-intervention minus pre-intervention) in either of the groups. Twelve weeks after application, the

MWM group demonstrated improvement in relative change of scores of the FAOS pain and ADL subscales, as well as the medio-lateral sway velocity of anterior excursion and antero-posterior sway velocity of postero-medial excursion of the SEBT across time. The placebo group demonstrated improvements in the FAAM sports subscale and eyes close balance, both antero-posterior sway velocity and antero-posterior sway area per second (Table 7.3).

7.3.6 Effect on subgroups

None of the primary outcome measures were found to be significantly different between the subgroups with a displaced fibula (n=8) and non-displaced fibula (n=8) (Table 7.4), or between the subgroups with a mechanically unstable ankle (n=8) and mechanically stable ankle (n=8) (Table 7.5).

Variable		Descr	iptive statis	stics (me	an, SD)		Interaction	Main	Between-
	Pre-inter	vention	Immedia post-inte	tely rvention	12 wee	ek follow-up	effect (p value, effect size)	effects (p value, effect size)	subject effect (p value, effect size)
	Exp	Con	Exp	Con	Exp	Con			
fibular position (mean, SD) cm	11.5	10.6	11.9	11.3	N/A	N/A	0.83	0.37	0.61
	(3.0)	(4.6)	(3.4)	(5.3)			(< 0.01)	(0.03)	(0.01)
normalised fibular position (mean, SD) %	26.2	25.3	27.2	26.7	N/A	N/A	0.82	0.34	0.81
	(5.7)	(9.2)	(6.9)	(10.1)			(< 0.01)	(0.03)	(< 0.01)
weight-bearing DFROM (mean, SD) cm	7.1 (3.3)	9.7 (2.0)	8.3 (3.3)	9.8	8.3	9.5	0.37	0.31	0.05
				(2.4)	(2.9)	(2.5)	(0.07)	(0.08)	(0.13)
pain intensity (mean, SD)	27.1	31.4	18.8	15.9	12.9	17.8	0.10	0.03*	0.70
	(19.9)	(28.4)	(14.3)	(15.0)	(11.1)	(19.6)	(0.16)	(0.23)	(< 0.01)
PPT (mean, SD) kPa									

Table 7.2 Effect of MWM with taping on outcomes in individuals with CAI. The results of the general linear model repeated measures are presented as the mean (SD) of groups, mean (SD) difference within groups, and mean (95% CI) difference between groups (n, MWM=13; placebo=17)

over Anterior Tibialis	20.7	17.9	22.2	18.4	21.4	21.0	0.50	0.60	0.52
	(10.4)	(6.5)	(14.5)	(9.8)	(11.3)	(12.9)	(0.05)	(0.04)	(0.02)
over medial ligament	25.7	19.7	25.1	19.9	23.0	18.2	0.93	0.25	0.11
	(8.6)	(9.9)	(12.5)	(8.6)	(10.7)	(10.7)	(0.01)	(0.10)	(0.09)
over lateral ligament	22.6	15.2	23.8	18.3	22.0	19.0	0.26	0.44	0.09
	(8.7)	(5.5)	(13.1)	(7.9)	(10.0)	(9.6)	(0.10)	(0.06)	(0.10)
Non normalised dynamic balance (mean, SD)	cm								
SEBT Anterior	55.1	50.6	55.4	53.5	55.1	53.4	0.15	0.08	0.27
	(8.3)	(5.8)	(8.2)	(6.7)	(7.1)	(6.7)	(0.13)	(0.17)	(0.04)
SEBT Post-lateral	74.2	68.1	77.0	72.4	75.9	71.6	0.79	0.05	0.15
	(10.6)	(9.1)	(11.7)	(9.4)	(11.6)	(8.8)	(0.02)	(0.2)	(0.07)
SEBT Post-medial	58.4	49.0	62.2	59.5	64.6	59.1	0.32	<0.01*	0.21
	(15.5)	(11.8)	(16.7)	(11.0)	(18.6)	(13.7)	(0.08)	(0.34)	(0.06)
Normalised dynamic balance (mean, SD) %									
SEBT Anterior	62.0	59.1	62.4	62.4	62.2	62.3	0.15	0.08	0.71
	(8.5)	(6.2)	(8.1)	(7.7)	(8.6)	(7.0)	(0.13)	(0.17)	(<0.01)

SEBT Post-lateral	83.5	79.6	86.8	84.1	85.7	83.6	0.85	0.05	0.46
	(10.0)	(11.1)	(12.9)	(11.1)	(13.4)	(9.8)	(0.01)	(0.19)	(0.02)
SEBT Post-medial	65.5	58.8	70.0	69.8	72.9	69.6	0.40	<0.01*	0.51
	(15.5)	(13.6)	(18.1)	(14.1)	(20.9)	(14.6)	(0.67)	(0.33)	(0.02)
FAAM score (mean, SD)									
ADL	72.6	74.4	74.8	80.8	75.6	80.5	0.80	0.36	0.26
	(8.7)	(15.8)	(13.0)	(12.5)	(11.7)	(14.1)	(0.02)	(0.07)	(0.05)
sports	51.7	54.3	54.7	60.0	59.2	68.1	0.69	0.02*	0.24
	(12.0)	(16.3)	(18.8)	(17.4)	(18.8)	(18.8)	(0.03)	(0.24)	(0.05)
FAOS score (mean, SD)									
symptoms	58.5	63.9	59.6	71.0	58.8	70.1	0.66	0.46	0.07
	(12.3)	(19.0)	(17.6)	(13.9)	(17.5)	(19.8)	(0.03)	(0.06)	(0.11)
pain	67.7	72.5	65.0	77.8	70.8	78.8	0.26	0.16	0.02*
	(13.9)	(13.8)	(11.9)	(11.5)	(11.3)	(15.7)	(0.10)	(0.13)	(0.18)
sports	50.4	59.1	55.4	69.1	60.4	73.2	0.85	0.03*	<0.01*
	(14.9)	(16.1)	(16.3)	(15.7)	(15.2)	(20.2)	(0.01)	(0.22)	(0.23)

ADL	77.2	80.6	76.9	85.5	82.0	84.9	0.17	0.20	0.16
	(14.1)	(15.6)	(10.9)	(11.1)	(10.3)	(14.3)	(0.12)	(0.11)	(0.07)
QOL	39.9	41.9	45.7	55.5	46.2	60.3	0.43	0.03*	0.11
	(16.6)	(15.1)	(18.0)	(18.2)	(22.0)	(22.3)	(0.06)	(0.22)	(0.09)
CoP measures during SEBT									
anterior reach excursion-sway velocity AP	76.7	72.8	74.9	99.7	81.6	79.5	0.09	0.11	0.40
(mm/s)	(16.1)	(15.9)	(23.8)	(45.7)	(23.8)	(25.9)	(0.16)	(0.15)	(0.03)
anterior reach excursion-sway velocity ML	50.6	57.3	44.2	57.4	53.2	51.6	0.21	0.72	0.20
(mm/s)	(14.7)	(19.5)	(12.0)	(24.5)	(15.0)	(15.9)	(0.11)	(0.02)	(0.006)
postero-lateral reach excursion-sway velocity	65.8	69.1	68.0	86.6	81.5	84.2	0.23	0.03*	0.30
AP (mm/s)	(17.4)	(19.9)	(19.9)	(31.3)	(31.4)	(36.2)	(0.10)	(0.24)	(0.04)
postero-lateral reach excursion-sway velocity	60.1	49.7	56.6	59.1	77.2	60.0	0.46	0.31	0.37
ML (mm/s)	(26.1)	(14.7)	(21.6)	(41.5)	(48.5)	(44.1)	(0.06)	(0.08)	(0.03)
postero-medial reach excursion-sway velocity	77.0	72.0	72.8	72.9	79.8	79.0	0.89	0.20	0.80
AP (mm/s)	(30.5)	(19.5)	(17.7)	(24.3)	(15.9)	(27.0)	(0.01)	(0.11)	(<0.01)
postero-medial reach excursion-sway velocity	58.7	55.3	52.7	61.0	52.8	62.1	0.35	1.00	0.31
ML (mm/s)	(18.4)	(19.8)	(15.9)	(28.1)	(14.0)	(23.6)			

							(0.08)	(<0.01)	(0.40)
CoP measures during static balance tests									
eyes open- sway velocity AP (mm/s)	33.5	34.8	27.4	30.6	28.6	33.7	0.52	0.13	0.32
	(12.7)	(10.1)	(6.6)	(13.6)	(6.1)	(11.8)	(0.05)	(0.14)	(0.04)
eyes open-sway velocity ML (mm/s)	33.8	38.5	29.5	33.9	30.3	35.8	0.93	0.19	0.11
	(8.8)	(9.5)	(6.3)	(13.9)	(6.3)	(11.8)	(<0.01)	(0.12)	(0.09)
eyes open-sway area AP (mm/s)	7.5 (3.2)	7.3 (2.7)	18.0	6.3	6.5	6.9 (2.8)	0.52	0.34	0.29
			(43.6)	(2.3)	(1.3)		(0.05)	(0.08)	(0.04)
eyes open-sway area ML (mm/s)	7.7 (4.2)	9.6	5.6 (1.2)	5.4	5.8	5.9 (2.1)	0.85	0.26	0.65
		(13.7)		(1.4)	(1.9)		(0.01)	(0.10)	(0.01)
eyes close-sway velocity AP (mm/s)	63.7	77.8	60.6	65.0	61.0	70.0	0.39	0.10	0.23
	(10.7)	(33.9)	(5.6)	(19.9)	(12.8)	(32.5)	(0.07)	(0.16)	(0.05)
eyes close-sway velocity ML (mm/s)	72.7	83.9	63.1	69.2	64.6	96.1	0.12	0.03*	0.12
	(25.9)	(37.6)	(18.6)	(24.3)	(16.0)	(51.1)	(0.15)	(0.23)	(0.09)
eyes close-sway area AP (mm/s)	14.1	14.7	82.2	12.8	12.8	16.7	0.26	0.56	0.29
	(5.4)	(4.9)	(250.9)	(4.4)	(3.5)	(7.6)	(0.09)	(0.04)	(0.04)

eyes close-sway area ML (mm/s)	12.4	38.6	11.4	11.5	9.9	11.6	0.45	0.51	0.37
	(4.4)	(110.0)	(2.9)	(3.1)	(2.0)	(5.0)	(0.06)	(0.05)	(0.03)

ADL, activities of daily living; AP, antero-posterior; CAI, chronic ankle instability; Con, control group (placebo, detuned laser); CoP, centre of pressure; Exp, experimental group (MWM, mobilisation with movement); FAAM, foot and ankle ability measure; FAOS, foot and ankle outcome score; ML, mediolateral; N/A, not applicable; PPT, pressure pain threshold; QOL, quality of life; SEBT, star excursion balance test; weight-bearing DFROM, weightbearing dorsiflexion range of motion

*significantly different, p< 0.05

Table 7.3 Effect of MWM with taping on the changes in outcomes at each time point in individuals with CAI.

Data are presented as mean (SD) of groups, mean (SD) difference within groups, and mean (95% CI) difference between groups

Outcome	Groups						Differe	ence within te	i groups (p st)	aired t	Difference between groups (independent-t test)		
Pre- inter		ntion	Immed po interv	diately st- ention	12 week u	a follow- p	Imme po intervo Pre-inte	diately st- ention- rvention	12 week up- interv	x follow- Pre- cention	Immediately post- intervention- Pre- intervention	12 week follow- up-Pre intervention	
	Exp	Con	Exp	Con	Exp	Con	Exp	Con	Exp	Con	Exp minus	Exp minus	
	(n=16)	(n=18)	(n=13)	(n=17)	(n=13)	(n=16)	(n=13)	(n=17)	(n=13)	(n=17)	Con	Con	
fibular position (cm)	11.3 (3.1)	10.4 (4.5)	11.9 (3.4)	11.3 (5.3)	N/A	N/A	0.4 (3.5)	0.7 (2.8)	N/A	N/A	-0.3 (-2.6 to 2.1)	N/A	
normalised fibular position (%)	25.6 (5.9)	25.0 (9.1)	27.2 (6.9)	26.7 (10.1)	N/A	N/A	0.9 (7.4)	1.5 (5.8)	N/A	N/A	-0.5 (-5.5 to 4.4)	N/A	
weight-bearing DFROM (cm)	7.2 (3.2)	9.8 (2.1)	8.3 (3.3)	9.8 (2.4)	8.3 (2.9)	9.5 (2.5)	0.8 ^m (2.4)	-0.4 ^p (3.0)	0.7 ^m (1.9)	0.4 ^p (1.3)	1.7 (-0.5 to 3.9)	0.2 ^{m,p} (-1.0 to 1.5)	

pain intensity	23.4	32.0	18.8	15.9	12.9	17.8	-8.2	-14.2	15.5	-13.5	7.2	-0.6
	(19.6)	(27.7)	(14.3)	(15.0)	(11.1)	(19.6)	(18.5)	(21.6)	(25.2)	(29.4)	(-9.1 to 23.6)	(-20.5 to 19.2)
PPT (kPa)												
over tibialis	20.7	18.0	22.2	18.4	21.4	21.0	1.5	0.4	0.7	3.0	1.1	-2.3
anterior	(10.4)	(6.5)	(14.5)	(9.8)	(11.3)	(12.9)	(8.3)	(9.2)	(5.9)	(11.7)	(-5.6 to 7.7)	(-9.6 to 4.9)
over medial	25.7	19.7	25.1	19.9	23.0	18.2	-0.6	0.2	-2.7	-1.4	-0.9	-1.2
ligament	(8.6)	(9.9)	(12.5)	(8.6)	(10.7)	(10.7)	(9.4)	(6.8)	(7.7)	(9.2)	(-6.9 to 5.2)	(-7.7 to 5.3)
over lateral	20.8	15.4	23.8	18.3	22.0	19.0	1.2	3.1	-0.5	3.8	-1.9	-4.3
ligament	(9.8)	(5.4)	(13.1)	(7.9)	(10.0)	(9.6)	(11.5)	(6.4)	(6.7)	(8.5)	(-8.7 to 4.9)	(-10.2 to 1.6)
non-normalised	SEBT (ci	n)										
anterior	54.8	51.2	55.4	53.5	55.1	53.4	0.3	2.8	0.0	2.8	-2.5	-2.8
	(7.6)	(6.1)	(8.2)	(6.7)	(7.1)	(6.7)	(3.4)	(4.2)	(5.5)	(5.3)	(-5.4 to 0.4)	(-6.8 to 1.3)
postero-lateral	73.6	68.8	77.0	72.4	75.9	71.6	2.8	4.3	1.7	3.5	-1.5	-1.8
	(9.8)	(9.4)	(11.7)	(9.4)	(11.6)	(8.8)	(7.3)	(7.4)	(7.3)	(7.4)	(-7.1 to 4.0)	(-7.4 to 3.7)
postero-medial	58.1	50.1	62.2	59.5	64.6	59.1	3.8	10.5	6.2	10.1	-6.7	-3.8
	(15.5)	(12.4)	(16.7)	(11.0)	(18.6)	(13.7)	(11.4)	(11.6)	(11.0)	(15.4)	(-15.4 to 2.0)	(-14.1 to 6.5)

normalised SEB	ST (%)	-										
anterior	62.2	59.8	62.4	62.4	62.2	62.3	0.3	3.3	0.2	3.2	-3.0	-3.0
	(7.8)	(6.8)	(8.1)	(7.7)	(8.6)	(7.0)	(4.0)	(4.9)	(6.2)	(6.4)	(-6.4 to 0.4)	(-7.8 to 1.8)
postero-lateral	83.4	80.6	86.8	84.1	85.7	83.6	3.3	4.5	2.1	4.0	-1.2	-1.9
	(9.3)	(11.5)	(12.9)	(11.1)	(13.4)	(9.8)	(8.3)	(8.2)	(8.3)	(8.6)	(-7.4 to 5.0)	(-8.3 to 4.6)
postero-medial	65.7 (14.2)	60.1 (14.3)	70.0 (18.1)	69.8 (14.1)	72.9 (20.9)	69.6 (14.6)	4.7 (12.5)	10.9 (12.6)	7.4 (13.0)	10.8 (17.3)	-6.5 (-15.9 to 3.0)	-3.4 (-15.2 to 8.4)
FAAM score												
ADL	72.9	73.7	74.8	80.8	75.6	80.5	2.2	6.4	3.0	6.1	-4.2	-3.1
	(8.7)	(15.6)	(13.0)	(12.5)	(11.7)	(14.1)	(8.4)	(21.4)	(10.2)	(20.5)	(-17.1 to 8.7)	(-15.8 to 9.6)
sports	51.7	54.3(16.	54.7	60.0	59.2	68.1	3.0	5.7	7.5	13.8*	-2.7	-6.3
	(12.0)	3)	(18.8)	(17.4)	(18.8)	(18.8)	(16.6)	(24.7)	(18.3)	(27.2)	(-19.0 to 13.6)	(23.4 to 10.7)
FAOS score												
symptoms	58.5	64.3	59.6	71.0	58.8	70.1	1.1	7.1	0.3	6.3	-6.0	-6.1
	(12.3)	(18.5)	(17.6)	(13.9)	(17.5)	(19.8)	(14.8)	(19.3)	(13.2)	(26.0)	(-19.2 to 7.2)	(-21.0 to 8.9)

$\begin{array}{ c c c c c c c c c c c c c c c c c c c$													
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	pain	69.8	73.0	65.0	77.8	70.8	78.8	-2.8	5.3	3.1*	6.2	-8.0	-3.1
$\frac{11.5}{11.5} = 1.1 + 1.5 + $	P	(13.7)	(13.6)	(11.9)	(11.5)	(11.3)	(15.7)	(14.8)	(20.2)	(13.1)	(22.9)	(-21.7 to 5.6)	(-16.8 to 10.5)
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$. ,	(11.5)	. ,	. ,	. ,					
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	sports	50.4	59.1	55.4	69.1	60.4	73.2	(5.0)	10.1	10.0	14.2	-5.0	-4.2
ADL 78.1 81.0 76.9 85.5 82.0 84.9 -0.2 4.8 4.8* 4.3 -5.0 0.5 (13.5) (15.2) (10.9) (11.1) (10.3) (14.3) (14.1) (21.0) (9.9) (23.0) (-18.9 to 8.8) (-12.3 to 13 QOL 40.3 43.4 45.7 55.5 46.2 60.3 5.7 13.6 6.3 18.4 -7.9 -12.1 (15.1) (16.0) (18.0) (18.2) (22.0) (22.3) (16.8) (23.9) (18.0) (28.7) (-23.8 to 7.4) (-30.8 to 6.1) CoP measures during SEBT anterior reach excursion sway velocity 75.0 73.5 75.0 100.0 81.6 79.5 -1.7 26.8 5.0 6.7 -28.5* -1.7 AP (mm/s) (15.0) (15.7) (23.8) (45.7) (23.8) (25.9) (25.4) (38.0) (28.7) (26.5) (-53.6 to -3.4) (-22.3 to 9.5) sway velocity 50.3 55.4 44.2 57.4 53.2 51.6 -6.4 0.2 2.6* -5.7 -6.6 8.3 ML (mm/s) (14.6) (20.6) (12.0) (24.5) (15.0) (15.9) (21.7) (19.4) (21.7) (20.1) (-22.0 to 8.9) (-9.0 to 25.5) postero-lateral excursion	sperie	(14.9)	(16.1)	(16.3)	(15.7)	(15.2)	(20.2)	(21.7)	(25.5)	(16.8)	(27.7)	(22.1 + 12.0)	(-22.0 to 13.7)
ADL $\boxed{78.1 \ 81.0 \ 76.9 \ 85.5 \ 82.0 \ 84.9 \ -0.2 \ 4.8 \ 4.8 \ 4.3 \ -5.0 \ 0.5 \ (-12.3 \text{ to } 13)} \\ (13.5) \ (15.2) \ (15.2) \ (10.9) \ (11.1) \ (10.3) \ (14.3) \ (14.1) \ (21.0) \ (9.9) \ (23.0) \ (-18.9 \text{ to } 8.8) \ (-12.3 \text{ to } 13)} \\ (-12.3 \text{ to } 13) \ (14.1) \ (21.0) \ (9.9) \ (23.0) \ (-18.9 \text{ to } 8.8) \ (-12.3 \text{ to } 13)} \\ (-12.3 \text{ to } 13) \ (14.1) \ (21.0) \ (9.9) \ (23.0) \ (-18.9 \text{ to } 8.8) \ (-12.3 \text{ to } 13) \ (14.1) \ (21.0) \ (9.9) \ (23.0) \ (-18.9 \text{ to } 8.8) \ (-12.3 \text{ to } 13) \ (-12.1 \ (-30.8 \text{ to } 6.3 \ 18.4 \ -7.9 \ (-23.8 \text{ to } 7.4) \ (-30.8 \text{ to } 6.5) \ (15.1) \ (16.0) \ (18.0) \ (18.2) \ (22.0) \ (22.3) \ (16.8) \ (23.9) \ (18.0) \ (28.7) \ (-23.8 \text{ to } 7.4) \ (-30.8 \text{ to } 6.5) \ (-22.3 \text{ to } 9.5) \ (-22.0 \text{ to } 8.9) \ (-22.3 \text{ to } 9.5) \ (-22.0 \text{ to } 8.9) \ (-9.0 \text{ to } 25.5) \ (-20.1 \text{ to } 8.9) \ (-9.0 \text{ to } 25.5) \ (-9.0 \text{ to } 25.5)$												(-23.1 to 13.0)	
ADL 78.1 81.0 76.9 85.5 82.0 84.9 -0.2 4.8 4.8^* 4.3 -5.0 0.5 (13.5) (15.2) (10.9) (11.1) (10.3) (14.3) (14.1) (21.0) (9.9) (23.0) $(-18.9 to 8.8)$ $(-12.3 to 13)$ QOL 40.3 43.4 45.7 55.5 46.2 60.3 5.7 13.6 6.3 18.4 -7.9 -12.1 (15.1) (16.0) (18.0) (18.2) (22.0) (22.3) (16.8) (23.9) (18.0) (28.7) $(-23.8 to 7.4)$ $(-30.8 to 6.5)$ CoP measures during SEBT anterior reach excursion sway velocity 75.0 73.5 75.0 100.0 81.6 79.5 -1.7 26.8 5.0 6.7 -28.5^* -1.7 AP (mm/s) (15.0) (15.7) (23.8) (45.7) (23.8) (25.9) (25.4) (38.0) (28.7) (26.5) $(-53.6 to -3.4)$ $(-22.3 to 9.4)$ sway velocity 50.3 55.4 44.2 57.4 53.2 51.6 -6.4 0.2 2.6^* -5.7 -6.6 8.3 ML (mm/s) (14.6) (20.6) (12.0) (24.5) (15.0) (15.9) (21.7) (19.4) (21.7) (20.1) $(-22.0 to 8.9)$ $(-9.0 to 25.7)$ postero-lateral excursion													
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	ADL	78.1	81.0	76.9	85.5	82.0	84.9	-0.2	4.8	4.8*	4.3	-5.0	0.5
QOL 40.3 43.4 45.7 55.5 46.2 60.3 5.7 13.6 6.3 18.4 -7.9 -12.1 (15.1) (16.0) (18.0) (18.2) (22.0) (22.3) (16.8) (23.9) (18.0) (28.7) $(-23.8 \text{ to } 7.4)$ $(-30.8 \text{ to } 6.8)$ CoP measures during SEBT anterior reach excursionsway velocity 75.0 73.5 75.0 100.0 81.6 79.5 -1.7 26.8 5.0 6.7 -28.5^* -1.7 AP (mm/s) (15.0) (15.7) (23.8) (45.7) (23.8) (25.9) (25.4) (38.0) (28.7) (26.5) $(-53.6 \text{ to } -3.4)$ $(-22.3 \text{ to } 9.8)$ sway velocity 50.3 55.4 44.2 57.4 53.2 51.6 -6.4 0.2 2.6^* -5.7 -6.6 8.3 ML (mm/s) (14.6) (20.6) (12.0) (24.5) (15.0) (15.9) (21.7) (19.4) (21.7) (20.1) $(-22.0 \text{ to } 8.9)$ $(-9.0 \text{ to } 25.9)$ postero-lateral excursion		(13.5)	(15.2)	(10.9)	(11.1)	(10.3)	(14.3)	(14.1)	(21.0)	(9.9)	(23.0)	(-18.9 to 8.8)	(-12.3 to 13.4)
QOL $40.3 \\ (15.1)$ $43.4 \\ (16.0)$ $45.7 \\ (18.0)$ $55.5 \\ (18.2)$ $46.2 \\ (22.0)$ $60.3 \\ (22.3)$ $5.7 \\ (16.8)$ $13.6 \\ (23.9)$ $6.3 \\ (18.0)$ $18.4 \\ (-23.8 \text{ to } 7.4)$ $-12.1 \\ (-30.8 \text{ to } 6.7)$ CoP measures during SEBTanterior reach excursionsway velocity $75.0 \\ (15.0)$ $73.5 \\ (15.0)$ $75.0 \\ (15.7)$ $100.0 \\ (23.8)$ $81.6 \\ 79.5 \\ (25.9)$ $-1.7 \\ 26.8 \\ 5.0 \\ (28.7)$ $6.7 \\ (-28.5) \\ (-53.6 \text{ to } -3.4)$ $(-22.3 \text{ to } 9.8) \\ (-22.3 \text{ to } 9.8) \\ (-22.0 \text{ to } 8.9) \\ (-9.0 \text{ to } 25.8) \\ $													
QOL 40.3 43.4 45.7 55.5 46.2 60.3 5.7 13.6 6.3 18.4 -7.9 -12.1 (15.1) (16.0) (18.0) (18.2) (22.0) (22.3) (16.8) (23.9) (18.0) (28.7) (-23.8 to 7.4) (-30.8 to 6.2) CoP measures during SEBT anterior reach excursion sway velocity 75.0 73.5 75.0 100.0 81.6 79.5 -1.7 26.8 5.0 6.7 -28.5* -1.7 AP (mm/s) (15.0) (15.7) (23.8) (45.7) (23.8) (25.9) (25.4) (38.0) (28.7) (26.5) (-53.6 to -3.4) (-22.3 to 9.2) sway velocity 50.3 55.4 44.2 57.4 53.2 51.6 -6.4 0.2 2.6* -5.7 -6.6 8.3 ML (mm/s) (14.6) (20.6) (12.0) (24.5) (15.0) (15.9) (21.7) (19.4) (21.7) (20.1) (-22.0 to 8.9) (-9.0 to 25.2)													
(15.1) (16.0) (18.0) (18.2) (22.0) (22.3) (16.8) (23.9) (18.0) (28.7) (-23.8 to 7.4) (-30.8 to 6.7) CoP measures during SEBT anterior reach excursion sway velocity 75.0 73.5 75.0 100.0 81.6 79.5 -1.7 26.8 5.0 6.7 -28.5* -1.7 AP (mm/s) (15.0) (15.7) (23.8) (45.7) (23.8) (25.9) (25.4) (38.0) (28.7) (26.5) (-53.6 to -3.4) (-22.3 to 9.8) sway velocity 50.3 55.4 44.2 57.4 53.2 51.6 -6.4 0.2 2.6* -5.7 -6.6 8.3 ML (mm/s) (14.6) (20.6) (12.0) (24.5) (15.0) (15.9) (21.7) (19.4) (21.7) (20.1) (-22.0 to 8.9) (-9.0 to 25.9) postero-lateral excursion	QOL	40.3	43.4	45.7	55.5	46.2	60.3	5.7	13.6	6.3	18.4	-7.9	-12.1
CoP measures during SEBT anterior reach excursion sway velocity 75.0 75.0 100.0 81.6 79.5 -1.7 26.8 5.0 6.7 -28.5* -1.7 AP (mm/s) (15.0) (15.7) (23.8) (45.7) (23.8) (25.9) (25.4) (38.0) (28.7) (26.5) (-53.6 to -3.4) (-22.3 to 9.9) sway velocity 50.3 55.4 44.2 57.4 53.2 51.6 -6.4 0.2 2.6* -5.7 -6.6 8.3 ML (mm/s) (14.6) (20.6) (12.0) (24.5) (15.0) (15.9) (21.7) (19.4) (21.7) (20.1) (-22.0 to 8.9) (-9.0 to 25.9) postero-lateral excursion		(15.1)	(16.0)	(18.0)	(18.2)	(22.0)	(22.3)	(16.8)	(23.9)	(18.0)	(28.7)	(-23.8 to 7.4)	(-30.8 to 6.5)
CoP measures during SEBT anterior reach excursion sway velocity 75.0 73.5 75.0 100.0 81.6 79.5 -1.7 26.8 5.0 6.7 -28.5* -1.7 AP (mm/s) (15.0) (15.7) (23.8) (45.7) (23.8) (25.9) (25.4) (38.0) (28.7) (26.5) (-53.6 to -3.4) (-22.3 to 9.5) sway velocity 50.3 55.4 44.2 57.4 53.2 51.6 -6.4 0.2 2.6* -5.7 -6.6 8.3 ML (mm/s) (14.6) (20.6) (12.0) (24.5) (15.0) (15.9) (21.7) (19.4) (21.7) (20.1) (-22.0 to 8.9) (-9.0 to 25.5) postero-lateral excursion													
anterior reach excursion sway velocity 75.0 73.5 75.0 100.0 81.6 79.5 -1.7 26.8 5.0 6.7 -28.5* -1.7 AP (mm/s) (15.0) (15.7) (23.8) (45.7) (23.8) (25.9) (25.4) (38.0) (28.7) (26.5) (-53.6 to -3.4) (-22.3 to 9.0) sway velocity 50.3 55.4 44.2 57.4 53.2 51.6 -6.4 0.2 2.6* -5.7 -6.6 8.3 ML (mm/s) (14.6) (20.6) (12.0) (24.5) (15.0) (15.9) (21.7) (19.4) (21.7) (20.1) (-22.0 to 8.9) (-9.0 to 25.0) postero-lateral excursion 50.0	CoP measures	during SE	BT										
sway velocity 75.0 73.5 75.0 100.0 81.6 79.5 -1.7 26.8 5.0 6.7 -28.5* -1.7 AP (mm/s) (15.0) (15.7) (23.8) (45.7) (23.8) (25.9) (25.4) (38.0) (28.7) (26.5) (-53.6 to -3.4) (-22.3 to 9.9) sway velocity 50.3 55.4 44.2 57.4 53.2 51.6 -6.4 0.2 2.6* -5.7 -6.6 8.3 ML (mm/s) (14.6) (20.6) (12.0) (24.5) (15.0) (15.9) (21.7) (19.4) (21.7) (20.1) (-22.0 to 8.9) (-9.0 to 25.9) postero-lateral excursion 50.0	anterior reach ex	cursion											
AP (mm/s) (15.0) (15.7) (23.8) (45.7) (23.8) (25.9) (25.4) (38.0) (28.7) (26.5) (-53.6 to -3.4) (-22.3 to 9.5) sway velocity 50.3 55.4 44.2 57.4 53.2 51.6 -6.4 0.2 2.6* -5.7 -6.6 8.3 ML (mm/s) (14.6) (20.6) (12.0) (24.5) (15.0) (15.9) (21.7) (19.4) (21.7) (20.1) (-22.0 to 8.9) (-9.0 to 25.4) postero-lateral excursion 50.4 <td< td=""><td>sway velocity</td><td>75.0</td><td>73.5</td><td>75.0</td><td>100.0</td><td>81.6</td><td>79.5</td><td>-1.7</td><td>26.8</td><td>5.0</td><td>6.7</td><td>-28.5*</td><td>-1.7</td></td<>	sway velocity	75.0	73.5	75.0	100.0	81.6	79.5	-1.7	26.8	5.0	6.7	-28.5*	-1.7
Image: construction (15.0) (15.7) (25.3) (45.7) (25.3) (25.7) (26.3) (26.3) (25.3) (22.3 to f) sway velocity 50.3 55.4 44.2 57.4 53.2 51.6 -6.4 0.2 2.6* -5.7 -6.6 8.3 ML (mm/s) (14.6) (20.6) (12.0) (24.5) (15.0) (15.9) (21.7) (19.4) (21.7) (20.1) (-22.0 to 8.9) (-9.0 to 25.4 postero-lateral excursion $=$	AP (mm/s)	(15.0)	(15.7)	(23.8)	(45.7)	(23.8)	(25.9)	(25.4)	(38.0)	(28.7)	(26.5)	(-53.6 to -3.4)	(-223 to 92)
sway velocity 50.3 55.4 44.2 57.4 53.2 51.6 -6.4 0.2 2.6* -5.7 -6.6 8.3 ML (mm/s) (14.6) (20.6) (12.0) (24.5) (15.0) (15.9) (21.7) (19.4) (21.7) (20.1) (-22.0 to 8.9) (-9.0 to 25.4) postero-lateral excursion		(15.0)	(13.7)	(25.0)	(+3.7)	(25.0)	(23.7)	(23.4)	(30.0)	(20.7)	(20.5)	(-55.0 to -5.4)	(-22.5 to 5.2)
ML (mm/s) (14.6) (20.6) (12.0) (24.5) (15.0) (15.9) (21.7) (19.4) (21.7) (20.1) (-22.0 to 8.9) (-9.0 to 25.0 postero-lateral excursion	sway velocity	50.3	55.4	44.2	57.4	53.2	51.6	-6.4	0.2	2.6*	-5.7	-6.6	8.3
postero-lateral excursion	ML (mm/s)	(14.6)	(20.6)	(12.0)	(24.5)	(15.0)	(15.9)	(21.7)	(19.4)	(21.7)	(20.1)	(-22.0 to 8.9)	(-9.0 to 25.6)
postero-lateral excursion	` ´ ´	(1	(=0.0)	(1=.0)	(=)	(10.0)	(1017)	()	(1))	()	(=0.1)	(==::: ::: ::: ::: ::: ::: ::: ::: :::	())
	postero-lateral e	xcursion											

sway velocity	62.5	68.8	68.0	86.6	81.5	84.2	2.2	17.5	15.7	15.1	-15.3	0.7
AP (mm/s)	(17.3)	(19.4)	(14.3)	(31.3)	(31.4)	(36.2)	(20.0)	(27.5)	(29.8)	(34.0)	(-33.8 to 3.2)	(-23.7 to 25.0)
sway velocity	56.9	49.9	56.6	59.1	77.2	60.0	-3.5	9.4	17.1	10.3	-13.0	6.8
ML (mm/s)	(24.6)	(14.3)	(21.6)	(41.5)	(48.5)	(44.1)	(23.6)	(40.8)	(47.4)	(45.7)	(-39.0 to 13.1)	(-28.2 to 41.9)
postero-medial e	excursion											
sway velocity	79.8	71.2	72.8	72.9	79.8	79.0	-3.8	1.0	3.2*	7.1	-4.7	-3.9
AP (mm/s)	(29.3)	(19.2)	(17.7)	(24.3)	(15.9)	(27.0)	(28.4)	(20.6)	(31.2)	(30.7)	(-23.1 to 25.0)	(-27.3 to 19.5)
sway velocity	61.6	54.8	52.7	61.0	52.8	62.1	-5.9	5.7	-5.8	6.8	-11.7	-12.6
ML (mm/s)	(19.5)	(19.3)	(15.9)	(28.1)	(14.0)	(23.6)	(21.6)	(27.5)	(17.5)	(36.3)	(-30.6 to 7.3)	(-33.3 to 8.1)
CoP measures of	during sta	tic balanc	e tests									
eyes open												
sway velocity	31.3	34.0	27.4	30.6	28.6	33.7	-6.1	-4.2	-4.9	-1.0	-2.0	-3.9
AP (mm/s)	(12.5)	(10.3)	(6.6)	(13.6)	(6.1)	(11.85)	(10.4)	(15.6)	(8.0)	(10.5)	(-12.3 to 8.3)	(-11.1 to 3.3)
sway velocity	32.5	37.5	29.5	33.9	30.3	35.8	-4.3	-4.6	-3.5	-2.8	0.3	-0.7
ML (mm/s)	(8.5)	(10.2)	(6.3)	(13.9)	(6.3)	(11.8)	(10.3)	(14.7)	(8.7)	(10.6)	(-9.5 to 10.1)	(-8.2 to 6.7)

sway area per	7.2	7.1	18.0	6.3	6.5	6.9	-1.5 ^m	-0.9	-0.8 ^m	-0.4	-0.5	-0.6
second AP (mm/s)	(2.9)	(2.7)	(43.6)	(2.3)	(1.3)	(2.8)	(3.7)	(2.7)	(3.1)	(3.1)	(-3.1 to 2.0) ^m	(-2.9 to 1.8)
sway area per	7.2	9.4	5.6	5.4	5.8	5.9	-2.1	-0.9 ^p	-1.9	-0.4 ^p	-1.2	-1.5
second ML (mm/s)	(4.0)	(13.3)	(1.2)	(1.4)	(2.1)	(2.1)	(4.6)	(2.7)	(4.0)	(3.1)	(-4.0 to 1.6) ^p	(-4.2 to 3.4) ^p
eyes close												
sway velocity	66.8	81.1	63.1	69.2	64.6	96.1	-9.6	-14.7	-8.1	12.2*	5.1	-20.3
AP (mm/s)	(226.5)	(38.2)	(18.6)	(24.3)	(16.0)	(51.1)	(23.2)	(30.4)	(22.7)	(38.8)	(-12.3 to 8.3)	(-43.5 to 3.0)
sway velocity	60.4	75.7	60.6	65.0	61.0	70.0	-3.2	-12.8	-2.7	-7.9	9.6	5.1
ML (mm/s)	(12.2)	(34.0)	(5.6)	(19.9)	(12.8)	(32.5)	(11.2)	(24.0)	(18.9)	(26.5)	(-4.0 to 23.2)	(-12.6 to 22.9)
sway area per	13.4	14.5	82.2	12.8	12.8	16.7	-1.6 ^m	-1.9	-1.4 ^m	-0.4*	0.3	-3.3
second AP (mm/s)	(5.7)	(4.9)	(12.8)	(4.4)	(3.5)	(7.6)	(6.9)	(5.9)	(7.5)	(4.6)	(-4.6 to 5.2) ^m	(-9.0 to 2.4)
sway area per	11.8	36.8	11.4	11.5	9.9	11.6	-1.0	-0.4 ^p	-2.5	-0.2 ^p	-0.6	-2.3
second ML (mm/s)	(4.4)	(106.9)	(2.9)	(3.1)	(2.0)	(5.0)	(5.6)	(4.6)	(4.3)	(6.1)	(-4.5 to 3.2) ^p	(-6.4 to 1.8) ^p

ADL, activities of daily living; AP, antero-posterior; CAI, chronic ankle instability; Con, control group (placebo, detuned laser); CoP, centre of pressure; Exp, experimental group (MWM with taping); FAAM, foot and ankle ability measure; FAOS, foot and ankle outcome score; ML, medio-lateral; N/A, not applicable; PPT, pressure pain threshold; QOL, quality of life; SEBT, star excursion balance test; weight-bearing DFROM, weight-bearing dorsiflexion range of motion

*significantly different, p< 0.05; ^m extreme outlier was identified in MWM group and removed (n=1); ^p extreme outlier was identified in placebo laser group and removed (n=1)

Table 7.4 Effects of MWM with taping on individuals with a displaced fibula (i.e. normalised fibular position \geq 27), compared to individuals with a normally positioned fibular on the changes of primary outcomes at each time point

Outcome			Difference between groups (Mann - Whitney t test)					
	Pre-interver	ntion	Immediately post- intervention		12 week follow-up		Immediately post- intervention-	12 week follow- up-Pre- intervention
	Displaced fibula (n=8)	Non- displaced fibula (n=8)	Displaced fibula (n=7)	Non- displaced fibula (n=6)	Displaced fibula (n=7)	Non- displaced fibula (n=6)	Pre- intervention (p value)	(p value)
fibular position (mm)	13.8 (1.8)	8.8 (1.9)	12.5 (3.9)	11.2 (3.0)	N/A	N/A	0.10	N/A
normalised fibular position (%)	30.9 (2.0)	20.3 (2.6)	28.4 (7.5)	25.8 (6.6)	N/A	N/A	0.10	N/A
FAAM score ADL	72.4	73.5	73 3	76.7	73.4	78.2	1.00	0.63
	(8.7)	(9.2)	(13.3)	(13.5)	(12.0)	(11.8)	1.00	0.05

sports	52.8	56.3	50.6	59.5	53.6	65.7	0.95	0.63
	(16.7)	(7.5)	(22.4)	(13.9)	(18.1)	(18.9)		
FAOS score								
symptoms	57.6	57.6	58.1	61.3	57.7	60.1	0.63	0.63
	(12.9)	(12.7)	(20.9)	(14.5)	(21.4)	(13.4)		
pain	68.7	70.8	65.5	64.4	71.0	70.6	0.73	0.45
	(13.7)	(14.5)	(10.2)	(14.7)	(12.7)	(10.7)		
sports	50.6	52.5	58.6	51.7	53.6	68.3	0.37	0.45
	(18.0)	(12.5)	(15.7)	(17.5)	(16.5)	(9.3)		
ADL	76.3	80.0	78.8	74.8	80.0	84.3	0.84	0.45
	(10.3)	(16.6)	(10.9)	(11.4)	(11.1)	(9.8)		
QOL	34.4	46.1	42.8	49.0	37.5	56.3	0.73	0.95
	(18.3)	(8.8)	(20.6)	(15.5)	(19.1)	(22.3)		

ADL, activities of daily living; FAAM, foot and ankle ability measure; FAOS, foot and ankle outcome score; MWM, mobilisation with movement; N/A,

not applicable; QOL, quality of life

Table 7.5 Effects of MWM with taping on individuals with a mechanically unstable ankle (i.e. positive anterior drawer stress test) compared to individuals with a mechanically stable ankle on the changes of primary outcomes at each time point

Outcome			Difference between groups (Mann - Whitney t test)					
	Pre-intervention		Immediately post- intervention		12 week follow-up		Immediately post- intervention- Pre-	12 week follow- up-Pre- intervention (n value)
	Mechanically unstable (n=8)	Mechanically stable (n=8)	Mechanically unstable (n=6)	Mechanically stable (n=7)	Mechanically unstable (n=6)	Mechanically stable (n=7)	intervention (p value)	× /
fibular position (mm)	8.7 (1.7)	13.9* (1.7)	10.2 (2.3)	13.4 (3.7)	N/A	N/A	0.45	N/A
normalised fibular position (%)	20.7 (3.2)	30.6* (2.7)	24.2 (6.5)	29.7 (6.7)	N/A	N/A	0.45	N/A
FAAM	74.4	71.5	73.5	76.0	77.0	74.4	0.14	0.95
ADL	(9.6)	(8.0)	(12.8)	(14.0)	(11.0)	(12.9)		
sports	59.0	50.0	57.5	52.3	64.7	54.4	0.45	0.95

	(6.3)	(16.0)	(12.0)	(23.9)	(18.1)	(19.4)		
FAOS	59.4	55.8	62.5	57.1	63.1	55.1	1.00	0.45
symptoms	(12.5)	(12.8)	(14.1)	(20.9)	(14.1)	(20.0)		
pain	74.0	65.6	61.1	68.3	71.5	70.2	0.30	1.00
	(14.8)	(11.9)	(13.2)	(10.6)	(11.6)	(12.0)		
sports	55.6	47.5	50.8	59.3	66.7	55.0	0.07	0.73
	(13.5)	(16.3)	(16.9)	(15.9)	(8.8)	(18.0)		
ADL	81.1	75.2	73.5	79.8	82.8	81.3	0.37	0.84
	(16.9)	(9.3)	(10.6)	(11.0)	(8.9)	(12.1)		
QOL	48.5	32.1	47.9	43.7	53.2	40.2	0.37	0.53
	(8.7)	(16.2)	(14.1)	(21.7)	(19.2)	(23.9)		

ADL, activities of daily living; FAAM, foot and ankle ability measure; FAOS, foot and ankle outcome score; MWM, mobilisation with movement; N/A,

not applicable; QOL, quality of life

*significantly different, p< 0.05

7.4 Discussion

This is the first mixed RCT which utilised a pragmatic application of 'Mulligan's MWM with fibular repositioning taping' for participants with CAI. In addition, this is the first mixed RCT to assess the capacity of MWM with taping to reverse a displaced fibular position. This study aimed to evaluate the effectiveness of this common clinical technique on fibular position and functional and other clinical outcomes in individuals with CAI, both immediately and at 12 weeks after treatment. On average, the participants in the MWM group underwent six sessions with four sets of eight repetitions of MWM and taping, over a maximum of four weeks. A recent meta-analysis has shown that six sessions of manual therapy is sufficient to improve functional performance of the ankle in individuals with CAI (Shi et al., 2019). Therefore this dosage of MWM with taping could be considered sufficient to demonstrate any improvement in outcomes resulting from this manual therapy treatment.

The results of the current study indicate that Mulligan's MWM with fibular repositioning taping has no immediate or short-term (12 weeks post intervention) effect on fibular position or clinical outcomes including weight-bearing DFROM, PPT, static and dynamic balance, pain intensity and any sub-scales of the self-reported function questionnaires. Interestingly, scores for subscales of self-reported function (FAOS pain and sports) were immediately improved after application of detuned laser. Further, the antero-posterior sway velocity of anterior excursion of the SEBT was found to be significantly higher in the placebo group immediately after intervention, whilst the MWM group remained unchanged. Postural performance has been suggested to be characterised by the ability to reduce postural sway (Thompson et al., 2017). However, the increased sway velocity observed in the placebo group may simply be explained as a possible Type 1 error. There were no statistically significant effects identified

on any of the other outcomes or change of any outcomes at any of the time points. Without a comparative 'no treatment' control group, it is unknown whether the outcomes for participants in either or both groups were superior to that of natural recovery (Bialosky et al., 2011). Ethical considerations precluded the inclusion of such a group in the study design. Further, it is also possible that MWM may be more or less effective in different stages of recovery following an ankle sprain (e.g., the sub-acute phase) and this may be worthy of future investigation.

Similarly, the subgroup analyses (displaced fibula vs non-displaced fibula; mechanically unstable vs mechanically stable ankle) failed to find any significant effect of MWM on the primary outcomes of fibular position and self-reported function. However, these findings should be considered as preliminary because the sample size was not powered for either subgroup analysis. The lack of any detectable effect of MWM on fibular position could also be due to the sample containing participants with both mechanically stable and unstable ankles, as demonstrated on the anterior drawer stress test (mechanically unstable, n=6; mechanically stable, n=7).

The findings of our study are supported by some previous investigations in which MWM and associated fibular repositioning taping have been assessed separately. In these studies, MWM applied in isolation was found to not demonstrate any significant immediate improvement in weight-bearing DFROM (Gilbreath et al., 2014), dynamic balance (Gilbreath et al., 2014) or self-reported function assessed using the FAAM in participants with CAI (Gilbreath et al., 2014). Taping alone has also not shown any significant immediate improvement in static balance (Alves et al., 2018; Hopper et al.,

2009), dynamic balance (Delahunt, Mcgrath, et al., 2010; Hopper et al., 2009) or dynamic functional tasks (Alves et al., 2018). However in other studies, MWM in isolation improved weight-bearing DFROM (Cruz-Diaz et al., 2015; Marron-Gomez et al., 2015; Reid et al., 2007; Vicenzino et al., 2006) and dynamic balance (Cruz-Diaz et al., 2015) immediately and/or in the short-term, and taping alone was shown to improve dynamic balance (Someeh et al., 2015a) and the performance of dynamic functional tasks (Someeh et al., 2015b). These contrasting results may to some degree be due to the different dosages (ranging from one to four sets of mobilization in one session) of intervention used (Gilbreath et al., 2014) or perhaps due to different control groups in each trial. Further, these findings may be influenced by differences in the baseline participant demographics and clinical characteristics.

Some limitations of our study need to be acknowledged. First, the method may have been strengthened if fibular position and the pain outcome measures had been included as inclusion criteria during recruitment. Similarly, variable duration of injury and ongoing pain or stiffness in participants may have had a potential impact on treatment response, and it is suggested that in future research greater homogeneity of the sample may better replicate clinical practise. Second, the effectiveness of blinding of participants may have been sub-optimal because the placebo intervention notably differed from the experimental treatment. However, the mechanism of fibular displacement with ankle injuries and the mechanical effects of MWM are not clearly understood, so it was not desirable to introduce any external force potentially creating a mechanical change to the ankle/fibula, as might occur with laying on of hands in other placebo or sham manual therapy treatments. Further, 52.9% of the laser group participants perceived that they had received the active intervention (compared with 46.2% of the MWM group participants), suggesting the placebo was effective. Third, the practitioner who applied

the treatment was not a Mulligan Concept Practitioner or Teacher as certified by the Mulligan Concept Teachers Association (Mulligan Concept Teachers Association, 2019). However, it is arguable that for a manual therapy intervention to be widely beneficial to a population with a given condition, it should be demonstrably effective when applied by a post-professionally trained and registered musculoskeletal/manual therapist, and not require significant additional educational certification. Notably, only one prior study stated that it utilized a certified Mulligan Concept Practitioner or Teacher as the treating practitioner (Reid et al., 2007). Further, the current study used a sustained anteroposterior glide of the fibula as the active intervention, and a different MWM technique may have produced different results. Finally, there remained the potential for anatomical rotation in the X-ray imaging, although the same instructions were given to all participants and the leg position was carefully monitored for this. If any leg rotation was observed on imaging, the radiograph was repeated.

In conclusion, individuals with CAI who underwent an average of six sessions of Mulligan's MWM with fibular repositioning taping over a maximum four week period, demonstrated no significant differences in any of the outcomes measured compared to a placebo intervention. Further powered trials with homogenous samples are recommended in order to assess the effectiveness of MWM with taping in specific CAI subgroups. Also, it may be worthwhile exploring fibular position changes in the longer term in future research.

Chapter 8 Discussion and conclusions

This chapter summarises the major overarching findings of the various studies comprising this thesis, and considers their collective clinical and research implications. This thesis includes a systematic review with meta-analysis, a case-control study, and an RCT with a one year follow-up in progress. The systematic review described in Chapter 3 is the first published study to assess the clinical benefits of joint mobilisation as the sole intervention in the management of ankle ligament sprains at all stages of recovery. It included the first meta-analysis undertaken to quantitatively evaluate the clinical benefits of joint mobilisation for ankle sprains. The case-control study assessed displaced fibular position as an anatomical characteristic of CAI (Chapter 4), and also some common clinical characteristics of CAI (Chapter 5). Fibular position was assessed using weight-bearing lateral X-rays, whereas previous studies had only assessed fibular position in non-weight-bearing positions. The weight-bearing position is more functional and easier to apply clinically than the methods used for measuring fibular position in prior research. In the RCT for which the protocol was published (Chapter 6), the effectiveness of an MWM intervention was evaluated in individuals with CAI immediately post-treatment, at 12 weeks, and at 12 months post treatment. This trial has a longer follow-up (one year) than any other study of MWM for CAI published to date. However, within this thesis outcomes data are presented only up to the 12 week follow-up, as described in Chapter 7. Twelve month followup data will be analysed when these are fully collected, and the findings will be added to the prepared manuscript (as presented in Chapter 7) before submission to a peer-reviewed journal. Further, this RCT is the first RCT to investigate the effects of MWM on antero-posterior fibular displacement relative to the tibia in CAI.

8.1 Summary of study findings

Chronic ankle instability and recurrent ankle sprains are extremely common in the community, both in Australia and around the world (Hertel, 2002; Hiller et al., 2012; Weerasekara & Hiller, 2017). Therefore, the findings of this series of studies have the potential to be advantageous for many people with this debilitating condition. Eligible participants were randomly recruited from the university population and general community in the Newcastle area of New South Wales, Australia and thus the findings can be generalised to the wider community.

The key findings presented in this thesis include:

From a systematic review and meta-analysis (Chapter 3) of 23 studies of joint mobilisation only for ankle sprains:

- 1. There are immediate (measured immediately after the intervention) benefits of joint mobilisation for improving dynamic balance in chronic ankle sprains including CAI.
- There are short-term (measured up to 3 months after the intervention) benefits of joint mobilisation for improving ROM (specifically, weight-bearing DFROM) in chronic ankle sprains including CAI.
- 3. There are no immediate improvements in static balance, pain intensity, and weight-bearing DFROM following ankle joint mobilisation in chronic ankle sprains including CAI.

- 4. Other clinical outcomes (proprioception, PPT, pain intensity and quality, function, talar stiffness, postural sway, and patient's confidence toward stability) that have been reported after ankle sprains demonstrate an inconsistent response to joint mobilisation.
- 5. The combination of Mulligan MWM and associated fibular repositioning taping is more likely to produce a clinical benefit than Maitland joint mobilisation with and without traction, or joint manipulation in individuals with ankle sprains including CAI.

From a case-control study (Chapters 4 and 5) of 33 participants with CAI and 33 matched control participants:

- 6. A more anteriorly positioned fibula in relation to the tibia was observed in participants with CAI compared with healthy controls in weight-bearing radiographs.
- 7. Weight-bearing lateral radiographic measurements of fibular position can be performed reliably and reproducibly.
- 8. The normalised fibular position has very little ability to predict CAI alone.
- CAI exhibits a multi-faceted nature including pain and impaired weight-bearing DFROM, static balance, dynamic balance, and self-reported function compared to individuals with healthy ankles.
- 10. The presence of localised peripheral pain (nociceptive mechanism) is associated with persistent pain in CAI, rather than central sensitisation.

 Weight-bearing DFROM and eyes open medio-lateral sway velocity are most predictive of being in the CAI group.

From an RCT (Chapters 6 and 7) that included 34 participants with CAI:

- 12. Individuals with CAI, who have undergone an average of six sessions of 'Mulligan's MWM with fibular repositioning taping' demonstrated no significant immediate difference on fibular position compared to a placebo intervention.
- 13. Individuals with CAI, who have undergone an average of six sessions of 'Mulligan's MWM with fibular repositioning taping' compared to a placebo intervention, reported no significant immediate or short-term differences in any of the outcomes measured (weightbearing DFROM, static or dynamic balance, pain intensity, PPT or self-reported function).

The studies completed in this thesis therefore answered the following four hypotheses as identified in Chapter 1:

- 1. Joint mobilisation has multiple clinical benefits for people with CAI.
- Individuals with CAI are anatomically different with a more anteriorly positioned fibula, compared to individuals with healthy ankles.
- 3. Individuals with CAI are clinically different, with greater pain intensity and more impaired weight-bearing DFROM, dynamic balance, self-reported function and static balance (single leg

stance eyes open, measured using CoP measures on a force plate), compared to individuals with healthy ankles.

 There are no immediate or short-term effects of MWM combined with fibular repositioning taping on anatomical or clinical outcomes of individuals with CAI, compared to a placebo intervention.

8.2 Significance of the findings of the thesis

The published systematic review presented in Chapter 3 of the thesis was the first systematic review to assess the clinical benefits of joint mobilisation as the sole intervention in the management of ankle sprains at all stages of recovery. The findings of this study provide compelling evidence to inform clinical practice that joint mobilisation may be effective in improving balance immediately and in increasing weight-bearing DFROM in the short-term in chronic ankle sprains, including CAI.

The case-control study described in this thesis presents the difference in fibular position between individuals with CAI and healthy ankles in a weight-bearing position, and adequately demonstrates the diagnostic utility of this imaging method of fibular position (Chapter 4). This imaging technique may be clinically relevant due to its easier application and more functional nature, and therefore could be of value in the therapeutic identification of CAI. The presence of an anteriorly positioned fibular may be a factor contributing to the persistence of pain, ROM restriction, and other symptoms and signs of CAI. This finding also lends support to Mulligan's hypothesis of a fibular 'positional fault' in chronic ankle sprains (Vicenzino et al., 2011). While the mechanism leading to the observed anterior fibular

positional change in CAI is unclear, it may be associated with mechanical instability in some cases. This case-control study was also the first to explore the involvement of central nervous system sensitisation in persistent pain in individuals with CAI, using PPT measures (Chapter 5). The main finding from this investigation was that there was evidence for the presence of localised peripheral sensitisation (nociceptive pain), with no evidence of central sensitisation (nociplastic pain). This suggests that management of CAI is less likely to require interventions to address psychosocial factors, as opposed to the physical impairments identified. However, future research should assess psychosocial factors (such as fear avoidance behaviour) to further explore this. Overall, this casecontrol study adds support for the multi-faceted anatomical and biological nature of CAI.

The final component of the thesis was an RCT to explore the effects of MWM with fibular repositioning taping on ankle instability (Chapters 6 and 7). This is the first RCT to investigate the potential of Mulligan's MWM with associate fibular repositioning taping on altering fibular position in CAI. The pragmatic RCT design used in this study utilised a real-world application of MWM with taping for participants with CAI, thus enhancing the generalisability of the findings to clinical practice. The RCT found no significant immediate or short-term (12 weeks) differences after undergoing a course of 'Mulligan's MWM with fibular repositioning taping' in any of the outcomes measured. The heterogeneous nature of the sample, including participants with both mechanical and functional instability, and differing dosages as judged by the treating practitioner may have had an impact on the findings to date. Further, some participants with CAI may have experienced transient changes in fibular position (i.e. temporary alterations in fibular position due to recurrent sprains), and this may have led to intermittent and recurrent symptoms (Gilbreath et al., 2014). Despite these considerations,

and although the long-term findings are yet to be fully collected and analysed, those findings to date presented in Chapter 7 suggest that MWM with taping is unlikely to be any more effective than a placebo intervention in treating CAI. Interestingly, the few significant differences found in this RCT were in favour of the placebo group, suggesting that detuned laser may have a strong placebo effect (Howick et al., 2013). Therefore, it may be considered as a potential placebo for future research in manual therapy. There is currently no consensus on a placebo intervention for manual therapy studies.

8.3 Strengths and limitations of the studies

This thesis is comprised of four studies employing three study designs: i) a systematic review and meta-analysis (Weerasekara et al., 2018), ii) a case-control study, and iii) a RCT (using a pragmatic approach) (Weerasekara, Osmotherly, Snodgrass, et al., 2019b). The strengths and limitations of each study are discussed individually below.

8.3.1 Systematic review

A major strength of the systematic review reported in Chapter 3 was that data were able to be pooled for meta-analysis. This quantitatively summarised the results of all of the included studies and increased the precision of the estimate of the treatment effect of the interventions (Bartolucci & Hillegass, 2010). In addition, joint mobilisation was evaluated as a sole intervention to ensure homogeneity and to increase the precision of the findings on the effects of joint mobilisation. However, the methodologies used by all studies included in the systematic review were found to be of moderate quality (Weerasekara et al., 2018), which limits the conclusions that can be drawn on the evidence for the effectiveness of joint mobilisation for ankle sprains. Some other limitations of this systematic review should be acknowledged, including the wide variety of follow-up time points that were defined as short-term (from one day up to three months), the wide variety of different mobilisation techniques in the included studies, and the insufficiency of data for quantitative analysis of some outcome measures (Weerasekara et al., 2018).

8.3.2 Case-control study

The case-control study compared individuals with CAI with age and gender matched individuals with healthy ankles. The major strengths of this study were randomisation of the measurements and the careful selection of the study sample according to IAC recommendations on selection criteria for participants with CAI in controlled research (Gribble et al., 2013). All the measurements were obtained in a randomised order within a single data collection session for each participant. The body sites of the measurements were also randomised when it was applicable (e.g. test locations in PPT measures). Further, the use of a force plate helped to assess balance more objectively, and also could precisely quantify the balance measurements.

In terms of limitations, two-thirds of the CAI group had bilateral ankle injuries, and this may have exerted a potential effect in all comparisons carried out within the CAI group. In addition, the study was not designed to perform a subgroup analysis to examine differences (anatomical and clinical)
between participants with mechanical and functional instabilities. Thus, the sample size was not adequately powered for this subgroup analysis and none of the outcomes were able to be compared between these two subgroups using appropriate statistical methods.

8.3.3 Randomised controlled trial

A strength of the RCT presented in this thesis is that a pragmatic approach was used. This utilised a pragmatic RCT design which allowed a routine clinical application of MWM combined with fibular repositioning taping (Alsop et al., 2016). This enhances the external validity and also controls for some threats to internal validity. Real-world application of the MWM, its dosage and the number of treatment visits were judged by the treating practitioner based on individual participant responses, and this enhanced the study's external validity by improving the extent the findings could be generalised to clinical practice. Internal validity of a study is typically threatened by bias (commonly selection bias, performance bias, detection bias, and attrition bias) and random error (Spieth et al., 2016). In this RCT, random allocation of the participants to the groups, concealed allocation, random allocation of the measurement locations, blinding of both the assessor and the participant, and ITT analysis strengthened the internal validity by controlling for the above sources of bias. Further, MWM combined with fibular repositioning taping was assessed as a unique intervention with no potential confounding effects from adjunct interventions, such as self-mobilisation or other home exercises.

Another strength was that a detuned laser was used as the placebo and has previously been shown to have a strong placebo effect (Reid et al., 2008). Though there was a different physical appearance to

the interventions, using a detuned laser prevents any potential direct mechanical effect to the ankle being treated and also does not activate somatosensory receptors. This avoids an external force potentially creating a therapeutic or deleterious mechanical change to the ankle/fibula, as might occur with the 'laying on of hands' in other placebo or sham manual therapy techniques. Due to the fact that alteration of fibular position in CAI is not well understood, precautions were taken to not introduce any external force which may potentially create a mechanical change to the fibula or ankle by manual handling or placing the participant in a different position, a common potential error in the placebo or control groups used in some other RCTs (Hancock et al., 2006; Pellow & Brantingham, 2001; Reid et al., 2008). This was to some degree controlled by the scheduling of participant appointments in a way that avoided interaction between participants.

As with any RCT in the field of manual therapy, an unavoidable design limitation is that the intervention practitioner could not be blinded, which had the potential to introduce bias if the practitioner has particular beliefs about the intervention. Another limitation was that there might also have been some outcome measures which were underpowered to detect a significant difference between groups, because the sample size was estimated using fibular position and self-reported function. Finally, if we had included some major measures such as fibular position in the inclusion criteria (i.e. by screening all potential participants radiographically), we may have recruited a more homogeneous group. However, this would not have been as generalisable to the clinical setting, as patients presenting for physiotherapy are not usually screened radiologically. Further, in future research the screening of people with CAI for ongoing pain or stiffness might better replicate clinical practise and ensure a more homogeneous sample. Similarly, the sample size was inadequate for

subgroup (mechanical and functional instability) comparisons as this was not a planned inclusion criteria, and any findings regarding subgroups should be considered preliminary in nature.

8.4 Clinical implications of findings

Clinicians may consider applying joint mobilisation when treating patients with chronic ankle sprains, as the meta-analysis from the systematic review revealed that it is immediately (measured immediately after the intervention) beneficial for dynamic balance and for DFROM in the short-term (measured up to three months after the intervention). Evidence for long-term (measured at three or more months after the intervention) benefits of joint mobilisation is lacking in the literature due to a paucity of RCTs assessing the long-term efficacy of joint mobilisation on CAI.

The case-control study presented in this thesis has led to the introduction of a simple and reliable method to assess fibular position in a weight-bearing position in individuals with CAI using plain film radiography. As it is not presently possible to precisely diagnose CAI with a single diagnostic test (Guillo et al., 2013; Rodriguez-Merchan, 2012), careful consideration (in terms of ease of obtaining an accurate X-ray and time spent on obtaining fibular position measures) of an anteriorly displaced fibula evident on weight-bearing lateral X-rays may improve the accurate diagnosis of patients with this condition. In addition, clinicians should consider the multi-faceted nature of CAI (such as balance, function and ROM deficits) when developing rehabilitation programs for individuals with CAI. They should be mindful when assessing dynamic balance with the SEBT in patients with CAI, that it may be impacted by restricted weight-bearing DFROM. It is unknown whether the observed differences in

clinical measures in the case-control study presented in this thesis may differ between subgroups of CAI, therefore each patient with CAI should be individually assessed.

The results of the RCT to date indicate that Mulligan's MWM with fibular repositioning taping is not superior to a placebo treatment in producing any immediate or short-term effect on fibular position or any of the clinical outcomes measured. However, this was based on a sample of participants with different sub-categories of CAI (i.e. this sample is a mix of mechanically unstable, functionally unstable, both mechanically and functionally unstable; fibular displaced and non-displaced ankles). It is possible the effects of MWM with taping in different subgroups may differ.

8.5 Future research

The series of studies described in this thesis raise a number of questions that could be addressed in future research. Identification of CAI subgroups with common clinical characteristics may in the future be linked to treatment outcomes. One or more CAI subgroups may respond better to MWM than others, for example MWM may be able to improve reduced DFROM in individuals with a displaced fibula by reversing the displacement. In this case, the presence of a displaced fibular position may be a common characteristic of mechanical instability. Therefore careful investigation of the use of normalised fibular position combined with other diagnostic tools in diagnosing individuals with this condition, may improve the accuracy of diagnosis of CAI or at least some subgroups of CAI. Further, research using an adequate sample size of individuals with CAI who have a displaced fibula with specific clinical deficits (e.g. reduced DFROM and impaired balance), may reveal differences between

individuals with CAI and healthy ankles not detected in this thesis. Therefore, more research is required to further determine the efficacy of joint mobilisation in the rehabilitation of CAI across its various subgroups. It is also possible that MWM may be more or less effective in different stages of recovery following an ankle sprain (e.g., the sub-acute phase) and could contribute to the prevention or reduction of recurrence of ankle instability, and this may be worthy of future investigation. In addition, researchers may also consider investigating the pros and cons of detuned laser as a placebo intervention in manual therapy studies.

8.6 Conclusions

Chronic ankle instability is a common musculoskeletal disorder that significantly affects the quality of life of affected individuals, and is also responsible for recurrent and persistent disability which potentially causes a substantial financial burden on society. It is an umbrella term which includes several different types of CAI, including mechanically unstable, functionally unstable, and both mechanically and functionally unstable ankles. This thesis on CAI and joint mobilisation including MWM, assesses the anatomical and clinical characteristics of individuals with CAI compared to individuals with healthy ankles, and also examines the efficacy of joint mobilisation in improving the outcomes of CAI. The series of studies presented in this thesis provide evidence that; (1) joint mobilisation provides both immediate and short-term benefits for people with chronic ankle sprains, including CAI; (2) there is an anteriorly positioned fibula in individuals with CAI compared to those with healthy ankles; (3) CAI is a multi-faceted disorder with residual symptoms such as impaired balance, function and ROM; and (4) MWM with fibular repositioning taping may not be helpful in improving a displaced fibula or common clinical outcomes of individuals with CAI. The dissemination

of results from this thesis will lead to benefits for patients suffering from CAI by improving our understanding of the nature of this condition and by providing high level evidence for its effective management using manual therapy.

References

- Aagesen, A. L., & Melek, M. (2013). Choosing the right diagnostic imaging modality in musculoskeletal diagnosis. *Primary Care; Clinics in Office Practice*, 40(4), 849-861, viii. doi:10.1016/j.pop.2013.08.005
- Adal, S. A., Mackey, M., Pourkazemi, F., & Hiller, C. E. (2019). The relationship between pain and associated characteristics of chronic ankle instability: a retrospective study. *Journal of Sport & Health Science*. doi:https://doi.org/10.1016/j.jshs.2019.07.009
- Ajis, A., & Maffulli, N. (2006). Conservative management of chronic ankle instability. *Foot & Ankle Clinics*, *11*(3), 531-537. doi:https://doi.org/10.1016/j.fcl.2006.07.004
- Alanson, N. (2012). The immediate effects of manual therapy on dorsiflexion and joint position sense at the talocrural joint in participants with a history of lateral ankle sprain. (Master of Osteopathy), Unitec Institute of Technology, Auckland, New Zealand.
- Alghadir, A. H., Anwer, S., Iqbal, A., & Iqbal, Z. A. (2018). Test-retest reliability, validity, and minimum detectable change of visual analog, numerical rating, and verbal rating scales for measurement of osteoarthritic knee pain. *Journal of Pain Research*, 11, 851-856. doi:10.2147/jpr.S158847
- Alsop, J., Scott, M., & Archey, W. (2016). The mixed randomized trial: combining randomized, pragmatic and observational clinical trial designs. *Journal of Comparative Effectiveness Research*, 5(6), 569-579. doi:10.2217/cer-2016-0034
- Alves, Y., Ribeiro, F., & Silva, A. G. (2018). Effect of fibular repositioning taping in adult basketball players with chronic ankle instability: a randomized, placebo-controlled, crossover trial. *The Journal of Sports Medicine & Physical Fitness*, 58(10), 1465-1473. doi:10.23736/s0022-4707.17.07472-2
- Ambarish, A. A., Chitra, J., & Subhash, K. M. (2008). Comparative effectiveness ofMulligan's mobilization in weight bearing and non-weight bearing in the treatment of

ankle sprains-a randomized clinical trial. *Indian Journal of Physiotherapy & Occupational Therapy*, 2(4), 1-4

- American Physical Therapy Association. (2003). *Position on thrust joint manipulation provided by physical therapists* (Revised 2nd ed.): Alexandria.
- Andersen, K., Albers, C., MÃ,ller-Madsen, B., Niedermann, B., Simonsen, O., & Snorrason, K. (1987). Conservative treatment of severe ankle sprains. *Acta Orthopaedica Scandinavica*, 58(6), 697-698.
- Andersen, S., Fryer, G. A., & McLaughlin, P. (2003). The effect of talo-crural joint manipulation on range of motion at the ankle joint in subjects with a history of ankle injury. *Australasian Chiropractic & Osteopathy*, 11(2), 57-62.
- Angthong, C. Validity and reliability of Thai version of the foot and ankle outcome score (FAOS) in patients with arthritis of the foot and ankle. *Foot & Ankle Surgery, 22*(4), 224-228. doi:10.1016/j.fas.2015.09.006
- Arendt-Nielsen, L., Nie, H., Laursen, M. B., Laursen, B. S., Madeleine, P., Simonsen, O. H., & Graven-Nielsen, T. (2010). Sensitization in patients with painful knee osteoarthritis. *Pain, 149*(3), 573-581. doi:http://dx.doi.org/10.1016/j.pain.2010.04.003
- Armijo-Olivo, S., Pitance, L., Singh, V., Neto, F., Thie, N., & Michelotti, A. (2016).
 Effectiveness of manual therapy and therapeutic exercise for temporomandibular disorders: systematic review and meta-analysis. *Physical Therapy*, *96*(1), 9-25. doi:10.2522/ptj.20140548
- Arnold, B. L., De La Motte, S., Linens, S., & Ross, S. E. (2009). Ankle instability is associated with balance impairments: a meta-analysis. *Medicine & Science in Sports & exercise*, 41(5), 1048-1062. doi:10.1249/MSS.0b013e318192d044
- Arunakul, M., Arunakul, P., Suesiritumrong, C., Angthong, C., & Chernchujit, B. (2015).
 Validity and reliability of Thai version of the foot and ankle ability measure (FAAM) subjective form. *Journal of the Medical Association of Thailand*, 98(6), 561-567.

- Atkins, D., Best, D., Briss, P. A., Eccles, M., Falck-Ytter, Y., Flottorp, S., Guyatt, G. H., Harbour, R. T., Haugh, M. C., Henry, D., Hill, S., Jaeschke, R., Leng, G., Liberati, A., Magrini, N., Mason, J., Middleton, P., Mrukowicz, J., O'Connell, D., Oxman, A. D., Phillips, B., Schunemann, H. J., Edejer, T., Varonen, H., Vist, G. E., Williams, J. W., Jr., & Zaza, S. (2004). Grading quality of evidence and strength of recommendations. *British Medical Journal*, 328(7454), 1490. doi:10.1136/bmj.328.7454.1490
- Attenborough, A. S., Hiller, C. E., Smith, R. M., Stuelcken, M., Greene, A., & Sinclair, P. J. (2014). Chronic ankle instability in sporting populations. *Sports Medicine*, 44(11), 1545-1556. doi:10.1007/s40279-014-0218-2
- Ayad, K., Abdelaal, A., Elazizi, H., & Abdelgalil, A. (2015). Efficacy of mobilization with movement in treatment of chronic ankle sprain. *Physiotherapy*, 101, eS96-eS97. doi:http://dx.doi.org/10.1016/j.physio.2015.03.230
- Baker, R. T., Jr. (2013). An examination of the effects of mechanical manual therapies on clinician and patient-oriented outcomes measures: a dissertation of clinical practice improvement. (Doctor of Athletic Training), University of Idaho, Ann Arbor.
- Bansbach, H. (2017). Evaluation of an inertial sensor to quantify postural stability assessments in young healthy individuals and individuals with chronic ankle instability. (Doctor of Philosophy), University of Pittsburgh, Pittsburgh.
- Bartolucci, A. A., & Hillegass, W. B. (2010). Evidence-Based Practice: Toward Optimizing Clinical Outcomes (F. Chiappelli et al. Ed.): Springer-Verlag Berlin Heidelberg
- Bartoníček, J., Rammelt, S., & Naňka, O. (2018). Anatomy of the subtalar joint. *Foot & Ankle Clinics*, 23(3), 315-340. doi:https://doi.org/10.1016/j.fcl.2018.04.001
- Basnett, C. R., Hanish, M. J., Wheeler, T. J., Miriovsky, D. J., Danielson, E. L., Barr, J. B., & Grindstaff, T. L. (2013). Ankle dorsiflexion range of motion influences dynamic balance in individuals with chronic ankle instability. *International Journal of Sports Physical Therapy*, 8(2), 121-128.

- Baumbach, S. F., Brumann, M., Binder, J., Mutschler, W., Regauer, M., & Polzer, H. (2014).
 The influence of knee position on ankle dorsiflexion a biometric study. *BMC Musculoskeletal Disorders*, *15*(1), 246. doi:10.1186/1471-2474-15-246
- Beazell, J. R., Grindstaff, T. L., Sauer, L. D., Magrum, E. M., Ingersoll, C. D., & Hertel, J. (2012). Effects of a proximal or distal tibiofibular joint manipulation on ankle range of motion and functional outcomes in individuals with chronic ankle instability. *Journal* of Orthopaedic & Sports Physical Therapy, 42(2), 125-134.
- Bedson, J., Hill, J., White, D., Chen, Y., Wathall, S., Dent, S., Cooke, K., & van der Windt, D. (2019). Development and validation of a pain monitoring app for patients with musculoskeletal conditions (The Keele pain recorder feasibility study). *BMC Medical Informatics & Decision Making*, 19(1), 24-24. doi:10.1186/s12911-019-0741-z
- Beninato, M., Fernandes, A., & Plummer, L. S. (2014). Minimal clinically important difference of the functional gait assessment in older adults. *Physical Therapy*, 94(11), 1594-1603. doi:10.2522/ptj.20130596
- Bennell, K. L., Talbot, R. C., Wajswelner, H., Techovanich, W., Kelly, D. H., & Hall, A. J. (1998). Intra-rater and inter-rater reliability of a weight-bearing lunge measure of ankle dorsiflexion. *Australian Journal of Physiotherapy*, 44(3), 175-180.
- Berkowitz, M. J., & Kim, D. H. (2004). Fibular position in relation to lateral ankle instability. *Foot Ankle International*, 25(5), 318-321.
- Bernier, J. N., Perrin, D. H., & Rijke, A. (1997). Effect of unilateral functional instability of the ankle on postural sway and inversion and eversion strength. *Journal of Athletic Training*, 32(3), 226-232.
- Beynnon, B. D., Webb, G., Huber, B. M., Pappas, C. N., Renström, P., & Haugh, L. D. (2005). Radiographic measurement of anterior talar translation in the ankle: determination of the most reliable method. *Clinical Biomechanics*, 20(3), 301-306. doi:http://dx.doi.org/10.1016/j.clinbiomech.2004.11.011

- Bialosky, J. E., Bishop, M. D., George, S. Z., & Robinson, M. E. (2011). Placebo response to manual therapy: something out of nothing? *The Journal of Manual & Manipulative Therapy*, 19(1), 11-19. doi:10.1179/2042618610Y.0000000001
- Bielska, I. A., Wang, X., Lee, R., & Johnson, A. P. (2017). The health economics of ankle and foot sprains and fractures: a systematic review of English-language published papers.
 Part 2: The direct and indirect costs of injury. *The Foot.* doi:10.1016/j.foot.2017.07.003
- Bleakley, C. M., Glasgow, P., & MacAuley, D. C. (2012). PRICE needs updating, should we call the POLICE? *British Journal of Sports Medicine*, 46(4), 220-221. doi:10.1136/bjsports-2011-090297
- Bleakley, C. M., McDonough, S. M., & MacAuley, D. C. (2008). Some conservative strategies are effective when added to controlled mobilisation with external support after acute ankle sprain: a systematic review. *Australian Journal of Physiotherapy*, 54(1), 7-20.
- Borloz, S., Crevoisier, X., Deriaz, O., Ballabeni, P., Martin, R. L., & Luthi, F. (2011).
 Evidence for validity and reliability of a French version of the FAAM. *BMC Musculoskeletal Disorders*, *12*, 40-40. doi:10.1186/1471-2474-12-40
- Bridgman, S. A., Clement, D., Downing, A., Walley, G., Phair, I., & Maffulli, N. (2003).
 Population based epidemiology of ankle sprains attending accident and emergency units in the West Midlands of England, and a survey of UK practice for severe ankle sprains. *Emergency Medicine Journal, 20*(6), 508-510.
- Brockett, C. L., & Chapman, G. J. (2016). Biomechanics of the ankle. *Orthopaedics & Trauma, 30*(3), 232-238. doi:10.1016/j.mporth.2016.04.015
- Brown, C. N., & Mynark, R. (2007). Balance deficits in recreational athletes with chronic ankle instability. *Journal of Athletic Training*, 42(3), 367-373.
- Browne, J. E., & O'Hare, N. J. (2001). Review of the different methods for assessing standing balance. *Physiotherapy*, 87(9), 489-495. doi:10.1016/S0031-9406(05)60696-7

- Carcia, C. R., Martin, R. L., & Drouin, J. M. (2008). Validity of the foot and ankle ability measure (FAAM) in athletes with chronic ankle instability. *Journal of Athletic Training*, 43(2), 179-183.
- Celik, D., Malkoc, M., & Martin, R. (2016). Evidence for reliability, validity and responsiveness of Turkish foot and ankle ability measure (FAAM). *Rheumatology International*, 36(10), 1469-1476. doi:10.1007/s00296-016-3485-4
- Cervera-Garvi, P., Ortega-Avila, A. B., Morales-Asencio, J. M., Cervera-Marin, J. A., Martin,
 R. R., & Gijon-Nogueron, G. (2017). Cross-cultural adaptation and validation of
 Spanish version of the foot and ankle ability measures (FAAM-Sp). *Journal of Foot & Ankle Research*, 10, 39. doi:10.1186/s13047-017-0221-6
- Chaibi, A., Šaltytė Benth, J., & Bjørn Russell, M. (2015). Validation of placebo in a manual therapy randomized controlled trial. *Scientific Reports*, 5, 11774. doi:10.1038/srep11774
- Chan, K. W., Ding, B. C., & Mroczek, K. J. (2011). Acute and chronic lateral ankle instability in the athlete. *Bulletin of the NYU Hospital for Joint Diseases, 69*(1), 17-26.
- Chen, B., Wang, X., Gao, N. Y., Lin, X., Chen, Y. C., Li, L. H., Zhan, H. S., & Shi, Y. Y.
 (2012). Clinical observation on Tuina manipulation for ankle sprain in Wushu athletes. Shanghai Zhong Yi Yao Za Zhi [Shanghai Journal of Traditional Chinese Medicine] 2012;46(8):56-58.
- Chen, H., Li, H. Y., Zhang, J., Hua, Y. H., & Chen, S. Y. (2014). Difference in postural control between patients with functional and mechanical ankle instability. *Foot Ankle International*, 35(10), 1068-1074. doi:10.1177/1071100714539657
- Chesterton, L. S., Sim, J., Wright, C. C., & Foster, N. E. (2007). Interrater reliability of algometry in measuring pressure pain thresholds in healthy humans, using multiple raters. *The Clinical Journal of Pain 23*(9), 760-766. doi:10.1097/AJP.0b013e318154b6ae

- Chien, A., Eliav, E., & Sterling, M. (2009). Hypoaesthesia occurs with sensory hypersensitivity in chronic whiplash-further evidence of a neuropathic condition. *Manual Therapy*, 14(2), 138-146. doi:10.1016/j.math.2007.12.004
- Childs, J. (2016). The effects of a lateral ankle sprain on balance and jumping performance in Varsity athletes. (Master of Science Kinesiology), Wilfrid Laurier University, Waterloo, Canada.
- Chimenti, R. L., Frey-Law, L. A., & Sluka, K. A. (2018). A mechanism-based approach to physical therapist management of pain. *Physical therapy*, 98(5), 302-314. doi:10.1093/ptj/pzy030
- Chimera, N. J., Smith, C. A., & Warren, M. (2015). Injury history, sex, and performance on the functional movement screen and Y balance test. *Journal of Athletic Training*, 50(5), 475-485. doi:10.4085/1062-6050-49.6.02
- Chiu, Y. L., Tsai, Y. J., Lin, C. H., Hou, Y. R., & Sung, W. H. (2017). Evaluation of a smartphone-based assessment system in subjects with chronic ankle instability. *Computer Methods & Programs in Biomedicine, 139*, 191-195. doi:10.1016/j.cmpb.2016.11.005
- Clark, M. S. (2007). The Unilateral Forefoot Balance Test: Reliability and validity for measuring balance in late midlife women. *New Zealand Journal of Physiotherapy*, 35(3), 110-118.
- Clinch, J., Deere, K., Sayers, A., Palmer, S., Riddoch, C., Tobias, J. H., & Clark, E. M. (2011).
 Epidemiology of generalized joint laxity (hypermobility) in fourteen-year-old children from the UK: a population-based evaluation. *Arthritis & Rheumatology*, 63(9), 2819-2827. doi:10.1002/art.30435
- Clinch, P. (1986). The effect of grade II mobilization on range of movement in acutely sprained ankles. *Australian Journal of Physiotherapy*, *32*(3), 205.
- Cohen, J. (1988). Statistical Power Analysis for the Behavioral Sciences (2nd ed.). Hillsdale, NJ.

- Collins, N., Teys, P., & Vicenzino, B. (2004). The initial effects of a Mulligan's mobilization with movement technique on dorsiflexion and pain in subacute ankle sprains. *Manual Therapy*, *9*(2), 77-82.
- Coronado, R. A., Simon, C. B., Valencia, C., & George, S. Z. (2014). Experimental pain responses support peripheral and central sensitization in patients with unilateral shoulder pain. *The Clinical Journal of Pain*, 30(2), 143-151. doi:10.1097/AJP.0b013e318287a2a4
- Corrêa, J. B., Costa, L. O. P., de Oliveira, N. T. B., Sluka, K. A., & Liebano, R. E. (2015). Central sensitization and changes in conditioned pain modulation in people with chronic nonspecific low back pain: a case–control study. *Experimental Brain Research*, 233(8), 2391-2399. doi:10.1007/s00221-015-4309-6
- Cosby, N. L. (2012). Effects of anterior to posterior joint mobilizations in individuals with subacute lateral ankle sprains. *Dissertation Abstracts International: Section B: The Sciences & Engineering*, 73(2-B), 925.
- Cosby, N. L., Koroch, M., Grindstaff, T. L., Parente, W., & Hertel, J. (2011). Immediate effects of anterior to posterior talocrural joint mobilizations following acute lateral ankle sprain. *Journal of Manual & Manipulative Therapy*, 19(2), 76-83. doi:http://dx.doi.org/10.1179/2042618610Y.0000000005
- Cotler, H. B., Chow, R. T., Hamblin, M. R., & Carroll, J. (2015). The Use of low level laser therapy (LLLT) for musculoskeletal pain. *MOJ Orthopedics & Rheumatology*, 2(5), 00068. doi:10.15406/mojor.2015.02.00068
- Cox, J. S. (1985). Surgical and nonsurgical treatment of acute ankle sprains. *Clinical Orthopaedics & Related Research* (198), 118-126.
- Croy, T., Saliba, S. A., Saliba, E., Anderson, M. W., & Hertel, J. (2012). Differences in lateral ankle laxity measured via stress ultrasonography in individuals with chronic ankle instability, ankle sprain copers, and healthy individuals. *Journal of Orthopaedic & Sports Physical Therapy*, 42(7), 593-600. doi:10.2519/jospt.2012.3923

- Cruz-Diaz, D., Hita-Contreras, F., Lomas-Vega, R., Osuna-Perez, M. C., & Martinez-Amat,
 A. (2013). Cross-cultural adaptation and validation of the Spanish version of the
 Cumberland ankle instability tool (CAIT): an instrument to assess unilateral chronic
 ankle instability. *Clinical Rheumatology*, *32*(1), 91-98. doi:10.1007/s10067-012-2095-0
- Cruz-Diaz, D., Lomas Vega, R., Osuna-Perez, M. C., Hita-Contreras, F., & Martinez-Amat, A. (2015). Effects of joint mobilization on chronic ankle instability: a randomized controlled trial. *Disability & Rehabilitation*, 37(7), 601-610. doi:http://dx.doi.org/10.3109/09638288.2014.935877
- Davenport, T. E., Kulig, K., & Fisher, B. E. (2010). Ankle manual therapy for individuals with post-acute ankle sprains: description of a randomized, placebo-controlled clinical trial. BMC Complementary & Alternative Medicine, 10, 59. doi:10.1186/1472-6882-10-59
- de Bie, R. A., de Vet, H. C. W., Lenssen, T. F., van den Wildenberg, F. A. J. M., Kootstra, G., & Knipschild, P. G. (1998). Low-level laser therapy in ankle sprains: a randomized clinical trial. *Archives of Physical Medicine & Rehabilitation*, *79*(11), 1415-1420. doi:https://doi.org/10.1016/S0003-9993(98)90237-4
- De Noronha, M., Refshauge, K. M., Kilbreath, S. L., & Figueiredo, V. G. (2008). Crosscultural adaptation of the Brazilian-Portuguese version of the Cumberland ankle instability tool (CAIT). *Disability & Rehabilitation*, 30(26), 1959-1965. doi:10.1080/09638280701809872
- de Vries, J. S. (2009). Management of chronic lateral ankle instability : alternatives for diagnosis and treatment. (Doctor of Philosophy), University of Amsterdam, Amsterdam, Netherlands.
- de Vries, J. S., Kingma, I., Blankevoort, L., & van Dijk, C. N. (2010). Difference in balance measures between patients with chronic ankle instability and patients after an acute ankle inversion trauma. *Knee Surgery, Sports Traumatology, Arthroscopy, 18*(5), 601-606. doi:10.1007/s00167-010-1097-1

- de Vries, J. S., Krips, R., Sierevelt, I. N., Blankevoort, L., & van Dijk, C. N. (2011). Interventions for treating chronic ankle instability. *Cochrane Database Of Systematic Reviews*(8), CD004124. doi:10.1002/14651858.CD004124.pub3
- de Vries, J. S., Krips, R., & van Dijk, C. N. (2003). Treatment of chronic anterolateral ankle instability in athletes. *Schweizerische Zeitschrift Fur Sportmedizin Und Sporttraumatologie*, 51(2), 100–105.
- Del Buono, A., Florio, A., Boccanera, M. S., & Maffulli, N. (2013). Syndesmosis injuries of the ankle. *Current Reviews in Musculoskeletal Medicine*, 6(4), 313-319. doi:10.1007/s12178-013-9183-x
- Delahunt, E., Coughlan, G. F., Caulfield, B., Nightingale, E. J., Lin, C. W., & Hiller, C. E. (2010). Inclusion criteria when investigating insufficiencies in chronic ankle instability. *Medicine & Science In Sports & Exercise*, 42(11), 2106-2121. doi:10.1249/MSS.0b013e3181de7a8a
- Delahunt, E., Cusack, K., Wilson, L., & Doherty, C. (2013). Joint mobilization acutely improves landing kinematics in chronic ankle instability. *Medicine & Science In Sports & Exercise*, 45(3), 514-519. doi:10.1249/MSS.0b013e3182746d0a
- Delahunt, E., McGrath, A., Doran, N., & Coughlan, G. F. (2010). Effect of taping on actual and perceived dynamic postural stability in persons with chronic ankle instability. *Archives of Physical Medicine & Rehabilitation*, *91*(9), 1383-1389. doi:10.1016/j.apmr.2010.06.023
- Delco, M. L., Kennedy, J. G., Bonassar, L. J., & Fortier, L. A. (2017). Post-traumatic osteoarthritis of the ankle: A distinct clinical entity requiring new research approaches. *Journal of Orthopaedic Research : Official Publication of the Orthopaedic Research Society*, 35(3), 440-453. doi:10.1002/jor.23462
- Deyle, G. D., Gill, N. W., Rhon, D. I., Allen, C. S., Allison, S. C., Hando, B. R., Petersen, E. J., Dusenberry, D. I., & Bellamy, N. (2016). A multicentre randomised, 1-year

comparative effectiveness, parallel-group trial protocol of a physical therapy approach compared to corticosteroid injections. *British Medical Journal*, *6*(3).

- Diebschlag, W. & Nocker, W. (1987). Treatment of acute sprains of the ankle joint: a placebo controlled double blind study. *Münchener medizinische Wochenschrift*, 129(44), 803-806.
- Doherty, C., Bleakley, C., Delahunt, E., & Holden, S. (2017). Treatment and prevention of acute and recurrent ankle sprain: an overview of systematic reviews with metaanalysis. *British Journal of Sports Medicine*, 51(2), 113-125. doi:10.1136/bjsports-2016-096178
- Doherty, C., Bleakley, C., Hertel, J., Caulfield, B., Ryan, J., & Delahunt, E. (2016). Dynamic balance deficits in individuals with chronic ankle instability compared to ankle sprain copers 1 year after a first-time lateral ankle sprain injury. *Knee Surgery, Sports Traumatology, Arthroscopy, 24*(4), 1086-1095. doi:10.1007/s00167-015-3744-z
- Doherty, C., Bleakley, C. M., Hertel, J., Caulfield, B., Ryan, J., & Delahunt, E. (2015).
 Laboratory measures of postural control during the star excursion balance test after acute first-time lateral ankle sprain. *Journal of Athletic Training*, 50(6), 651-664. doi:10.4085/1062-6050-50.1.09
- Doherty, C., Delahunt, E., Caulfield, B., Hertel, J., Ryan, J., & Bleakley, C. (2014). The incidence and prevalence of ankle sprain injury: a systematic review and meta-analysis of prospective epidemiological studies. *Sports Medicine*, 44(1), 123-140. doi:10.1007/s40279-013-0102-5
- Drewes, L. K., McKeon, P. O., Kerrigan, D. C., & Hertel, J. (2009). Dorsiflexion deficit during jogging with chronic ankle instability. *Journal of Science & Medicine in Sport*, 12(6), 685-687. doi:10.1016/j.jsams.2008.07.003
- Dubin, J. C., Comeau, D., McClelland, R. I., Dubin, R. A., & Ferrel, E. (2011). Lateral and syndesmotic ankle sprain injuries: a narrative literature review. *Journal of Chiropractic Medicine*, 10(3), 204-219. doi:10.1016/j.jcm.2011.02.001

- Dutton, M. (2017). *Dutton's Orthopaedic Examination, Evaluation, and Intervention* (4thg ed.). New York: mCGraw-Hill Companies.
- Eisenhart, A. W., Gaeta, T. J., & Yens, D. P. (2003). Osteopathic manipulative treatment in the emergency department for patients with acute ankle injuries. *Journal of the American Osteopathic Association*, 103(9), 417-421.
- El-Liethy, N., & Kamal, H. (2016). High resolution ultrasonography and magnetic resonance imaging in the evaluation of tendino-ligamentous injuries around ankle joint. *The Egyptian Journal of Radiology & Nuclear Medicine*, 47(2), 543-555. doi:https://doi.org/10.1016/j.ejrnm.2016.02.009
- Ellis, S. J., Williams, B. R., Pavlov, H., & Deland, J. (2011). Results of anatomic lateral ankle ligament reconstruction with tendon allograft. *The Musculoskeletal Journal of Hospital for Special Surgery*, 7(2), 134-140. doi:10.1007/s11420-011-9199-y
- Eren, O. T., Kucukkaya, M., Kabukcuoglu, Y., & Kuzgun, U. (2003). The role of a posteriorly positioned fibula in ankle sprain. *American Journal of Sports Medicine*, 31(6), 995-998.
- Estrade, J. L. (2013). Therapie manuelle et entorse de cheville. Pratique basee sur les preuves. [Manual physical therapy and ankle sprain. Evidence-based practice]. *Kinesitherapie*, *13*(144), 12-15. doi:http://dx.doi.org/10.1016/j.kine.2013.10.008
- Fallat, L., Grimm, D. J., & Saracco, J. A. (1998). Sprained ankle syndrome: prevalence and analysis of 639 acute injuries. *The Journal of Foot and Ankle Surgery*, 37(4), 280-285. doi:https://doi.org/10.1016/S1067-2516(98)80063-X
- Fay, F., & Egerod, S. (1984). The effects of joint mobilizations on swelling in the acutely sprained ankle joint: a pilot study. *Australian Journal of Physiotherapy*, *31*(4), 168.
- Feger, M. A., Glaviano, N. R., Donovan, L., Hart, J. M., Saliba, S. A., Park, J. S., & Hertel, J. (2017). Current trends in the management of lateral ankle sprain in the United States. *Clinical Journal of Sport Medicine*, 27(2), 145-152. doi:10.1097/jsm.00000000000321

- Ferran, N. A., & Maffulli, N. (2006). Epidemiology of sprains of the lateral ankle ligament complex. *Foot & Ankle Clinics*, 11(3), 659-662. doi:10.1016/j.fcl.2006.07.002
- Ferreira-Valente, M. A., Pais-Ribeiro, J. L., & Jensen, M. P. (2011). Validity of four pain intensity rating scales. *Pain*, 152(10), 2399-2404. doi:10.1016/j.pain.2011.07.005
- Fisher, B. E., Davenport, T. E., Kulig, K., & Wu, A. D. (2009). Identification of potential neuromotor mechanisms of manual therapy in patients with musculoskeletal disablement: rationale and description of a clinical trial. *BMC Neurology*, 9, 20. doi:http://dx.doi.org/10.1186/1471-2377-9-20
- Fingleton, C., Smart, K., Moloney, N., Fullen, B. M., & Doody, C. (2015). Pain sensitization in people with knee osteoarthritis: a systematic review and meta-analysis. *Osteoarthritis Cartilage*, 23(7), 1043-1056. doi:10.1016/j.joca.2015.02.163
- Fong, D. T. P., Hong, Y., Chan, L.-K., Yung, P. S.-H., & Chan, K.-M. (2007). A Systematic review on ankle injury and ankle sprain in sports. *Sports Medicine*, 37(1), 73-94. doi:10.2165/00007256-200737010-00006
- Fong, D. T. P., Chan, Y. Y., Mok, K. M., Yung, P. S. H., & Chan, K. M. (2009).
 Understanding acute ankle ligamentous sprain injury in sports. *Sports Medicine, Arthroscopy, Rehabilitation, Therapy & Technology, 1*, 14. doi:10.1186/1758-2555-1-14
- Frank, L., McLaughlin, P., & Vaughan, B. (2013). The repeatability of pressure algometry in asymptomatic individuals over consecutive days. *International Journal of Osteopathic Medicine*, 16(3), 143-152. doi:10.1016/j.ijosm.2012.10.004
- Freeman, M. A. (1965). Instability of the foot after injuries to the lateral ligament of the ankle. *Journal of Bone & Joint Surgery*, 47(4), 669-677.
- Fukuhara, T., Sakamoto, M., Nakazawa, R., & Kato, K. (2012). Anterior positional fault of the fibula after sub-acute anterior talofibular ligament injury. *Journal of Physical Therapy Science*, 24(1), 115-117. doi:10.1589/jpts.24.115

- Garrick, J. G. (1977). The frequency of injury, mechanism of injury, and epidemiology of ankle sprains. *American Journal of Sports Medicine*, *5*(6), 241-242.
- Giannini, S., Ruffilli, A., Pagliazzi, G., Mazzotti, A., Evangelisti, G., Buda, R., & Faldini, C. (2014). Treatment algorithm for chronic lateral ankle instability. *Muscles, Ligaments & Tendons Journal, 4*(4), 455-460.
- Giesbrecht, R. J. S., & Batti , M. C. (2005). A comparison of pressure pain detection thresholds in people with chronic low back pain and volunteers without pain. *Physical Therapy*, 85(10), 1085-1092. doi:10.1093/ptj/85.10.1085
- Gigi, R., Haim, A., Luger, E., Segal, G., Melamed, E., Beer, Y., Nof, M., Nyska, M., & Elbaz, A. (2015). Deviations in gait metrics in patients with chronic ankle instability: a case control study. *Journal of Foot & Ankle Research*, 8(1), 1-8. doi:10.1186/s13047-014-0058-1
- Gilbreath, J. P., Gaven, S. L., Van Lunen, L., & Hoch, M. C. (2014). The effects of mobilization with movement on dorsiflexion range of motion, dynamic balance, and self-reported function in individuals with chronic ankle instability. *Manual Therapy*, 19(2), 152-157.
- Golanó, P., Vega, J., de Leeuw, P. A. J., Malagelada, F., Manzanares, M. C., Götzens, V., & van Dijk, C. N. (2010). Anatomy of the ankle ligaments: a pictorial essay. *Knee Surgery, Sports Traumatology, Arthroscopy*, 18(5), 557-569.
- Golriz, S., Hebert, J. J., Foreman, K. B., & Walker, B. F. (2012). The reliability of a portable clinical force plate used for the assessment of static postural control: repeated measures reliability study. *Chiropractic & Manual Therapies*, 20(1), 14-14. doi:10.1186/2045-709X-20-14
- Gómez, D. M., Fernándezb, A. L. R., & Urrialdeb, J. A. M. (2015). The effect of two mobilization techniques on dorsiflexion in people with chronic ankle instability. *Physical Therapy in Sport : Official Journal of the Association of Chartered Physiotherapists in Sports Medicine*, 16(1), 10–15.

- Gould, N., Seligson, D., & Gassman, J. (1980). Early and late repair of lateral ligament of the ankle. *Foot & Ankle, 1*(2), 84-89. doi:10.1177/107110078000100206
- Green, T., Refshauge, K., Crosbie, J., & Adams, R. (2001). A randomized controlled trial of a passive accessory joint mobilization on acute ankle inversion sprains. *Physical Therapy*, 81(4), 984-994.
- Gribble, P. A. (2019). Evaluating and differentiating ankle instability. *Journal of Athletic Training*, *54*(6), 617-627. doi:10.4085/1062-6050-484-17
- Gribble, P. A., Bleakley, C. M., Caulfield, B. M., Docherty, C. L., Fourchet, F., Fong, D. T., Hertel, J., Hiller, C. E., Kaminski, T. W., McKeon, P. O., Refshauge, K. M., Verhagen, E. A., Vicenzino, B. T., Wikstrom, E. A., & Delahunt, E. (2016a). 2016 consensus statement of the International ankle consortium: prevalence, impact and long-term consequences of lateral ankle sprains. *British Journal of Sports Medicine, 50*(24), 1493-1495. doi:10.1136/bjsports-2016-096188
- Gribble, P. A., Bleakley, C. M., Caulfield, B. M., Docherty, C. L., Fourchet, F., Fong, D. T., Hertel, J., Hiller, C. E., Kaminski, T. W., McKeon, P. O., Refshauge, K. M., Verhagen, E. A., Vicenzino, B. T., Wikstrom, E. A., & Delahunt, E. (2016b). Evidence review for the 2016 International ankle consortium consensus statement on the prevalence, impact and long-term consequences of lateral ankle sprains. *British Journal of Sports Medicine*, 50(24), 1496-1505. doi:10.1136/bjsports-2016-096189
- Gribble, P. A., Delahunt, E., Bleakley, C., Caulfield, B., Docherty, C., Fourchet, F., Hertel, J.,
 Hiller, C., Kaminski, T., McKeon, P., Refshauge, K., van der Wees, P., Vincenzino,
 B., & Wikstrom, E. (2013). Selection criteria for patients with chronic ankle instability
 in controlled research: a position statement of the International ankle consortium. *Journal of Orthopaedic & Sports Physical Therapy*, 43(8), 585-591.
 doi:10.1136/bjsports-2013-093175

- Gribble, P. A., & Hertel, J. (2003). Considerations for normalizing measures of the star excursion balance test. *Measurement in Physical Education & Exercise Science*, 7(2), 89–100.
- Gribble, P. A., Hertel, J., & Plisky, P. (2012). Using the star excursion balance test to assess dynamic postural-control deficits and outcomes in lower extremity injury: a literature and systematic review. *Journal of Athletic Training*, *47*(3), 339-357.
- Groeneweg, R., van Assen, L., Kropman, H., Leopold, H., Mulder, J., Smits-Engelsman, B. C.
 M., . . . van Tulder, M. W. (2017). Manual therapy compared with physical therapy in patients with non-specific neck pain: a randomized controlled trial. *Chiropractic & Manual Therapies*, 25(1), 12. doi:10.1186/s12998-017-0141-3
- Groters, S., Groen, B. E., van Cingel, R., & Duysens, J. (2013). Double-leg stance and dynamic balance in individuals with functional ankle instability. *Gait & Posture*, 38(4), 968-973. doi:10.1016/j.gaitpost.2013.05.005
- Guillén-Rogel, P., San Emeterio, C., & Marín, P. J. (2017). Associations between ankle dorsiflexion range of motion and foot and ankle strength in young adults. *Journal of Physical Therapy Science*, 29(8), 1363-1367. doi:10.1589/jpts.29.1363
- Guillo, S., Bauer, T., Lee, J. W., Takao, M., Kong, S. W., Stone, J. W., Mangone, P. G., Molloy, A., Perera, A., Pearce, C. J., Michels, F., Tourné, Y., Ghorbani, A., & Calder, J. (2013). Consensus in chronic ankle instability: aetiology, assessment, surgical indications and place for arthroscopy. *Orthopaedics & Traumatology: Surgery & Research, 99*(8, Supplement), S411-S419. doi:http://dx.doi.org/10.1016/j.otsr.2013.10.009
- Guyatt, G. H., Oxman, A. D., Vist, G. E., Kunz, R., Falck-Ytter, Y., Alonso-Coello, P., & Schünemann, H. J. (2008). GRADE: an emerging consensus on rating quality of evidence and strength of recommendations. *British Medical Journal*, 336(7650), 924.
- Hadadi, M., Ebrahimi Takamjani, I., Ebrahim Mosavi, M., Aminian, G., Fardipour, S., & Abbasi, F. (2017). Cross-cultural adaptation, reliability, and validity of the Persian

version of the Cumberland ankle instability tool. *Disability & Rehabilitation, 39*(16), 1644-1649. doi:10.1080/09638288.2016.1207105

- Haefeli, M., & Elfering, A. (2006). Pain assessment. European Spine Journal : Official Publication of the European Spine Society, The European Spinal Deformity Society, and the European Section of the Cervical Spine Research Society, 15(Suppl 1), S17-S24. doi:10.1007/s00586-005-1044-x
- Hallaceli, H., Uruc, V., Uysal, H. H., Ozden, R., Hallaceli, C., Soyuer, F., Ince Parpucu, T., Yengil, E., & Cavlak, U. (2014). Normal hip, knee and ankle range of motion in the Turkish population. *Acta Orthopaedica et Traumatologica Turcica*, 48(1), 37-42. doi:10.3944/aott.2014.3113
- Harkey, M., McLeod, M., Van, S. A., Terada, M., Tevald, M., Gribble, P., & Pietrasimone, B. (2014). The immediate effects of an anterior-to-posterior talar mobilization on neural excitability, dorsiflexion range of motion, and dynamic balance in patients with chronic ankle instability. *Journal of Sport Rehabilitation*, 23(4), 351-359.
- Hart, L. E., & Macintyre, J. (2002). Passive joint mobilization for acute ankle inversion sprains. *Clinical Journal of Sport Medicine*, 12(1), 54. doi:http://dx.doi.org/10.1097/00042752-200201000-00016
- Hawker, G. A., Mian, S., Kendzerska, T., & French, M. (2011). Measures of adult pain: visual analog scale for pain (VAS pain), numeric rating scale for pain (NRS pain), McGill pain questionnaire (MPQ), short-form McGill pain questionnaire (SF-MPQ), chronic pain grade scale (CPGS), short form-36 bodily pain scale (SF-36 BPS), and measure of intermittent and constant osteoarthritis pain (ICOAP). *Arthritis Care & Research, 63*(S11), S240-S252. doi:10.1002/acr.20543
- Hedlund, S., Nilsson, H., Lenz, M., & Sundberg, T. (2014). Effect of chiropractic manipulation on vertical jump height in young female athletes with talocrural joint dysfunction:asingle-blind randomized clinical pilot trial. *Journal of Manipulative & Physiological Therapeutics*, 37(2).

- Hefford, C., Abbott, J. H., Arnold, R., & Baxter, G. D. (2012). The patient-specific functional scale: validity, reliability, and responsiveness in patients with upper extremity musculoskeletal problems. *Journal of Orthopaedic & Sports Physical Therapy*, 42(2), 56-65. doi:10.2519/jospt.2012.3953
- Herb, C. C., & Hertel, J. (2014). Current concepts on the pathophysiology and management of recurrent ankle sprains and chronic ankle instability. *Current Physical Medicine & Rehabilitation Reports*, 2(1), 25-34. doi:10.1007/s40141-013-0041-y
- Hershkovich, O., Tenenbaum, S., Gordon, B., Bruck, N., Thein, R., Derazne, E., Tzur, D., Shamiss, A., & Afek, A. (2015). A large-scale study on epidemiology and risk factors for chronic ankle instability in young adults. *The Journal Of Foot & Ankle Surgery*, 54(2), 183-187. doi:http://dx.doi.org/10.1053/j.jfas.2014.06.001
- Hertel, J. (2002). Functional anatomy, pathomechanics, and pathophysiology of lateral ankle instability. *Journal of Athletic Training*, *37*(4), 364-375.
- Hertel, J., & Corbett, R. O. (2019). An updated model of chronic ankle instability. *Journal of Athletic Training*, 54(6), 572-588. doi:10.4085/1062-6050-344-18
- Hertling, D., & Kessler, R. M. (2006). Management of Common Musculoskeletal Disorders; Physical Therapy Principles and Methods (Pam Lappies & David Payne Eds. 4 ed.): Lippincott Williams & Wilkins.
- Higgins, J., Green, S., & (editors). (2011). Cochrane Handbook for Systematic Reviews of Interventions Version 5.1.0 [updated March 2011]. The Cochrane Collaboration, 2011. available from www.handbook.cochrane.org.
- Hiller, C. E., Kilbreath, S. L., & Refshauge, K. M. (2011). Chronic ankle instability: evolution of the model. *Journal of Athletic Training*, 46(2), 133-141. doi:10.4085/1062-6050-46.2.133
- Hiller, C. E., Nightingale, E. J., Raymond, J., Kilbreath, S. L., Burns, J., Black, D. A., & Refshauge, K. M. (2012). Prevalence and impact of chronic musculoskeletal ankle

disorders in the community. *Archives of Physical Medicine & Rehabilitation*, 93(10), 1801-1807. doi:10.1016/j.apmr.2012.04.023

- Hiller, C. E., Refshauge, K. M., Bundy, A. C., Herbert, R. D., & Kilbreath, S. L. (2006). The Cumberland ankle instability tool: a report of validity and reliability testing. *Archives of Physical Medicine & Rehabilitation*, 87(9), 1235-1241.
 doi:10.1016/j.apmr.2006.05.022
- Hing, W., Hall, T., Rivett, D. A., Vicenzino, B., & Mulligan, B. (2015). *The Mulligan Concept* of Manual Therapy (Vol. 13). Australia: Elsevier

Hintermann, B. (2005). Total ankle arthroplasty. New York: Springer.

- Hoch, J. M., Legner, J. L., Lorete, C., & Hoch, M. C. (2016). The validity of the quick-FAAM in patients seeking treatment for an acute or sub acute foot or ankle health condition. *Journal of Sport Rehabilitation. Technical Notes*, 1-13. doi:10.1123/jsr.2016-0089
- Hoch, M. C. (2011). The effect of joint mobilization on functional outcomes associated with chronic ankle instability. (Doctor of Philosophy) University of Kentucky, Lexington, Kentucky.
- Hoch, M. C., Andreatta, R. D., Mullineaux, D. R., English, R. A., Medina McKeon, J. M., Mattacola, C. G., & McKeon, P. O. (2012). Two-week joint mobilization intervention improves self-reported function, range of motion, and dynamic balance in those with chronic ankle instability. *Journal of Orthopaedic Research*, 30(11), 1798-1804. doi:http://dx.doi.org/10.1002/jor.22150
- Hoch, M. C., Farwell, K. E., Gaven, S. L., & Weinhandl, J. T. (2015). Weight-bearing dorsiflexion range of motion and landing biomechanics in individuals with chronic ankle instability. *Journal of Athletic Training*, 50(8), 833-839. doi:10.4085/1062-6050-50.5.07
- Hoch, M. C., & McKeon, P. O. (2010). The effectiveness of mobilization with movement at improving dorsiflexion after ankle sprain. *Journal of Sport Rehabilitation. Technical Notes, 19*(2), 226-232.

- Hoch, M. C., & McKeon, P. O. (2011). Joint mobilization improves spatiotemporal postural control and range of motion in those with chronic ankle instability. *Journal of Orthopaedic Research*, 29(3), 326-332. doi:http://dx.doi.org/10.1002/jor.21256
- Hoch, M. C., Mullineaux, D. R., Andreatta, R. D., English, R. A., Medina-McKeon, J. M., Mattacola, C. G., & McKeon, P. O. (2014). Effect of a 2-week joint mobilization intervention on single-limb balance and ankle arthrokinematics in those with chronic ankle instability. *Journal of Sport Rehabilitation*, 23(1), 18-26. doi:http://dx.doi.org/10.1123/jsr.2012-0125
- Hoch, M. C., Staton, G. S., McKeon, J. M. M., Mattacola, C. G., & McKeon, P. O. (2012).
 Dorsiflexion and dynamic postural control deficits are present in those with chronic ankle instability. *Journal of Science & Medicine in Sport*, 15(6), 574-579.
 doi:http://dx.doi.org/10.1016/j.jsams.2012.02.009
- Hoffman, E., Paller, D., Koruprolu, S., Drakos, M., Behrens, S. B., Crisco, J. J., &
 DiGiovanni, C. W. (2011). Accuracy of plain radiographs versus 3D analysis of ankle stress test. *Foot & Ankle International*, 32(10), 994-999. doi:10.3113/FAI.2011.0994
- Hogan, M. V., Mani, S. B., Chan, J. Y., Do, H., Deland, J. T., & Ellis, S. J. (2016). Validation of the foot and ankle outcome score (FAOS) for hallux rigidus. *The Musculoskeletal Journal of Hospital for Special Surgery*, 12(1), 44-50. doi:10.1007/s11420-015-9466-4
- Holmer, P., Sondergaard, L., Konradsen, L., Nielsen, P. T., & Jorgensen, L. N. (1994). Epidemiology of sprains in the lateral ankle and foot. *Foot & Ankle International*, 15(2), 72-74.
- Hopkinson, W. J., St Pierre, P., Ryan, J. B., & Wheeler, J. H. (1990). Syndesmosis sprains of the ankle. *Foot & Ankle, 10*(6), 325-330. doi:10.1177/107110079001000607
- Hopper, D., Samsson, K., Hulenik, T., Ng, C., Hall, T., & Robinson, K. (2009). The influence of Mulligan ankle taping during balance performance in subjects with unilateral chronic ankle instability. *Physical Therapy in Sport, 10*(4), 125-130. doi:http://dx.doi.org/10.1016/j.ptsp.2009.07.005

- Houston, M. N., Hoch, J. M., & Hoch, M. C. (2015). Patient-reported outcome measures in individuals with chronic ankle instability: a systematic review. *Journal of Athletic Training*, 50(10), 1019-1033. doi:10.4085/1062-6050-50.9.01
- Houston, M. N., McKeon, P. O., & Hoch, M. C. (2013). Foot and ankle ability measure scores in patients with chronic ankle instability following joint mobilization. *International Journal of Athletic Therapy & Training*, 18(2), 4-7.
- Howick, J., Friedemann, C., Tsakok, M., Watson, R., Tsakok, T., Thomas, J., Perera, R.,
 Fleming, S., & Heneghan, C. (2013). Are treatments more effective than placebos? a systematic review and meta-analysis. *PloS One*, 8(5), e62599-e62599.
 doi:10.1371/journal.pone.0062599
- Hubbard, T. J., & Cordova, M. (2009). Mechanical instability after an acute lateral ankle sprain. Archives of Physical Medicine & Rehabilitation, 90(7), 1142-1146. doi:10.1016/j.apmr.2009.01.020
- Hubbard, T. J., & Hertel, J. (2008). Anterior positional fault of the fibula after sub-acute lateral ankle sprains. *Manual Therapy*, 13(1), 63-67. doi:http://dx.doi.org/10.1016/j.math.2006.09.008
- Hubbard, T. J., Hertel, J., & Sherbondy, P. (2006). Fibular position in individuals with selfreported chronic ankle instability. *Journal of Orthopaedic & Sports Physical Therapy*, 36(1), 3-9. doi:10.2519/jospt.2006.36.1.3
- Hubbard, T. J., & Hicks-Little, C. A. (2008). Ankle ligament healing after an acute ankle sprain: an evidence-based approach. *Journal of Athletic Training*, 43(5), 523-529. doi:10.4085/1062-6050-43.5.523
- Irnich, D., Behrens, N., Molzen, H., Konig, A., Gleditsch, J., Krauss, M., Natalis, M., Senn, E., Beyer, A., & Schops, P. (2001). Randomised trial of acupuncture compared with conventional massage and "sham" laser acupuncture for treatment of chronic neck pain. *British Medical Journal*, 322(7302), 1574-1578.

- Isakov, E., & Mizrahi, J. (1997). Is balance impaired by recurrent sprained ankle? *British journal of Sports Medicine*, *31*(1), 65-67.
- Jaber, H., Lohman, E., Daher, N., Bains, G., Nagaraj, A., Mayekar, P., Shanbhag, M., & Alameri, M. (2018). Neuromuscular control of ankle and hip during performance of the star excursion balance test in subjects with and without chronic ankle instability. *PloS One, 13*(8), e0201479. doi:10.1371/journal.pone.0201479
- Jaeschke, R., Guyatt, G. H., & Sackett, D. L. (1994). Users' guides to the medical literature. III. How to use an article about a diagnostic test. B. What are the results and will they help me in caring for my patients? The evidence-based medicine working group. *The Journal of the American Medical Association*, 271(9), 703-707.
- Jensen, M. P., Chen, C., & Brugger, A. M. (2003). Interpretation of visual analog scale ratings and change scores: a reanalysis of two clinical trials of postoperative pain. *Journal of Pain*, 4(7), 407-414.
- Jensen, M. P., Karoly, P., & Braver, S. (1986). The measurement of clinical pain intensity: a comparison of six methods. *Pain*, 27(1), 117-126.
- Johannsen, A. (1978). Radiological diagnosis of lateral ligament lesion of the ankle. a comparison between talar tilt and anterior drawer sign. *Acta Orthop Scand*, *49*(3), 295-301.
- Jolman, S., Robbins, J., Lewis, L., Wilkes, M., & Ryan, P. (2017). Comparison of magnetic resonance imaging and stress radiographs in the evaluation of chronic lateral ankle instability. *Foot & Ankle International*, 38(4), 397-404. doi:10.1177/1071100716685526
- Joseph, L. C., de Busser, N., Brantingham, J. W., Globe, G. A., Cassa, T. K., Korporaal, C., & Bonello, R. (2010). The comparative effect of muscle energy technique versus manipulation for the treatment of chronic recurrent ankle sprain. *Journal of the American Chiropractic Association 2010 Oct;47(5):8-22.*

- Kaminski, T. W., Buckley, B. D., Powers, M. E., Hubbard, T. J., & Ortiz, C. (2003). Effect of strength and proprioception training on eversion to inversion strength ratios in subjects with unilateral functional ankle instability. *British Journal of Sports Medicine*, 37(5), 410-415; discussion 415. doi:10.1136/bjsm.37.5.410
- Kamper, S. J., Ostelo, R. W. J. G., Knol, D. L., Maher, C. G., de Vet, H. C. W., & Hancock, M. J. (2010). Global perceived Effect scales provided reliable assessments of health transition in people with musculoskeletal disorders, but ratings are strongly influenced by current status. *Journal of Clinical Epidemiology*, *63*(7), 760-766.e761. doi:https://doi.org/10.1016/j.jclinepi.2009.0909
- Karlsson, A., & Frykberg, G. (2000). Correlations between force plate measures for assessment of balance. *Clinical Biomechanics*, 15(5), 365-369.
- Karlsson, J., Bergsten, T., Lansinger, O., & Peterson, L. (1989). Surgical treatment of chronic lateral instability of the ankle joint. *The American Journal of Sports Medicine*, 17(2), 268-274. doi:10.1177/036354658901700220
- Kavanagh, J. (1999). Is there a positional fault at the inferior tibiofibular joint in patients withacute or chronic ankle sprains compared to normals? *Manual Therapy*, 4(1), 19-24. doi:http://dx.doi.org/10.1016/S1356-689X(99)80005-8
- Kingsley, J. D., Demchak, T., & Mathis, R. (2014). Low-level laser therapy as a treatment for chronic pain. *Frontiers in Physiology*, 5, 306. doi:10.3389/fphys.2014.00306
- Kivlan, B. R., Martin, R. L., & Wukich, D. K. (2011). Responsiveness of the foot and ankle ability measure (FAAM) in individuals with diabetes. *Foot (Edinb)*, 21(2), 84-87. doi:10.1016/j.foot.2011.04.004
- Ko, J., Rosen, A. B., & Brown, C. N. (2015). Cross-cultural adaptation and validation of the korean version of the Cumberland ankle instability tool. *International Journal of Sports Physical Therapy*, 10(7), 1007-1014.
- Kobayashi, T., & Gamada, K. (2014). Lateral ankle sprain and chronic ankle instability: a critical review. *Foot & Ankle Specialist*, 7(4), 298-326.

- Kobayashi, T., Suzuki, E., Yamazaki, N., Suzukawa, M., Akaike, A., Shimizu, K., & Gamada,
 K. (2014). Fibular malalignment in individuals with chronic ankle instability. *Journal* of Orthopaedic & Sports Physical Therapy, 44(11), 872-878.
 doi:10.2519/jospt.2014.5217
- Kohne, E., Andrew, J., Charmaine, K., Jennifer, P. L., James, W. B., & Gary, G. (2007). A prospective, single-blinded, randomized, controlled clinical trial of the effects of manipulation on proprioception and ankle dorsiflexion in chronic recurrent ankle sprain. *Journal of the American Chiropractic Association*, 44(5), 7-17.
- Konor, M. M., Morton, S., Eckerson, J. M., & Grindstaff, T. L. (2012). Reliability of three measures of ankle dorsiflexion range of motion. *International Journal of Sports physical therapy. 2012; 7(3): 279–287., 7*(3), 279–287.
- Konradsen, L., Bech, L., Ehrenbjerg, M., & Nickelsen, T. (2002). Seven years follow-up after ankle inversion trauma. *Scandinavian Journal of Medicine & Science In Sports*, 12(3), 129-135.
- Kosik, K. B., McCann, R. S., Terada, M., & Gribble, P. A. (2016). Therapeutic interventions for improving self-reported function in patients with chronic ankle instability: a systematic review. *British Journal of Sports Medicine*. doi:10.1136/bjsports-2016-096534
- Krabak, B. J., & Baima, J. (2008). Ankle sprain. In *Essentials of Physical Medicine And Rehabilitation* (2 ed.). Philadelphia: Saunders.
- Kremer, E., Atkinson, J. H., & Ignelzi, R. J. (1981). Measurement of pain: patient preference does not confound pain measurement. *Pain*, 10(2), 241-248.
- Kumari, N., Megha, N. A., & Paresh, P. (2014). Efficacy of weight bearing distal tibiofibular joint mobilization with movement (MWM) in improving pain, dorsiflexion range and function in patients with postacute lateral ankle sprain. *International Journal of Physiotherapy and Research,* 2(3), 542-548.

- Kunugi, S., Masunari, A., Noh, B., Mori, T., Yoshida, N., & Miyakawa, S. (2017). Cross-cultural adaptation, reliability, and validity of the Japanese version of the Cumberland ankle instability tool. *Disability & Rehabilitation*, 39(1), 50-58. doi:10.3109/09638288.2016.1138555
- Lalkhen, A. G., & McCluskey, A. (2008). Clinical tests: sensitivity and specificity. Continuing Education in Anaesthesia Critical Care & Pain, 8(6), 221-223. doi:10.1093/bjaceaccp/mkn041
- Landrum, E. L., Kelln, B. M., Parente, W. R., Ingersoll, C. D., & Hertel, J. (2008). Immediate effects of anterior-to-posterior talocrural joint mobilization after prolonged ankle immobilization: a preliminary study. *The Journal of Manual & Manipulative Therapy* 2008;16(2):100-105.
- Leanderson, J., Wykman, A., & Eriksson, E. (1993). Ankle sprain and postural sway in basketball players. *Knee Surgery, Sports Traumatology, Arthroscopy, 1*(3-4), 203-205.
- Lee, D. Y., & Kim, S. J. (2005). Clinical study of traction therapy for ankle sprain. *Journal of Oriental Rehabilitation Medicine*, 15(4), 39-46
- Lee, K. M., Chung, C. Y., Kwon, S.-S., Chung, M. K., Won, S. H., Lee, S. Y., & Park, M. S. (2013). Relationship between stress ankle radiographs and injured ligaments on MRI. *Skeletal Radiology*, 42(11), 1537-1542. doi:10.1007/s00256-013-1704-6
- Lee, K. Y., Lee, H. J., Kim, S. E., Choi, P. B., Song, S. H., & Jee, Y. S. (2012). Short term rehabilitation and ankle instability. *International Journal of Sports Medicine*, 33(6), 485-496. doi:http://dx.doi.org/10.1055/s-0032-1301933
- Li, H. Y., Zhou, R. S., Hua, Y. H., & Chen, S. Y. (2017). MRI identification of the fibular and talus position in patients with mechanical ankle instability. *International Journal of Sports Medicine*, 38(7), 546-550. doi:10.1055/s-0043-106741
- Linens, S. W., Ross, S. E., Arnold, B. L., Gayle, R., & Pidcoe, P. (2014). Postural-stability tests that identify individuals with chronic ankle instability. *Journal of Athletic Training*, 49(1), 15-23. doi:10.4085/1062-6050-48.6.09

- Lohrer, H., Nauck, T., Gehring, D., Wissler, S., Braag, B., & Gollhofer, A. (2015).
 Differences between mechanically stable and unstable chronic ankle instability subgroups when examined by arthrometer and FAAM-G. *Journal of Orthopaedic Surgery And Research*, 10, 32. doi:10.1186/s13018-015-0171-2
- Lopez-Rodriguez, S., de-Las-Penas, C. F., burquerque-Sendin, F., Rodriguez-Blanco, C., & Palomeque-del-Cerro, L. (2007). Immediate effects of manipulation of the talocrural joint on stabilometry and baropodometry in patients with ankle sprain. *Journal of Manipulative & Physiological Therapeutics*, 30(3), 186-192.
- Loudon, J. K., Reiman, M. P., & Sylvain, J. (2014). The efficacy of manual joint mobilisation/manipulation in treatment of lateral ankle sprains: a systematic review. *British Journal of Sports Medicine, 48*(5), 365-370 366p. doi:10.1136/bjsports-2013-092763
- Lubbe, D., Lakhani, E., Brantingham, J., Parkin-Smith, G., Cassa, T., Globe, G., & Korporaal, C. (2015). Manipulative therapy and rehabilitation for recurrent ankle sprain with functional instability: a short-term, assessor-blind, parallel-group randomized trial. *Journal of Manipulative & Physiological Therapeutics, 38*(1), 22-34.
- Lynch, S. A. (2002). Assessment of the injured ankle in the athlete. *Journal of Athletic Training*, *37*(4), 406-412.
- Maitland, G. D. (2005). *Maitland's Vertebral Manipulation* (K. B. Hengeveld & K. English Eds. 7th ed.). Philadelphia, PA. : Elsevier.
- Mani, S. B., Brown, H. C., Nair, P., Chen, L., Do, H. T., Lyman, S., Deland, J. T., & Ellis, S. J. (2013). Validation of the foot and ankle outcome score (FAOS) in adult acquired flatfoot deformity. *Foot & Ankle International*, *34*(8), 1140-1146. doi:10.1177/1071100713483117
- Mani, S. B., Do, H., Vulcano, E., Hogan, M. V., Lyman, S., Deland, J. T., & Ellis, S. J. (2015). Evaluation of the foot and ankle outcome score in patients with osteoarthritis of the ankle. *Bone & Joint Journal*, *97-B*(5), 662.

- Marron-Gomez, D., Rodriguez-Fernandez, A., & Martin-Urrialde, J. (2015). The effect of two mobilization techniques on dorsiflexion in people with chronic ankle instability. *Physical Therapy in Sport, 16*(1), 10-15.
- Martin, R. L., Davenport, T. E., Paulseth, S., Wukich, D. K., & Godges, J. J. (2013). Ankle stability and movement coordination impairments: ankle ligament sprains. *Journal of Orthopaedic & Sports Physical Therapy*, 43(9), A1-A40. doi:10.2519/jospt.2013.0305
- Martin, R. L., Irrgang, J. J., Burdett, R. G., Conti, S. F., & Van Swearingen, J. M. (2005). Evidence of validity for the foot and ankle ability measure (FAAM). *Foot & Ankle International*, 26(11), 968-983. doi:10.1177/107110070502601113
- Mau, H., & Baker, R. T. (2014). A modified mobilization-with-movement to treat a lateral ankle sprain. *International Journal of Sports Physical Therapy*, *9*(4), 540-548.
- Mavi, A., Yildirim, H., Gunes, H., Pestamalci, T., & Gumusburun, E. (2002). The fibular incisura of the tibia with recurrent sprained ankle on magnetic resonance imaging. *Saudi Medical Journal*, 23(7), 845-849.
- Mazaheri, M., Salavati, M., Negahban, H., Sohani, S. M., Taghizadeh, F., Feizi, A., Karimi, A., & Parnianpour, M. (2010). Reliability and validity of the Persian version of foot and ankle ability measure (FAAM) to measure functional limitations in patients with foot and ankle disorders. *Osteoarthritis & Cartilage, 18*(6), 755-759. doi:https://doi.org/10.1016/j.joca.2010.03.006
- McCann, R. S., & Gribble, P. A. (2016). Resilience and Self-Efficacy: A Theory-based model of chronic ankle instability. *International Journal Of Athletic Therapy & Training*, 21(3), 32-37.
- McCarthy, C., Bialosky, J., & Rivett, D. (2015). Spinal manipulation. In G. Jull, A. Moore, D. Falla, J. Lewis, C. McCarthy, & M. Sterling (Eds.), *Grieve's Modern Musculoskeletal Physiotherapy* (4th ed., pp. 278-282): Elsevier.

- McCriskin, B. J., Cameron, K. L., Orr, J. D., & Waterman, B. R. (2015). Management and prevention of acute and chronic lateral ankle instability in athletic patient populations. *World Journal of Orthopedics*, 6(2), 161-171. doi:10.5312/wjo.v6.i2.161
- McKay, G. D., Goldie, P. A., Payne, W. R., & Oakes, B. W. (2001). Ankle injuries in basketball: injury rate and risk factors. *British Journal of Sports Medicine*, 35(2), 103-108. doi:10.1136/bjsm.35.2.103
- McKeon, P. O., & Hertel, J. (2008). Systematic review of postural control and lateral ankle instability, part i: can deficits be detected with instrumented testing. *Journal of Athletic Training*, 43(3), 293-304.
- McKiernan, S., Fenech, M., Fox, D., & Stewart, I. (2017). Sonography of the ankle: The lateral ankle and ankle sprains. *Sonography*, *4*(4), 146-155. doi:10.1002/sono.12120
- Merlin, D. J., McEwan, I. M., & Thom, J. M. (2005). Mulligan's mobilisation with movement technique for lateral ankle pain and the use of magnetic resonance imaging to evaluate the "positional fault" hypothesis. Paper presented at the *International Congress on Sports Rehabilitation & Traumatology* Bologna, Italy.
- Miklovic, T. M., Donovan, L., Protzuk, O. A., Kang, M. S., & Feger, M. A. (2018). Acute lateral ankle sprain to chronic ankle instability: a pathway of dysfunction. *The Physician & Sports Medicine*, 46(1), 116-122. doi:10.1080/00913847.2018.1409604
- Moreira, T. S., Magalhaes Lde, C., Silva, R. D., Martin, R. L., & Resende, M. A. (2016).
 Translation, cross-cultural adaptation and validity of the Brazilian version of the foot and ankle ability measure questionnaire. *Disability & Rehabilitation, 38*(25), 2479-2490. doi:10.3109/09638288.2015.1137979
- Mulligan, B. R. (1993). Mobilisations with movement (MWM'S). *Journal of Manual & Manipulative Therapy*, *1*(4), 154-156. doi:10.1179/jmt.1993.1.4.154
- Mulligan, B. R. (1995). *Manual Therapy: NAGS, SNAGS, MWMs, etc* (6th ed ed.). Plane View Services, Wellington.

- Mulligan Concept Teachers Association. (2019). Become a practitioner, The certified Mulligan concept practitioner (CMP) certification. Accessed 06.07.2019, Retrieved from https://bmulligan.com/certified-practitioners/become-a-practitioner/
- Mutlu, E. K., & Ozdincler, A. R. (2015). Reliability and responsiveness of algometry for measuring pressure pain threshold in patients with knee osteoarthritis. *Journal of Physical Therapy Science*, 27(6), 1961-1965. doi:10.1589/jpts.27.1961
- Nakagawa, L., & Hoffman, M. (2004). Performance in static, dynamic, and clinical tests of postural control in individuals with recurrent ankle sprains. *Journal of Sport Rehabilitation*, 13(3), 255-268.
- Nambi, G., & Shah, B. T. (2012). Kinesio taping versus Mulligan's mobilization with movement in sub-acute lateral ankle sprain in secondary school Hockey players comparative study. *International Journal of Pharmaceutical Science & Health Care*, 2(2).
- National Institutes of Health. (2014). Quality assessment tool for observational cohort and cross-sectional studies.
- Nauck, T., & Lohrer, H. (2011). Translation, cross-cultural adaption and validation of the German version of the foot and ankle ability measure for patients with chronic ankle instability. *British Journal of Sports Medicine*, 45(10), 785-790. doi:10.1136/bjsm.2009.067637
- Negahban, H., Mazaheri, M., Salavati, M., Sohani, S. M., Askari, M., Fanian, H., & Parnianpour, M. (2010). Reliability and validity of the foot and ankle outcome score: a validation study from Iran. *Clinical Rheumatology*, 29(5), 479-486. doi:10.1007/s10067-009-1344-3
- Nijs, J., Van Houdenhove, B., & Oostendorp, R. A. (2010). Recognition of central sensitization in patients with musculoskeletal pain: application of pain neurophysiology in manual therapy practice. *Manual Therapy*, 15(2), 135-141. doi:10.1016/j.math.2009.12.001

- Nilsson, S. (1983). Sprains of the lateral ankle ligaments. An epidemiological and clinical study with reference to different forms of conservative treatment. *Journal Oslo City Hospitals, 33*(2-3), 13-36.
- Norkus, S. A., & Floyd, R. T. (2001). The anatomy and mechanisms of syndesmotic ankle sprains. *Journal of Athletic Training*, *36*(1), 68-73.
- Nyska, M., & Mann, G. (2002). The Unstable Ankle. Champaign, IL: Human Kinetics.
- O'Brien, T., & Vicenzino, B. (1998). A study on the effects of Mulligan's mobilization with movement treatment of lateral ankle pain using a case study design. *Manual Therapy*, *3*(2), 78-84.
- Oatis, C. A. (2004). *Kinesiology; the Mechanics & Pathomechanics of Human Movement* (David Payne Ed. 2 ed.): Lippincott Williams & Wilkins.
- Owens, J. E., & Menard, M. (2011). The Quantification of placebo effects within a general model of health care outcomes. *Journal of Alternative & Complementary Medicine*, 17(9), 817-821. doi:10.1089/acm.2010.0566
- Parker, R., & Jelsma, J. (2010). The prevalence and functional impact of musculoskeletal conditions amongst clients of a primary health care facility in an under-resourced area of Cape Town. *BMC Musculoskelet Disorders*, 11, 2. doi:10.1186/1471-2474-11-2
- PEDro-scale. (1999). Accessed 05.12.2015, Retrieved from http://www.pedro.org.au/english/downloads/pedro-scale/.
- Pellow, J. E., & Brantingham, J. W. (2001). The efficacy of adjusting the ankle in the treatment of subacute and chronic grade I and grade II ankle inversion sprains. *Journal* of Manipulative & Physiological Therapeutics, 24(1), 17-24. doi:10.1067/mmt.2001.112015
- Penso, M. (2008). The effectiveness of mobilisation with movement for chronic medial ankle pain: a case study. *South African Journal of Physiotherapy*, *64*(1), 13-16.
- Peters, J. W., Trevino, S. G., & Renstrom, P. A. (1991). Chronic lateral ankle instability. *Foot & Ankle, 12*(3), 182-191.
- Pettman, E. (2007). A history of manipulative therapy. *The Journal of Manual & Manipulative therapy*, 15(3), 165-174.
- Pionnier, R., Decoufour, N., Barbier, F., Popineau, C., & Simoneau-Buessinger, E. (2016). A new approach of the star excursion balance test to assess dynamic postural control in people complaining from chronic ankle instability. *Gait & Posture, 45*, 97-102. doi:10.1016/j.gaitpost.2016.01.013
- Plante, J. E., & Wikstrom, E. A. (2013). Differences in clinician-oriented outcomes among controls, copers, and chronic ankle instability groups. *Physical Therapy in Sport*, 14(4), 221-226. doi:10.1016/j.ptsp.2012.09.005
- Plante, P. D. (2012). The effects of talar mobilization in individuals without previous history of ankle sprains and those with chronic ankle instability. (Master of Science), Wilfrid Laurier University, Canada.
- Pope, M., Chinn, L., Mullineaux, D., McKeon, P. O., Drewes, L., & Hertel, J. (2011). Spatial postural control alterations with chronic ankle instability. *Gait & Posture*, 34(2), 154-158. doi:10.1016/j.gaitpost.2011.04.012
- Pope, R., Herbert, R., & Kirwan, J. (1998). Effects of ankle dorsiflexion range and preexercise calf muscle stretching on injury risk inarmy recruits. *Australian Journal of Physiotherapy*, 44(3), 165-172.
- Powden, C. J., Hoch, J. M., & Hoch, M. C. (2015). Reliability and minimal detectable change of the weight-bearing lunge test: a systematic review. *Manual Therapy*, 20(4), 524-532. doi:10.1016/j.math.2015.01.004
- Powden, C. J., Hogan, K. K., Wikstrom, E. A., & Hoch, M. C. (2016). The Effect of two forms of talocrural joint traction on dorsiflexion range of motion and postural control in those with chronic ankle instability. *Journal of Sport Rehabilitation. Technical Notes*, 1-19. doi:10.1123/jsr.2015-0152

- Powers, C. M., Ghoddosi, N., Straub, R. K., & Khayambashi, K. (2017). Hip strength as a predictor of ankle sprains in male soccer players: a prospective study. *Journal of Athletic Training*, 52(11), 1048-1055. doi:10.4085/1062-6050-52.11.18
- Prado, M. P., Fernandes, T. D., Camanho, G. L., Mendes, A. A. M., & Amodio, D. T. (2013). Mechanical instability after acute ankle ligament injury: randomized prospective comparison of two forms of conservative treatment. *Revista Brasileira de Ortopedia*, 48, 307-316.
- Price, D. D., McGrath, P. A., Rafii, A., & Buckingham, B. (1983). The validation of visual analogue scales as ratio scale measures for chronic and experimental pain. *Pain*, 17(1), 45-56.
- Rabin, A., & Kozol, Z. (2012). Weightbearing and nonweightbearing ankle dorsiflexion range of motion: are we measuring the same thing? *Journal of the American Podiatric Medical Association*, 102(5), 406-411.
- Ramiro-Gonzalez, M. D., Cano-de-la-Cuerda, R., De-la-Llave-Rincon, A. I., Miangolarra-Page, J. C., Zarzoso-Sanchez, R., & Fernandez-de-Las-Penas, C. (2012). Deep tissue hypersensitivity to pressure pain in individuals with unilateral acute inversion ankle sprain. *Pain Medicine*, 13(3), 361-367. doi:10.1111/j.1526-4637.2011.01302.x
- Rashid, S., Salick, K., Kashif, M., Ahmad, A., & Sarwar, K. (2013). To evaluate the efficacy of mobilization techniques in post-traumatic stiff ankle with and without paraffin wax bath. *Pakistan Journal of Medical Sciences, 29*(6). doi:http://dx.doi.org/10.12669/pjms.296.4127
- Reb, C. W., Saini, S. S., Stenson, J. F., Albana, M. F., Pedowitz, D. I., Raikin, S. M., & Daniel, J. N. (2017). Content relevance of the foot and ankle ability measure in patients with achilles tendon diseases. *Foot & Ankle Specialist*, 1938640017718342. doi:10.1177/1938640017718342
- Rebbeck, T., Moloney, N., Azoory, R., Hubscher, M., Waller, R., Gibbons, R., & Beales, D.(2015). Clinical Ratings of pain sensitivity correlate with quantitative measures in

people with chronic neck pain and healthy controls: cross-sectional study. *Physical Therapy*, *95*(11), 1536-1546. doi:10.2522/ptj.20140352

- Rehabilitation Measures Database. (2010). Rehab measures: lower quarter Y-balance test. Accessed 14/07/2016, 2016. Retrieved from http://www.rehabmeasures.org/Lists/RehabMeasures/PrintView.aspx?ID=1271
- Reid, A., Birmingham, T. B., & Alcock, G. (2007). Efficacy of mobilization with movement for patients with limited dorsiflexion after ankle sprain: a crossover trial. *Physiotherapy Canada*, 59(3), 166-172 167p.
- Reid, S. A., Callister, R., Snodgrass, S. J., Katekar, M. G., & Rivett, D. A. (2015). Manual therapy for cervicogenic dizziness: Long-term outcomes of a randomised trial. *Manual Therapy*, 20(1), 148-156. doi:10.1016/j.math.2014.08.003
- Reid, S. A., Rivett, D. A., Katekar, M. G., & Callister, R. (2008). Sustained natural apophyseal glides (SNAGs) are an effective treatment for cervicogenic dizziness. *Manual Therapy*, 13(4), 357-366. doi:10.1016/j.math.2007.03.006
- Renström, P. A., & Konradsen, L. (1997). Ankle ligament injuries. British Journal of Sports Medicine, 31(1), 11-20.
- RevMan. (2014). Review Manager (Version Version 5.3). Copenhagen: The Nordic Cochrane Centre: The Cochrane Collaboration.
- Ribbans, W. J., & Garde, A. (2013). Tibialis posterior tendon and deltoid and spring ligament injuries in the elite athlete. *Foot & Ankle Clinics*, 18(2), 255-291. doi:10.1016/j.fcl.2013.02.006
- Riddle, D. L., Pulisic, M., Pidcoe, P., & Johnson, R. E. (2003). Risk factors for plantar fasciitis: a matched case-control study. *Journal of Bone & Joint Surgery*, 85-a(5), 872-877.
- Roaas, A., & Andersson, G. B. (1982). Normal range of motion of the hip, knee and ankle joints in male subjects, 30-40 years of age. *Acta Orthopaedica Scandinavica*, 53(2), 205-208.

- Robinson, R. H., & Gribble, P. A. (2008). Support for a reduction in the number of trials needed for the star excursion balance test. *Archives of Physical Medicine & Rehabilitation*, 89(2), 364-370. doi:10.1016/j.apmr.2007.08.139
- Rodriguez-Merchan, E. C. (2012). Chronic ankle instability: diagnosis and treatment. *Archives* of Orthopaedic & Trauma Surgery, 132(2), 211-219. doi:10.1007/s00402-011-1421-3
- Roos, E. M., Brandsson, S., & Karlsson, J. (2001). Validation of the foot and ankle outcome score for ankle ligament reconstruction. *Foot & Ankle International*, 22(10), 788-794. doi:10.1177/107110070102201004
- Ross, S. E., & Guskiewicz, K. M. (2004). Examination of static and dynamic postural stability in individuals with functionally stable and unstable ankles. *Clinical Journal of Sport Medicine*, 14(6), 332-338.
- Ross, S. E., Guskiewicz, K. M., Gross, M. T., & Yu, B. (2009). Balance measures for discriminating between functionally unstable and stable ankles. *Medicine & Science In Sports & Exercise*, 41(2), 399-407. doi:10.1249/MSS.0b013e3181872d89
- Rubin, G., & Witten, M. (1960). The talar-tilt angle and the fibular collateral ligaments. A method for the determination of talar tilt. *Journal of Bone & Joint Surgery*, 42(2), 311-326.
- Rundle, E. (1988). The effect of grade IV mobilisation on the range of dorsiflexion in the weight-bearing position in non-acute sprained ankles. *Australian Journal of Physiotherapy*, 34(1), 56.
- Salkind, N. (2010). Last observation carried forward. *Encyclopedia of Research Design*, 686-687. doi:10.4135/9781412961288
- Saltzman, C. L., Salamon, M. L., Blanchard, G. M., Huff, T., Hayes, A., Buckwalter, J. A., & Amendola, A. (2005). Epidemiology of ankle arthritis: report of a consecutive series of 639 patients from a tertiary orthopaedic center. *The Iowa Orthopaedic Journal, 25*, 44-46.

- Sartorio, F., Vercelli, S., Bravini, E., Bargeri, S., Moroso, M., Plebani, G., & Ferriero, G. (2014). Foot and ankle ability measure: cross-cultural translation and validation of the Italian version of the ADL module (FAAM-I/ADL). *La Medicina Del Lavoro, 105*(5), 357-365.
- Schuenke, M., Schulte, E., & Schumacher, U. (2009). *Atlas of Anatomy* (A. M. Gilroy, B. R. MacPherson, & L. M. Ross Eds. 2nd ed.). Canada: Thieme Medical Publishers, Inc.
- Scranton, P. E., Jr., McDermott, J. E., & Rogers, J. V. (2000). The relationship between chronic ankle instability and variations in mortise anatomy and impingement spurs. *Foot & Ankle International*, 21(8), 657-664.
- Shi, X., Han, J., Witchalls, J., Waddington, G., & Adams, R. (2019). Does treatment duration of manual therapy influence functional outcomes for individuals with chronic ankle instability: a systematic review with meta-analysis? *Musculoskeletal Science & Practice*, 40, 87-95. doi:https://doi.org/10.1016/j.msksp.2019.01.015
- Shiravi, Z., Shadmehr, A., Moghadam, S. T., & Moghadam, B. A. (2017). Comparison of dynamic postural stability scores between athletes with and without chronic ankle instability during lateral jump landing. *Muscles, Ligaments & Tendons Journal, 7*(1), 119-124. doi:10.11138/mltj/2017.7.1.119
- Silva, R. D., Teixeira, L. M., Moreira, T. S., Teixeira-Salmela, L. F., & de Resende, M. A. (2017). Effects of anteroposterior talus mobilization on range of motion, pain, and functional capacity in participants with subacute and chronic ankle injuries: a controlled trial. *Journal of Manipulative & Physiological Therapeutics*, 40(4), 273-283. doi:10.1016/j.jmpt.2017.02.003

- Simon, J., Hall, E., & Docherty, C. (2014). Prevalence of chronic ankle instability and associated symptoms in university dance majors: an exploratory study. *Journal of Dance Medicine & Science, 18*(4), 178-184. doi:10.12678/1089-313x.18.4.178
- Sisson, L., Croy, T., Saliba, S., & Hertel, J. (2011). Comparison of ankle arthrometry to stress ultrasound imaging in the assessment of ankle laxity in healthy adults. *International Journal of Sports Physical Therapy*, 6(4), 297-305.
- Smits-Engelsman, B., Klerks, M., & Kirby, A. (2011). Beighton score: a valid measure for generalized hypermobility in children. *The Journal of Pediatrics*, 158(1), 119-123, 123 e111-114. doi:10.1016/j.jpeds.2010.07.021
- Soboroff, S. H., Pappius, E. M., & Komaroff, A. L. (1984). Benefits, risks, and costs of alternative approaches to the evaluation and treatment of severe ankle sprain. *Clinical Orthopaedics & Related Research* (183), 160-168.
- Someeh, M., Norasteh, A. A., Daneshmandi, H., & Asadi, A. (2015a). Immediate effects of Mulligan's fibular repositioning taping on postural control in athletes with and without chronic ankle instability. *Physical Therapy in Sport*, 16(2), 135-139.
- Someeh, M., Norasteh, A. A., Daneshmandi, H., & Asadi, A. (2015b). Influence of Mulligan Ankle taping on functional performance tests in healthy athletes and athletes with chronic ankle instability. *International Journal of Athletic Therapy & Training, 20*(1), 39-45.
- Southerst, D., Yu, H., Randhawa, K., Cote, P., D'Angelo, K., Shearer, H. M., Wong, J. J.,
 Sutton, D., Varatharajan, S., Goldgrub, R., Dion, S., Cox, J., Menta, R., Brown, C. K.,
 Stern, P. J., Stupar, M., Carroll, L. J., & Taylor-Vaisey, A. (2015). The effectiveness of
 manual therapy for the management of musculoskeletal disorders of the upper and
 lower extremities: a systematic review by the Ontario protocol for traffic injury
 management (OPTIMa) collaboration. *Chiropractic & Manual Therapies, 23*, 30.
 doi:http://dx.doi.org/10.1186/s12998-015-0075-6

- Spieth, P. M., Kubasch, A. S., Penzlin, A. I., Illigens, B. M.-W., Barlinn, K., & Siepmann, T. (2016). Randomized controlled trials - a matter of design. *Neuropsychiatric Disease & Treatment*, 12, 1341-1349. doi:10.2147/NDT.S101938
- Starkweather, A. R., Heineman, A., Storey, S., Rubia, G., Lyon, D. E., Greenspan, J., & Dorsey, S. G. (2016). Methods to measure peripheral and central sensitization using quantitative sensory testing: a focus on individuals with low back pain. *Applied Nursing Research*, 29, 237-241. doi:10.1016/j.apnr.2015.03.013
- Sterling, M., Jull, G., & Wright, A. (2001). Cervical mobilisation: concurrent effects on pain, sympathetic nervous system activity and motor activity. *Manual Therapy*, 6(2), 72-81. doi:10.1054/math.2000.0378
- Stratford, P., Gill, C., Westaway, M., & Binkley, J. (1995). Assessing disability and change on individual patients: a report of a patient specific measure. *Physiotherapy Canada*, 47, 258-263.
- Teixeira, L. M., Pires, T., Silva, R. D., & de Resende, M. A. (2013). Immediate effect of a single anteroposterior talus mobilization on dorsiflexion range of motion in participants with orthopedic dysfunction of the ankle and foot. *Journal of Manipulative & Physiological Therapeutics*, 36(6), 369-375. doi:http://dx.doi.org/10.1016/j.jmpt.2013.06.003
- Teo, T. K. B., Tay, K. H., Lin, S. E., Tan, S. G., Lo, R. H., Taneja, M., Irani, F. G., Sebastien, M. G., Lim, K. H., & Tan, B. S. (2010). Endovenous Laser therapy in the treatment of lower-limb venous ulcers. *Journal of Vascular & Interventional Radiology*, 21(5), 657-662. doi:https://doi.org/10.1016/j.jvir.2010.01.029
- Terada, M., Harkey, M. S., Wells, A. M., Pietrosimone, B. G., & Gribble, P. A. (2014). The influence of ankle dorsiflexion and self-reported patient outcomes on dynamic postural control in participants with chronic ankle instability. *Gait & Posture, 40*(1), 193-197. doi:10.1016/j.gaitpost.2014.03.186

- Terada, M., Pietrosimone, B. G., & Gribble, P. A. (2013). Therapeutic interventions for increasing ankle dorsiflexion after ankle sprain: a systematic review. *Journal of Athletic Training*, 48(5):696-709.
- Thompson, L. A., Badache, M., Cale, S., Behera, L., & Zhang, N. (2017). Balance performance as observed by center-of-pressure parameter characteristics in male soccer athletes and non-athletes. *Sports*, 5(4). doi:10.3390/sports5040086
- Tourné, Y., Besse, J. L., & Mabit, C. (2010). Chronic ankle instability. which tests to assess the lesions? which therapeutic options? *Orthopaedics & Traumatology: Surgery & Research*, 96(4), 433-446. doi:https://doi.org/10.1016/j.otsr.2010.04.005
- Trojian, T. H., & McKeag, D. B. (2006). Single leg balance test to identify risk of ankle sprains. *British Journal of Sports Medicine*, 40(7), 610-613. doi:10.1136/bjsm.2005.024356
- Tropp, H., Odenrick, P., & Gillquist, J. (1985). Stabilometry recordings in functional and mechanical instability of the ankle joint. *International Journal of Sports Medicine*, 6(3), 180-182. doi:10.1055/s-2008-1025836
- Tummala, S. V., Hartigan, D. E., Makovicka, J. L., Patel, K. A., & Chhabra, A. (2018). 10year epidemiology of ankle injuries in men's and women's collegiate basketball. *Orthopaedic Journal of Sports Medicine*, 6(11), 2325967118805400-2325967118805400. doi:10.1177/2325967118805400
- Uematsu, D., Suzuki, H., Sasaki, S., Nagano, Y., Shinozuka, N., Sunagawa, N., & Fukubayashi, T. (2015). Evidence of validity for the japanese version of the foot and ankle ability measure. *Journal of Athletic Training*, 50(1), 65-70. doi:10.4085/1062-6050-49.3.42
- Valentine, J. C., & Cooper, H. (2013). Effect size substantive interpretation guidelines: issues in the interpretation of effect sizes. In (pp. 1-7). Washington, DC: What Works Clearinghouse.

- Vallandingham, R. A., Gaven, S. L., & Powden, C. J. (2019). changes in dorsiflexion and dynamic postural control after mobilizations in individuals with chronic ankle instability: a systematic review and meta-analysis. *Journal of Athletic Training*, 54(4), 403-417. doi:10.4085/1062-6050-380-17
- van den Akker-Scheek, I., Seldentuis, A., Reininga, I. H. F., & Stevens, M. (2013). Reliability and validity of the Dutch version of the foot and ankle outcome score (FAOS). *BMC Musculoskeletal Disorders*, 14(1), 183. doi:10.1186/1471-2474-14-183
- van den Bekerom, M. P., Kerkhoffs, G. M., McCollum, G. A., Calder, J. D., & van Dijk, C. N. (2013). Management of acute lateral ankle ligament injury in the athlete. *Knee Surgery, Sports Traumatology, Arthroscopy, 21*(6), 1390-1395. doi:10.1007/s00167-012-2252-7
- van der Wees, P. J., Lenssen, A. F., Hendriks, E. J., Stomp, D. J., Dekker, J., & de Bie, R. A. (2006). Effectiveness of exercise therapy and manual mobilisation in ankle sprain and functional instability: a systematic review. *Australian Journal of Physiotherapy*, 52(1), 27-37.
- van der Windt, D. A., Koes, B. W., Deville, W., Boeke, A. J., de Jong, B. A., & Bouter, L. M. (1998). Effectiveness of corticosteroid injections versus physiotherapy for treatment of painful stiff shoulder in primary care: randomised trial. *British Medical Journal*, 317(7168), 1292-1296.
- van Lieshout, R., Reijneveld, E. A. E., van den Berg, S. M., Haerkens, G. M., Koenders, N. H., de Leeuw, A. J., van Oorsouw, R. G., Paap, D., Scheffer, E., Weterings, S., & Stukstette, M. J. (2016). Reproducibility of the modified star excursion balance test composite and specific reach direction scores. *International Journal of Sports Physical Therapy*, 11(3), 356-365.
- Van Ochten, J. M., Van Middelkoop, M., Meuffels, D., & Bierma-Zeinstra, S. M. A. (2014). Chronic complaints after ankle sprains: a systematic review on effectiveness of

treatments. Journal of Orthopaedic & Sports Physical Therapy, 44(11), 862-871. doi:10.2519/jospt.2014.5221

- Vicenzino, B., Branjerdporn, M., Teys, P., & Jordan, K. (2006). Initial changes in posterior talar glide and dorsiflexion of the ankle after mobilization with movement in individuals with recurrent ankle sprain. *Journal of orthopaedic & Sports Physical Therapy*, 36(7), 464-471. doi:10.2519/jospt.2006.2265
- Vicenzino, B., Hing, W., Rivett, D. A., & Hall, T. (2011). *Mobilisation with Movement: The Art & the Science*: Elsevier.
- Vicenzino, B., Jordan, K., Brandjerporn, M., Teys, P., & Bisset, L. (2005). Mobilization with movement treatment of the ankle changes joint position sense in subjects with recurrent sprains: a preliminary report. *Journal of Orthopaedic & Sports Physical Therapy*, 35(5), A-21.
- Vicenzino, B., Paungmali, A., & Teys, P. (2007). Mulligan's mobilization-with-movement, positional faults and pain relief: current concepts from a critical review of literature. *Manual Therapy*, 12(2), 98-108. doi:http://dx.doi.org/10.1016/j.math.2006.07.012
- Viera, A. J., & Garrett, J. M. (2005). Understanding interobserver agreement: the kappa statistic. *Family Medicine*, *37*(5), 360-363.
- Vuurberg, G., Kluit, L., & van Dijk, C. N. (2016). The Cumberland ankle instability tool (CAIT) in the Dutch population with and without complaints of ankle instability. *Knee Surgery, Sports Traumatology, Arthroscopy*. doi:10.1007/s00167-016-4350-4
- Waterman, B. R., Belmont, P. J., Jr., Cameron, K. L., Deberardino, T. M., & Owens, B. D.
 (2010). Epidemiology of ankle sprain at the United States military academy. *American Journal of Sports Medicine*, 38(4), 797-803. doi:10.1177/0363546509350757
- Waterman, B. R., Owens, B. D., Davey, S., Zacchilli, M. A., & Belmont, P. J., Jr. (2010). The epidemiology of ankle sprains in the United States. *The Journal of Bone & Joint Surgery*, 92(13), 2279-2284.

- Weel, H., Zwiers, R., Azim, D., Sierevelt, I. N., Haverkamp, D., van Dijk, C. N., & Kerkhoffs, G. M. (2016). Validity and reliability of a Dutch version of the foot and ankle ability measure. *Knee Surgery, Sports Traumatology, Arthroscopy, 24*(4), 1348-1354. doi:10.1007/s00167-014-3480-9
- Weerasekara, I., & Hiller, C. E. (2017). Chronic musculoskeletal ankle disorders in Sri Lanka. BMC Musculoskeletal Disorders, 18, 219. doi:10.1186/s12891-017-1580-7
- Weerasekara, I., Osmotherly, P., Snodgrass, S., Marquez, J., de Zoete, R., & Rivett, D. A. (2018). Clinical benefits of joint mobilization on ankle sprains: a systematic review and meta-analysis. *Archives of Physical Medicine & Rehabilitation*, 99(7), 1395-1412.e1395. doi:https://doi.org/10.1016/j.apmr.2017.07.019
- Weerasekara, I., Osmotherly, P. G., Snodgrass, S. J., Tessier, J., & Rivett, D. A. (2019).
 Effects of mobilisation with movement (MWM) on anatomical and clinical characteristics of chronic ankle instability: a randomised controlled trial protocol. *BMC Musculoskeletal Disorders*, 20(1), 75. doi:10.1186/s12891-019-2447-x
- Weerasekara, I., Osmotherly, P., Snodgrass, S., Tessier, J., & Rivett, D. (2019, May 10-13).
 Fibular position in chronic ankle instability (CAI) and the reliability of weigh-bearing radiographic measurements of fibular position. Paper presented at the *World Confederation for Physical Therapy Congress* Geneva, Switzerland.
- Weerasekara, I., Tennakoon, S. U. B., & Suraweera, H. J. (2019). Pain level, range of motion, and foot volume do not differ in recurrent and first-time ankle sprains. *Foot & Ankle Specialist*, 1938640019843331. doi:10.1177/1938640019843331
- Weerasekara, R. M. I. M., Tennakoon, S. U. B., & Suraweera, H. J. (2016). Contrast therapy and heat therapy in subacute stage of grade i and ii lateral ankle sprains. *Foot & Ankle Specialist*, 9(4), 307-323. doi:10.1177/1938640016640885
- Wells, A. M. (2012). Effects of joint mobilization on ankle dorsiflexion range of motion, dynamic postural control and self-reported patient outcomes in individuals with

chronic ankle instability. (Master of Science in Exercise Sciences), The University of Toledo, Toledo, Ohio.

- Westad, K., Tjoestolvsen, F., & Hebron, C. (2019). The effectiveness of Mulligan's mobilisation with movement (MWM) on peripheral joints in musculoskeletal (MSK) conditions: A systematic review. *Musculoskeletal Science & Practice, 39*, 157-163. doi:https://doi.org/10.1016/j.msksp.2018.12.001
- Wheeless, C. R. (March 4, 2018). Wheeless' Textbook of Orthopaedics. Accessed August 10, 2019. Retrieved from http://www.wheelessonline.com/
- Wikstrom, E. A., & Brown, C. N. (2014). Minimum reporting standards for copers in chronic ankle instability research. *Sports Medicine*, 44(2), 251-268. doi:10.1007/s40279-013-0111-4
- Wikstrom, E. A., Fournier, K. A., & McKeon, P. O. (2010). Postural control differs between those with and without chronic ankle instability. *Gait & Posture*, 32(1), 82-86. doi:http://dx.doi.org/10.1016/j.gaitpost.2010.03.015
- Wikstrom, E. A., Tillman, M. D., Chmielewski, T. L., Cauraugh, J. H., & Borsa, P. A. (2007).
 Dynamic postural stability deficits in subjects with self-reported ankle instability. *Medicine & Science in Sports & Exercise*, 39(3), 397-402.
 doi:10.1249/mss.0b013e31802d3460
- Wikstrom, E. A., Tillman, M. D., Chmielewski, T. L., Cauraugh, J. H., Naugle, K. E., & Borsa, P. A. (2010). Dynamic postural control but not mechanical stability differs among those with and without chronic ankle instability. *Scandinavian Journal of Medicine & Science in Sports, 20*(1), e137-144. doi:10.1111/j.1600-0838.2009.00929.x
- Wikstrom, E. A., Tillman, M. D., Chmielewski, T. L., Cauraugh, J. H., Naugle, K. E., & Borsa, P. A. (2012). Discriminating between copers and people with chronic ankle instability. *Journal of Athletic Training*, 47(2), 136-142.

- Wikstrom, E. A. & McKeon, P. O. (2011). Manipulative therapy effectiveness following acute lateral ankle sprains: a systematic review. *Athletic Training & Sports Health Care*, 3(6), 271-279. doi:http://dx.doi.org/10.3928/19425864-20110131-02
- Woodman, R., Berghorn, K., Underhill, T., & Wolanin, M. (2013). Utilization of mobilization with movement for an apparent sprain of the posterior talofibular ligament: a case report. *Manual Therapy*, 18(1), e1-7. doi:10.1016/j.math.2012.03.014
- Woolf, A. D. P., B. (2003). Burden of major musculoskeletal conditions. Bulletin of the World Health Organization, 81(9), 646-656.
- Woolf, C. J. (1983). Evidence for a central component of post-injury pain hypersensitivity. *Nature, 306*(5944), 686-688.
- Wright, A. (1999). Recent concepts in the neurophysiology of pain. *Manual Therapy*, 4(4), 196-202. doi:10.1054/math.1999.0207
- Wright, C. J., Arnold, B. L., Ross, S. E., & Linens, S. W. (2014). Recalibration and validation of the Cumberland ankle instability tool cut-off score for individuals with chronic ankle instability. *Archives of Physical Medicine & Rehabilitation*, 95(10), 1853-1859. doi:10.1016/j.apmr.2014.04.017
- Wright, C. J., Linens, S. W., & Cain, M. S. (2017). Establishing the minimal clinical important difference and minimal detectable change for the Cumberland ankle instability tool. *Archives of Physical Medicine & Rehabilitation, 98*(9), 1806-1811. doi:10.1016/j.apmr.2017.01.003
- Yeo, H. K., & Wright, A. (2011). Hypoalgesic effect of a passive accessory mobilisation technique in patients with lateral ankle pain. *Manual Therapy*, 16(4), 373-377. doi:http://dx.doi.org/10.1016/j.math.2011.01.001
- Young, R., Nix, S., Wholohan, A., Bradhurst, R., & Reed, L. (2013). Interventions for increasing ankle joint dorsiflexion: a systematic review and meta-analysis. *Journal of Foot & Ankle Research, 6*, 46-46. doi:10.1186/1757-1146-6-46

Yu, X. H. (2007). Clinical observation on acupuncture and Chinese naprapathy for the treatment of ankle sprain. *Zhongguo Zhongyi Gushangke Zazhi [Chinese Journal of Traditional Medicine Traumatology & Orthopedics]*, 12(2), 14-15.

Appendices

Appendix 1 Statements of collaboration from co-authors

Statement from Darren A. Rivett relating to papers published with Ishanka Weerasekara

I, Darren A. Rivett, attest that Research Higher Degree candidate, Ishanka Weerasekara contributed to the listed publications by contributing to the conception and design of the studies, conducting and writing up the literature review, the collection of data, undertaking the statistical analysis, description and interpretation of the results, and writing the discussion and conclusions.

- Weerasekara, I., Osmotherly, P., Snodgrass, S., Marquez, J., de Zoete, R., & Rivett, D. A. (2018). Clinical benefits of joint mobilisation on ankle sprains: a systematic review and metaanalysis. *Archives of physical medicine and rehabilitation*, 99(7), 1395-1412.e1395. doi:https://doi.org/10.1016/j.apmr.2017.07.019
- Weerasekara, I., Osmotherly, P. G., Snodgrass, S. J., Tessier, J., & Rivett, D. A. (2019). Effects of mobilisation with movement (MWM) on anatomical and clinical characteristics of chronic ankle instability: a randomised controlled trial protocol. *BMC musculoskeletal disorders*, 20(1), 75. doi:10.1186/s12891-019-2447-x
- Weerasekara, I., Osmotherly, P., Snodgrass, S., Tesseir, J., & Rivett, D. A. (2019). Is the fibula positioned anteriorly in weight-bearing in individuals with chronic ankle instability? (under review).

 Weerasekara, I., Snodgrass, S., Osmotherly, P., Tessier, J., & Rivett, D. A. (2019). Clinical characteristics of chronic ankle instability: a cross sectional study (under review).

.....

Date: 15 August 2019

Professor Darren A. Rivett

.....

Date: 15 August 2019

Ishanka Weerasekara

Date: 15 August 2019

Dr Lesley MacDonald-Wicks

Statement from Peter G. Osmotherly relating to papers published with Ishanka Weerasekara

I, Peter G. Osmotherly, attest that Research Higher Degree candidate, Ishanka Weerasekara contributed to the listed publication by contributing to the conception and design of the studies, conducting and writing up the literature review, the collection of data, undertaking the statistical analysis, description and interpretation of the results, and writing the discussion and conclusions.

- Weerasekara, I., Osmotherly, P., Snodgrass, S., Marquez, J., de Zoete, R., & Rivett, D. A. (2018). Clinical benefits of joint mobilization on ankle sprains: a systematic review and metaanalysis. *Archives of physical medicine and rehabilitation*, 99(7), 1395-1412.e1395. doi:https://doi.org/10.1016/j.apmr.2017.07.019
- Weerasekara, I., Osmotherly, P. G., Snodgrass, S. J., Tessier, J., & Rivett, D. A. (2019). Effects of mobilisation with movement (MWM) on anatomical and clinical characteristics of chronic ankle instability: a randomised controlled trial protocol. *BMC musculoskeletal disorders*, 20(1), 75. doi:10.1186/s12891-019-2447-x
- Weerasekara, I., Osmotherly, P., Snodgrass, S., Tesseir, J., & Rivett, D. A. (2019). Is the fibula positioned anteriorly in weight-bearing in individuals with chronic ankle instability? (under review).
- Weerasekara, I., Snodgrass, S., Osmotherly, P., Tessier, J., & Rivett, D. A. (2019). Clinical characteristics of chronic ankle instability: a cross sectional study (under review).

.....

Date: 14 August 2019

Dr. Peter G. Osmotherly

.....

Date: 15 August 2019

Ishanka Weerasekara

.....

Date: 15 August 2019

Dr Lesley MacDonald-Wicks

Statement from Suzanne J. Snodgrass relating to papers published with Ishanka Weerasekara

I, Suzanne J. Snodgrass, attest that Research Higher Degree candidate, Ishanka Weerasekara contributed to the listed publication by contributing to the conception and design of the studies, conducting and writing up the literature review, the collection of data, undertaking the statistical analysis, description and interpretation of the results, and writing the discussion and conclusions.

- Weerasekara, I., Osmotherly, P., Snodgrass, S., Marquez, J., de Zoete, R., & Rivett, D. A. (2018). Clinical benefits of joint mobilization on ankle sprains: a systematic review and metaanalysis. *Archives of physical medicine and rehabilitation*, *99*(7), 1395-1412.e1395. doi:https://doi.org/10.1016/j.apmr.2017.07.019
- Weerasekara, I., Osmotherly, P. G., Snodgrass, S. J., Tessier, J., & Rivett, D. A. (2019). Effects of mobilisation with movement (MWM) on anatomical and clinical characteristics of chronic ankle instability: a randomised controlled trial protocol. *BMC musculoskeletal disorders*, 20(1), 75. doi:10.1186/s12891-019-2447-x
- Weerasekara, I., Osmotherly, P., Snodgrass, S., Tesseir, J., & Rivett, D. A. (2019). Is the fibula positioned anteriorly in weight-bearing in individuals with chronic ankle instability? (under review).
- Weerasekara, I., Snodgrass, S., Osmotherly, P., Tessier, J., & Rivett, D. A. (2019). Clinical characteristics of chronic ankle instability: a cross sectional study (under review).

Date: 14 August 2019

A/ Professor Suzanne J. Snodgrass

.....

Date: 15 August 2019

Ishanka Weerasekara

.....

Date: 15 August 2019

Dr Lesley MacDonald-Wicks

Statement from John Tessier relating to papers published with Ishanka Weerasekara

I, John Tessier, attest that Research Higher Degree candidate, Ishanka Weerasekara contributed to the listed publication by contributing to the conception and design of the studies, conducting and writing up the literature review, the collection of data, undertaking the statistical analysis, description and interpretation of the results, and writing the discussion and conclusions.

- Weerasekara, I., Osmotherly, P. G., Snodgrass, S. J., Tessier, J., & Rivett, D. A. (2019). Effects of mobilisation with movement (MWM) on anatomical and clinical characteristics of chronic ankle instability: a randomised controlled trial protocol. *BMC musculoskeletal disorders*, 20(1), 75. doi:10.1186/s12891-019-2447-x
- Weerasekara, I., Osmotherly, P., Snodgrass, S., Tesseir, J., & Rivett, D. A. (2019). Is the fibula positioned anteriorly in weight-bearing in individuals with chronic ankle instability? (under review).
- Weerasekara, I., Snodgrass, S., Osmotherly, P., Tessier, J., & Rivett, D. A. (2019). Clinical characteristics of chronic ankle instability: a cross sectional study (under review).

John Tessier

Date: 14 August 2019

Date: 15 August 2019

Ishanka Weerasekara

Date: 15 August 2019

Dr Lesley MacDonald-Wicks

Statement from Jodie Marquez relating to a paper published with Ishanka Weerasekara

I, Jodie Marquez, attest that Research Higher Degree candidate, Ishanka Weerasekara contributed to the listed publication by developing the systematic review protocol, conducting and writing up the literature review including interpreting the literature and writing the discussion and conclusions.

 Weerasekara, I., Osmotherly, P., Snodgrass, S., Marquez, J., de Zoete, R., & Rivett, D. A. (2018). Clinical benefits of joint mobilization on ankle sprains: a systematic review and metaanalysis. *Archives of physical medicine and rehabilitation*, 99(7), 1395-1412.e1395. doi:https://doi.org/10.1016/j.apmr.2017.07.019

.....

Date: 15 August 2019

Dr. Jodie Marquez

Date: 15 August 2019

Ishanka Weerasekara

.....

Date: 15 August 2019

Dr Lesley MacDonald-Wicks

Statement from Rutger de Zoete relating to a paper published with Ishanka

Weerasekara

I, Rutger de Zoete, attest that Research Higher Degree candidate, Ishanka Weerasekara contributed to the listed publication by developing the systematic review protocol, conducting and writing up the literature review including interpreting the literature and writing the discussion and conclusions.

 Weerasekara, I., Osmotherly, P., Snodgrass, S., Marquez, J., de Zoete, R., & Rivett, D. A. (2018). Clinical benefits of joint mobilization on ankle sprains: a systematic review and metaanalysis. *Archives of physical medicine and rehabilitation*, 99(7), 1395-1412.e1395. doi:https://doi.org/10.1016/j.apmr.2017.07.019

.....

Dr. Rutger de Zoete

Date: 14 August 2019

Date: 15 August 2019

Ishanka Weerasekara

Date: 15 August 2019

Dr Lesley MacDonald-Wicks

Appendix 2 Human research ethics committee approval: case-control study



HUMAN RESEARCH ETHICS COMMITTEE

Notification of Expedited Approval

To Chief Investigator or Project Supervisor:	Professor Darren Rivett
Cc Co-investigators / Research Students:	Doctor Peter Osmotherly Mr John Tessier Mrs Ishanka Madhurangani Rajapaksha Mudiyanselage Associate Professor Suzanne Snodgrass
Re Protocol:	Anatomical and clinical characteristics of the ankles of individuals with chronic ankle instability
Date:	19-Sep-2017
Reference No:	H-2017-0217
Date of Initial Approval:	19-Sep-2017

Thank you for your **Response to Conditional Approval (minor amendments)** submission to the Human Research Ethics Committee (HREC) seeking approval in relation to the above protocol.

Your submission was considered under Expedited review by the Ethics Administrator.

I am pleased to advise that the decision on your submission is Approved effective 19-Sep-2017.

In approving this protocol, the Human Research Ethics Committee (HREC) is of the opinion that the project complies with the provisions contained in the National Statement on Ethical Conduct in Human Research, 2007, and the requirements within this University relating to human research.

Approval will remain valid subject to the submission, and satisfactory assessment, of annual progress reports. If the approval of an External HREC has been "noted" the approval period is as determined by that HREC.

The full Committee will be asked to ratify this decision at its next scheduled meeting. A formal *Certificate of Approval* will be available upon request. Your approval number is H-2017-0217.

If the research requires the use of an Information Statement, ensure this number is inserted at the relevant point in the Complaints paragraph prior to distribution to potential participants You may then proceed with the research.

Conditions of Approval

This approval has been granted subject to you complying with the requirements for Monitoring of Progress, Reporting of Adverse Events, and Variations to the Approved Protocol as <u>detailed below</u>.

PLEASE NOTE:

In the case where the HREC has "noted" the approval of an External HREC, progress reports and reports of adverse events are to be submitted to the External HREC only. In the case of Variations to the approved protocol, or a Renewal of approval, you will apply to the External HREC for approval in the first instance and then Register that approval with the University's HREC.

Monitoring of Progress

Other than above, the University is obliged to monitor the progress of research projects involving human participants to ensure that they are conducted according to the protocol as approved by the HREC. A progress report is required on an annual basis. Continuation of your HREC approval for this project is conditional upon receipt, and satisfactory assessment, of annual progress reports. You will be advised when a report is due.

Reporting of Adverse Events

- 1. It is the responsibility of the person first named on this Approval Advice to report adverse events.
- Adverse events, however minor, must be recorded by the investigator as observed by the investigator or as volunteered by a participant in the research. Full details are to be documented, whether or not the investigator, or his/her deputies, consider the event to be related to the research substance or procedure.
- 3. Serious or unforeseen adverse events that occur during the research or within six (6) months of completion of the research, must be reported by the person first named on the Approval Advice to the (HREC) by way of the Adverse Event Report form (via RIMS at <u>https://rims.newcastle.edu.au/login.asp</u>) within 72 hours of the occurrence of the event or the investigator receiving advice of the event.
- 4. Serious adverse events are defined as:
 - · Causing death, life threatening or serious disability.
 - · Causing or prolonging hospitalisation.
 - Overdoses, cancers, congenital abnormalities, tissue damage, whether or not they are judged to be caused by the investigational agent or procedure.
 - Causing psycho-social and/or financial harm. This covers everything from perceived invasion of privacy, breach of confidentiality, or the diminution of social reputation, to the creation of psychological fears and trauma.
 - · Any other event which might affect the continued ethical acceptability of the project.
- 5. Reports of adverse events must include:
 - Participant's study identification number;
 - date of birth;
 - date of entry into the study;
 - treatment arm (if applicable);
 - date of event;
 - details of event;
 - · the investigator's opinion as to whether the event is related to the research procedures; and
 - · action taken in response to the event.
- Adverse events which do not fall within the definition of serious or unexpected, including those reported from other sites involved in the research, are to be reported in detail at the time of the annual progress report to the HREC.

Variations to approved protocol

If you wish to change, or deviate from, the approved protocol, you will need to submit an *Application for Variation to Approved Human Research* (via RIMS at <u>https://rims.newcastle.edu.au/login.asp</u>). Variations may include, but are not limited to, changes or additions to investigators, study design, study population, number of participants, methods of recruitment, or participant information/consent documentation. **Variations must be approved by the (HREC) before they are implemented** except when Registering an approval of a variation from an external HREC which has been designated the lead HREC, in which case you may proceed as soon as you receive an acknowledgement of your Registration.

Linkage of ethics approval to a new Grant

HREC approvals cannot be assigned to a new grant or award (ie those that were not identified on the application for ethics approval) without confirmation of the approval from the Human Research Ethics Officer on behalf of the HREC.

Best wishes for a successful project.

Associate Professor Helen Warren-Forward Chair, Human Research Ethics Committee

For communications and enquiries: Human Research Ethics Administration

Research & Innovation Services Research Integrity Unit The University of Newcastle Callaghan NSW 2308 T+612 492 17894 Human-Ethics@newcastle.edu.au

RIMS website - https://RIMS.newcastle.edu.au/login.asp

Linked University of Newcastle administered funding:

Funding body

Funding project title

First named investigator Grant Ref

Appendix 3 Human research ethics committee approval: randomised controlled trial



HUMAN RESEARCH ETHICS COMMITTEE

Notification of Expedited Approval

To Chief Investigator or Project Supervisor:	Professor Darren Rivett
Cc Co-investigators / Research Students:	Doctor Peter Osmotherly Associate Professor Suzanne Snodgrass Mrs Ishanka Madhurangani Rajapaksha Mudiyanselage Mr John Tessier
Re Protocol:	The effect of Mobilisation with Movement on Anatomical and Clinical Characteristics of Chronic Ankle Instability (CAI)
Date:	01-Feb-2018
Reference No:	H-2017-0354
Date of Initial Approval:	01-Feb-2018

Thank you for your Response to Conditional Approval (minor amendments) submission to the Human Research Ethics Committee (HREC) seeking approval in relation to the above protocol.

Your submission was considered under Expedited review by the Ethics Administrator.

I am pleased to advise that the decision on your submission is Approved effective 01-Feb-2018.

In approving this protocol, the Human Research Ethics Committee (HREC) is of the opinion that the project complies with the provisions contained in the National Statement on Ethical Conduct in Human Research, 2007, and the requirements within this University relating to human research.

Approval will remain valid subject to the submission, and satisfactory assessment, of annual progress reports. If the approval of an External HREC has been "noted" the approval period is as determined by that HREC.

The full Committee will be asked to ratify this decision at its next scheduled meeting. A formal Certificate of Approval will be available upon request. Your approval number is H-2017-0354.

If the research requires the use of an Information Statement, ensure this number is inserted at the relevant point in the Complaints paragraph prior to distribution to potential participants You may then proceed with the research.

Conditions of Approval

This approval has been granted subject to you complying with the requirements for Monitoring of Progress, Reporting of Adverse Events, and Variations to the Approved Protocol as <u>detailed below</u>.

PLEASE NOTE:

In the case where the HREC has "noted" the approval of an External HREC, progress reports and reports of adverse events are to be submitted to the External HREC only. In the case of Variations to the approved protocol, or a Renewal of approval, you will apply to the External HREC for approval in the first instance and then Register that approval with the University's HREC.

Monitoring of Progress

Other than above, the University is obliged to monitor the progress of research projects involving human participants to ensure that they are conducted according to the protocol as approved by the HREC. A progress report is required on an annual basis. Continuation of your HREC approval for this project is conditional upon receipt, and satisfactory assessment, of annual progress reports. You will be advised when a report is due.

• Reporting of Adverse Events

- 1. It is the responsibility of the person first named on this Approval Advice to report adverse events.
- Adverse events, however minor, must be recorded by the investigator as observed by the investigator or as volunteered by a participant in the research. Full details are to be documented, whether or not the investigator, or his/her deputies, consider the event to be related to the research substance or procedure.
- 3. Serious or unforeseen adverse events that occur during the research or within six (6) months of completion of the research, must be reported by the person first named on the Approval Advice to the (HREC) by way of the Adverse Event Report form (via RIMS at <u>https://rims.newcastle.edu.au/login.asp</u>) within 72 hours of the occurrence of the event or the investigator receiving advice of the event.
- 4. Serious adverse events are defined as:
 - · Causing death, life threatening or serious disability.
 - Causing or prolonging hospitalisation.
 - Overdoses, cancers, congenital abnormalities, tissue damage, whether or not they are judged to be caused by the investigational agent or procedure.
 - Causing psycho-social and/or financial harm. This covers everything from perceived invasion of privacy, breach of confidentiality, or the diminution of social reputation, to the creation of psychological fears and trauma.
 - Any other event which might affect the continued ethical acceptability of the project.
- 5. Reports of adverse events must include:
 - Participant's study identification number;
 - · date of birth;
 - date of entry into the study;
 - treatment arm (if applicable);
 - date of event;
 - · details of event;
 - · the investigator's opinion as to whether the event is related to the research procedures; and
 - · action taken in response to the event.
- Adverse events which do not fall within the definition of serious or unexpected, including those reported from other sites involved in the research, are to be reported in detail at the time of the annual progress report to the HREC.

Variations to approved protocol

If you wish to change, or deviate from, the approved protocol, you will need to submit an Application for Variation to Approved Human Research (via RIMS at https://rims.newcastle.edu.au/login.asp). Variations may include, but are not limited to, changes or additions to investigators, study design, study population, number of participants, methods of recruitment, or participant information/consent documentation. Variations must be approved by the (HREC) before they are implemented except when Registering an approval of a variation from an external HREC which has been designated the lead HREC, in which case you may proceed as soon as you receive an acknowledgement of your Registration.

Linkage of ethics approval to a new Grant

HREC approvals cannot be assigned to a new grant or award (ie those that were not identified on the application for ethics approval) without confirmation of the approval from the Human Research Ethics Officer on behalf of the HREC.

Best wishes for a successful project

Associate Professor Helen Warren-Forward Chair, Human Research Ethics Committee

For communications and enquiries: Human Research Ethics Administration

Research & Innovation Services Research Integrity Unit The University of Newcastle Callaghan NSW 2308 T+612 492 17894 Human-Ethics@newcastle.edu.au

RIMS website - https://RIMS.newcastle.edu.au/login.asp

Linked University of Newcastle administered funding:

Funding body

Funding project title

First named investigator Grant Ref
Appendix 4 Safety review notification



University Drive, Callaghan NSW 2308 AUSTRALIA HUMAN RESOURCE SERVICES HEALTH AND SAFETY

Contact Person: Liz Pilorim Telephone 02-4921 6542 Fax 02-4921 6982 E-mail: safetyclearance@newcastle.edu.au

14 February 2018

TO: Professor D Rivett, School of Health Sciences

COPY TO: A/Professor S Dempsey, HOS, School of Health Sciences Professor D Rivett, A/Professor S Snodgrass, Dr P Osmotherly, Dr S Walmsley, Mr J Tessier, Ms I Weerasekara Ms R Gibbins, Human Research Ethics Officer Dr N Gerrand, Manager Research Ethics and Governance, HNE Research Ethics Unit FROM: Ms Jane Hamson, Chairperson, Chemical/Radiation Technical Committee SUBJECT: SAFETY REVIEW NOTIFICATION

Based on the information provided the following Research Project has been reviewed and may proceed subject to compliance with the review conditions and there being no variation to the research processes that have been indicated in the application. If there is any variation to the protocol that affects the safety outcomes an additional application for safety review is necessary.

PROJECT TITLE	CHEMICAL	INSTITUTIONAL BIOSAFETY COMMITTEE (IBC)	RADIATION	HRE NUMBER	GRANTING BODY
THE EFFECT OF MOBILISATION WITH MOVEMENT ON ANATOMICAL AND CLINICAL CHARACTERISTICS OF CHRONIC ANKLE INSTABILITY (CAI) REF NO 11/2018	NA	NA	REVIEWED 07.02.2018	H-2017-0354	NA

In order to comply with the Work Health and Safety Act 2011 Chief Investigators must ensure that all, reasonably foreseeable, occupational health and safety risks arising out of their research activities are effectively controlled. A risk assessment must be completed to achieve this control. Effective controls follow on through the elimination (preferable), or minimisation, of these risks. Risk assessments are only validated once they have been signed and dated by the author and authorising supervisor. They must be reviewed annually and the review process needs to be documented (signed and dated).

Control measures for research activities must include (but are not limited to):

- A site (documented) orientation/induction to be given to all personnel when they are attending a facility/location for the first time. This induction will include; local rules (eating, drinking, storage of bags, standard of dress, hand washing and Personnel Protective Equipment [PPE] required etc), location of amenities, location of emergency procedures flipchart (which must contain correct contact information), rundown of evacuation procedure including the location of the assembly area.
- Ongoing inductions as appropriate to ensure current legislative requirements are reinforced to all facility users. Written standard operating procedures (SOP's) for equipment and processes
- Current SDS's must be readily available for all hazardous substances associated with the activity
- Please ensure the chemicals in your facility have been entered onto the Chemical Manifest.
- Training in SOP's for all personnel engaged in hazardous operations (with appropriate records)
- All documented control measures must be implemented.
- The minimum requirements for PPE in a laboratory are laboratory clothing (lab coat), protective eyewear (safety glasses), and closed shoes unless lesser requirements can be justified by a risk assessment AS/NZS 2243.1 Safety in laboratories Part 1: Planning and operational aspects.

U: SRA2018\Rivett, Professor D - 11/2018 - Safety review notification

- When ordering hazardous substances with a hazard rating of high or extreme, consideration should be given to using safer alternate reagents (where available) and only amounts required for the project should be purchased- to prevent unnecessary stockpiles of hazardous reagents.
- Any injury/incident occurring during the activity is reported via the University Online Incident Reporting System

Please ensure that all staff involved in working on this project is aware of the requirements of this review and a copy of this memorandum is to be kept in the Laboratory Safety Manual/s of the facility/s where the work is performed.

If you have any enquiries in relation to this safety review (implications) please do not hesitate to contact me.

MS JANE HAMSON CHAIRPERSON, CHEMICAL/RADIATION TECHNICAL COMMITTEE

U: SRA2018\Rivett, Professor D - 11/2018 - Safety review notification

Appendix 5 Dosimetry report: case-control study

Medical Physics Group Nick Hille, Area Radiation Safety Officer

Jenny Diffey, Senior Medical Physics Specialist Tel: 4921 3380 Fax: 4921 3392 Email: jennifer.diffey@health.nsw.gov.au

Dr John Tessier School of Health Sciences University of Newcastle Callaghan NSW 2308



Dear John,

Please find below a dosimetry report for the research study entitled "Anatomical and clinical characteristics of the ankles of individuals with chronic ankle instability"

Dosimetry information for a study involving exposure to an external source of ionising radiation:

> 18 years old
X-ray
All participants: Weight bearing lateral x-ray
Symptomatic participants: Lateral stress views of both ankles

Examination	kVp	mAs	Entrance Skin Dose (mGy)	Effective Dose ¹ (mSv)
Lateral Ankle X-ray	55	2.1	0.1	0.0001

Total Effective Dose from Study = 0.0001 mSv (asymptomatic) or 0.0003 mSv (symptomatic participants)

The dose received from this study, not including any exposures incurred as part of the normal clinical management of the participant, does not exceed the relevant dose constraint in Table 1 of the ARPANSA Code of Practice (RPS8)².

The recommended statement for inclusion in the patient information sheet for this study is as follows:

"This research study involves exposure to a very small amount of radiation. As part of everyday living, everyone is exposed to naturally occurring background radiation and receives a dose of about 2 to 3 millisieverts (mSv) each year. The effective dose from this study is about 0.0003 mSv. At this dose level, no harmful effects of radiation have been demonstrated as any effect is too small to measure. The risk is believed to be minimal."

26 May 2017

Dr Jenny Diffey, Senior Medical Physics Specialist NSW EPA Consulting Radiation Expert No. 5025022 ACPSEM Accreditation # R00069

References

1. Hart D et al. Frequency and Collective Dose for Medical and Dental X-ray Examinations in the UK, 2008. Health Protection Agency Report HPA-CRCE-012

2. ARPANSA Code of Practice RPS 8: Exposure of Humans to Ionizing Radiation for Research Purposes

Hunter New England Area Health Service Hunter New England Imaging Locked Bag 1 HRMC Newcastle NSW 2310 Appendix 6 Dosimetry report: randomised controlled trial

Medical Physics Group Nick Hille, Area Radiation Safety Officer

Jenny Diffey, Senior Medical Physics Specialist Tel: 4921 3380 Fax: 4921 3392 Email: jennifer.diffey@health.nsw.gov.au



Dr John Tessier School of Health Sciences University of Newcastle Callaghan NSW 2308

Dear John,

Please find below a dosimetry report for the research study entitled "The effect of mobilisation with movement on anatomical and clinical characteristics of chronic ankle instability (CAI)"

Dosimetry information for a study involving exposure to an external source of ionising radiation:

Participant Age Range: > 18 years old; all participants are symptomatic. Source of Exposure: X-ray

Examinations will be carried out on a Philips Dura Diagnost X-ray unit with DR detector.

Exa	amination	kVp	mAs	Entrance Skin Dose (mGy)	Effective Dose ¹ (mSv)
1	Pre-intervention weight bearing lateral x-rays of the symptomatic ankle	55	2.1	0.1	0.0001
2	Lateral stress views of symptomatic ankle	55	2.1	0.1	0.0001
3	Lateral stress views of asymptomatic ankle	55	2.1	0.1	0.0001
4	Post-intervention weight bearing lateral x-rays of the symptomatic ankle	55	2.1	0.1	0.0001

Total Effective Dose from Study = 0.0004 mSv*

* Please note: some of the study participants will have taken part in a previous study entitled "Anatomical and clinical characteristics of the ankles of individuals with chronic ankle instability" (see Dosimetry Report dated 26 May 2017). As part of this previous study, they will have received Examinations 1 – 3 and will only receive Examination 4 as part of the current study. Therefore, the maximum dose received by a participant of both studies will be 0.0004 mSv.

The dose received from this study, not including any exposures incurred as part of the normal clinical management of the participant, does not exceed the relevant dose constraint in Table 1 of the ARPANSA Code of Practice (RPS8)².

The recommended statement for inclusion in the patient information sheet for this study is as follows:

"This research study involves exposure to a very small amount of radiation. As part of everyday living, everyone is exposed to naturally occurring background radiation and receives a dose of about 2 to 3

> Hunter New England Area Health Service Hunter New England Imaging Locked Bag 1 HRMC Newcastle NSW 2310

Appendix 7 PROSPERO systematic review registration notification

Dear Mrs Weerasekara

Thank you for submitting details of your systematic review *Clinical benefits of* passive joint mobilisation on ankle sprains: a systematic review and meta-analysis to the PROSPERO register. We are pleased to confirm that the record has been published on the database.

Your registration number is: CRD42016030194

You are free to update the record at any time, all submitted changes will be displayed as the latest version with previous versions available to public view. Please also give brief details of the key changes in the Revision notes facility. You can log in to PROSPERO and access your records at <u>http://www.crd.york.ac.uk/PROSPERO</u>

An email reminder will be sent to you on the anticipated completion date, prompting you to update the record.

Comments and feedback on your experience of registering with PROSPERO are welcome at: <u>crd-register@york.ac.uk</u>

Best wishes for the successful completion of your review. Yours sincerely

Lesley Indge

PROSPERO Administrator Centre for Reviews and Dissemination University of York York YO10 5DD t: +44 (0) 1904 321040 f: +44 (0) 1904 321041 e: <u>CRD-register@york.ac.uk</u> www.york.ac.uk/inst/crd

PROSPERO is funded by the National Institute for Health Research and produced by CRD, an academic department of the University of York. Appendix 8 ANZCTR clinical trial registration notification

Dear Ishanka Weerasekara and Darren Rivett,

Re: Joint Mobilisation in Chronic Ankle Instability

Thank you for submitting the above trial for inclusion in the Australian New Zealand Clinical Trials Registry (ANZCTR).

Your trial has now been successfully registered and allocated the ACTRN: ACTRN12617001467325p

Web address of your trial: http://www.ANZCTR.org.au/ACTRN12617001467325p.aspx Date submitted: 10/10/2017 12:04:43 AM Date registered: 17/10/2017 12:10:22 PM Registered by: Ishanka Weerasekara Principal Investigator: Darren Rivett

If you have already obtained Ethics approval for your trial, please send a copy of at least one Ethics Committee approval letter to info@actr.org.au or by fax to (+61 2) 9565 1863, attention to ANZCTR.

Note that updates should be made to the registration record as soon as any trial information changes or new information becomes available. Updates can be made at any time and the quality and accuracy of the information provided is the responsibility of the trial's primary sponsor or their representative (the registrant). For instructions on how to update please see http://www.anzctr.org.au/Support/HowToUpdate.aspx.

Please also note that the original data lodged at the time of trial registration and the tracked history of any changes made as updates will remain publicly available on the ANZCTR website.

The ANZCTR is recognised as an ICMJE acceptable registry (http://www.icmje.org/faq.pdf) and a Primary Registry in the WHO registry network (http://www.who.int/ictrp/network/primary/en/index.html).

If you have any enquiries please send a message to info@actr.org.au or telephone +61 2 9562 5333.

Kind regards, ANZCTR Staff T: +61 2 9562 5333 F: +61 2 9565 1863 E: info@actr.org.au W: www.ANZCTR.org.au



Appendix 9 Information statement: case-control study

Professor Darren A. Rivett School of Health Sciences Faculty of Health and Medicine The University of Newcastle University Drive Callaghan NSW 2308 Phone: 02 49215642 Fax: 02 49217053 Darren.Rivett@newcastle.edu.au



Information Statement for the Research Project: Anatomical and clinical characteristics of the ankles of individuals with chronic ankle instability (CAI)

Version 2 (04/09/17)

You are invited to participate in the research project identified above which is being conducted in the School of Health Sciences at The University of Newcastle. The research is part of Ishanka Weerasekara's PhD studies and is supervised by Professor Darren Rivett, Dr. Peter Osmotherly, and Associate Professor Suzanne Snodgrass. Your eligibility to participate in the clinical/ lab data collection is dependent on the score obtained by the questionnaires attached herewith. You will be informed (over the phone or via email) as to your eligibility to participate once you return the attached questionnaires to the researcher, according to your preferred method of contact.

Why is the research being done?

The purpose of this study is to determine the basic characteristics of ankle joint instability compared to healthy ankles.

The feeling of ankle joint instability is 'the situation whereby during activities of daily living (ADL) and sporting activities the participant feels that the ankle joint is unstable and is usually associated with the fear of sustaining an acute ligament sprain."

Who can participate in the research?

There are two groups of participants in this study, who are 18 years of age or older. The first group is composed of people who have ankle joint instability as described above. The second group are those with no ankle symptoms.

If you have the feeling of ankle joint instability and would like to participate in the study, you must have:

- A history of at least one significant ankle sprain
- A history of the previously injured ankle joint "giving way" and/or recurrent sprain and/or "feelings of instability."

If you have a healthy ankle, you may participate in the study if you have:

- Not had ankle problems or surgeries previously.
- Not had treatments for an ankle problem previously.
- No pain or problems in or around the ankle while performing daily activities currently

You may not be able to participate in the study if you have or have had any of the following:

- A history of fracture, or surgery to your ankle
- Any recent injury other than ankle instability (within 3 months) to your lower limb
- Any previous injury still causing problems to the lower limbs
- Any neuromuscular disorders that may result in lower limb soft tissue anomalies i.e. muscle weakness
- Any conditions for which x-ray imaging is contraindicated (e.g., pregnancy)
- Inability to speak or write English

If you are unsure if one of these criteria applies to you, please ask the investigator.

What choice do you have?

Participation in this research is entirely your choice. Only those people who give their informed consent will be included in the project. Whether or not you decide to participate, your decision will not disadvantage you.

If you do decide to participate, you may withdraw from the project at any time without giving a reason and have the option of withdrawing any data which identifies you.

What would you be asked to do?

If you agree to participate, you will be asked to:

- Attend one data collection session lasting approximately one hour at the School of Health Sciences Research Laboratories at the University of Newcastle at Callaghan.
- During this session you will:
 - · Have your age, gender, height and weight recorded by the researchers
 - · Have a registered radiographer take up to three x-rays of your ankle
 - Undergo few clinical measures including general hypermobility, ankle range of motion, balance tests, pain scales and pressure pain thresholds.

What measurements will be performed?

If you are eligible to participate, you will be invited to the study at the Discipline of Physiotherapy in the Hunter building at the University of Newcastle at Callaghan. During the study, we will be recording your basic data including age, gender, height, weight, side of the dominant leg and completing clinical measures. Your ankles will be also x-ray filmed by a registered radiographer.

If you have the feeling of ankle joint instability:

- Two x-rays of the affected ankle will be obtained (one in lying position and one in standing) to determine the bony involvement related to the ankle condition. One x-ray will be obtained from the other ankle to compare the difference.
- Clinical measures will be taken including ankle range of motion, balance tests, pain scales and pressure pain thresholds.

If you have a healthy ankle:

- One x-ray of the dominant side ankle will be obtained (in standing position) to determine the bony alignment of the ankle.
- Clinical measures will be taken including ankle range of motion, balance tests, pain scales and pressure pain thresholds.

All these tests will be stopped if you feel pain.

How much time will it take?

. The session will take about an hour.

What are the risks and benefits of participating?

The procedures in this study are used routinely in clinical practice, and there have been no published reports of adverse effects or problems with any of these procedures, to the best of our knowledge.

You will be benefitted from participating in this research, as this is an opportunity to have your ankle evaluated using x-ray measurement. During the process of reviewing your x-ray, if any abnormality is detected you will be advised to seek further review with your local medical practitioner. In the case of any abnormality being found, you will be provided with a copy of the x-ray/s. You will be provided with a letter detailing the findings and a copy of them so that you can take them to your medical practitioner if you choose to do so. If you request, you will be provided an electronic copy of your x-ray/s. In addition, at the completion of the study you can receive, if you wish, a summary of the study results. There will be no cost to you for any of the procedures you may receive in this trial or any part of your participation.

The risks to participating are minimal. However, if you do experience any unpleasant sensations, you can stop the assessment procedures at any time and withdraw from the study. The researchers, who are registered health professionals, will discuss any uncomfortable sensations with you, and if appropriate, will provide you with a letter for your medical practitioner.

This research study involves exposure to a very small amount of radiation. As part of everyday living, everyone is exposed to naturally occurring background radiation and receives a dose of about 2 to 3 millisieverts (mSv) each year. The effective dose from this study is about 0.0003 mSv. At this dose level, no harmful effects of radiation have been demonstrated as any effect is too small to measure. The risk is believed to be minimal. Also the ankle is away from vital organs of the body, which will not be exposed to radiation. The risk is, therefore, believed to be very low.

How will your privacy be protected?

Personally identifiable information that is provided will be confidential to the researchers. Non-identifiable data may be also be shared with other parties to encourage scientific scrutiny, and to contribute to further research and public knowledge, or as required by law. No identifying information will be included in any reports presented at a scientific meeting or published in a journal or thesis. Your x-ray images will be stored on a secure database only accessible to the researchers listed above. Paper copies of questionnaires and the other measurements will be stored in a locked cabinet in the School of Health Sciences offices and the information will only be accessed by the researchers unless you consent otherwise, except as required by law. All data collected from you will be labelled with a code on the day it is collected. Coded data will be stored in the ownCloud data management facility of The University of Newcastle that is password protected and is only accessible to the researchers listed above. If you choose to complete the questionnaires online, your information will be securely collected and transferred to ownCloud via the Qualtrics platform, and only you and the researchers have access to your information on this platform. The contact details that you provide will be stored separately to the data in a locked cabinet in the office of the School of Health Sciences. At the conclusion of the study all data will be downloaded to CD and stored in a locked cabinet in the office of the School of Health Sciences at the University of Newcastle. All information will be stored in the School of Health Sciences at The University of Newcastle for a minimum of 5 years, after which time it will be appropriately destroyed.

How will the information collected be used?

The results of this study will be submitted in a thesis for, Ishanka Weerasekara's PhD, and may be presented at conferences and in scientific journals. Individual participants will not be identified in any reports arising from this project. Further, de-identified data may be used in future research upon your consent. A summary of the study results can be provided to you, if you wish to receive this, at the completion of the study.

What do you need to do to participate?

If you would like to participate, please telephone or email your contact details to the research team. Once you contact the researchers, a preliminary telephone screening will be carried out and then, if you are potentially eligible to participate, you will be sent the information statement, consent form and three screening questionnaires. These documents will either be posted as hard copies or a link will be emailed to enable you to access them online, depending on your preference.

Please read this Information Statement and be sure you understand its contents before you consent to participate. If there is anything you do not understand, or you have questions, contact the researchers.

Please complete the attached consent form and return it in the reply paid envelope (or submit the online form) provided. Researchers will then contact you to arrange a time convenient to you for the screening. Also you can contact one of the researchers listed below either by telephone, email or in person. They will then answer any further queries, and confirm an appointment time for you to attend your data collection session.

Funding

This research received no specific grant from any funding agency in the public, commercial, or not- for profit sectors.

Further information

If you would like further information, you may contact the researchers directly: Ishanka Weerasekara: tel. 0424208114 email: <u>Ishanka.weerasekara@uon.edu.au</u> Professor Darren Rivett: tel: 049217220 email: <u>Darren.rivett@newcastle.edu.au</u>

Thank you for considering this invitation.

Professor Darren Rivett

Ishanka Weerasekara

Project supervisor

Research Higher Degree Candidate

Complaints about this research

This project has been approved by the University's Human Research Ethics Committee, Approval No.H-2017-0217.

Should you have concerns about your rights as a participant in this research, or you have a complaint about the manner in which the research is conducted, it may be given to the researcher, or, if an independent person is preferred, to the Human Research Ethics Officer, Research & Innovation Services, The University of Newcastle, University Drive, Callaghan NSW 2308, Australia, telephone (02) 49216333, email <u>Human-Ethics@newcastle.edu.au</u>.

Appendix 10 Consent form: case-control study



Consent Form for the Research Project: Anatomical and clinical characteristics of the ankles of individuals with chronic ankle instability (CAI) Version 2 (04/09/17)

Chief Investigators: Professor Darren Rivett, Dr Peter Osmotherly, A/ Professor Suzanne Snodgrass Student Researcher: Ms Ishanka Weerasekara

I agree to participate in the above research project and give my consent freely.

I understand that the project will be conducted as described in the Information Statement, a copy of which I have retained.

I understand I can withdraw from the project at any time and do not have to give any reason for withdrawing.

I consent to:

Professor Darren A. Rivett School of Health Sciences Faculty of Health and Medicine The University of Newcastle

Phone: 02 49215642 Fax: 02 49217053

Darren.Rivett@newcastle.edu.au

University Drive

Callaghan NSW 2308

- Attend one data collection session lasting approximately one hour at the School of Health Sciences Research ٠ Laboratories at the University of Newcastle Callaghan Campus.
- Being tested to determine the whether I have chronic ankle instability. ٠
- Have my age, gender, height and weight and hypermobility measurements recorded ٠
- Have the clinical measures (balance, range of ankle, pain intensity and pain threshold) /assessments (x- rays of my ankle) explained in the information sheet recorded by the researchers.

I understand that my personal information will remain confidential to the researchers.

I have had the opportunity to have questions answered to my satisfaction.

Print Name:	
Contact Phone:	
Contact Address:	
Email:	_
Signature:	
Date:	

Do you consent to be contacted for about potential participation in future research?	Yes / No
Do you consent to reuse of your de-identified data if required in future research?	Yes / No
Please indicate (circle) whether you would like a summary of the study findings	Yes / No
If you indicate yes, in any of the questions above, please select the preferred method of contact	Email / Post

Appendix 11 Information statement: randomised controlled trial

Professor Darren A. Rivett School of Health Sciences Faculty of Health and Medicine The University of Newcastle University Drive Callaghan NSW 2308 Phone: 02 49215642 Fax: 02 49217053 Darren.Rivett@newcastle.edu.au



Information Statement for the Research Project: The effect of Mobilisation with Movement on Anatomical and Clinical Characteristics of Chronic Ankle Instability (CAI) Version 3 (23/01/18)

You are invited to participate in the research project identified above which is being conducted in the School of Health Sciences at The University of Newcastle. The research is part of Ishanka Weerasekara's PhD studies and is supervised by Professor Darren Rivett, Dr. Peter Osmotherly, and Associate Professor Suzanne Snodgrass. Your eligibility to participate in the clinical/ lab data collection is dependent on the score obtained by the questionnaires attached. You will be informed (over the phone or via email) as to your eligibility to participate once you return the attached questionnaires to the researcher, according to your preferred method of contact. If you do participate, you are advised to retain this Information Statement for at least 5 years so that it can be provided to researchers in any future research project involving radiation.

Why is the research being done?

The purpose of this study is to determine the effectiveness of two procedures performed by physiotherapists as treatment for ankle joint instability.

The feeling of ankle joint instability is 'the situation whereby during activities of daily living (ADL) and sporting activities the participant feels that the ankle joint is unstable and is usually associated with the fear of sustaining an acute ligament sprain."

Who can participate in the research?

The group of participants is composed of people who have ankle joint instability as described above, who are more than 18 years of age.

If you would like to participate in the study, you must have:

- A history of at least one significant ankle sprain
- A history of the previously injured ankle joint "giving way" and/or recurrent sprain and/or "feelings of instability."

You may not be able to participate in the study if you have or have had any of the following:

- A history of fracture, or surgery to your lower limb
- Any recent injury other than ankle instability (within 3 months) to your lower limb
- Any previous injuries (other than ankle instability), still causing problems to the lower limb
- Any neuromuscular disorders that may result in lower limb soft tissue anomalies ie. muscle weakness
- Any conditions for which manual therapy is not recommended (e. g.: tumour, rheumatoid arthritis, or osteoporosis)
- Any conditions for which x-ray imaging is not recommended (e.g., pregnancy)
- Any allergies to adhesive strapping tape
- Inability to speak or write English

If you are unsure if one of these criteria applies to you, please ask the investigator.

What choice do you have?

Participation in this research is entirely your choice. Only those people who give their informed consent will be included in the project. Whether or not you decide to participate, your decision will not disadvantage you.

If you do decide to participate, you may withdraw from the project at any time without giving a reason and have the option of withdrawing any data which identifies you. Also it will not disadvantage you or impact on any relationship you may have with the University of Newcastle and its staff.

What would you be asked to do?

If you agree to participate, you will be asked to:

- Complete screening questionnaires, and if it is determined that you are eligible you will then
 proceed to the next phase of the study.
- If you are eligible to participate, you will be invited to attend a data collection session at the Physiotherapy Research Laboratory in the Hunter Building at the University of Newcastle at Callaghan. During the first visit you will:
 - Have your age, gender, height, weight and general joint mobility recorded by the researchers.
 - Have a registered radiographer (Mr John Tessier) take three x-rays of your ankle.
 - Undergo clinical measures including ankle range of motion, balance tests, pain scales, self- reported function questionnaires and pressure pain thresholds.
- Following the measurements, you will be:
 - Randomly allocated to one of the two possible groups. Each group will receive a different intervention from a registered Physiotherapist. You are requested not to discuss your intervention with other participants in the study.
 - Given this same intervention 2-8 times over 4 weeks. The number of sessions you attend will be determined by the physiotherapist based on your response to treatment, and sessions will be scheduled at a time convenient for you.
- Then follow up data will be collected during three occasions; at the 4th week after the enrolment (lab data collection), at the 12th week (lab data collection) and 12th month after the intervention (questionnaire based data collection).
 - At the 4th week after enrolment: One x-ray of your ankle will be obtained by a registered radiographer. Clinical measures including ankle range of motion, balance tests, pain scales, pressure pain thresholds and self- reported function questionnaires will be obtained.
 - At the 12th week of intervention: Clinical measures including ankle range of motion, balance tests, pain scales, self- reported function questionnaires and pressure pain thresholds will be obtained.
 - At the 52nd week of intervention: Fill three questionnaires to measure ankle instability, function and pain intensity (on hard copies posted or online survey link via email according to your preference).

You may stop these tests at any time you feel pain. You will be requested not to receive any additional treatment for the ankle during the study period, and the most recent treatment should have been received at least a week prior to study enrolment.

Your activity level, re-injuries and also regarding other treatments during the procedures will be recorded at each treatment visit and further you will be phone interviewed monthly for up to one year to record the same details.

(Note: If you have already participated in the study on 'Anatomical and clinical characteristics of the ankles of individuals with chronic ankle instability (CAI)' (ethics reference no: H-2017-0217), you do not have to undergo the measurements collected during the first visit, and will be directed to the treatment phase directly.)

The treatment procedures

You will be randomly allocated to one of two intervention groups, and you will not be able to choose which intervention you will receive. Some of you will receive an active intervention of unproven benefit, though this intervention is routinely used by physiotherapists and believed to help ankle symptoms. There is a chance of not receiving an intervention with any therapeutic benefit.

One group will receive laser to the lateral aspect of the ankle. Normally there is no sensation felt with laser. The other group will receive a mobilisation technique to the ankle region, which involves gentle pressure on one of your ankle bones while you move your ankle.

A registered and experienced physiotherapist will perform all interventions. If you experience any pain during either intervention you must tell the physiotherapist.

If you choose to participate in this research by completing and returning the consent form, you are agreeing to participate in all aspects of the study, as detailed above. You will be provided with parking vouchers or free parking for all treatment and measurement sessions that you attend. There will be no other reimbursements or payments for participating in this research.

How much time will it take?

To be involved in the study, you will need to be able to attend the 2-8 separate intervention sessions over a period of 4 weeks, and measurement sessions on 3 separate occasions over a period of twelve (12) months. The table below provides details on how long each session will take and where each will be held.

Week/s	Session	Time	Location
0	Questionnaire screening (3 questionnaires)	10 minutes	Online/ post
1	Measurement sessions (3 x-rays, clinical measures)	1.15 hours	Hunter Building, The University of Newcastle
1, 2, 3, 4	Treatment sessions (2-8 sessions)	15 minutes	Hunter Building, The University of Newcastle
	Injury, activity and treatment record		
4	Measurement sessions (1 x-ray, 2 questionnaires, clinical measures)	1 hour	Hunter Building, The University of Newcastle
12	Measurement sessions (2 questionnaires, clinical measures)	45 minutes	Hunter Building, The University of Newcastle
52	Measurement sessions (3 questionnaires)	10 minutes	Online/ post
4, 8, 12, 16, 20, 24, 27, 32, 36, 40, 44, 48, 52	Injury, activity and treatment record (phone interview/online survey)	7 minutes	Over the phone

What are the risks and benefits of participating?

The procedures and interventions in this study are used routinely used in clinical practice, and there have been no published reports of adverse effects or problems with any of these procedures, to the best of our knowledge. Some participants will receive an active intervention of unproven benefit, though this intervention is routinely used by physiotherapists and believed to help ankle symptoms. Other participants will undergo an intervention that is not designed to be of any therapeutic benefit. Therefore researchers anticipate that some participants will benefit from the interventions used, but not all. If the active intervention is shown to be beneficial and you did not receive it, you will be offered the active intervention free of charge, after completion of your participation in the study. The request should be made within 12 months after their enrolment by the participant. Otherwise, there will be no continued access to the trial intervention you have received during the trial, after the completion of the trial. If treatment is offered after your participation in the study, it will be conducted at the University of Newcastle, Callaghan campus.

You will benefit from participating in this research, as this is an opportunity to have your ankle evaluated using x-ray measurement. During the process of reviewing your x-ray, if any abnormality is detected you will be provided with a copy of your x-ray/s and a letter detailing the findings, and advised to seek further review with your local medical practitioner. In addition, at the completion of the study you can request to receive a summary of the study results. There will be no cost to you for any of the procedures you may receive in this trial or any part of your participation.

The risks to participating are minimal. However, if you do experience any unpleasant sensations, you can stop the assessment procedures at any time and withdraw from the study. The researchers, who are registered health professionals, will discuss any uncomfortable sensations with you, and if appropriate, will provide you with a letter for your medical practitioner. This research study involves exposure to a very small amount of radiation. As part of everyday living, everyone is exposed to naturally occurring background radiation and receives a dose of about 2 to 3 millisieverts (mSv) each year. The effective dose from this study is about 0.0004 mSv. At this dose level, no harmful effects of radiation have been demonstrated as any effect is too small to measure. The risk is believed to be minimal. Also the ankle is away from vital organs of the body, which will not be exposed to radiation. The risk of harm from exposure to radiation is, therefore, believed to be very low.

How will your privacy be protected?

Personally identifiable information that is provided will be confidential to the researchers. Non-identifiable data may be also be shared with other parties to encourage scientific scrutiny, and to contribute to further research and public knowledge, or as required by law. No identifying information will be included in any reports presented at a scientific meeting or published in a journal or thesis. Your x-ray images will be stored on a secure database only accessible to the researchers listed above. Paper copies of questionnaires and the other measurements will be stored in a locked cabinet in the School of Health Sciences offices and the information will only be accessed by the researchers unless you consent otherwise, except as required by law. All data collected from you will be labelled with a code on the day it is collected. Coded data will be stored in the ownCloud data management facility of The University of Newcastle that is password protected and is only accessible to the researchers listed above. If you choose to complete the questionnaires online, your information will be securely collected and transferred to ownCloud via the Qualtrics platform, and only you and the researchers have access to your information on this platform. The contact details that you provide will be stored separately to the data in a locked cabinet in the office of the School of Health Sciences. At the conclusion of the study all data will be downloaded to CD and stored in a locked cabinet in the office of the School of Health Sciences at the University of Newcastle, All information will be stored in the School of Health Sciences at The University of Newcastle for a minimum of 5 years, after which time it will be appropriately destroyed.

How will the information collected be used?

The results of this study will be submitted in a thesis for, Ishanka Weerasekara's PhD, and may be presented at conferences and in scientific journals. Individual participants will not be identified in any reports arising from this project. Further, de-identified data may be used in future research upon your consent. A summary of the study results can be provided to you, if you wish to receive this, at the completion of the study.

What do you need to do to participate?

Please read this Information Statement and be sure you understand its contents before you consent to participate. If there is anything you do not understand, or if you have any questions, please contact the researchers.

If you wish to participate, please complete the consent form and three screening questionnaires, and return these in the reply-paid envelope provided (or alternatively complete the documents online using the provided link). Once these are received, the researchers will contact you to confirm your eligibility and arrange an appointment time for you to attend your first session.

Funding

This research received no specific grant from any funding agency in the public, commercial, or not-for profit sectors.

Further information

If you would like further information, you may contact the researchers directly: Ishanka Weerasekara: tel. 0424208114 email: <u>Ishanka.weerasekara@uon.edu.au</u> Professor Darren Rivett: tel: 49217220 email: <u>Darren.rivett@newcastle.edu.au</u>

Thank you for considering this invitation.

Professor Darren Rivett Project supervisor Ishanka Weerasekara Research Higher Degree Candidate

Complaints about this research

This project has been approved by the University's Human Research Ethics Committee, Approval No. H-2017-0354.

Should you have concerns about your rights as a participant in this research, or you have a complaint about the manner in which the research is conducted, it may be given to the researcher, or, if an independent person is preferred, to the Human Research Ethics Officer, Research & Innovation Services, The University of Newcastle, University Drive, Callaghan NSW 2308, Australia, telephone (02) 49216333, email Human-Ethics@newcastle.edu.au.

Appendix 12 Consent form: randomised controlled trial



Consent Form for the Research Project: The effect of Mobilisation with Movement on Anatomical and Clinical Characteristics of Chronic Ankle Instability (CAI) Version 2 (15/12/2017)

Chief Investigators: Professor Darren Rivett, Dr Peter Osmotherly, A/ Professor Suzanne Snodgrass Student Researcher: Ms Ishanka Weerasekara

I agree to participate in the above research project and give my consent freely.

I understand that the project will be conducted as described in the Information Statement, a copy of which I have retained.

I understand I can withdraw from the project at any time and do not have to give any reason for withdrawing.

I consent to:

Professor Darren A. Rivett School of Health Sciences Faculty of Health and Medicine The University of Newcastle

Phone: 02 49215642 Fax: 02 49217053

Darren.Rivett@newcastle.edu.au

University Drive

Callaghan NSW 2308

- Attend 3 data collection sessions lasting approximately one and half hour at the School of Health Sciences Research Laboratories at the University of Newcastle Callaghan Campus.
- · Being tested to determine the whether I have chronic ankle instability.
- Attend 2-8 treatment sessions to receive an intervention which may treat my ankle instability
- · Have my age, gender, height and weight and hypermobility measurements recorded
- · Have the clinical measures (balance, range of ankle, pain intensity, function and pain threshold) recorded
- Have my ankle x-rayed and recorded
- Being contacted to record the details about my activity level, re-injuries and other treatments at each treatment visit and also to contact monthly after the treatments for one year.

I understand that my personal information will remain confidential to the researchers.

I have had the opportunity to have questions answered to my satisfaction.

Print Name:		
Contact Phone:		
Contact Address:	 	
Email:		
Signature:		
Date:		

Do you consent to reuse of your de-identified data if required in future research?	Yes / No
Please indicate (circle) whether you would like a summary of the study findings	Yes / No
If you indicate yes, in any of the questions above, please select the preferred method of contact	Email / Post

Appendix 13 Participant recruitment flyers: case-control study



SEEKING HEALTHY ANKLES ...!

ARE YOU ...

- More than 18 years old
- Without any ankle pain or problem in the ankle now
- Without any history of previous ankle surgeries or any recent ankle problems

This University of Newcastle based project on 'Anatomical and clinical characteristics of the ankles of individuals with chronic ankle instability (CAI)' will require you to attend the School of Health Sciences Research Laboratories in the Hunter Building to have ankle movement, balance, function, pain assessments, and X-ray examination measured.

If you would like to find our more information about the study please phone Ishanka in the Discipline of Physiotherapy, the University of Newcastle.

Project Supervision
Prof. Darren Rivett
T: 49217220
E: Darren.rivett@newcastle.edu.au

This project has been approved by the University of Newcastle Human Research Ethics Committee, HREC Approval No. H-2017-0217

Ankle instability study Participants with without Ankle instability Ishanka- 0424208114 Ishanka.weerasekara@uon.edu.au
Ankle instability study Partopants with/without Ankle instability Ishanka-0424208114 Ishanka.weerasekara@uon.edu.au
Ankle instability study Partopans with without Ankle instability Ishanka- 0424208114 Ishanka.weerasekara@uon.edu.au
Ankle instability study Partopants with without Ankle instability Ishanka- 0424208114 Ishanka.weerasekara@uon.edu.au
Ankle instability study Parcipants with without Arkle instability Ishanka- 0424208114 Ishanka.weerasekara@uon.edu.au
Ankle instability study Parcepans with without Ankle instability Ishanka- 0424208114 Ishanka.weerasekara@uon.edu.au
Ankle instability study Participants with without Ankle instability Ishanka-0424208114 Ishanka.weerasekara@uon.edu.au



KEEP ON SPRAINING YOUR ANKLE?

IS THIS YOU ...?

- · More than 18 years old
- Have a history of the previously injured ankle joint "giving way" and/or recurrent sprain and/or "feelings of instability."

This University of Newcastle based project on 'Anatomical and clinical characteristics of the ankles of individuals with chronic ankle instability (CAI)' will require you to attend the School of Health Sciences Research Laboratories in the Hunter Building to have ankle movement, balance, function, pain assessments, and X-ray examination measured.

If you would like to find our more information about the study please phone Ishanka in the Discipline of Physiotherapy, the University of Newcastle.

Research	Higher	Degree	Candidate	Projec
	-	-		

Ishanka Weerasekara	F
T: 0424208114	1
E: Ishanka.weerasekara@uon.edu.au	E

Project Supervision Prof. Darren Rivett T: 49217220

E: Darren.rivett@newcastle.edu.au

This project has been approved by the University of Newcastle Human Research Ethics Committee, HREC Approval No. H-2017-0217

Appendix 14 Participant recruitment flyers: randomised controlled trial



Does this sound like you?

- More than 18 years old
- Have you had repeated ankle sprains? and / or does your ankle 'give way' or feel unstable?
- Would you like to have the chance of free physiotherapy treatment for your ankle?

We are recruiting individuals to participate in a study to evaluate the effectiveness of two interventions in the management of chronic ankle instability.

This University of Newcastle based project on 'The effect of Mobilisation with Movement on Anatomical and Clinical Characteristics of Chronic Ankle Instability (CAI)' will require you to attend the School of Health Sciences Research Laboratories in the Hunter Building to have **ankle movement**, **balance**, **function**, **pain assessments**, **and an x-ray examination** measured before and after the treatment. You will **receive 2-8 intervention sessions** depending on the condition of your ankle.

If you would like to find out more information about the study please phone Ishanka in the Discipline of Physiotherapy, University of Newcastle on 0424208114.

Research Higher Degree Candidate

Ishanka Weerasekara on 0424208114 or <u>Ishanka.weerasekara@uon.edu.au</u> Project Supervisor

Professor Darren Rivett on 49217220 or Darren.rivett@newcastle.edu.au

This project has been approved by the University's Human Research Ethics Committee, Approval No. H-2017-0354.

shaaka ka
--



Does this sound like you?

- More than 18 years old
- Have you had repeated ankle sprains? and/ or does your ankle 'give way' or feel unstable?
- Would you like to have the chance of free physiotherapy treatment for your ankle?

We are recruiting individuals to participate in a study to evaluate the effectiveness of two interventions in the management of chronic ankle instability.

This University of Newcastle based project on 'The effect of Mobilisation with Movement on Anatomical and Clinical Characteristics of Chronic Ankle Instability (CAI)' will require you to attend the School of Health Sciences Research Laboratories in the Hunter Building to have **ankle movement**, **balance**, **function**, **pain assessments**, and an x-ray examination measured before and after the treatment. You will receive 2-8 intervention sessions depending on the condition of your ankle.

If you would like to find out more information about the study please phone **Ishanka** in the Discipline of Physiotherapy, University of Newcastle on **0424208114**.

Research Higher Degree Candidate

Ishanka Weerasekara on 0424208114 or <u>Ishanka.weerasekara@uon.edu.au</u> Project Supervisor Professor Darren Rivett on 49217220 or <u>Darren.rivett@newcastle.edu.au</u>

This project has been approved by the University's Human Research Ethics Committee, Approval No. H-2017-0354.

kle instability study	kle instability study	kle instability study	kle instability study					
Ianka- 0424208114	anka- 0424208114	Ianka- 0424208114	anka- 0424208114	tanka - 0424208114	anka- 0424208114	anka- 0424208114	anka- 0424208114	anka- 0424208114
Ianka.weerasekara@uon.edu.au	anka weerasekara@uon.edu.au	Ianka.weerasekara@uon.edu.au	anka.weerasekara@uon.edu.au	tanka weeras ekara@uon.edu.	anka.weerasekara@uon.edu.au	anka.weerasekara@uon.edu.au	anka.weerasekara@uon.edu.au	anka.weerasekara@uon.edu.au
Ankle	Ankle	Ankle	Ankle	Ankle	Ankle	Ankle	Ankle	Ankle
Ishan	Ishan	Ishan	Ishan	Ishan	Ishani	Ishani	Ishani	Ishani
Ishan	Ishan	Ishan	Ishan	Ishan	Ishani	Ishani	Ishani	Ishani

Appendix 15 Data collection forms (both case-control and RCT studies) including questionnaires

Data Collection sheet



The information on this form will be kept strictly confidential and is used to research purposes.

Before subject arrives:						
Randomise the tests: (and keep a note here)						
A-DFROM TEST (good leg first > affected next)						
B-BALANCE TEST (good leg first > affected next)						
C- PPT TEST (good leg first > affected next)						
Randomise the three sites of algometer reading: (and keep a note here)						
X- Lateral ankle						
Y- Medial ankle						
Z- Ant tibialis						
PPT:						
Ensure devices are charged and Freedom USB is attached to the laptop (Shortcut to open Tracker 5), N/s using a 1 cm ² indenter tip, corresponding to 40kPa/s. Software records the maximum force (N) generated; may want to convert to kPa for analysis for consistency with UQ literature; some others have used N.						
Zero calibrate – to be done daily prior to patient arrival:						
Select System (top left) – zero calibrate – algometer, Wake up device by pushing the on switch on the PPT, Click on "Zero" and then OK						
BALANCE:						
Ensure Force plate is on and the USB attached. Short cut to MARS. Open the add details of the participants to be filled with height/weight/birthday.						

Brief introduction about the study and the process

Data Collection sheet



Participant No: ----

The information on this form will be kept strictly confidential and is used to research purposes.

Date: ----General Information
Gender: ----Height: ----Weight: ----Dominant leg: ----Affected leg: ----Right leg length: ----Left leg length: ----Beighton test:

De	scription	Right	Left	Scoring max.
				points
1.	Passive hyperextension of the knee ≥ 10			(2)
	degrees			
2.	Passive dorsiflexion of the fifth			(2)
	metacarpophalangeal joint to ≥ 90 degrees			
3.	Passive hyperextension of the elbow ≥ 10			(2)
	degrees			
4.	Passive apposition of the thumb to the flexor			(2)
	side of the forearm, while shoulder is flexed			
	90 degrees, elbow is extended, and hand is			
	pronated			
5.	Forward flexion of the trunk, with the knees		•	(1)
	straight, so that the hand palms rest easily			
	on the floor			
Tot	tal	•		(9)

Weight bearing lunge test (cm)

Attempts	Right	Left
Read 1		
Read 2		
Read 3		

Data Collection sheet



The information on this form will be kept strictly confidential and is used to research purposes.

Balance tests

SEBT test (cm)

Side	Right	Left
Anterior		
Postero medial		
Postero lateral		

Pressure pain threshold (Kpa)

Set up patient:

Select Perform exam (bottom of screen) Select Ankle study for the database Enter patient enter their study number as their first and last name (or find them in the list if they have come before) Enter Anklestudy1 as the incident Enter the name of the exam into Exam: baseline

Collect PPT

Select 'Instrumented' Select the "lightbulb" which is algometry Click on custom tests Select the practice test on the first round, and then select all three tests (left and right versions) **in the order they are to be performed** (check data collection sheet for randomised order). We will randomised the 3 positions, and then do left and right in each position (also randomised), but will do one position at a time to reduce moving around. Rest time is set to 10 seconds (between repeated tests)

Location	Right			Left		
	Read 1	Read 2	Read 3	Read 1	Read 2	Read 3
Anterior to the lateral malleolus over the anterior talo-fibular ligament (ATFL)						
Inferior to the medial malleolus over the deltoid ligament						
over the proximal third of tibialis anterior muscle belly						


The information on this form will be kept strictly confidential and is used to research purposes.

Patient-Specific Functional Scale (PSFS) and body chart

At rest: (with 'no pain' on the left hand of the line and 'worst possible' on the right.)

L		Left leg
No pain	Worst possible pain	
No pain	Worst possible pain	Right leg
List the three activities causes most pain?		
Activity 1:		
Activity 2:		
Activity 3:		



The information on this form will be kept strictly confidential and is used to research purposes.

Mark the level of pain of each activity in below lines (with 'unable to perform the activity' on the left hand of the line and 'able to perform activity at the same levels as before injury or problem' on the right.)

Activity 1:







The information on this form will be kept strictly confidential and is used to research purposes.

On the diagram below, please indicate the areas in which you are currently feeling symptoms.

- 1. First, shade (colour) the areas in which you are feeling pain.

 Next, Circle the area in which you are feeling <u>tingling pricking</u>, or <u>burning</u>.
 Finally, place an 'N' near the area where you are feeling <u>numbness</u>, <u>heaviness or other</u>. sensations.





The information on this form will be kept strictly confidential and is used to research purposes.

On the diagram below, please indicate the areas in which you are currently feeling symptoms.

1. First, shade (colour) the areas in which you are feeling pain.

2. Next (circle the area in which you are feeling tingling pricking, or burning.

3. Finally, place an 'N' near the area where you are feeling <u>numbness</u>, <u>heaviness or other</u> sensations.



THE CAIT QUESTIONNAIRE

Please tick the ONE statement in EACH question that BEST describes your ankles.

1. I have pain in my ankle

	Left Ankle	Right Ankle
Never	0	0
During sport	0	•
Running on uneven surfaces	0	0
Running on level surfaces	0	0
Walking on uneven surfaces	0	0
Walking on level surfaces	0	0

2. My ankle feels UNSTABLE

	Left Ankle	Right Ankle
Never	0	0
Sometimes during sport (not every time)	0	0
Frequently during sport (every time)	0	0
Sometimes during daily activity	0	0
Frequently during daily activity	0	٥

3. When I make SHARP turns, my ankle feels UNSTABLE

	Left Ankle	Right Ankle
Never	0	0
Sometimes when running	0	0
Often when running	0	0
When walking	0	0

4. When going down the stairs, my ankle feels UNSTABLE

	Left Ankle	Right Ankle
Never	0	0
If I go fast	0	0
Occasionally	0	0
Always	0	0

5. My ankle feels UNSTABLE when standing on ONE leg

	Left Ankle	Right Ankle
Never	0	0
On the ball of my foot	0	0
With my foot flat	0	0

6. My ankle feels UNSTABLE when

	Left Ankle	Right Ankle
Never	0	0
I hop from side to side	0	0
I hop on the spot	0	0
When I jump	0	0

7. My ankle feels UNSTABLE when

	Left Ankle	Right Ankle
Never	0	0
I run on uneven surfaces	0	0
I jog on uneven surfaces	0	0
I walk on uneven surfaces	0	0
I walk on a flat surface	0	0

8. TYPICALLY, when I start to roll over (or "twist") on my ankle, I can stop it

	Left Ankle	Right Ankle
Immediately	0	0
Often	0	0
Sometimes	0	0
Never	0	0
I have never rolled over on my ankle	0	0

9. After a TYPICAL incident of my ankle rolling over, my ankle returns to "normal"

	Left Ankle	Right Ankle
Almost immediately	0	0
Less than one day	0	0
1–2 days	0	0
More than 2 days	0	0
I have never rolled over on my ankle	0	0

Foot and Ankle Ability Measure (FAAM-ADL)

Please answer <u>every question</u> with <u>one response</u> that most closely describes your condition within the past week.

If the activity in question is limited by something other than your foot or ankle mark "Not Applicable" (N/A).

	No Difficulty at all	Slight Difficulty	Moderate Difficulty	Extreme Difficulty	Unable to do	× Not applicable
Standing	0	0	0	0	0	0
Walking on even ground	0	0	0	0	0	0
Walking on even ground without shoes	0	0	0	0	0	0
Walking up hills	0	0	0	0	0	0
Walking down hills	0	0	0	0	0	0
Going up stairs	0	0	0	0	0	0
Going down stairs	0	0	0	0	0	0
Walking on uneven ground	0	0	0	0	0	0
Stepping up and down curbs	0	0	0	0	0	0
Squatting	0	0	0	0	0	0
Coming up on your toes	0	0	0	0	0	0
Walking initially	0	0	0	0	0	0
Walking 5 minutes or less	0	0	0	0	0	0
Walking approximately 10 minutes	0	0	0	0	0	0
Walking 15 minutes or greater	0	0	0	0	0	0

Because of your foot and ankle how much difficulty do you have with:

	No difficulty at all	Slight Difficulty	Moderate difficulty	Extreme Difficulty	Unable to do	× Not applicable
Home responsibilities	0	0	0	0	0	0
Activities of daily living	0	0	0	0	•	0
Personal care	0	0	0	0	•	0
Light to moderate work (standing, walking)	0	0	0	0	0	0
Heavy work (push/pulling, climbing, carrying	0	0	0	0	0	0
Recreational activities	0	•	0	•	0	•

How would you rate your current level of function during you usual activities of daily living from 0 to 100 with **100 being your level of function prior to your foot or ankle problem** and **0 being the inability to perform** any of your usual daily activities?



Foot and Ankle Ability Measure (FAAM-SPORTS)

Because of your foot and ankle how much difficulty do you have with:

	No Difficulty	Slight Difficulty	Moderate Difficulty	Extreme Difficulty	Unable to do	× Not applicable
Running	0	0	0	0	0	0
Jumping	0	0	0	0	0	0
Landing	0	0	0	0	0	0
Starting and stopping quickly	0	0	0	0	0	0
Cutting/lateral movements	0	0	0	0	0	0
Low impact activities	0	0	0	0	0	0
Ability to perform activity with your normal technique	0	0	0	0	0	0
Ability to participate in your desired sport as long as you like	0	0	0	0	0	0

How would you rate your current level of function during your sports related activities from 0 to 100 with **100 being your level of function prior to your foot or ankle problem** and **0 being the inability to perform** any of your usual daily activities?

	0	10	20	30	40	50	60	70	80	90	100
Percentage 9	6										

Overall, how would you rate your current level of function?

Normal	Nearly normal	Abnormal	Severely abnormal
0	0	0	0

Foot and Ankle Outcome Score (FAOS) questionnaire

This survey asks for your view about your foot/ankle. Answer every question by ticking the appropriate box, only one box for each question. If you are unsure about how to answer a question, please give the best answer you can.

Symptoms

These questions should be answered thinking of your foot/ankle symptoms during the last week.

	Never	Rarely	Sometimes	Often	Always
S1. Do you have swelling in your foot/ankle?	0	0	0	0	0
S2. Do you feel grinding, hear clicking or any other type of noise when your foot/ankle moves?	0	0	0	۲	0
S3. Does your foot/ankle catch or hang up when moving?	0	0	0	0	0

These questions should be answered thinking of your foot/ankle symptoms during the last week.

	Always	Often	Somtimes	Rarely	Never
S4. Can you straighten(ankle move towards ground) your foot/ankle fully?	0	0	0	0	0
S5. Can you bend (ankle move towards your face)your foot/ankle fully?	0	0	0	0	0

Stiffness

The following questions concern the amount of joint stiffness you have experienced during the <u>last week</u> in your foot/ankle. Stiffness is a sensation of restriction or slowness in the ease with which you move your joints.

	None	Mild	Moderate	Severe	Click to write Scale point 5
S6. How severe is your foot/ankle stiffness after first wakening in the morning?	0	0	0	0	0
S7. How severe is your foot/ankle stiffness after sitting, lying or resting later in the day?	0	0	0	0	0

Pain

	Never	Monthly	Weekly	Daily	Always
P1. How often do you experience foot/ankle pain?	0	0	0	0	0

What amount of foot/ankle pain have you experienced the last week during the following activities?

	None	Mild	Moderate	Severe	Extreme
P2. Twisting/pivoting on your foot/ankle	0	0	0	0	0
P3. Straightening foot/ankle fully	0	\odot	0	0	0
P4. Bending foot/ankle fully	0	0	0	0	0
P5. Walking on flat surface	0	0	0	0	0
P6. Going up or down stairs	0	0	0	0	0
P7. At night while in bed	0	0	0	0	0
P8. Sitting or lying	0	0	0	0	0
P9. Standing upright	0	0	0	0	0

Function, daily living

The following questions concern your physical function. By this we mean your ability to move around and to look after yourself. For each of the following activities please indicate the degree of difficulty you have experienced in the <u>last week</u> due to your foot/ankle.

	None	Mild	Moderate	Severe	Extreme
A1. Descending stairs	0	0	0	0	0
A2. Ascending stairs	0	0	0	0	0

For each of the following activities please indicate the degree of difficulty you have experienced in the last week due to your foot/ankle.

	None	Mild	Moderate	Severe	Extreme
A3. Rising from sitting	0	0	0	0	0
A4. Standing	0	0	0	0	0
A5. Bending to floor/pick up an object	0	\odot	0	0	0
Aδ. Walking on flat surface	0	\odot	0	0	0
A7. Getting in/out of car	0	\odot	0	0	0
A8. Going shopping	0	\odot	0	0	0
A9. Putting on socks/stockings	0	\odot	0	0	0
A10. Rising from bed	0	\odot	0	0	0
A11. Taking off socks/stockings	0	\odot	0	0	0
A12. Lying in bed (turning over, maintaining foot/ankle position)	0	0	0	0	0
A13. Getting in/out of bath	0	0	0	0	0
A14. Sitting	0	0	0	0	0
A15. Getting on/off toilet	0	0	0	0	0

For each of the following activities please indicate the degree of difficulty you have experienced in the <u>last week</u> due to your foot/ankle.

	None	Mild	Moderate	Severe	Extreme
A16. Heavy domestic duties (moving heavy boxes, scrubbing floors, etc)	0	0	0	0	0
A17. Light domestic duties (cooking, dusting, etc)	0	\odot	0	0	0

Function, sports and recreational activities

The following questions concern your physical function when being active on a higher level. The questions should be answered thinking of what degree of difficulty you have experienced during the <u>last</u> <u>week</u> due to your foot/ankle.

	None	Mild	Moderate	Severe	Extreme
SP1. Squatting	0	0	0	0	0
SP2. Running	0	0	0	0	0
SP3. Jumping	0	0	0	0	0
SP4. Twisting/pivoting on your injured foot/ankle	0	0	0	0	0
SP5. Kneeling	0	0	0	0	0

Quality of Life

	Never	Monthly	Weekly	Daily	Constantly
Q1. How often are you aware of your foot/ankle problem?	0	0	0	0	0

	Not at			0	
	all	Mildly	Moderately	Severely	Extremely
Q2. Have you modified your life style to avoid potentially damaging activities to your foot/ankle?	0	0	0	0	0
Q3. How much are you troubled with lack of confidence in your foot/ankle?	0	0	0	0	0

	None	Mild	Moderate	Severe	Exrteme
Q4. In general, how much difficulty do you have with your foot/ankle?	0	0	0	0	0



Perception about the intervention (to be distributed at the end of the intervention phase)

Appendix 16 Journal publications



journal homepage: www.archives-pmr.org Archives of Physical Medicine and Rehabilitation 2018;99:1395-412

REVIEW ARTICLE (META-ANALYSIS)

Clinical Benefits of Joint Mobilization on Ankle Sprains: A Systematic Review and Meta-Analysis



Ishanka Weerasekara, MPhil, Peter Osmotherly, PhD, Suzanne Snodgrass, PhD, Jodie Marquez, MMgt(Health), Rutger de Zoete, MSc, Darren A. Rivett, PhD

From the School of Health Sciences, Faculty of Health and Medicine, The University of Newcastle, Callaghan, Australia.

Abstract

Objective: To assess the clinical benefits of joint mobilization for ankle sprains.

Data Sources: MEDLINE, MEDLINE In-Process, Embase, AMED, PsycINFO, CINAHL, Cochrane Library, PEDro, Scopus, SPORTDiscus, and Dissertations and Theses were searched from inception to June 2017.

Study Selection: Studies investigating humans with grade I or II lateral or medial sprains of the ankle in any pathologic state from acute to chronic, who had been treated with joint mobilization were considered for inclusion. Any conservative intervention was considered as a comparator. Commonly reported clinical outcomes were considered such as ankle range of movement, pain, and function. After screening of 1530 abstracts, 56 studies were selected for full-text screening, and 23 were eligible for inclusion. Eleven studies on chronic sprains reported sufficient data for meta-analysis.

Data Extraction: Data were extracted using the participants, interventions, comparison, outcomes, and study design approach. Clinically relevant outcomes (dorsiflexion range, proprioception, balance, function, pain threshold, pain intensity) were assessed at immediate, short-term, and long-term follow-up points.

Data Synthesis: Methodological quality was assessed independently by 2 reviewers, and most studies were found to be of moderate quality, with no studies rated as poor. Meta-analysis revealed significant immediate benefits of joint mobilization compared with comparators on improving posteromedial dynamic balance (P=.0004), but not for improving dorsiflexion range (P=.16), static balance (P=.96), or pain intensity (P=.45). Joint mobilization was beneficial in the short-term for improving weight-bearing dorsiflexion range (P=.003) compared with a control.

Conclusions: Joint mobilization appears to be beneficial for improving dynamic balance immediately after application, and dorsiflexion range in the short-term. Long-term benefits have not been adequately investigated.

Archives of Physical Medicine and Rehabilitation 2018;99:1395-412

© 2017 by the American Congress of Rehabilitation Medicine

Ankle sprains are a common injury in sports and the general community, and may lead to chronic pain, functional limitations, and physical disability.^{1,2} Epidemiologic studies conducted in various countries highlight the high incidence of ankle sprains during sports training and competition, with rates reported as 7 per

1000 in Denmark, 6.09 per 1000 in the United Kingdom, and 2.15 per 1000 in the United States in person-years.³⁻⁵ Plantarflexion inversion sprain or lateral ankle sprain is the most common type of ankle sprain.⁶ It typically results in an injury of the inferior tibiofibular ligament, anterior tibiofibular ligament, or the bifurcate ligament.⁷ Eversion injuries often result in damage to the deltoid and spring ligaments of the medial aspect of the ankle.⁷

According to the clinical practice guidelines linked to the *International Classification of Functioning, Disability and Health* from the Orthopaedic Section of the American Physical Therapy Association, manual therapy is recommended for both the acute and progressive loading phases of rehabilitation.⁸ Management of

Presented as an abstract to Sports Medicine Australia October 12-15, 2016 in Melbourne, Victoria, Australia.

Supported by an Australian Government Research Training Program Scholarship. International Prospective Register of Systematic Reviews (PROSPERO) Registration No.: CRD42016030194.

Disclosures: none.

ankle sprains commonly involves mobilization procedures applied to the joint, such as nonthrust joint mobilization, high-velocity thrust manipulation, and mobilization with movement (MWM).

The mechanisms by which these techniques are purported to work are biomechanical (such as stretching/tearing tissue, inducing cavitation within the joint, reducing muscle hypertonic-ity/stiffness) and neurophysiological, potentially including spinal cord and supraspinally mediated mechanisms.^{9,10}

Several studies¹¹⁻²¹ have investigated the effects of manual therapy on ankle sprains using a variety of outcome measures including pain, range of motion (ROM), and function from the acute to chronic stages of recovery, with different results reported. Several systematic reviews²²⁻²⁶ have attempted to collate this evidence but have been limited by their narrow focus on lateral ankle sprains and restricted outcome measures. Previous systematic reviews have all included some studies that involved other interventions such as "rest, ice, compression, and elevation" (RICE) and home exercise programs as an adjunct to mobilization. Therefore, they have not actually assessed mobilization as the sole intervention. Moreover, the clinical benefits of joint mobilization have not yet been evaluated through meta-analysis, despite it being a common intervention used in the rehabilitation of a number of ankle conditions and despite the growing body of empirical literature.

The present systematic review aims to address these limitations by synthesizing and meta-analyzing the available evidence for ankle joint mobilization (including high-velocity thrust manipulation) in grade I or II ankle sprains of the medial or lateral ligaments in the acute/subacute/chronic stages of rehabilitation in any ambulant setting.

Methods

Registration

The protocol for this systematic review was registered with the International Prospective Register of Systematic Reviews on January 12, 2016 (CRD42016030194).

Search strategy

A search of electronic databases, including MEDLINE, MED-LINE In-Process, Embase, AMED, PsycINFO, CINAHL, Cochrane Library, PEDro, Scopus, SPORTDiscus, and Dissertations and Theses was conducted from inception to June 2017. In addition to the database search, a hand search of the reference lists of identified studies was also carried out. A search strategy (supplemental appendix S1, available online only at http://www. archives-pmr.org/) was developed for the main search strings of ankle sprain and mobilization. Keywords used for "ankle sprain" included sprain, talocrural joint, ligament injuries, lateral

List of	abbreviations:
CI	confidence interval
DFROM	dorsiflexion range of motion
MCID	minimal clinically important difference
MD	mean difference
MWM	mobilization with movement
RICE	rest, ice, compression, and elevation
ROM	range of motion
SEBT	star excursion balance test
SMD	standardized mean difference

ligament, medial ligament, deltoid ligament, collateral ligament, anterior talofibular ligament, posterior talofibular ligament, sprain and strain, and ankle twist. Keywords used for "mobilization" included manual therapy, joint mobilization, manipulation, MWM, Maitland, Mulligan, and rehabilitation. These terms were used alone and in combinations during the search.

Identification and selection of studies

Full-text randomized controlled trials, crossover studies, crosssectional studies, cohort studies, and case series published in peerreviewed journals and dissertations were considered for the present review. Studies were not restricted by language, provided the title and abstract were in English. Studies not involving live human participants (eg, model-based, animal, and cadaveric investigations) were excluded. Conference proceedings, commentaries, research notes, editorials, and letters were also excluded. To be included, studies were required to meet the criteria that follow.

Participants

Participants were live humans (without any age limitation) with a grade I or II lateral or medial ligament sprain of the ankle at any stage of recovery (acute to chronic) in any ambulant setting who have been treated with joint mobilization. Studies involving grade III sprains, fractures (other than Weber type A), and syndesmotic injuries were excluded from this review.

Intervention

Studies reporting any type of joint mobilization techniques applied to the talocrural joint, subtalar joint, or inferior tibiofibular joint by a physiotherapist, medical practitioner, osteopath, chiropractor, or athletic trainer were eligible for inclusion in the review. Interventions other than therapist-performed joint mobilization were excluded from the review.

Comparators

Studies reporting any conservative intervention for comparison, such as exercise therapy, elevation and icing, supportive strapping, sham intervention, or no treatment, were eligible for inclusion. Control groups with healthy subjects were also eligible as a comparator. Studies that compared mobilization techniques to surgical interventions were excluded.

Outcome measures

All commonly reported clinical impairments (pain, swelling, balance, proprioception, strength, stability, gait), activity restriction and self-reported confidence, community participation, quality of life, reinjury rate, function, and return to sport were considered for the review. The primary outcomes of interest were ankle ROM, pain, quality of life, and function.

Timing of the measurement of the outcomes was categorized as "immediate," measured immediately after the intervention²⁷; "short-term," measured up to 3 months after the intervention²⁸; and "long-term," measured at 3 or more months²² after the intervention.

Identified studies were exported to reference management software (EndNote X7.3.1^a), and duplicate records were manually removed. Study titles and abstracts were initially screened by 2 independent reviewers, followed by screening of full-text articles, to determine the eligibility of the identified studies. Disagreement between the reviewers was resolved by consensus or involvement of a third reviewer. The level of agreement between reviewers was assessed using Cohen's kappa.²⁹

Joint
mobilization
in
ankle
sprain
management

www.archives-pmr.org

			Intervention and		Measurement		
Study	Design	Sample	Dosage	Comparator	Time Points	Outcomes	Results
Alanson et al, ³⁶ 2012	RCT	17 (10M) Grade 1/2 Chronic lateral ankle sprains	TCJ (anteroposterior) mobilization + TCJ traction 30s	Sham	Immediate	Non—weight-bearing DFROM, proprioception (joint position sense)	 Non-weight-bearing DFROM, significantly improved across time (P=.04). Joint position sense significantly improved across time at target angle 10° plantarflexion (P=.03).
Beazell et al, ³⁷ 2012	RCT	43 Chronic ankle sprains	Distal TFJ manipulation + HVLA thrust 1 repetition	No intervention	Immediate, short-term (1, 2, and 3 wk)	Weight-bearing DFROM, static balance (single-limb stance), function (step down test, self-reported function, FAAM sports)	Weight-bearing DFROM not significant $(P=.82)$. Single-limb stance not significant $(P=.42)$. Function not significant—step down test $(P=.76)$, self-reported function $(P=.61)$, FAAM sports $(P=.83)$
Collins et al, ¹³ 2004	Randomized crossover	16 (8M) Grade 2 Subacute lateral ankle sprains	Weight- bearing MWM TCJ (posterior talar glide, posteroanterior tibial glide) 3 sets of 10 repetitions	Placebo No intervention	Immediate	Weight- bearing DFROM, pressure-pain threshold, thermal pain threshold	Weight- bearing DFROM significantly improved across time (P =.013) and no significant group difference (vs placebo, P =.202; vs control, P =.208). Pressure-pain threshold and thermal pain threshold not significant (P <.05).
Cruz-Díaz et al, ¹⁴ 2015	RCT	81 (47M) Chronic ankle sprains	Weight—bearing MWM TCJ (posterior talar glide, posteroanterior tibial glide) 2 sets of 10 repetitions, 2 sessions/wk for 3wk	Sham, no intervention	Immediate, short-term (3wk), long- term (6mo)	Weight—bearing DFROM, dynamic balance (SEBT)	<pre>Weight-bearing DFROM significantly improved—P<.0001 (at each time point). Dynamic balance significantly improved—P<.0001 (each direction of SEBT)</pre>
Gilbreath et al, ²¹ 2014	Prospective longitudinal	11 (5M) Chronic ankle sprains	Weight—bearing MWM TCJ (posterior talar glide, posteroanterior tibial glide) 2 sets of 4 repetitions 4min of MWM×3 sessions over 1wk	No control group	Short-term (after 24—48h)	Weight-bearing DFROM, dynamic balance (SEBT), function (FAAM)	Weight-bearing DFROM not significant $(P=.69)$. Dynamic balance not significant (SEBT-anterior, $P=.99$; posteromedial, $P=.15$; posterolateral, $P=.24$). FAAM ADL not significant $(P=.19)$. FAAM Sports significantly improved across time $(P=.01)$.
Harkey et al, ³⁸ 2014	RCT	30 (14M) Chronic ankle sprains	Maitland mobilization TCJ (anteroposterior grade III) 3 sets of 60s	No intervention	Immediate	Non—weight-bearing DFROM, dynamic balance (SEBT)	Non-weight-bearing DFROM significantly improved (P =.049). Dynamic balance no improvement (P >.05). (continued on next page)

Table 1 (continued)							
Study	Design	Sample	Intervention and Dosage	Comparator	Measurement Time Points	Outcomes	Results
Hoch and McKeon, ³⁹ 2011	Randomized crossover	20 (9M) Chronic ankle sprains	Maitland mobilization TCJ (anterior posterior grade III) 50±5 of 1-s oscillations ×2	No intervention	Immediate	Weight- bearing DFROM, static balance, dynamic balance (SEBT), talar stiffness	Weight-bearing DFROM significantly improved (P =.01). Static balance significantly improved. Time to boundary anteroposterior minima significantly improved (P <.0001). Dynamic balance—not significant— P =.98 (normalized reach distance). Talar stiffness not significant (P =.08).
Hoch et al, ⁴⁰ 2012	Prospective longitudinal	12 (6M) Chronic ankle sprains	Maitland mobilization TCJ (anteroposterior grade III) + TCJ traction 2 sets of 2min traction and 4 sets of 2min mobilization	No control group	Short-term (24—48h and 1-wk follow-up)	Weight-bearing DFROM, dynamic balance, function (FAAM)	Weight-bearing DFROM significantly improved across time (P <.0001). Dynamic balance significantly improved across time (SEBT anterior, P <.0001; posteromedial, P=.003; posterolateral, P <.0001). FAAM ADL and Sports significantly improved across time (P =.001).
Hoch et al, ⁴¹ 2014	Prospective longitudinal	12 (6M) Chronic ankle sprains	Maitland mobilization TCJ (anteroposterior grade III) + TCJ traction 2 sets of 2min traction and 4 sets of 2min	No control group	Short-term (24—48h, and 1-wk follow-up)	Static balance, talar stiffness	 Static balance not significant. Time to boundary anteroposterior and time to boundary mediolateral not significant (<i>P</i>>.05). Talar stiffness not significant (<i>P</i>>.05).
Hopper et al, ⁴² 2009	Randomized controlled Within-subjects design	20 (8M) Chronic ankle sprains	Mulligan ankle taping Not explicitly stated	Injured taped, Injured untaped, Uninjured taped, Uninjured untaped	Immediate	Static balance, dynamic balance (wandering, overshoot, reaction time)	Static balance significantly improved in postural sway recovery across time (P <.001). Single-limb stance not significant- (P =.792), Dynamic tracking balance not significant; wandering (P =.559), overshoot (P =.547), reaction time (P =.142).
Houstan et al, ⁴³ 2013	Prospective longitudinal	12 (6M) Chronic ankle sprains	 Maitland mobilization TCJ (anteroposterior grade III) + TCJ traction 4min of traction and 8min of mobilization 6 sessions over 2wk. 	No control group	Immediate, short-term (1-wk follow-up)	Function (FAAM Sports)	FAAM ADL some components significantly improved across time; walking on even ground (P =.06); going down stairs (P =.07); walking on uneven ground (P =.03); light to moderate work (P =.06); heavy work (P =.03); recreational activity (P =.07). (continued on next page)

1398

1399

Table 1 (continued)							
			Intervention and		Measurement		
Study	Design	Sample	Dosage	Comparator	Time Points	Outcomes	Results
							FAAM Sports some components significantly improved across time; landing (P =.03); low impact activities (P =.07); cutting (P =.02).
Joseph et al, ⁴⁴ 2010	RCT	40 (19M) Grade 1/2 Chronic lateral ankle sprains	Ankle axial elongation TCJ (superior inferior)- HVLA thrust 6 sessions over 3wk	Muscle energy technique	Short-term (1mo)	DFROM, plantarflexion ROM, static balance, pain quality and intensity, function (Functional Evaluation Scale)	DFROM significantly improved across time (P <.001) and no significant group differences (P =.713). Plantarflexion ROM significantly improved across time (P <.001) and no significant group differences (P =.300). Single-limb stance—eyes closed significantly improved across time (P <.001) and no significant group differences (P =.344). Single-limb stance—eyes open significantly improved across time (P <.001) and no significant group differences (P =.413). McGill significantly improved across time (P <.001) and no significant group differences (P =.077). Functional Evaluation Scale significantly improved across time (P <.001) and no significant group differences (P =.16()
Kohne, et al, ⁴⁵ 2007	RCT	30 (21M) Grade 1/2 Chronic, recurrent lateral ankle sprains	Ankle axial elongation TCJ (superior inferior by a mortise separation) —(6 manipulations over 4wk)	Single manipulation treatment	Short-term (fifth week follow- up)	DFROM, proprioception (joint position sense), pressure-pain threshold, pain intensity	DFROM significantly improved: P = .028 (across time). Joint position sense at 5° plantarflexion error significantly improved: $P = .029$ (across time). Pressure-pain threshold (P value not reported), pain intensity (P value not reported)
Lopez-Rodriguez et al, ⁴⁶ 2007	Randomized controlled within-subject repeated measures	52 (35M) Grade 2 Chronic lateral ankle sprains	TCJ Manipulation (caudal) HVLA thrust + posterior gliding manipulation TCJ -HVLA thrust 1min	Placebo	Immediate	Proprioception	Proprioception significantly improved; load support—bilateral posterior load (P =.016), anterior load (P =.04), posterior load (P =.043), posterior-anterior load (P =.016). (continued on next page)

	/		T 1 1				
Study	Design	Sample	Intervention and Dosage	Comparator	Measurement Time Points	Outcomes	Results
Marron-Gomez et al, ¹⁵ 2015	RCT	52 (31M) Chronic ankle sprains	Weight-bearing MWM TCJ (posterior talar glide, posteroanterior tibial glide) 1 set of 10 repetitions TCJ HVLA distraction thrust ×3	Placebo	Immediate, short-term (24 and 48h)	Weight-bearing DFROM	MWM—Weight-bearing DFROM significantly greater than placebo -P < .05 (immediately and short- term). HVLA—weight-bearing DFROM significantly greater than placebo -P < .001 (immediately) and P = .001(short-term).
Pellow and Brantingham, ⁴⁷ 2001	RCT	30 (19M) Grade 1/2 subacute and chronic lateral ankle sprains	Ankle axial elongation (TCJ— superior inferior by a mortise separation) 8 manipulations over 4wk	Detuned ultrasound treatment	Short-term (1-mo follow-up)	Non—weight-bearing DFROM, pain threshold, pain quality and intensity, function (Functional Evaluation Scale)	Non-weight-bearing DFROM significantly improved across time (P=.001) and between groups (P=.001). Pain threshold significantly improved across time $(P=.002)$ and no significant group differences (P=.395). McGill significantly improved across time $(P=.001)$ and between group (P=.004). Pain intensity significantly improved across time $(P=.002)$ and between groups $(P=.004)$. Functional Evaluation Scale significantly improved across time (P=.001) and between groups (P=.001) and between groups
Plante, ⁴⁸ 2012	RCT	20 (12M) Chronic ankle sprains	TCJ (anteroposterior) 10 oscillations	Healthy subjects	Immediate	Weight-bearing DFROM, static balance, function (dynamic functional tasks)	 Weight-bearing DFROM significantly improved across time (<i>P</i><.0001). Single-limb stance, center of pressure significantly improved (<i>P</i><.04). Dynamic functional task (center of pressure medial-lateral during jump task significantly improved [<i>P</i><.001]; center of pressure medial-lateral during squat significantly improved [<i>P</i><.022]; center of pressure medial-lateral during stance task significantly improved [<i>P</i><.039]).

www.archives-pmr.org

0
.⊒•
1
–
1
띭
Ц.
Li.
N
بم
Ξ.
0
n
=.
_
a
X
<u> </u>
5
2
പ്
<u> </u>
-
-
1
<u>a</u>
<u> </u>
<u>n</u>
е
- +

Table 1 (continued)							
Study	Design	Sample	Intervention and Dosage	Comparator	Measurement Time Points	Outcomes	Results
Reid et al, ⁴⁹ 2007	Randomized crossover	23 (8M) Chronic lateral ankle	Weight- bearing MWM (posterior talar glide, posteroanterior tibial glide) 10 repetitions ×2	Sham	Immediate	Weight-bearing DFROM	Weight-bearing DFROM significantly improved (<i>P</i> =.02).
Someeh et al, ⁵⁰ 2015	Experimental study design—within subjects	32 (20M) Chronic ankle sprains	Mulligan ankle taping/ fibular repositioning taping Not explicitly stated	Healthy subjects	Immediate	Dynamic balance (SEBT)	Dynamic balance significantly improved across time—SEBT overall reach (P=.001).
Someeh et al, ⁵¹ 2015	Experimental study design—within subjects	32 (20M) Chronic ankle sprains	Mulligan ankle taping Not explicitly stated	Healthy subjects	Immediate	Function (dynamic functional tasks), participants' perceptions of stability and confidence	Function significantly improved across time; single leg hopping (P =.014); figure-of-8 hopping (P =.05); side hopping (P =.001). Confidence in abovementioned functional tests significantly improved across time; consequently, P =.023, .048, and .038, respectively.
Vicenzino et al, ¹⁶ 2006	Randomized crossover	16 (8M) Chronic lateral ankle sprains	 Non-weight-bearing MWM (anteroposterior talar glide for dorsiflexion), 4 glides of 10s 4 sets Weight-bearing MWM (posterior talar glide, posteroanterior tibial glide) 4 sets of 10 glides 	No intervention	Immediate	Weight-bearing DFROM, talar stiffness	<pre>Weight-bearing DFROM significantly improved (P=.017). Talar glide significantly improved (P<.001).</pre>
Wells, ⁵² 2012	RCT	17 (7M) Chronic ankle sprains	Maitland mobilization (TCJ—anteroposterior grade IV) 3 repetitions, 60s	No intervention	Immediate	Weight -bearing DFROM, non—weight-bearing DFROM, dynamic balance, pain intensity, static balance, stiffness, function (self- reported function)	Weight -bearing DFROM not significant $(P=.95)$. Non-weight-bearing DFROM not significant $(P=.1)$. Dynamic balance not significant; SEBT composite $(P=.8)$; anterior $(P=.07)$; posteromedial $(P=.79)$; posterolateral $(P=.73)$. Pain not significant $(P=.06)$. Stiffness not significant $(P=.59)$. Stability not significant $(P=.40)$. Function (visual analog scale) not significant $(P=.44)$.

(continued on next page)

1401

Table 1 (continued)							
Study	Design	Sample	Intervention and Dosage	Comparator	Measurement Time Points	Outcomes	Results
Yeo et al, ⁵³ 2011	Randomized, controlled, within-subject repeated measures	13 (10M) Grade 2 Subacute lateral sprain	Maitland mobilization (distal TFJ anteroposterior) 3 sets of 1min mobilization	Placebo No intervention	Immediate	Weight-bearing DFROM, pressure-pain threshold, pain intensity, function (Functional Evaluation Scale)	Weight-bearing DFROM significantly improved (P <.0001). Pressure-pain threshold significantly improved (P <.0001). Pain intensity not significant (P =.369). Functional Evaluation Scale not significant (P =.475).
NOTE. "Immediate," m Abbreviations: ADL, ac	neasured immediately a tivities of daily living;	fter the intervention. FAAM, Foot and Ankle	"Short-term," measured up Ability Measure; HVLA, higl	to 3 months after t h velocity, low ampli	he intervention. "Lo tude; M, male; RCT, r	ng-term," measured at 3 or 1 andomized controlled trial; T	more months after the intervention. CJ, talocrural joint; TFJ, tibiofibular joint.

Assessment of methodological quality

The methodological quality of individual studies was assessed using the PEDro Scale for randomized controlled trials and the Quality Assessment Tool for Observational Cohort and Crosssectional Studies.³⁰⁻³² Two independent reviewers assessed the methodological quality, and the level of agreement between reviewers was assessed using Cohen's kappa.

Assessment of quality of evidence

The overall quality of evidence was assessed at the stage of metaanalysis, using the Grading of Recommendations, Assessment, Development, and Evaluation approach.³³ The quality of the evidence was classified as either high, moderate, low, or very low.³⁴ Risk of bias, consistency of results, directness (eg, generalizability), and precision (eg, sufficient data) were considered in assessing the overall quality.³⁵

Data extraction and statistical analysis

Descriptive data were extracted using an extraction table (table 1). Authors were contacted if possible where there were difficulties extracting data from the published article. Where feasible, study data that were comparable in terms of participant characteristics, outcome measures, and follow-up periods were pooled, and a meta-analysis was performed.

For the meta-analysis, the standardized mean difference (SMD) was calculated for the outcomes where the means and SDs were provided pre- and postintervention. This conversion of the data to a common scale permitted comparison of studies that used different tools to measure the same outcome. This review followed the general practice of interpretation for small, medium, and large effect sizes (0.2, small effect; 0.5, medium effect; 0.8, large effect).54,55 The mean difference (MD) was calculated for studies using the same instrument for measurement. The results were reported in forest plots with 95% confidence intervals (CIs). The minimal clinically important difference (MCID) was used to interpret the clinical meaningfulness of the findings. Inconsistency was quantified by calculating I^2 and interpreted as follows: 30% to 59% may represent moderate heterogeneity, 60% to 89% substantial heterogeneity, and 90% to 100% considerable heterogeneity between studies. If I^2 was >30%, a random-effects model was used to incorporate intertrial heterogeneity.³¹

In the instance of multiple comparison groups, the sham group was selected as the control condition. For the outcome of "static balance," studies with eyes-closed balance were selected to maintain the homogeneity of the analysis. Further, in studies with multiple time points, measurements taken at 2 to 3 weeks were selected for the meta-analysis (eg, if effects were measured at the time points of 2d, 3wk, and 2mo in a single study, data from measurements at 3wk were selected for the analysis). All statistical analyses were conducted using RevMan 5.3.^b

Results

Selection and characteristics of included studies

The database search identified 1521 studies after duplicate removal, and a further 9 studies were identified through citation tracking and hand searching of reference lists (fig 1). After the first



Fig 1 Flow chart of study selection.

stage of screening (using study title and abstract), 56 studies (database search, n=47; hand search, n=9) were identified as eligible for inclusion from the original 1530 (database search, n=1521; hand search, n=9) studies. Common reasons for exclusion after title and abstract screening included ineligible study design, joint mobilization not assessed in isolation, and study aim not relevant to the review research question. A further 33 studies were excluded in second stage (full text) screening, and reasons for exclusion included study aim not relevant to research question^{12,18,19,56-71} (n=19); conference proceedings, commentaries, and research notes⁷²⁻⁷⁸ (n=7); not peer reviewed⁷⁹⁻⁸¹ (n=3); full text not available^{82,83} (n=2); study protocol only⁸⁴ (n=1); and thesis removed as the relevant published article was included⁸⁵ (n=1). Twenty-three studies (including 3 theses) were therefore included in the current review. The interreviewer agreement for the title/abstract and full-text screenings was considered to be very good ($\kappa = .80$; 95% CI, .72–.89) and good (κ = .71; 95% CI, .52–.90), respectively. All disagreements were resolved by consensus. The data from 11 studies (including 2 theses^{48,52}) were available and deemed appropriate for inclusion in the meta-analysis (see fig 1). Publication bias was visually observed using funnel plots (supplemental appendix S2, available online only at http://www.archives-pmr.org/).

The included studies were conducted in 7 countries (Australia, Canada, Iran, New Zealand, South Africa, Spain, United States) and involved a total of 585 participants. Twenty-one studies evaluated chronic ankle sprains, and 3 studies investigated subacute sprains. Outcomes measured varied widely and included dorsiflexion range of motion (DFROM), proprioception, stability/ balance, pain threshold (pressure and thermal), pain intensity and quality, function, talar stiffness, postural sway, and patient confidence. A range of joint mobilization techniques were used, and these included MWM in weight bearing and/or non-weight bearing $(n=6)^{13-16,21,49}$; anteroposterior talocrural mobilization (Maitland grades III and IV)⁸⁶ $(n=4)^{38,39,48,52}$; high-velocity, lowamplitude ankle axial elongation manipulation and manipulation of the talocrural joint $(n=6)^{15,36,44-47}$; Mulligan ankle taping $(n=3)^{42,50,51}$; distal tibiofibular joint manipulation or mobilization $(n=2)^{37,53}$; and combined mobilization and traction of the talocrural joint (n=4).^{36,40,41,43} Mulligan ankle taping was included because it aims to mimic a MWM by sustaining the fibula glide during daily activities.⁷ These techniques were variously applied by physiotherapists, medical practitioners, chiropractors, and athletic trainers. Table 1 describes the participants, interventions, comparators, outcome measures, and results of the included studies.



n=number of trials MWM=mobilisation with movement

Percentage of outcome evaluations with positive findings

Percentage of outcome evaluations without positive findings

Fig 2 Percentage and number of outcome evaluations with and without positive findings following each technique combination of mobilisation for any clinically relevant outcome at any time point.

The immediate effects of joint mobilization were evaluated in 17 studies, short-term effects in 10 studies, and the long-term effects were assessed in only 1 study (see table 1). No studies evaluating effects on gait parameters, quality of life, reinjury rate, or strength were located in our search. In this systematic review, participants with chronic ankle sprains were included in 21 studies, and 3 studies included participants with subacute sprains. No studies measuring the effectiveness of mobilization in isolation for acute ankle sprains were able to be found. A meta-analysis was conducted using 11 studies, all involving participants with chronic ankle sprains.

Common mobilization techniques used in rehabilitation of ankle sprains

Five combinations of mobilization techniques were used in the 23 studies, including Mulligan MWM and taping techniques, Maitland mobilization with and without traction, and manipulation. The number of studies with positive effects on any clinically relevant outcome is contrasted against the number of studies with no positive effects, for each mobilization technique (fig 2). The findings also suggest that the combination of Mulligan MWM and taping is more likely to produce a clinical benefit than the other 3 mobilization combinations, as more (17)

of the studies using MWM techniques found positive outcomes compared with other techniques (Maitland mobilization 12, manipulation 14). Further, studies reporting no positive outcomes with MWM techniques are fewer in number (6) compared with the other techniques (Maitland mobilization 14, manipulation 13).

Quality of studies

Because of differences in study design, 2 tools were used to assess the methodological quality of the included studies. PEDro was used for the assessment of randomized controlled trials (n=19), and the Quality Assessment Tool for Observational Cohort and Cross-Sectional Studies was used for all other study types (n=4). The level of agreement between reviewers for the quality assessment was considered to be high (κ =.63; 95% CI, .53–.73), and all disagreements were resolved by consensus.

Most studies scored well on random allocation, adequate follow-up, and for providing both point measures and measures of variability for at least 1 key outcome. In studies assessed using the PEDro Scale (fig 3), the most common risk of bias was for therapist and subject blinding. For the Quality Assessment Tool for Observational Cohort and Cross-Sectional Studies, all 4 studies demonstrated bias in terms of insufficient time frame, different



PEDro Scores per Criteria

🛚 Yes 📕 No

Fig 3 PEDro scores for assessment of quality of individual criteria.³⁰ Details about criteria: 1, Eligibility criteria were specified (Explanation: This criterion influences external validity, but not the internal or statistical validity of the trial. It has been included in the PEDro Scale so that all items of the Delphi Scale are represented on the PEDro Scale. This item is not used to calculate the PEDro score) (PEDro Scale); 2, Participants were randomly allocated to groups; 3, Allocation was concealed; 4, Groups were similar at baseline regarding most important prognostic indicators; 5, Blinding of all participants; 6, Blinding of therapists who administered the therapy; 7, Blinding of all assessors who measured at least 1 key outcome; 8, Measures of at least 1 key outcome were obtained from more than 85% of the participants; 9, All participants for whom outcome measures were available received the treatment or control condition as allocated; 10, Results of between-group statistical comparisons are reported for at least 1 key outcome; 11, Study provides both point measures and measures of variability for at least 1 key outcome.

levels of exposures as related to the outcome examined, and clearly defined valid and reliable exposure measures (fig 4). All studies scored at least moderate in terms of the overall quality of

the methodology for both the scales used (supplemental appendixes S3 and S4, available online only at http://www. archives-pmr.org/).



Quality Assessment Tool for Observational Cohort and Cross-Sectional Studies Score per Criteria

Fig 4 Quality Assessment Tool for Observational Cohort and Cross-Sectional Studies.³²

[■]Yes ■No

	Positive Findings									
	Imme	ediate	Short	-Term	Long-Term					
Outcome	Yes	No	Yes	No	Yes	No				
1. DFROM	11	3	4	4	1	0				
Weight-bearing DFROM	9	2	3	2	1	0				
Non-weight-bearing DFROM	2	1	0	1	0	0				
Unspecified	0	0	1	1	0	0				
2. Proprioception	2	0	1	0	0	0				
3. Stability/balance	3	7	3	3	1	0				
Static balance	1	3	1	3	0	0				
Dynamic balance	2	4	2	0	1	0				
4. Pain threshold	1	1	1	1	0	0				
5. Pain intensity	0	2	2	1	0	0				
6. Functional outcomes	2	4	4	2	0	0				
7. Talar stiffness	1	2	0	1	0	0				
8. Recovery from postural sway	1	0	0	0	0	0				
9. Patient's confidence toward stability	1	0	0	0	0	0				

NOTE. "Immediate," measured immediately after the intervention. "Short-term," measured up to 3 months after the intervention. "Long-term," measured at 3 or more months after the intervention.

Effects of mobilization on subacute/chronic ankle sprains

evaluation, indicating positive effects of mobilization at each of the 3 time points of interest.

The outcome measures of DFROM, proprioception, stability/balance, pain threshold, pain intensity and quality, function, talar stiffness, postural sway, and patient's confidence toward stability were assessed at varying time points across the studies after application of joint mobilization. Table 2 lists each outcome Eleven studies on chronic sprains reported quantitative data on 5 different outcomes, including weight-bearing DFROM, static balance, dynamic balance, pain intensity, and pain threshold. However, because of study heterogeneity and a lack of useable data for some outcomes, data could only be pooled for weightbearing DFROM, static balance, dynamic balance, and pain

Experimental Control Mean Difference Mean Difference	
Study or Subgroup Mean SD Total Mean SD Total Weight IV, Random, 95% CI IV, Random, 95% CI	
1.2.1 Postero-medial SEBT	
Cruz-diaz et al, 2015(14) 88.51 5.43 30 85.52 2.86 31 14.7% 2.99 [0.80, 5.18]	
Harkey et al, 2014(73) 83 11 15 78 9.7 15 4.2% 5.00 [-2.42, 12.42]	
Hoch & McKeon, 2011(74) 94.06 8.15 20 93.3 8.48 20 7.1% 0.76 [-4.39, 5.91]	
Someeh et al, 2015(81) 101.8 8.3 16 96.4 7.2 16 6.7% 5.40 [0.02, 10.78]	
Wells, 2012(69) 85.04 7.55 9 79.12 11 8 3.1% 5.92 [-3.16, 15.00]	
Subtotal (95% CI) 90 90 35.8% 3.22 [1.43, 5.01]	
Heterogeneity: Tau ² = 0.00; Chi ² = 2.11, df = 4 (P = 0.72); l ² = 0%	
Test for overall effect: Z = 3.52 (P = 0.0004)	
1.2.2 Anterior SEB1	
Cruz-diaz et al, 2015(14) 84.72 5.07 30 77.09 6.01 31 12.8% 7.63 [4.84, 10.42]	
Harkey et al, 2014(73) 72.5 7.5 15 64.6 6.8 15 7.1% 7.90 [2.78, 13.02]	
Hoch & McKeon, 2011(74) 79.44 4.73 20 78.91 5.51 20 11.6% 0.53 [-2.65, 3.71]	
Wells, 2012(69) 64.82 6.12 9 65.07 6.42 8 5.8% -0.25[-6.23, 5.73]	
Subtotal (95% Cl) 74 74 37.4% 4.10 [-0.35, 8.54]	
Heterogeneity: Tau ² = 15.83; Chi ² = 14.94, df = 3 (P = 0.002); i ² = 80%	
Test for overall effect: Z = 1.80 (P = 0.07)	
1.2.3 Postero-lateral SEBT	
Cruz-diaz et al. 2015(14) 89.28 3.04 30 87.11 3.25 31 16.5% 2.17 [0.59, 3.75]	
Harkey et al. 2014(73) 77.6 13.9 15 70.4 12.2 15 2.9% 7.20 (-2.16, 16, 56)	
Hoch & McKeon, 2011(74) 87.48 10.55 20 86.89 11.02 20 5.0% 0.59 [6.10, 7.28]	
Wells, 2012(69) 82.65 10.04 9 70.47 11.57 8 2.4% 12.18 [1.82, 22.54]	_
Subtotal (95% CI) 74 74 26.8% 3.55 [-0.18, 7.28]	
Heterogeneity: Tau ² = 5.81; Chi ² = 4.80, df = 3 (P = 0.19); i ² = 37%	
Test for overall effect: Z = 1.87 (P = 0.06)	
Total (95% Cl) 238 238 100.0% 3.73 [2.00, 5.46]	
Heteroneneity: Tau 2 = 4.06: Cbi 2 = 24.83: df = 12 (P = 0.02): P = 52%	—
Test for werall effect $7 = 4.23 (8 < 0.001)$ -20 -10 0 10 2	0
Test for subaround fifterences: Chile 0.14, df = 2 (P = 0.93) P = 0% Favours [control] Favours [contro	iental]

Fig 5 MD (95% CI) of the immediate effect of joint mobilization on dynamic balance by pooling data from 5 studies (n = 180). Abbreviations: df, degrees of freedom; IV, inverse variance.



Fig 6 SMD (95% CI) of the immediate effect of joint mobilization on static balance by pooling data from 3 studies (n = 100). Abbreviations: df, degrees of freedom; IV, inverse variance; Std., standardized.

intensity in order to evaluate immediate effects, and weightbearing DFROM was the only outcome measure available to assess the short-term effects of ankle mobilization.

Immediate effects of mobilization on ankle sprains

The immediate effects on DFROM were assessed in 14 outcome evaluations, of which 11 reported improvement with mobilization techniques (see table 2). The findings for other outcomes were less notable. Of the 10 studies that investigated the immediate effects of mobilization on stability/balance, $3^{14,39,50}$ had demonstrable improvement. Similarly, studies that assessed pain, talar stiffness, and function revealed inconsistent results. When considering the immediate effects of mobilization on functional outcomes, 2 outcome evaluations out of 6 demonstrated that it was effective.^{43,51} A summary of the reported immediate effects is provided in table 2.

Pooled data from 5 studies with a total of 180 participants were grouped for analysis of the effects of mobilization on each direction of the Star Excursion Balance Test (SEBT); anterior, posteromedial, and posterolateral. This analysis provided significant findings for the posteromedial direction of the SEBT (MD=3.22; 95% CI, 1.43–5.01; P=.0004); however, the posterolateral direction (MD=3.55; 95% CI, -.18 to 7.28; P=.06) and the anterior direction (MD=4.10; 95% CI, -.35 to 8.54; P=.07) results of the SEBT were not significant (fig 5). Pooled data for static balance from 3 studies with a total of 100 participants indicated there were no significant immediate benefits after mobilization of individuals with chronic sprains when compared with control participants (SMD=.01; 95% CI, -.38 to .40; P=.96) (fig 6).

Similarly, data from 7 studies with a total of 249 participants indicated there were no significant immediate effects of mobilization on the weight-bearing DFROM of individuals with chronic sprains (SMD=.66; 95% CI, -.25 to 1.58; P=.16) (fig 7). For pain intensity, pooled data from 2 studies with a total of 47 participants indicated mobilization had no immediate effect on individuals with chronic sprains (SMD=-.21; 95% CI, -.78 to .37;

P=.48) (fig 8). There were insufficient data to analyze the immediate benefits of mobilization on pain threshold.

Short-term effects of mobilization on ankle sprains

Half of the outcome evaluations reported that mobilization improved DFROM, stability/balance, and pain threshold in the short-term (see table 2). Demonstrable improvement was also observed in pain intensity and function (see table 2), and 2 studies^{41,45} that evaluated short-term outcomes on talar stiffness and proprioception reported improvements. No studies reported short-term findings on postural sway or patient's balance confidence.

Pooled data from 2 studies with 94 participants with chronic sprains indicated joint mobilization was effective in the short-term for improving weight-bearing DFROM (MD = 2.56; 95% CI, .89–4.23; P=.003) (fig 9). There were insufficient data evaluating static balance, dynamic balance, pain threshold, and pain intensity to permit analysis of the short-term benefits of mobilization on these outcomes.

Long-term effects of mobilization on ankle sprains

Only 1 study¹⁴ evaluated the long-term effects of mobilization on ankle sprains. Long-term improvement in DFROM and stability/ balance were reported in the single included study.

Quality of evidence

According to the Grading of Recommendations, Assessment, Development, and Evaluation (supplemental appendix S5, available online only at http://www.archives-pmr.org/), the evidence for DFROM (immediate and short-term), static balance, and dynamic balance can be considered to be of moderate quality. The evidence for pain was considered to be of low quality because of lack of generalizability of 1 of the included studies. Overall, the evidence included in this meta-analysis was considered to be of



Fig 7 SMD (95% CI) of the immediate effect of joint mobilization on weight-bearing DFROM by pooling data from 7 studies (n=249). Abbreviations: df, degrees of freedom; IV, inverse variance; Std., standardized.



Fig 8 SMD (95% CI) of the immediate effect of joint mobilization on pain intensity by pooling data from 2 studies (n=47). Abbreviations: df, degrees of freedom; IV, inverse variance; Std., standardized.

moderate quality, with the risk of bias and the level of heterogeneity the main factors influencing the quality of the evidence.

Discussion

This is the first systematic review to assess the clinical benefits of joint mobilization in the management of either lateral or medial ankle ligament sprains at all stages of recovery. Importantly, this is the first review to only include studies in which joint mobilization is the sole intervention. The current review did not identify any studies evaluating the clinical benefits of joint mobilization on acute ankle sprains, perhaps because mobilization is not typically the preferred choice of management in the acute stage of ankle sprains.⁸⁷ Findings about the clinical benefits of mobilization on most outcome measures were inconsistent across studies, and a lack of reported quantitative data, heterogeneity of subjects, and the differing types of joint mobilization applied made direct comparisons difficult. Despite this, meta-analysis indicated there are immediate benefits of mobilization for improving dynamic balance, and a short-term benefit in improving weight-bearing DFROM in chronic ankle sprains. These results provide compelling evidence that joint mobilization may be effective in improving balance immediately and in increasing DFROM in the short-term in chronic ankle sprains.

Dynamic balance and weight-bearing DFROM improvements after joint mobilization were both associated with clinically meaningful changes. The modified SEBT assesses performance during single-leg balance with reaching in 3 directions (anterior, posteromedial, posterolateral).^{88,89} The MCID for this test is reported as being 3.5%, and therefore the immediate effect on dynamic balance found in the meta-analysis (MD=3.73) can be considered as clinically meaningful.^{88,89} It is plausible that the immediate improvements in dynamic balance after joint mobilization may increase the individual's balance confidence and perhaps reduce the risk of reinjury. Clinically, this may assist the individual with an ankle sprain to more safely proceed to the next level of functional exercise in the rehabilitation process.

There were no immediate improvements in either anterior SEBT performance or DFROM. Interestingly, previous research⁹⁰ supports the existence of a correlation between anterior SEBT

performance and the weight-bearing lunge test. This correlation could help explain the current review's findings on immediate anterior SEBT performance and DFROM. Notably, the MCID for ankle DFROM has not been established.⁹¹ However, approximately 3.6° of DFROM is associated with 1cm in distance from the wall in the lunge test.³⁹ The MD in the short-term measurement of weight-bearing DFROM from the current meta-analysis was 2.56cm, and this equates to 9.2° of dorsiflexion, which can be considered as clinically meaningful given that the normal total range is only 15° to 20°.^{92,93}

Joint mobilization techniques are aimed at restoring the normal joint ROM,^{94,95} and indeed this review found DFROM improved after mobilization. However, the mechanisms by which restoring ankle ROM may assist other impairments are unclear, as are the underlying mechanisms by which mobilization may actually work.^{15,16} It has been proposed that increased ankle ROM is due to the correction of a bony positional fault.¹⁰ It is further postulated that the correct alignment of the articular surfaces may help to restore normal biomechanics as well as sensorimotor function.¹⁰ However, it may be that mobilization produces less impact on pain, as evidenced by the lack of improvement in ankle pain outcome measures in this review. Potential underlying central nervous system changes related to persistent pain in chronic sprains remain unclear, but central sensitization may be a possible factor for persistence of chronic pain. If central sensitization is actually a key factor contributing to chronic ankle sprain pain, then changing the bony alignment would be unlikely to improve pain in chronic sprains since it is not the usual localized pressurepain hypersensitivity⁹⁶ experienced immediately after a sprain.

According to the clinical practice guidelines linked to the *International Classification of Functioning, Disability and Health* from the American Physical Therapy Association, clinicians should use joint mobilization to improve ankle dorsiflexion, proprioception, and weight-bearing tolerance in patients recovering from a lateral sprain.⁸ Of these 3 outcomes, the findings of the current review only support the benefit of mobilization for dorsiflexion. There was insufficient research available to conclude whether mobilization is effective for improving proprioception or weight-bearing tolerance. However, the current review found clinically meaningful evidence



Fig 9 SMD (95% CI) of the short-term effect of joint mobilization on weight-bearing DFROM by pooling data from 2 studies (n=94). Abbreviations: df, degrees of freedom; IV, inverse variance; Std., standardized.

for the effect of mobilization on dynamic balance, an outcome not mentioned in the clinical practice guidelines from the American Physical Therapy Association. One explanation for this difference may be that the guidelines only included literature published before April 2012, while the current review has included 7 more recently published studies.

The inclusion and exclusion criteria of the current review differ in important ways from previous systematic reviews on this topic. In contrast to those prior reviews, our search criteria included both lateral and medial ligament sprains, covered all stages of recovery from acute to chronic, and encompassed all clinically relevant outcomes used to assess the effects of mobilization. Importantly, of the 6 prior reviews^{22,24-27,97} that have evaluated the efficacy of mobilization techniques on ankle sprains, all included studies that did not evaluate joint mobilization as a unique intervention, but rather as an adjunct to other interventions (such as home exercise programs, RICE protocol, and external supports) included in their review. The current review excluded these multimodal studies to ensure the homogeneity of the included studies and to increase the precision of the results in relation to the effects of joint mobilization. Compared with the recent review by Loudon et al,²⁴ the present review included almost 3 times more studies (23), with all of these only investigating the clinical effects of joint mobilization techniques in isolation. In the review by Loudon,²⁴ only 8 studies were included, and of those, mobilization was used as the sole intervention in only 5.24 This disparity in the number of included studies may be due to our searching a greater number of databases (11), including medial ankle sprains in the search criteria, by reviewing dissertations and theses, and by not limiting clinical outcomes.

This review includes the first meta-analysis undertaken to assess the clinical benefits of joint mobilization for ankle sprains. When comparing the findings of the current review to previous systematic reviews, there were some agreements and some inconsistent results. When considering the immediate effects of mobilization, the review by van der Wees et al26 reported an improvement in DFROM.²⁶ However, the current review did not support an immediate effect on weight-bearing DFROM, with mobilization providing only a short-term effect. Pain and function are concluded to improve immediately in the review by Southerst et al,²⁷ but in our review immediate pain relief was not evident, and inconclusive results were found for immediate function. When considering the short-term effects, the effectiveness of mobilization in increasing ankle ROM was supported in the review of Bleakly et al,²² and this was consistent with the findings of the current review.²² The review by van Ochten et al²⁸ reported positive changes in short-term pain and function in chronic sprains; however, the findings of the present review were inconclusive for both of these outcomes. When considering the long-term effects of mobilization, pain and function are improved according to the review by Southerst.²⁷ The findings of the current review on these outcomes were inconclusive because of a lack of data. Different definitions of inclusion criteria for mobilization techniques included within reviews (eg, including other therapies such as home exercise or RICE treatment along with mobilization), as well as differences in the databases searched and the periods of the data searches, are all factors contributing to these differing findings.

Study limitations

Limitations of this review include the wide variation in follow-up time points that we defined as short-term (from 1d to <3mo).

Additionally, the included studies have used a range of different mobilization techniques and comparators. It was beyond the scope of this review to attempt to determine the independent merits of individual techniques. In particular, there may be value in analyzing joint mobilization and high-velocity thrust manipulation techniques separately rather than together, but given the lack of available research at this time directly comparing these 2 manual therapy approaches, this level of scrutiny is not possible. In addition, it was not possible to pool data to analyze the effectiveness of mobilization for some important outcomes that were reported in single studies. Despite attempts to contact the authors of included studies, data were insufficient to analyze the immediate effects on the pressure-pain threshold, and the short-term effects on the pressure-pain threshold and pain intensity. Finally, no high-quality evidence was found to provide robust evidence for the effectiveness of joint mobilization for ankle sprains.

Further research is required to determine the mechanisms by which mobilization improves dynamic balance and weightbearing DFROM. Also, the long-term effects of mobilization on ankle sprains should be further investigated using clinically relevant outcomes.

Conclusions

Joint mobilization appears to clinically benefit individuals with chronic ankle sprains, improving dynamic balance immediately and weight-bearing DFROM in the short-term. It is unlikely to have an immediate effect on static balance, pain intensity, and weight-bearing DFROM. Other clinical outcomes that have been reported after mobilization demonstrate an inconsistent response to mobilization, and this may be a reflection of previous study designs or of the intervention itself.

Suppliers

- a. EndNote X7.3.1; Clarivate Analytics.
- RevMan (Review Manager) 5.3; The Nordic Cochrane Centre, The Cochrane Collaboration, 2014. Available at: http://ims. cochrane.org/revman.

Keywords

Ankle injuries; Ankle joint; Joint instability; Musculoskeletal manipulations; Physical therapy modalities; Rehabilitation

Corresponding author

Ishanka Weerasekara, MPhil, School of Health Sciences, Faculty of Health and Medicine, HA 06, Hunter Bldg, The University of Newcastle, University Drive, Callaghan, NSW 2308, Australia. *E-mail address:* ishanka.weerasekara@uon.edu.au.

References

 Woolf AD, Pfleger B. Burden of major musculoskeletal conditions. Bull World Health Organ 2003;81:646-56.

- Parker R, Jelsma J. The prevalence and functional impact of musculoskeletal conditions amongst clients of a primary health care facility in an under-resourced area of Cape Town. BMC Musculoskelet Disord 2010;11:2.
- Bridgman SA, Clement D, Downing A, Walley G, Phair I, Maffulli N. Population based epidemiology of ankle sprains attending accident and emergency units in the West Midlands of England, and a survey of UK practice for severe ankle sprains. Emerg Med J 2003;20:508-10.
- Holmer P, Sondergaard L, Konradsen L, Nielsen PT, Jorgensen LN. Epidemiology of sprains in the lateral ankle and foot. Foot Ankle Int 1994;15:72-4.
- 5. Waterman BR, Owens BD, Davey S, Zacchilli MA, Belmont PJ Jr. The epidemiology of ankle sprains in the United States. J Bone Joint Surg Am 2010;92:2279-84.
- **6.** Doherty C, Delahunt E, Caulfield B, Hertel J, Ryan J, Bleakley C. The incidence and prevalence of ankle sprain injury: a systematic review and meta-analysis of prospective epidemiological studies. Sports Med 2014;44:123-40.
- 7. Hing W, Hall T, Rivett DA, Vicenzino B, Mulligan B. The Mulligan concept of manual therapy. Chatswood: Elsevier; 2015.
- 8. Martin RL, Davenport TE, Paulseth S, Wukich DK, Godges JJ. Ankle stability and movement coordination impairments: ankle ligament sprains. J Orthop Sports Phys Ther 2013;43:A1-40.
- McCarthy C, Bialosky J, Rivett D. Spinal manipulation. In: Jull G, Moore A, Falla D, Lewis J, McCarthy C, Sterling M, editors. Grieve's modern musculoskeletal physiotherapy. 4th ed. Chatswood: Elsevier; 2015. p 278-82.
- 10. Vicenzino B, Hing W, Rivett DA, Hall T. Mobilisation with movement: the art and the science. Chatswood: Elsevier; 2011.
- 11. O'Brien T, Vicenzino B. A study of the effects of Mulligan's mobilization with movement treatment of lateral ankle pain using a case study design. Man Ther 1998;3:78-84.
- Green T, Refshauge K, Crosbie J, Adams R. A randomized controlled trial of a passive accessory joint mobilization on acute ankle inversion sprains. Phys Ther 2001;81:984-94.
- Collins N, Teys P, Vicenzino B. The initial effects of a Mulligan's mobilization with movement technique on dorsiflexion and pain in subacute ankle sprains. Man Ther 2004;9:77-82.
- Cruz-Diaz D, Lomas Vega R, Osuna-Perez MC, Hita-Contreras F, Martinez-Amat A. Effects of joint mobilization on chronic ankle instability: a randomized controlled trial. Disabil Rehabil 2015;37: 601-10.
- **15.** Marron-Gomez D, Rodriguez-Fernandez A, Martin-Urrialde J. The effect of two mobilization techniques on dorsiflexion in people with chronic ankle instability. Phys Ther Sport 2015;16:10-5.
- 16. Vicenzino B, Branjerdporn M, Teys P, Jordan K. Initial changes in posterior talar glide and dorsiflexion of the ankle after mobilization with movement in individuals with recurrent ankle sprain. J Orthop Sports Phys Ther 2006;36:464-71.
- Penso M. The effectiveness of mobilisation with movement for chronic medial ankle pain: a case study. S Afr J Physiother 2008;64: 13-6.
- Cosby NL, Koroch M, Grindstaff TL, Parente W, Hertel J. Immediate effects of anterior to posterior talocrural joint mobilizations following acute lateral ankle sprain. J Man Manip Ther 2011;19:76-83.
- Delahunt E, Cusack K, Wilson L, Doherty C. Joint mobilization acutely improves landing kinematics in chronic ankle instability. Med Sci Sports Exerc 2013;45:514-9.
- Woodman R, Berghorn K, Underhill T, Wolanin M. Utilization of mobilization with movement for an apparent sprain of the posterior talofibular ligament: a case report. Man Ther 2013;18:e1-7.
- 21. Gilbreath JP, Gaven SL, Van Lunen L, Hoch MC. The effects of mobilization with movement on dorsiflexion range of motion, dynamic balance, and self-reported function in individuals with chronic ankle instability. Man Ther 2014;19:152-7.
- 22. Bleakley CM, McDonough SM, MacAuley DC. Some conservative strategies are effective when added to controlled mobilisation with

external support after acute ankle sprain: a systematic review. Aust J Physiother 2008;54:7-20.

- Hoch MC, McKeon PO. The effectiveness of mobilization with movement at improving dorsiflexion after ankle sprain. J Sport Rehabil 2010;19:226-32.
- Loudon JK, Reiman MP, Sylvain J. The efficacy of manual joint mobilisation/manipulation in treatment of lateral ankle sprains: a systematic review. Br J Sports Med 2014;48:365-70.
- 25. Terada M, Pietrosimone BG, Gribble PA. Therapeutic interventions for increasing ankle dorsiflexion after ankle sprain: a systematic review. J Athl Train 2013;48:696-709.
- **26.** van der Wees PJ, Lenssen AF, Hendriks EJ, Stomp DJ, Dekker J, de Bie RA. Effectiveness of exercise therapy and manual mobilisation in ankle sprain and functional instability: a systematic review. Aust J Physiother 2006;52:27-37.
- 27. Southerst D, Yu H, Randhawa K, et al. The effectiveness of manual therapy for the management of musculoskeletal disorders of the upper and lower extremities: a systematic review by the Ontario Protocol for Traffic Injury Management (OPTIMa) Collaboration. Chiropr Man Ther 2015;23:30.
- Van Ochten JM, Van Middelkoop M, Meuffels D, Bierma-Zeinstra SM. Chronic complaints after ankle sprains: a systematic review on effectiveness of treatments. J Orthop Sports Phys Ther 2014;44:862-71.
- 29. Viera AJ, Garrett JM. Understanding inter observer agreement: the kappa statistic. Fam Med 2005;37:360-3.
- PEDro scale. University of Sydney. Available at: http://www.pedro. org.au/english/downloads/pedro-scale/. Accessed March 1, 2016.
- 31. Higgins JPT, Green S, editors. Cochrane Handbook for Systematic Reviews of Interventions Version 5.1.0 [updated March 2011]. The Cochrane Collaboration, 2011. Available from www.handbook. cochrane.org. Accessed October 3, 2017.
- 32. National Institutes of Health. Quality Assessment Tool for Observational Cohort and Cross-Sectional Studies. Available at https://www.nhlbi.nih.gov/health-pro/guidelines/in-develop/cardiovascular-risk-reduction/tools/cohort. Accessed March 1, 2016.
- Atkins D, Best D, Briss PA, et al. Grading quality of evidence and strength of recommendations. BMJ 2004;328:1490.
- 34. Guyatt GH, Oxman AD, Vist GE, et al. GRADE: an emerging consensus on rating quality of evidence and strength of recommendations. BMJ 2008;336:924.
- 35. Armijo-Olivo S, Pitance L, Singh V, Neto F, Thie N, Michelotti A. Effectiveness of manual therapy and therapeutic exercise for temporomandibular disorders: systematic review and meta-analysis. Phys Ther 2016;96:9-25.
- 36. Alanson N. The immediate effects of manual therapy on dorsiflexion and joint position sense at the talocrural joint in participants with a history of lateral ankle sprain [Thesis]. Mount Albert: Unitec Institute of Technology; 2012.
- 37. Beazell JR, Grindstaff TL, Sauer LD, Magrum EM, Ingersoll CD, Hertel J. Effects of a proximal or distal tibiofibular joint manipulation on ankle range of motion and functional outcomes in individuals with chronic ankle instability. J Orthop Sports Phys Ther 2012;42:125-34.
- 38. Harkey M, McLeod M, Van SA, et al. The immediate effects of an anterior-to-posterior talar mobilization on neural excitability, dorsiflexion range of motion, and dynamic balance in patients with chronic ankle instability. J Sport Rehabil 2014;23:351-9.
- **39.** Hoch MC, McKeon PO. Joint mobilization improves spatiotemporal postural control and range of motion in those with chronic ankle instability. J Orthop Res 2011;29:326-32.
- 40. Hoch MC, Andreatta RD, Mullineaux DR, et al. Two-week joint mobilization intervention improves self-reported function, range of motion, and dynamic balance in those with chronic ankle instability. J Orthop Res 2012;30:1798-804.
- 41. Hoch MC, Mullineaux DR, Andreatta RD, et al. Effect of a 2-week joint mobilization intervention on single-limb balance and ankle

arthrokinematics in those with chronic ankle instability. J Sport Rehabil 2014;23:18-26.

- Hopper D, Samsson K, Hulenik T, Ng C, Hall T, Robinson K. The influence of Mulligan ankle taping during balance performance in subjects with unilateral chronic ankle instability. Phys Ther Sport 2009;10:125-30.
- Houston MN, McKeon PO, Hoch MC. Foot and ankle ability measure scores in patients with chronic ankle instability following joint mobilization. Int J Athl Ther Train 2013;18:4-7.
- 44. Joseph LC, de Busser N, Brantingham JW, et al. The comparative effect of muscle energy technique vs. manipulation for the treatment of chronic recurrent ankle sprain. J Am Chiropr Assoc 2010; 47:8-22.
- 45. Kohne E, Andrew J, Charmaine K, Jennifer PL, James WB, Gary G. A prospective, single-blinded, randomized, controlled clinical trial of the effects of manipulation on proprioception and ankle dorsiflexion in chronic recurrent ankle sprain. J Am Chiropr Assoc 2007;44:7-17.
- 46. Lopez-Rodriguez S, de-Las-Penas CF, Burquerque-Sendin F, Rodriguez-Blanco C, Palomeque-del-Cerro L. Immediate effects of manipulation of the talocrural joint on stabilometry and baropodometry in patients with ankle sprain. J Manipulative Physiol Ther 2007;30:186-92.
- 47. Pellow JE, Brantingham JW. The efficacy of adjusting the ankle in the treatment of subacute and chronic grade I and grade II ankle inversion sprains. J Manipulative Physiol Ther 2001;24:17-24.
- 48. Plante PD. The effects of talar mobilization in individuals without previous history of ankle sprains and those with chronic ankle instability [thesis]. Ann Arbor: Wilfrid Laurier University; 2012.
- **49.** Reid A, Birmingham TB, Alcock G. Efficacy of mobilization with movement for patients with limited dorsiflexion after ankle sprain: a crossover trial. Physiother Can 2007;59:166-72.
- 50. Someeh M, Norasteh AA, Daneshmandi H, Asadi A. Immediate effects of Mulligan's fibular repositioning taping on postural control in athletes with and without chronic ankle instability. Phys Ther Sport 2015;16:135-9.
- 51. Someeh M, Norasteh AA, Daneshmandi H, Asadi A. Influence of Mulligan ankle taping on functional performance tests in healthy athletes and athletes with chronic ankle instability. Int J Athl Ther Train 2015;20:39-45.
- 52. Wells AM. Effects of joint mobilization on ankle dorsiflexion range of motion, dynamic postural control and self-reported patient outcomes in individuals with chronic ankle instability [thesis]. Ohio: University of Toledo; 2012.
- Yeo HK, Wright A. Hypoalgesic effect of a passive accessory mobilisation technique in patients with lateral ankle pain. Man Ther 2011;16:373-7.
- Cohen J. Statistical power analysis for the behavioral sciences. 2nd ed. Hillsdale, New York: Lawrence Earlbaum Associates; 1988.
- 55. Valentine JC, Cooper H. Effect size substantive interpretation guidelines: issues in the interpretation of effect sizes. Washington (DC): What Works Clearinghouse; 2013. p 1-7.
- 56. Ambarish AA, Chitra J, Subhash KM. Comparative effectiveness of Mulligan's mobilization in weight bearing and non-weight bearing in the treatment of ankle sprains—a randomized clinical trial. Indian J Physiother Occup Ther 2008;2:1-4.
- 57. Baker RT Jr. An examination of the effects of mechanical manual therapies on clinician and patient-oriented outcomes measures: a dissertation of clinical practice improvement [Dissertation]. Ann Arbor: University of Idaho; 2013.
- Cosby NL. Effects of anterior to posterior joint mobilizations in individuals with subacute lateral ankle sprains. Diss Abstr Int 2012; 73(2-B):925.
- 59. Eisenhart AW, Gaeta TJ, Yens DP. Osteopathic manipulative treatment in the emergency department for patients with acute ankle injuries. J Am Osteopath Assoc 2003;103:417-21.
- 60. Estrade JL. Therapie manuelle et entorse de cheville. Pratique basee sur les preuves [French]. Manual physical therapy and ankle sprain. Evidence-based practice Kinesitherapie 2013;13:12-5.

- 61. Fisher BE, Davenport TE, Kulig K, Wu AD. Identification of potential neuromotor mechanisms of manual therapy in patients with musculoskeletal disablement: rationale and description of a clinical trial. BMC Neurol 2009;9:20.
- 62. Hedlund S, Nilsson H, Lenz M, Sundberg T. Effect of chiropractic manipulation on vertical jump height in young female athletes with talocrural joint dysfunction: a single-blind randomized clinical pilot trial. J Manipulative Physiol Ther 2014;37:116-23.
- **63.** Kumari N, Megha NA, Paresh P. Efficacy of weight bearing distal tibiofibular joint mobilization with movement (MWM) in improving pain, dorsiflexion range and function in patients with post-acute lateral ankle sprain. Int J Physiother Res 2014;2:542-8.
- 64. Landrum EL, Kelln BM, Parente WR, Ingersoll CD, Hertel J. Immediate effects of anterior-to-posterior talocrural joint mobilization after prolonged ankle immobilization: a preliminary study. J Man Manip Ther 2008;16:100-5.
- Lee KY, Lee HJ, Kim SE, Choi PB, Song SH, Jee YS. Short term rehabilitation and ankle instability. Int J Sports Med 2012;33:485-96.
- 66. Lubbe D, Lakhani E, Brantingham J, et al. Manipulative therapy and rehabilitation for recurrent ankle sprain with functional instability: a short-term, assessor-blind, parallel-group randomized trial. J Manipulative Physiol Ther 2015;38:22-34.
- **67.** Nambi G, Shah BT. Kinesio taping versus Mulligan's mobilization with movement in sub-acute lateral ankle sprain in secondary school hockey players—comparative study. Int J Pharm Sci Health Care 2012;2:136-49.
- **68.** Nilsson S. Sprains of the lateral ankle ligaments. An epidemiological and clinical study with reference to different forms of conservative treatment. J Oslo City Hosp 1982;32:13-36.
- **69.** Rashid S, Salick K, Kashif M, Ahmad A, Sarwar K. To evaluate the efficacy of mobilization techniques in post-traumatic stiff ankle with and without paraffin wax bath. Pak J Med Sci 2013;29:1406-9.
- 70. Teixeira LM, Pires T, Silva RD, de Resende MA. Immediate effect of a single anteroposterior talus mobilization on dorsiflexion range of motion in participants with orthopedic dysfunction of the ankle and foot. J Manipulative Physiol Ther 2013;36:369-75.
- 71. Silva RD, Teixeira LM, Moreira TS, Teixeira-Salmela LF, de Resende MA. Effects of anteroposterior talus mobilization on range of motion, pain, and functional capacity in participants with subacute and chronic ankle injuries: a controlled trial. J Manipulative Physiol Ther 2017;40:273-83.
- Andersen K, Albers C, Moller-Madsen B, Niedermann B, Simonsen O, Snorrason K. Conservative treatment of severe ankle sprains. Acta Orthop Scand 1987;58:697-8.
- Ayad K, Abdelaal A, Elazizi H, Abdelgalil A. Efficacy of mobilization with movement in treatment of chronic ankle sprain. Physiotherapy 2015;101:e96-7.
- Clinch P. The effect of grade II mobilization on range of movement in acutely sprained ankles [abstract]. Aust J Physiother 1986;32:205.
- Fay F, Egerod S. The effects of joint mobilizations on swelling in the acutely sprained ankle joint: a pilot study. Aust J Physiother 1984;30: 168.
- Hart LE, Macintyre J. Passive joint mobilization for acute ankle inversion sprains. Clin J Sport Med 2002;12:54.
- Rundle E. The effect of grade IV mobilisation on the range of dorsiflexion in the weight-bearing position in non-acute sprained ankles [abstract]. Aust J Physiother 1988:56.
- 78. Vicenzino B, Jordan K, Brandjerporn M, Teys P, Bisset L. Mobilization with movement treatment of the ankle changes joint position sense in subjects with recurrent sprains: a preliminary report [abstract]. J Orthop Sports Phys Ther 2005;35:A-21.
- 79. Andersen S, Fryer GA, McLaughlin P. The effect of talo-crural joint manipulation on range of motion at the ankle joint in subjects with a history of ankle injury. Australas Chiropr Osteopathy 2003;11:57-62.
- 80. Chen B, Wang X, Gao NY, et al. [Clinical observation on Tuina manipulation for ankle sprain in Wushu athletes] [Chinese]. Shanghai Zhong Yi Yao Za Zhi [Shanghai Journal of Traditional Chinese Medicine] 2012;46:56-8.

- 81. Yu XH. [Clinical observation on acupuncture and Chinese naprapathy for the treatment of ankle sprain] [Chinese]. Zhongguo Zhongyi Gushangke Zazhi [Chinese Journal of Traditional Medicine Traumatology & Orthopedics] 2007;12:14-5.
- Diebschlag W, Nocker W. Treatment of acute sprains of the ankle joint: a placebo controlled double blind study. Munch Med Wschr 1987;129:803-6.
- Lee DY, Kim SJ. [Clinical study of traction therapy for ankle sprain]. [Korean]. J Orient Rehabil Med 2005;15:39-46.
- Davenport TE, Kulig K, Fisher BE. Ankle manual therapy for individuals with post-acute ankle sprains: description of a randomized, placebo-controlled clinical trial. BMC Complement Altern Med 2010;10:59.
- **85.** Hoch MC. The effect of joint mobilization on functional outcomes associated with chronic ankle instability [dissertation]. Kentucky: University of Kentucky UKnowledge; 2011.
- Maitland GD. Maitland's vertebral manipulation. 7th ed. Philadelphia: Elsevier; 2005.
- 87. van den Bekerom MP, Kerkhoffs GM, McCollum GA, Calder JD, van Dijk CN. Management of acute lateral ankle ligament injury in the athlete. Knee Surg Sports Traumatol Arthrosc 2013;21: 1390-5.
- Rehabilitation Institute of Chicago, Center for Rehabilitation Outcomes Research, Northwestern University Feinberg School of Medicine Department of Medical Social Sciences Informatics Group. Rehabilitation measures database. Rehab measures: Lower Quarter Y-Balance Test; 2010. Available at: http://www.rehabmeasures.org/Lists/RehabMeasures/PrintView.aspx?ID=1271. Accessed July 14, 2016.

- Chimera NJ, Smith CA, Warren M. Injury history, sex, and performance on the functional movement screen and Y Balance Test. J Athl Train 2015;50:475-85.
- Hoch MC, Staton GS, Medina McKeon JM, Mattacola CG, McKeon PO. Dorsiflexion and dynamic postural control deficits are present in those with chronic ankle instability. J Sci Med Sport 2012; 15:574-9.
- **91.** Young R, Nix S, Wholohan A, Bradhurst R, Reed L. Interventions for increasing ankle joint dorsiflexion: a systematic review and metaanalysis. J Foot Ankle Res 2013;6:46.
- 92. Roaas A, Andersson GB. Normal range of motion of the hip, knee and ankle joints in male subjects, 30-40 years of age. Acta Orthop Scand 1982;53:205-8.
- **93.** Hallaceli H, Uruc V, Uysal HH, et al. Normal hip, knee and ankle range of motion in the Turkish population. Acta Orthop Traumatol Turc 2014;48:37-42.
- **94.** Hertling D, Kessler RM. Management of common musculoskeletal disorders; physical therapy principles and methods. 4th ed. Philadelphia: Lippincott Williams & Wilkins; 2006.
- Oatis CA. Kinesiology; the mechanics & pathomechanics of human movement. 2nd ed. Baltimore: Lippincott Williams & Wilkins; 2004.
- 96. Ramiro-Gonzalez MD, Cano-de-la-Cuerda R, De-la-Llave-Rincon AI, Miangolarra-Page JC, Zarzoso-Sanchez R, Fernandez-de-Las-Penas C. Deep tissue hypersensitivity to pressure pain in individuals with unilateral acute inversion ankle sprain. Pain Med 2012;13:361-7.
- **97.** Wikstrom EA, McKeon PO. Manipulative therapy effectiveness following acute lateral ankle sprains: a systematic review. Athl Train Sports Health Care 2011;3:271-9.

Supplemental Appendix S1 Search strategy

_	
#	Searches
1	Ankle Injuries/
2	ankle sprain.mp.
3	(ankle* adj5 injur*).tw.
4	(ankle* adj5 sprain*).tw.
5	(ankle* adj5 twist*).tw.
6	(injur* adj5 ligament*).tw.
7	lateral ligament*.mp. or Collateral Ligaments/
8	Ankle Joint/or medial ligament*.mp.
9	Ankle Joint/or deltoid ligament*.mp.
10	ATFL.mp.
11	PTFL.mp.
12	"Sprains and Strains"/
13	talo crural.tw.
14	1 or 2 or 3 or 4 or 5 or 6 or 7 or 8 or 9 or
	10 or 11 or 12 or 13
15	Chiropractic/or Manipulation, Orthopedic/
16	musculoskeletal manipulation.mp. or Musculoskeletal
	Manipulations/
17	(joint* adj5 manipul*).tw.
18	(ankle* adj5 rehab*).tw.
19	Mulligan*.mp.
20	Maitland*.mp.
21	MWM*.mp.
22	manual therap*.mp.
23	manual technique*.mp.
24	(joint* adj5 mobili?ation*).tw.
25	15 or 16 or 17 or 18 or 19 or 20 or 21 or 22 or 23 or 24
26	Randomized controlled trial.pt.
27	clinical trial.pt.
28	random*.tw.
29	trial*.tw.
30	group*.tw.
31	case series.tw.
32	cross-over studies/
33	Cross-Sectional Studies/
34	exp Cohort Studies/
35	26 or 27 or 28 or 29 or 30 or 31 or 32 or 33 or 34
36	14 and 25 and 35
37	limit 36 to humans

Supplemental Appendix S2 Funnel Plots

Immediate effect of mobilization on weight-bearing dorsiflexion, pain, static balance, and dynamic balance



Short-term effect of mobilization on weight-bearing dorsiflexion



Abbreviations: MD, mean difference; PL, posterolateral; PM, posteromedial; SEBT, Star Excursion Balance Test; SMD, standardized mean difference.

Supplemental Appendix S3 PEDro scores for assessment of quality of individual intervention studies														
	PEDro Scale													
Study	1) Eligibility Criteria	2) Random Allocation	3) Concealed Allocation	4) Baseline Comparability	5) Blinding Subjects	6) Blinding Therapists	7) Blinding Assessors	8) Adequate Follow-up (>85% of Subjects)	9) Intention- to-Treat Analysis	10) Between- Group Comparisons	11) Point Measures and Variability	Total Score Out of 10		
Alanson, ³⁶ 2012	+	+	+	_	_	_	+	+	+	+	+	7		
Beazell et al, ³⁷ 2012	+	+	_	+	_	_	+	+	+	+	+	7		
Collins et al, ¹³ 2004	+	+	_	+	+	_	+	+	_	+	+	7		
Cruz-Diaz et al, ¹⁴ 2015	+	+	+	+	-	_	+	+	-	+	+	7		
Harkey et al, ³⁸ 2014	+	+	+	+	_	_	+	+	+	+	+	8		
Hoch and McKeon, ³⁹ 2011	+	+	+	+	-	_	+	+	+	+	+	8		
Hopper et al, ⁴² 2009	+	+	_	+	_	_	_	+	+	+	+	6		
Joseph et al, ⁴⁴ 2010	+	+	+	+	_	_	_	+	+	+	+	7		
Kohne et al, ⁴⁵ 2007	+	+	+	_	_	_	_	+	+	+	+	6		
Lopez-Rodriguez et al, ⁴⁶ 2007	+	_	_	+	-	_	_	+	+	+	+	5		
Marron-Gomez et al, ¹⁵ 2015	+	+	_	+	+	_	+	+	+	+	+	8		
Pellow and Brantingham, ⁴⁷ 2001	+	+	-	+	-	-	-	+	-	+	+	5		
Plante, ⁴⁸ 2012	+	+	_	+	_	_	_	+	+	+	+	6		
Reid et al, ⁴⁹ 2007	+	+	_	+	_	_	+	+	-	+	+	6		
Someeh et al, ⁵⁰ 2015	+	+	_	+	_	_	_	+	+	+	+	6		
Someeh et al, ⁵¹ 2015	+	+	_	+	_	_	_	+	+	+	+	6		
Vicenzino et al, ¹⁶ 2006	+	+	_	+	+	_	+	+	+	+	+	8		
Wells, ⁵² 2012	+	+	+	+	_	-	+	+	+	+	+	8		
Yeo and Wright, ⁵³ 2011	+	+	-	+	-	-	+	+	+	+	+	7		

NOTE. 1, Eligibility criteria were specified (Explanation: This criterion influences external validity, but not the internal or statistical validity of the trial. It has been included in the PEDro Scale so that all items of the Delphi Scale are represented on the PEDro Scale. This item is not used to calculate the PEDro score) (PEDro Scale); 2, Participants were randomly allocated to groups; 3, Allocation was concealed; 4, Groups were similar at baseline regarding most important prognostic indicators; 5, Blinding of all participants; 6, Blinding of therapists who administered the therapy; 7, Blinding of all assessors who measured at least 1 key outcome; 8, Measures of at least 1 key outcome were obtained from more than 85% of the participants; 9, All participants for whom outcome measures were available received the treatment or control condition as allocated; 10, Results of between-group statistical comparisons are reported for at least 1 key outcome; 11, Study provides both point measures and measures of variability for at least 1 key outcome.¹

Abbreviations: +, meet criteria; -, do not meet criteria.

Reference

 PEDro Scale. 1999. Available at: http://www.pedro.org.au/english/ downloads/pedro-scale/. Accessed May 12, 2015.

Supplemental Appendix S4 Quality assessment tool for observational cohort and cross-sectional studies scores for assessment of quality of individual cohort studies

		Quality Assessment Tool for Observational Cohort and Cross-Sectional Studies Scale													
Study	1	2	3	4	5	6	7	8	9	10	11	12	13	14	Score Out of 14
Gilbreath et al, ²¹ 2014	+	+	+	+	+	_	_	_	+	+	+	_	+	+	10
Hoch et al, ⁴⁰ 2012	+	+	+	+	+	_	_	-	+	+	+	+	+	+	11
Hoch et al, ⁴¹ 2014	+	+	+	+	+	_	_	_	+	+	+	_	+	+	10
Houston et al, ⁴³ 2013	+	+	+	+	—	_	_	_	+	-	+	_	+	_	7

NOTE. 1, Research question or objective clearly stated; 2, Study population clearly specified and defined; 3, Participation rate of eligible persons \geq 50%; 4, Subjects selected from same or similar population; 5, Sample size justification; 6, Exposure(s) of interest measured before outcome(s); 7, Time frame sufficient; 8, Different levels of exposures as related to the outcome are examined; 9, Exposure measures clearly defined, valid, and reliable; 10, Exposure(s) assessed more than once over time; 11, Outcome measures clearly defined, valid, and reliable; 12, Outcome assessors blinded to the exposure status; 13, Follow-up after baseline \leq 20%; 14, Adjusted for potential confounding variables. Total (0–14).^{1,2} Abbreviations: +, meet criteria; -, do not meet criteria.

References

- 1. National Institutes of Health. Quality Assessment Tool for Observational Cohort and Cross-Sectional Studies; 2014.
- 2. The Cochrane Collaboration. Cochrane handbook for systematic reviews of interventions; 2011.

Quality of Evidence

inconsistency)

inconsistency)

inconsistency)

inconsistency)

Moderate quality (low risk of bias and high

Moderate quality (low risk of bias and moderate

Moderate quality (moderate risk of bias and low

Low quality (moderate risk of bias, moderate inconsistency, and low indirectness)

Moderate quality (low risk of bias and high

Supplemental Appendix S5	Assessment of quality of eviden	ce		
No. of Studies (Sample Size, n)	Risk of Bias	Inconsistency	Indirectness	Imprecision
Immediate effects Outcome: DEROM				
7 studies (n; experimental=126: control=123)	Low risk of bias (PEDro scores: 6, 6, 7, 8, 8, 8, and 8)	 P value on test for heterogeneity P<.00001, I²=91% High inconsistency 	Low indirectness	Low imprecision
Outcome: dynamic balance				
5 studies (n; experimental=90: control=90)	Low risk of bias (PEDro scores: 6, 7, 8, 8, and 8)	P value on test for heterogeneity P=.02, I ² =52% Moderate inconsistency	Low indirectness	Low imprecision
Outcome: static balance		5		
3 studies (n; experimental=50: control=50)	Moderate risk of bias (PEDro scores: 6, 6, and 8)	<i>P</i> value on test for heterogeneity $P=.93$, $I^2=0\%$ Low inconsistency	Low indirectness	Low imprecision
Outcome: pain intensity		Ĵ		
2 studies (n; experimental=24: control=23)	Moderate risk of bias (PEDro scores: 5 and 8)	P value on test for heterogeneity $P=.73$, $I^2=0\%$ Low inconsistency	Moderate indirectness (less generalizable)	Low imprecision
Short-term effects Outcome: DFROM				
2 studies (n; experimental=48: control=46)	Low risk of bias (PEDro scores: 7 and 8)	P value on test for heterogeneity P<.0001, I ² =95%	Low indirectness	Low imprecision

High inconsistency

www.archives-pmr.org
STUDY PROTOCOL

Open Access



Effects of mobilisation with movement (MWM) on anatomical and clinical characteristics of chronic ankle instability: a randomised controlled trial protocol

Ishanka Weerasekara^{*}, Peter Grant Osmotherly, Suzanne Jordan Snodgrass, John Tessier and Darren Anthony Rivett

Abstract

Background: Up to 40% of individuals who sprain their ankle develop chronic ankle instability (CAI). One treatment option for this debilitating condition is joint mobilisation. There is preliminary evidence that Mulligan's Mobilisation With Movement (MWM) is effective for treating patients with CAI, but the mechanisms by which it works are unclear, with Mulligan suggesting a repositioning of the fibula. This randomised controlled trial aims to determine the effects of MWM on anatomical and clinical characteristics of CAI.

Methods: Participants 18 years or over with CAI will be accepted into the study if they satisfy the inclusion and exclusion criteria endorsed by the International Ankle Consortium. They will be randomised into the experimental group (MWM) or the placebo group (detuned laser) and will receive the assigned intervention over 4 weeks. General joint hypermobility and the presence of mechanical instability of the ankle will be recorded during the first visit. Further, position of the fibula, self-reported function, ankle dorsiflexion range, pressure pain threshold, pain intensity, and static and dynamic balance will be assessed at baseline, and at the conclusion of course of intervention. Follow-up data will be collected at the twelfth week and at the twelfth month following intervention.

Discussion: Effectiveness of MWM on clinically relevant outcomes, including long term benefits will be evaluated. The capacity of MWM to reverse any positional fault of the fibula and the association of any positional fault with other clinically important outcomes for CAI will be explored. Proposed biomechanical mechanisms of fibular positional fault and other neurophysiological mechanisms that may explain the treatment effects of MWM will be further explored. The long term effectiveness of MWM in CAI will also be assessed.

Trial registration: Australian New Zealand Clinical Trials Registry; ACTRN12617001467325 (17/10/2017).

Keywords: Ankle sprain, Chronic ankle instability, Joint mobilisation, Mobilisation with movement, Fibular positional fault, Mechanical instability, Functional instability

* Correspondence: ishanka.weerasekara@uon.edu.au; https://www.newcastle. edu.au/profile/ishankamadhurangani-rajapakshamudiyanselage-uon School of Health Sciences, Faculty of Health and Medicine, The University of Newcastle, Callaghan, Australia



© The Author(s). 2019 **Open Access** This article is distributed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated.

Background

Up to 40% of patients with an initial ankle sprain develop chronic ankle instability (CAI), which is frequently associated with recurrent sprains and persistent pain [1, 2]. A recurrent subjective perception of the ankle joint 'giving way' is clinically indicative of CAI [3], which is defined as "repetitive bouts of lateral ankle instability resulting in numerous ankle sprains" [4]. Clinical management of CAI often involves balance and sport-related activity training [5]. In a recent metaanalysis, preliminary evidence was also found supporting joint mobilisation as a clinically effective intervention in improving dynamic balance and dorsiflexion range of motion in CAI [6].

Several ankle joint mobilisation procedures have been developed and described by renowned manual therapists such as Geoffrey Maitland, Freddy Kaltenborn and Brian Mulligan, and are commonly used in rehabilitation [7]. These procedures are applied to a joint, either in the form of non-thrust passive joint mobilisations, high velocity thrust manipulation, or Mobilisation With Movement (MWM). MWM is defined as the application of a sustained passive accessory movement to a joint while the patient actively performs a task/movement that was previously identified as being painful or limited [8]. After manual application of MWM, tape is applied to help maintain the glide and corrected fibular position [9]. Biomechanically and neurophysiologically mediated mechanisms have been proposed to explain how these joint mobilisation procedures may work [8, 10]. The proposed neurophysiological mechanisms are based on animal [11] and human experiments [12] related to pain science and motor systems [8]. These have shown that joint manual therapy techniques including MWM, activate a descending pain inhibitory pathway which is non-opioid mediated [8]. One proposed biomechanical mechanism relates to a reduction of an entrapped meniscoid or synovial fringe by a specifically directed MWM glide particularly in those instances where only one repetition is required to bring about a substantial and long lasting effect [8].

Of our recent systematic review and meta-analysis identified greater effects for MWM and Mulligan taping compared to Maitland joint mobilisation (with and without traction) and joint thrust manipulation [6, 13]. Dorsiflexion range of motion and self-reported instability were some of the outcomes for which there was evidence of improvement from MWM, although the long term benefits were unclear [14–16]. Most of the previous studies on chronic ankle sprains have applied MWM to the talocrural joint [14–18], and few studies have applied MWM taping [19–21]. However Mulligan proposes that an anterior fibular positional fault commonly results from ankle inversion sprains, and that

a MWM using a posterior glide of the fibula to correct this should be trialled after 48 h following such an injury [22]. Patients with recurrent ankle sprains may also benefit from this MWM treatment combined with taping aimed at maintaining the posterior fibula glide, with reportedly less 'giving way' and greater confidence in using the ankle in patients with functional instability and pain [22]. Therefore this study is designed to evaluate the clinical benefits of the fibular posterior glide MWM with Mulligan taping, and whether it corrects any demonstrable positional fault which may exist. The prevalence of pain in people with CAI is high (60.1%) [23] and to our knowledge no studies have assessed the effect of MWM on pain. In addition, the present study will assess the effects of MWM on pressure pain threshold in CAI. The presence of localized peripheral sensitization has been previously identified in acute inversion ankle sprains [24] and in subacute ankle sprains [25]. Balance impairments in CAI are frequently reported in the literature and MWM has been found to be effective immediately after application, but there is presently insufficient research to determine the short term benefits of MWM for balance impairments [6]. The present study plans to address this deficiency in the literature as well.

A positional fault at the inferior tibio-fibular joint, is one arthrokinematic abnormality proposed to be related to persistent/recurrent symptoms and repetitive ankle sprains in CAI [4]. In the case of an ankle joint sprain, Mulligan suggests that the distal fibula is 'mal-positioned' anteriorly (anterior positional fault) following an inversion injury and that chronicity may result if this remains uncorrected [22, 26]. Preliminary evidence for such an anterior fibular positional fault was identified in Hubbard et al's study of individuals with CAI [27]. However it is unclear whether ankle instability caused the anterior fibular position or whether the fault itself was actually the predisposing factor to re-injury. Also, the clinical importance of an anterior fibular positional fault in relation to other potential contributors to CAI remains unclear. Further, the mechanism(s) of changes in CAI outcomes after MWM needs to be further investigated [15, 28]. It has been proposed by Mulligan in his positional fault hypothesis, that MWM effects an immediate and lasting improvement by correcting a minor bony incongruity which is the source of the patient's presenting problem [22]. However, the capacity of MWM to reverse any positional fault remains unclear and further studies are required to assess the effectiveness of this technique.

The objective of this study is to determine the effect of MWM on anatomical and clinical characteristics of CAI, and to determine the long term effectiveness of this treatment.

The specific aims of the study are therefore:

- 1. To evaluate the effectiveness of MWM on clinically relevant outcomes, including patient-reported outcomes (dorsiflexion range, pain intensity, self- reported function, pressure pain threshold, static and dynamic balance), including long lasting benefits assessed at 12 months post treatment.
- 2. To assess the effect of MWM on changing the fibular position relative to the position of the tibia in CAI.

Methods

Design

This randomised controlled study has been registered in the Australian New Zealand Clinical Trial Registry (ANZTR) and ethical approval has been granted by the Human Research Ethics Committee of The University of Newcastle, Australia (H-2017-0354). Informed consent will be obtained in writing from all participants.

Participants

Participants aged 18 years or over will be recruited from the general community in the Newcastle area of New South Wales, Australia through flyers posted on noticeboards in shopping centres, the University of Newcastle main campus, and various other public places. Recruitment advertising will also be via University of Newcastle social media channels. Volunteers with CAI will be accepted into the study if they satisfy the inclusion and exclusion criteria as endorsed by the International Ankle Consortium [29], except the time period for experiencing at least two episodes of giving way is changed from 6 months to 12 months to account for the seasonal nature of some sports (Table 1).

Data collection will be carried out at the physiotherapy and radiography research laboratories of the School of Health Sciences, The University of Newcastle, Australia.

Procedure

This trial will adopt a pragmatic randomized controlled trial design to allow for real world application of MWM in a randomized setting [30]. This design has been used by previously published trials of manual therapy to better reflect routine clinical practice [31–33]. It enhances the external validity, but still controls for threats to internal validity.

The initial screening will be performed over the telephone after the potential participant contacts the research team. The screening questions are to determine if the potential participant meets some of the inclusion/ exclusion criteria (Table 1). If a potential participant appears eligible following the telephone interview, further screening will be carried out using two standardised

questionnaires: the Foot and Ankle Ability Measure (FAAM) [34], which measures function, and the Cumberland Ankle Instability Tool (CAIT) [35], which measures ankle instability. A link to access these questionnaires on the Qualtrics online survey platform (Qualtrics, Provo, Utah, USA) will be sent to the potential participant, along with the participant information statement and the consent form, through an email. Once the potential participant returns their completed forms, their final eligibility will be determined according to their scores (FAAM: activities of daily living (ADL) subscale < 90%, sport subscale < 80%; CAIT \leq 24) on the two screening questionnaires. The participant will then be contacted to schedule an appointment for data collection.

Consenting participants will be randomised into two groups: an experimental group who will receive MWM, and a control group who will receive a placebo (detuned laser). All of the participants will be assessed for general joint hypermobility using the Beighton score [36]. Mechanical ankle instability will be tested separately for each ankle using an X-ray while undergoing an anterior drawer stress test [37, 38]. The clinically important outcome measures will include; radiological imaging of fibular position with respect to the tibia (positional fault), dorsiflexion range of motion (DFROM), pressure pain threshold (PPT), pain intensity, function, static balance and dynamic balance. These procedures and outcome measures are further explained below. The researcher who collects the clinical measurements, and the radiographer taking the X-rays, will be blinded to the participant's group (intervention) allocation. This researcher will remain blinded to the group allocation until the 3 month follow-up. The 12 month follow-up data will be collected using online questionnaires.

Each participant will be randomly allocated to a group to receive either MWM (active) treatment or detuned laser treatment (placebo) (these interventions are fully explained below). The participant will be blinded as to whether they are receiving an active or placebo intervention, however due to the nature of the interventions, the treating practitioner cannot be blinded. Participants will be randomly allocated to groups according to a computer generated (GraphPad Software, Inc., CA, USA) randomisation schedule by another researcher not involved in data collection using sealed opaque envelopes. Each envelope will contain a piece of paper printed with either '1' or '0', for which '1' denotes 'MWM' group and '0' denotes 'placebo' group. The treating practitioner will open the envelope and allocate the participant to a group according to the number in the envelope, and deliver the designated treatment accordingly.

Participants of both groups will attend for 2–8 treatment sessions over 4 weeks. The exact number of

Table 1 Inclusion and exclusion criteria

Inclusion criteria	Exclusion criteria
 A history of at least one significant ankle sprain; Initial sprain must have occurred at least 12 months prior to study enrolment Was associated with inflammatory symptoms Created at least one interrupted day of desired physical activity The most recent injury must have occurred more than 3 months prior to study enrolment 	 A history of previous surgeries to the musculoskeletal structures (i.e., bones, joint structures, nerves) in either lower extremity A history of a fracture in either lower extremity requiring realignment Acute injury to musculoskeletal structures of other joints of the lower extremity in the previous 3 months that impacted joint integrity and function (i.e., sprains, fractures), resulting in at least one interrupted day of desired physical activity Have conditions for which manual therapy is generally contraindicated (such as the presence of a tumour, fracture, rheumatoid arthritis, osteoporosis, prolonged history of steroid use, or severe vascular disease)
 A history of the previously injured ankle joint 'giving way' and/or recurrent sprain and/or 'feelings of instability' Participants should report at least two episodes of giving way in the 12 months prior to study enrolment 	Have conditions for which radiological imaging is contraindicated (e.g., pregnancy)
- Self-reported ankle instability should be confirmed with the Cumberland Ankle Instability Tool (CAIT) (\leq 24)	 Have conditions for which taping is contraindicated (e.g., allergy to strapping tape)
 General self-reported foot and ankle function questionnaire minimum score (Foot and Ankle Ability Measure (FAAM): activities of daily living (ADL) subscale < 90%, sport subscale < 80%) 	 Receiving concurrent treatment The most recent treatment for the ankle condition should have been received at least a week prior to study enrolment
	Inability to read English

(2019) 20:75

treatments needed to achieve an optimal change is not presently known, so a range allows the practitioner to exercise their clinical judgement. We have chosen two as the minimum number of treatments, because usual clinical practice would involve a minimum of two visits to enable re-assessment following the initial treatment [39]. The actual number of treatment sessions delivered to participants in each group will be determined according to the clinical judgement of the treating practitioner, who is a registered physiotherapist with a post-professional tertiary qualification in the field of manual therapy and more than 20 years of clinical experience in treating musculoskeletal conditions. The physiotherapist will also be individually instructed in the MWM intervention by an accredited member of the Mulligan Concept Teachers Association. The physiotherapist will conclude the course of intervention if the patient reports they have fully recovered or if no further improvement is possible up to a limit of eight sessions over a 4 week period. The number of sessions and the duration of each session will be recorded. The same measures taken at baseline will be repeated at the conclusion of the course of intervention, within a maximum of 4 days after the participant's last intervention session. Further measurements will be repeated at the twelfth week with the exception of the imaging, and only self-report outcomes will be assessed at 12 months. Participants will be contacted

by telephone every 4 weeks after finishing treatment for up to 1 year to record any new ankle injuries, any treatments undertaken, and their level of engagement in sport and other activities. Figure 1 describes the flow of the study.

Outcome measures

Measurement of fibular positional fault from radiograph

A weight-bearing (neutral ankle in standing position) X-ray (55 k Vp and 2.1 mAs) will be taken of the affected ankle of the participant. The participant will be asked to stand on the foot to be imaged on a wooden box with the knee slightly flexed to simulate mid-stance of the gait cycle, with the foot of the non-stance leg hanging in a relaxed manner. The imaged leg will be maintained ~ 2 cm away and parallel to the image receptor. The same instructions will be given to all participants and the participant's leg position will be monitored throughout the procedure. If any leg rotation is noted on imaging, the X-ray will be redone. The central ray will be centred at the base of the metatarsals and perpendicular to the image receptor, and the focal-film distance will be set to 110 cm. The participant will be allowed to hold on to body of the X-ray machine for balance if required.

Radiographic images will be digitally obtained using Merge PACS^{**} software (Merge Health Care, 2012). The fibular position will be measured as the distance between the anterior edge of the distal fibula and the anterior edge of the distal tibia [27] (Fig. 2). The test-retest reliability intra-class correlation coefficient (ICC)_{3,1} has been estimated as 0.98, with a SEM of 0.64 mm for this measurement, and for intra-tester reliability, the ICC_{3,1} is 0.92 and SEM is 0.72 mm [27].

Weight-bearing dorsiflexion range of movement

Weight-bearing DFROM will be measured using the weight-bearing lunge test. The participant will be



instructed to lunge towards the wall, touch their knee to the wall, and keep their heel in contact with the floor. Then the participant will be asked to move their foot away from the wall in 1 cm increments until the heel no longer maintains contact with the floor or the knee is no longer in contact with the wall. Maximal dorsiflexion will be considered to be the greatest distance between the great toe and wall with the participant's knee maintaining contact with the wall [18, 40]. Both inter-rater reliability (ICC = 0.80–0.99) and intra-rater reliability (ICC = 0.65 - 0.99) have been reported as high for this test [41]. The same procedure will be followed for the opposite leg. Each centimetre away from the wall in the lunge test represents approximately 3.6 degrees of dorsiflexion [42]. Three test attempts will be performed and the average value will be used for analysis.

Pressure pain threshold

PPT measurements will be obtained in each leg from two local points (to assess local hypersensitivity) and one remote body area (to assess central sensitisation), in accordance with the method used in a previous study on acute ankle sprain [24]. The points include anterior to the lateral malleolus over the anterior talo-fibular ligament, inferior to the medial malleolus over the deltoid ligament, and over the proximal third of the tibialis anterior muscle belly.

A Freedom Tracker hand-held algometer (JTECH Medical, Salt Lake City, UT, USA) will be used for measuring PPT. A probe (contact surface of 1 cm^2) will be placed perpendicular to the skin and pressure will be applied (40 kPa/s). The participant will be asked to indicate when the feeling of the stimulus changes from 'pressure only' to 'discomfort' by pressing an indicator switch [43, 44]. This process will be performed three consecutive times and a 10 s rest period will be allowed between each set of measurements. Pressure algometry is considered a stable and reliable measure of PPT [45]. The inter-rater reliability of pressure algometry has been reported to be high when the algometer pressure is applied at a consistent rate (ICC 0.91, 95% CI 0.82-0.97) [46].



Pain intensity

Current pain intensity will be assessed using the visual analogue scale (VAS) which consists of a 100 mm horizontal line, with 'no pain' anchored on the left of the line and 'worst possible pain' anchored on the right. The validity of the VAS for detecting changes in pain intensity has been supported by several studies [47, 48].

The participant will also be asked to indicate all areas in which they currently feel symptoms on a body chart. The areas in which they feel 'pain' will be shaded; the areas in which they feel 'tingling, pricking, or burning' will be circled; and the areas where they feel 'numbness, heaviness or other sensations' will be indicated on the chart by an 'N'.

Function

Self-reported physical function of the participant will be evaluated using the FAAM which consists of a 21-item ADL subscale and an 8-item sport subscale [34]. This tool has been documented as a reliable, responsive and valid measure of physical function for individuals with a broad range of musculoskeletal disorders of the lower leg, foot and ankle [34]. The Foot and Ankle Outcome Score (FAOS) questionnaire comprising 42 items will also be used, and has been reported as also being a reliable and valid measure (ICCs reported as 0.78, 0.86, 0.70, 0.85, 0.92 for the five subscales of pain, symptoms, ADL, sport and recreation function, and foot- and ankle-related quality of life, respectively) [49].

Further, the participant will be asked to identify up to three important activities that they are unable to perform or are having moderate to extreme difficulty performing due to pain. For each activity, the participant will be asked to rate between 0 and 10 the level of difficulty they experience performing that activity using the Patient-Specific Functional Scale (PSFS) [35]. The construct validity of the PSFS is well supported, and the test-retest reliability has been assessed as moderate to good (ICC2,1 = 0.713) [36].

Static balance

For static balance, the participant will stand barefoot on the centre of a force plate (KISTLER 9260AA6, Winterthur, Switzerland), assuming a standardized single leg stance position. The participant will then be instructed to flex the other leg slightly at the hip, with the knee flexed to 90 degrees. Their arms will be crossed at their chest with each hand resting on the opposite shoulder. Measurements will be recorded with both 'eyes open' and 'eyes closed'. For 'eyes open,' the participant will be asked to maintain a fixed gaze on a cross marked on the wall three metres in front of them and remain as still as possible for 10 s [50]. For 'eyes closed', the participant will be asked to close their eyes and remain as still as possible for 10 s [50]. If the participant is unable to stand for 10 s, the standing time achieved will be recorded. Only averaged Centre of Pressure (CoP) data including sway velocity, sway area per second, sway average amplitude and sway maximal amplitude will be used in the analysis to maintain consistency. CoP data obtained through the force platform will be acquired at 100 Hz [21].

Dynamic balance

Dynamic balance will be assessed using the Star Excursion Balance Test (SEBT) which has been shown to be a reliable measure to identify dynamic balance deficits in patients with a variety of lower extremity conditions [51]. The participant will be asked to establish a stable base of support on the stance limb in the middle of the testing grid on a force plate (KISTLER 9260AA6, Winterthur, Switzerland). While standing on a single limb, the participant will be asked to reach as far as possible with the reaching limb along each line (anterior, postero-medial and postero-lateral directions), lightly touching the line with the most distal portion of the reaching foot without shifting weight or coming to rest on the foot of the reaching limb. The participant will then be asked to return the reaching limb to the starting position in the centre of the grid. If the individual lifts or shifts any part of the foot of the stance limb during the trial, the trial will be not considered as complete [51].

After performing a maximum of four non-recorded trials for familiarisation, the next trial for each direction will be recorded for the purpose of analysis [52, 53]. Normalised SEBT values will be obtained by dividing the excursion distance by the participant's leg length (the distance between the anterior superior iliac spine and the ipsilateral medial malleolus), and then multiplying by 100 [52, 54]. Data for centre of pressure (CoP) velocity (V) to quantify spatio-temporal parameters (VCoP-total, VCoP-mediolateral, VCoP-anteroposterior) will be acquired at 100 Hz, under the foot during unipodal stance [52].

Perceptions of the credibility of the placebo

At the data collection session at the conclusion of course of intervention, the participant will be asked to indicate which intervention (active or placebo) they thought they had received during the last 4 weeks and to give a confidence rating on a scale of 0-10 (with 0='not at all confident' and 10 = 'extremely confident' [55]). Global perceived effect will also be measured using a self-assessment of improvement on a seven point rating scale (1 = completely recovered, 2 = much improved, 3 = slightly improved, 4 = not changed, 5 = slightly worsened, 6 = much worsened, 7 = worse than ever) in response to the question 'How would you rate the course of your ankle complaints since the start of this study?' [56, 57].

Other measures

Telephone interviews will be conducted monthly after enrolment up to 1 year to record any new injuries, any treatments undertaken, and the level of engagement in sports and other activities. These variables will be used as covariates in the analysis of the 12 month follow-up data as they are possible confounders. Further, the Beighton score for hypermobility and radiographic measurement of the anterior drawer stress test will be recorded.

Beighton score

Scoring for joint hypermobility will be undertaken according to previously published methods [36]. Each participant will be assessed in five test positions, as follows:

- Passive extension of the fifth metacarpophalangeal (MCP) joint to ≥90 degrees. The participant sits on a chair at the short side of the table with the shoulder in 80 degrees abduction, elbow flexed at 90 degrees, and the forearm resting on the table in a pronated position. The fifth MCP joint is passively extended by the researcher and a goniometer is used to measure the angle.
- Passive hyperextension of the elbow ≥10 degrees. The participant sits on a chair with the shoulder at 90 degrees of flexion and the forearm supinated. A goniometer is placed at the lateral epicondyle and the measurement is taken at maximum elbow extension.
- 3. Passive hyperextension of the knee ≥10 degrees. The participant lies supine with their legs in the horizontal plane. The goniometer is placed at the lateral femoral condyle and the measurement taken at maximum knee extension.
- 4. Passive apposition of the thumb to the flexor side of the forearm. The score is positive if the entire thumb touches the flexor side of the forearm while the shoulder is flexed at 90 degrees, the elbow extended, and the forearm pronated.
- 5. Forward flexion of the trunk with the knees straight. The score is positive if the participant's hand palms rest easily on the floor.

Anterior drawer stress test with radiographic measurement

Ankle joint mechanical instability will be assessed using a lateral x-ray to measure the amount of anterior translation of the talus during a ligament stress test for each ankle. The radiograph will be taken while the ankle is undergoing a simulated anterior drawer test using 125 N force [38]. The stress radiograph will be taken with the participant in a supine lying position with the foot relaxed in a resting position and the lower leg resting on a support, with the hip and knee each flexed approximately 45 degrees. The heel will be supported on a dynamometer (Lafayette Manual Muscle Tester, Model 01165, Lafayette, IN, USA) attached to a customised device which produces the anteriorly directed force. The distal tibia will be fixed on the support using a stabilising belt placed over the distal aspect of the tibia [58]. The central ray will be centred just above the tip of the lateral malleolus and perpendicular to the image receptor [59]. Then an anterior force of 125 N will be applied [38] to the heel of the participant at an angle of 20 degrees to the vertical plane as per recommended clinical practice [60], using the customised device. The force will be monitored using the digital display of the dynamometer attached to the customised device, and the radiograph will be taken at 125 N. The ankle radiograph will be taken at the focal-film distance of 110 cm [61] and will set to 55 kVp and 2.1 mAs. The same procedure will be applied to the other ankle. These images will be taken at the baseline data collection session to assess mechanical instability for use in subgroup analysis.

Radiographic images will be digitally obtained using Merge PACS[™] software (Merge Health Care, 2012). Anterior translation of the talus will be measured between the posterior lip of the tibial articular surface and the nearest point of the talar dome (Fig. 3) [61-64] to identify any mechanical instability. Anterior drawer stress radiographs have been found to have moderate sensitivity, high specificity and a high positive predictive value for the evaluation of lateral ankle instability [37]. A between-limb difference of 3 mm in anterior translation of the talus or an absolute value of 10 mm is considered clinically significant [37].

Application of the intervention

Participants in the experimental group will be treated with a manual MWM technique to the ankle and will be taped after the intervention using the Mulligan approach [22] to attempt to maintain the effects of the MWM. The control group will receive a detuned (inactive) therapeutic laser treatment to the lateral region of the ankle. The number of treatment sessions delivered for each participant will be based on their symptomatic response to treatment, as determined by the clinical judgement of the treating practitioner. Each participant will be asked to avoid concurrent interventions during their participation in the study.

MWM intervention

The participant's inferior tibio-fibular joint will be mobilised using Mulligan's fibula MWM for dorsiflexion and/ or inversion [22]. Initially, the technique will be performed in supine lying with the tibia resting on the treatment table and the foot unsupported off the table's edge. The practitioner applies a sustained pain-free anteroposterior glide with a slight cephalad and lateral inclination to the distal fibula (lateral malleolus). This glide is maintained while the participant performs active inversion or dorsiflexion (depending on which is more limited in range) to end of range. There should be no pain with the active movement. At the end of range, the practitioner will apply and sustain overpressure to the active movement for a few seconds (or the participant will do so after appropriate instruction) [16, 22]. If dorsiflexion remains restricted, this technique can be progressed and performed in partial and/or full weight-bearing.

One treatment session will consist of three to five sets, with six to ten repetitions of the active movement in each set, with the actual dosage depending on the individual response of the participant [22]. Participants will receive between two to eight sessions according to the clinical reasoning of the practitioner, over a period of 4 weeks. After each session, Mulligan MWM taping will be applied in an attempt to replicate the sustained fibula glide [8]. Non-elastic tape will be applied to the ankle starting 2 cm anterior to the fibula and 1 cm proximal to the tip of the lateral malleolus. The tape will be spiralled obliquely around the lower limb while the fibula glide is sustained, finishing on the anterior aspect of the leg [22]. The participant will be instructed to keep the tape on for 24 h. In the case of an adverse reaction, they will be advised to remove the tape immediately and note the length of time the tape was in place.

Detuned laser intervention

The placebo intervention will be applied using a detuned therapeutic laser device (Meyer Medical Electronics, Mordialloc, Australia) for 5 min to the lateral region of the ankle, maintaining the probe 0.5-1 cm away from the skin [31, 65-67]. The detuned laser device will appear to function normally (both audibly and visually) to participants, but no effective emission will be produced. Both the participant and the practitioner will be required to wear protective glasses as per normal clinical practice [66, 68]. Participants will receive two to eight treatments over 4 weeks, according to the clinical judgement of the treating practitioner. Detuned laser has been used in several other studies assessing manual therapy including for chronic ankle sprains. It avoids any possible direct mechanical effects to the ankle being treated and also does not activate somatosensory receptors [69–71]. Further, it has been shown to have a strong placebo effect [70]. Scheduling of participant appointments will be arranged to avoid interaction between participants.



Fig. 3 Anterior translation of the talus during the anterior drawer stress test is measured as the distance on X-ray from the posterior lip of the tibial joint surface to the nearest point of the talar dome (left ankle, 13.2 mm in this image)

Sample size and data analysis

Previously published data related to the primary outcome measure of function (FAAM subscales, ADL and sports) [18, 34] (MCID = 8.0, SD = 5.68; MCID = 9.0, SD = 7.42 respectively) were used in sample size calculations [18, 34, 72]. A sample size of 16 per group allowing for a 30% drop-out rate was estimated, for a minimal statistical power of 0.80 and an alpha significance level of 0.05. Secondary analysis based on the subgroups of ankle instability (mechanical, functional) will be preliminary in nature as the study is not powered for this aim.

Data will be analysed using SPSS Statistics for Windows (Version 23.0, Armonk, NY, IBM Corp). Continuous data will be assessed for normality using the Shapiro-Wilk test.

Baseline comparability between groups will be analysed using the independent t-test or non-parametric equivalent, as appropriate. Linear mixed models will be used to analyse the outcome measures. For the primary outcome measure, 'function' will be the outcome variable and time,

group and an interaction term for time by group will be the predictors. Any statistically significant difference in change in the outcome variable over time between the groups will be indicated by the *p* value for the interaction term. Pairwise Bonferroni comparisons will be performed to explore the differences between time points and between groups if a significant interaction is identified. Independent t-tests will be used to compare outcome measures between groups at each time point and the changes of the scores will be used to detect any changes in the outcomes of interest. Intention to treat analysis will be performed with all participants allocated to each group condition to evaluate the effect of the independent variable. For missing data in ITT analysis, a participant's last observation for each outcome measure will be carried forward. The average number and the average duration of intervention sessions between groups will be compared. If any significant difference observed, secondary analysis will be taken to find any correlation between the treatment volume and the outcomes.

Additional variables recorded during monthly phone interviews (new injuries, changes in activity level, and occurrence of other treatments) will be used as covariates in the analysis of the 12 month follow-up data as they are possible confounders. Further, the Beighton score for hypermobility will also be included in regression analysis as a covariate.

Radiographic measurement of the anterior drawer stress test will be used to differentiate subgroups of CAI in potential sub group analysis.

Discussion

One proposed anatomical mechanism underpinning MWM is theorised to be a correction of a minor bony incongruity (positional fault) which is at the source of the patient's presenting problem [22, 73]. The existence of an anterior fibular 'positional fault' in individuals with CAI has some preliminary radiological support [27]. There are also limited MRI data supporting Mulligan's positional fault hypothesis in cases of lateral ankle pain [72], however there is no evidence to date that MWM reverses any positional anomaly. Further, should any fibular positional anomaly be reversed immediately after the application of MWM, the length of time this reversal or correction is maintained is unknown. The proposed study protocol is designed to determine the presence of any positional fault of the fibula in CAI, and whether MWM can reverse this, and if so, whether this reversal evident 4 weeks after treatment commences. is Moreover, this study protocol will explore the correlation between an anatomical measure (fibular position) and other clinical outcomes (pain, function, pressure pain threshold, DFROM, static and dynamic balance). Potential relationships between these measures may help explain how changing an anatomical measure may effect a clinically meaningful outcome. The effect of MWM in CAI will also be explored in relation to the presence or not of radiologically measurable mechanical instability.

There are very few clinical trials with long term follow-ups which have assessed MWM for any musculoskeletal condition, and only one for CAI which had a 6 month follow-up [6, 15]. The proposed study protocol is therefore the first designed to evaluate the long term effectiveness of MWM on CAI. Moreover, the treatment effect may depend on the type of instability present (mechanical or functional), and this study protocol may evaluate the efficacy of MWM on these two subgroups of CAI. However, the subgroup analysis will be exploratory as the study was only powered to detect the main effect being the intervention on the functional outcome.

Abbreviations

ADL: Activities of daily living; ANZCTR: Australian New Zealand Clinical Trial Registry; AP: Antero-posterior; CAI: Chronic ankle instability; CAIT: Cumberland Ankle Instability Tool; CoP: Centre of pressure; DFROM: Dorsiflexion range of motion; FAAM: Foot and Ankle Ability Measure; FAOS: Foot and Ankle Outcome Score; MCID: Minimal clinically important difference; MCP: Metacarpophalangeal; ML: Medio-lateral; MWM: Mobilisation With Movement; PPT: Pressure pain threshold; PSFS: Patient-Specific Functional Scale; SD: Standard deviation; SEBT: Star Excursion Balance Test; VAS: Visual Analogue Scale

Acknowledgements

Ishanka Weerasekara is supported by an Australian Government Research Training Program Scholarship.

Funding

none.

Availability of data and materials not applicable.

Authors' contributions

IW, PO, JT, SS and DR contributed in designing the study. JT performed the radiological procedures. IW drafted the manuscript, and all authors (IW, PO, JT, SS and DR) read and approved the final manuscript.

Authors' information

- IW: https://www.newcastle.edu.au/profile/ishankamadhuranganirajapakshamudiyanselage-uon
- PO: https://www.newcastle.edu.au/profile/peter-osmotherly
- SS: https://www.newcastle.edu.au/profile/suzanne-snodgrass
- DR: https://www.newcastle.edu.au/profile/darren-rivett
- Л: https://www.newcastle.edu.au/profile/john-tessier

Ethics approval and consent to participate

This project has been approved by the Human Research Ethics Committee of The University of Newcastle, Australia (H-2017-0354). Written informed consent will be obtained from all participants.

Consent for publication

Not applicable.

Competing interests

Darren Rivett is an honorary member of the Mulligan Concept Teachers Association. The other authors declare that they have no competing interests.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Received: 23 January 2018 Accepted: 1 February 2019 Published online: 13 February 2019

References

- Hershkovich O, Tenenbaum S, Gordon B, Bruck N, Thein R, Derazne E, Tzur D, Shamiss A, Afek A. A large-scale study on epidemiology and risk factors for chronic ankle instability in young adults. J Foot Ankle Surg. 2015;54(2):183–7.
- Doherty C, Delahunt E, Caulfield B, Hertel J, Ryan J, Bleakley C. The incidence and prevalence of ankle sprain injury: a systematic review and meta-analysis of prospective epidemiological studies. Sports Med. 2014;44(1):123–40.
- Gigi R, Haim A, Luger E, Segal G, Melamed E, Beer Y, Nof M, Nyska M, Elbaz A. Deviations in gait metrics in patients with chronic ankle instability: a case control study. J Foot Ankle Res. 2015;8(1):1–8.
- 4. Hertel J. Functional anatomy, Pathomechanics, and pathophysiology of lateral ankle instability. J Athl Train. 2002;37(4):364–75.
- Martin RL, Davenport TE, Paulseth S, Wukich DK, Godges JJ. Ankle stability and movement coordination impairments: ankle ligament sprains. J Orthop Sports Phys Ther. 2013;43(9):A1–A40.
- Weerasekara I, Osmotherly P, Snodgrass S, Marquez J, de Zoete R, Rivett DA. Clinical Benefits of Joint Mobilization on Ankle Sprains: A Systematic Review and Meta-Analysis. Arch Phys Med Rehabil. 2018;99(7):1395–1412.e1395.

- Pettman E. A history of manipulative therapy. J Man Manip Ther. 2007;15(3):165–74.
- Vicenzino B, Hing W, Rivett DA, Hall T. Mobilisation with movement: the art and the science. Australia: Elsevier; 2011.
- 9. Mau H, Baker RT. A modified mobilization-with-movement to treat a lateral ankle sprain. Int J Sports Phys Ther. 2014;9(4):540–8.
- McCarthy C, Bialosky J, Rivett D. Spinal manipulation. In: Jull G, Moore A, Falla D, Lewis J, McCarthy C, Sterling M, editors. Grieve's Modern Musculoskeletal Physiotherapy. 4th edn: Australia: Elsevier; 2015. p. 278–82.
- Chien A, Eliav E, Sterling M. Hypoaesthesia occurs with sensory hypersensitivity in chronic whiplash--further evidence of a neuropathic condition. Man Ther. 2009;14(2):138–46.
- Sterling M, Jull G, Wright A. Cervical mobilisation: concurrent effects on pain, sympathetic nervous system activity and motor activity. Man Ther. 2001;6(2):72–81.
- Beazell JR, Grindstaff TL, Sauer LD, Magrum EM, Ingersoll CD, Hertel J. Effects of a proximal or distal tibiofibular joint manipulation on ankle range of motion and functional outcomes in individuals with chronic ankle instability. J Orthop Sports Phys Ther. 2012;42(2):125–34.
- Reid A, Birmingham TB, Alcock G. Efficacy of mobilization with movement for patients with limited dorsiflexion after ankle sprain: a crossover trial. Physiotherapy Canada. 2007;59(3):166–72 167p.
- Cruz-Diaz D, Lomas Vega R, Osuna-Perez MC, Hita-Contreras F, Martinez-Amat A. Effects of joint mobilization on chronic ankle instability: a randomized controlled trial. Disabil Rehabil. 2015;37(7):601–10.
- Vicenzino B, Branjerdporn M, Teys P, Jordan K. Initial changes in posterior talar glide and dorsiflexion of the ankle after mobilization with movement in individuals with recurrent ankle sprain. J Orthop Sports Phys Ther. 2006;36(7):464–71.
- Marron-Gomez D, Rodriguez-Fernandez A, Martin-Urrialde J. The effect of two mobilization techniques on dorsiflexion in people with chronic ankle instability. Phys Ther Sport. 2015;16(1):10–5.
- Gilbreath JP, Gaven SL, Van Lunen L, Hoch MC. The effects of mobilization with movement on dorsiflexion range of motion, dynamic balance, and self-reported function in individuals with chronic ankle instability. Man Ther. 2014;19(2):152–7.
- Someeh M, Norasteh AA, Daneshmandi H, Asadi A. Influence of Mulligan ankle taping on functional performance tests in healthy athletes and athletes with chronic ankle instability. Int J Athl Ther Train. 2015;20(1):39–45.
- Someeh M, Norasteh AA, Daneshmandi H, Asadi A. Immediate effects of Mulligan's fibular repositioning taping on postural control in athletes with and without chronic ankle instability. Phys Ther Sport. 2015;16(2):135–9.
- Hopper D, Samsson K, Hulenik T, Ng C, Hall T, Robinson K. The influence of Mulligan ankle taping during balance performance in subjects with unilateral chronic ankle instability. Phys Ther Sport. 2009;10(4):125–30.
- 22. Hing W, Hall T, Rivett DA, Vicenzino B, Mulligan B. The Mulligan concept of manual therapy, vol. 13. Australia: Elsevier; 2015.
- 23. Adal SA, Pourkazemi F, Mackey M, Hiller C. Presence of pain in people with chronic ankle instability. Br J Sports Med. 2017;51(Suppl 1):A14.
- Ramiro-Gonzalez MD, Cano-de-la-Cuerda R, De-la-Llave-Rincon AI, Miangolarra-Page JC, Zarzoso-Sanchez R, Fernandez-de-Las-Penas C. Deep tissue hypersensitivity to pressure pain in individuals with unilateral acute inversion ankle sprain. Pain Med. 2012;13(3):361–7.
- Collins N, Teys P, Vicenzino B. The initial effects of a Mulligan's mobilization with movement technique on dorsiflexion and pain in subacute ankle sprains. Man Ther. 2004;9(2):77–82.
- 26. Mulligan BR. Manual therapy: NAGS, SNAGS, MWMs, etc 6th ed edn. Wellington: Plane View Services; 1995.
- 27. Hubbard TJ, Hertel J, Sherbondy P. Fibular position in individuals with selfreported chronic ankle instability. J Orthop Sports Phys Ther. 2006;36(1):3–9.
- Hoch MC, Mullineaux DR, Andreatta RD, English RA, Medina-McKeon JM, Mattacola CG, McKeon PO. Effect of a 2-week joint mobilization intervention on single-limb balance and ankle arthrokinematics in those with chronic ankle instability. J Sport Rehabil. 2014;23(1):18–26.
- Gribble PA, Delahunt E, Bleakley C, Caulfield B, Docherty C, Fourchet F, Fong D, Hertel J, Hiller C, Kaminski T. Selection criteria for patients with chronic ankle instability in controlled research: a position statement of the international ankle consortium. J Orthop Sports Phys Ther. 2013;43(8):585–91.
- Alsop J, Scott M, Archey W. The mixed randomized trial: combining randomized, pragmatic and observational clinical trial designs. J Comp Effect Res. 2016;5(6):569–79.

- Reid SA, Callister R, Snodgrass SJ, Katekar MG, Rivett DA. Manual therapy for cervicogenic dizziness: long-term outcomes of a randomised trial. Man Ther. 2015;20(1):148–56.
- 32. Groeneweg R, van Assen L, Kropman H, Leopold H, Mulder J, Smits-Engelsman BCM, Ostelo RWJG, Oostendorp RAB, van Tulder MW. Manual therapy compared with physical therapy in patients with non-specific neck pain: a randomized controlled trial. Chiropr Man Therap. 2017;25(1):12.
- Deyle GD, Gill NW, Rhon DI, et al. A multicentre randomised, 1-year comparative effectiveness, parallel-group trial protocol of a physical therapy approach compared to corticosteroid injections. BMJ Open. 2016;6:e010528.
- Martin RL, Irrgang JJ, Burdett RG, Conti SF, Van Swearingen JM. Evidence of validity for the foot and ankle ability measure (FAAM). Foot Ankle Int. 2005;26(11):968–83.
- Hiller CE, Refshauge KM, Bundy AC, Herbert RD, Kilbreath SL. The Cumberland ankle instability tool: a report of validity and reliability testing. Arch Phys Med Rehabil. 2006;87(9):1235–41.
- Smits-Engelsman B, Klerks M, Kirby A. Beighton score: a valid measure for generalized hypermobility in children. J Pediatr. 2011;158(1):119–23 123 e111–114.
- Jolman S, Robbins J, Lewis L, Wilkes M, Ryan P. Comparison of magnetic resonance imaging and stress radiographs in the evaluation of chronic lateral ankle instability. Foot Ankle Int. 2017;1071100716685526.
- Hubbard TJ, Cordova M. Mechanical instability after an acute lateral ankle sprain. Arch Phys Med Rehabil. 2009;90(7):1142–6.
- 39. Maitland GD. Maitland's vertebral manipulation, 7 edn. Elesevier: UK; 2005.
- Hoch MC, Andreatta RD, Mullineaux DR, English RA, Medina McKeon JM, Mattacola CG, McKeon PO. Two-week joint mobilization intervention improves self-reported function, range of motion, and dynamic balance in those with chronic ankle instability. J Orthop Res. 2012;30(11):1798–804.
- Powden CJ, Hoch JM, Hoch MC. Reliability and minimal detectable change of the weight-bearing lunge test: a systematic review. Man Ther. 2015;20(4):524–32.
- Bennell KL, Talbot RC, Wajswelner H, Techovanich W, Kelly DH, Hall AJ. Intrarater and inter-rater reliability of a weight-bearing lunge measure of ankle dorsiflexion. Aust J Physiother. 1998;44(3):175–80.
- Arendt-Nielsen L, Nie H, Laursen MB, Laursen BS, Madeleine P, Simonsen OH, Graven-Nielsen T. Sensitization in patients with painful knee osteoarthritis. PAIN. 2010;149(3):573–81.
- Rebbeck T, Moloney N, Azoory R, Hubscher M, Waller R, Gibbons R, Beales D. Clinical ratings of pain sensitivity correlate with quantitative measures in people with chronic neck pain and healthy controls: cross-sectional study. Phys Ther. 2015;95(11):1536–46.
- Frank L, McLaughlin P, Vaughan B. The repeatability of pressure algometry in asymptomatic individuals over consecutive days. Int J Osteopath Med. 2013;16(3):143–52.
- Chesterton LS, Sim J, Wright CC, Foster NE. Interrater reliability of algometry in measuring pressure pain thresholds in healthy humans, using multiple raters. Clin J Pain. 2007;23(9):760–6.
- Ferreira-Valente MA, Pais-Ribeiro JL, Jensen MP. Validity of four pain intensity rating scales. Pain. 2011;152(10):2399–404.
- Price DD, McGrath PA, Rafii A, Buckingham B. The validation of visual analogue scales as ratio scale measures for chronic and experimental pain. Pain. 1983;17(1):45–56.
- Roos EM, Brandsson S, Karlsson J. Validation of the foot and ankle outcome score for ankle ligament reconstruction. Foot Ankle Int. 2001;22(10):788–94.
- 50. Trojian TH, McKeag DB. Single leg balance test to identify risk of ankle sprains. Br J Sports Med. 2006;40(7):610–3.
- Gribble PA, Hertel J, Plisky P. Using the star excursion balance test to assess dynamic postural-control deficits and outcomes in lower extremity injury: a literature and systematic review. J Athl Train. 2012;47(3):339–57.
- Pionnier R, Decoufour N, Barbier F, Popineau C, Simoneau-Buessinger E. A new approach of the star excursion balance test to assess dynamic postural control in people complaining from chronic ankle instability. Gait Posture. 2016;45:97–102.
- Robinson RH, Gribble PA. Support for a reduction in the number of trials needed for the star excursion balance test. Arch Phys Med Rehabil. 2008;89(2):364–70.
- 54. Gribble PA, Hertel J. Considerations for normalizing measures of the star excursion balance test. Meas Phys Educ Exerc Sci. 2003;7(2):89–100.
- Owens JE, Menard M. The quantification of placebo effects within a general model of health care outcomes. J Altern Complement Med. 2011;17(9):817–21.

 van der Windt DA, Koes BW, Deville W, Boeke AJ, de Jong BA, Bouter LM. Effectiveness of corticosteroid injections versus physiotherapy for treatment of painful stiff shoulder in primary care: randomised trial. Bmj. 1998; 317(7168):1292–6.

(2019) 20:75

- Kamper SJ, Ostelo RWJG, Knol DL, Maher CG, de Vet HCW, Hancock MJ. Global Perceived Effect scales provided reliable assessments of health transition in people with musculoskeletal disorders, but ratings are strongly influenced by current status. J Clin Epidemiol. 2010;63(7):760–766.e761.
- Seebauer CJ, Bail HJ, Rump JC, Hamm B, Walter T, Teichgräber UKM. Ankle laxity: stress investigation under MRI control. Am J Roentgenol. 2013;201(3):496–504.
- Johannsen A. Radiological diagnosis of lateral ligament lesion of the ankle. A comparison between talar tilt and anterior drawer sign. Acta Orthop Scand. 1978;49(3):295–301.
- 60. Dutton M. Dutton's Orthopaedic Examination, Evaluation, and Intervention, 4thg edn. New York: mCGraw-Hill Companies; 2017.
- Lee KM, Chung CY, Kwon S-S, Chung MK, Won SH, Lee SY, Park MS. Relationship between stress ankle radiographs and injured ligaments on MRI. Skelet Radiol. 2013;42(11):1537–42.
- Beynnon BD, Webb G, Huber BM, Pappas CN, Renström P, Haugh LD. Radiographic measurement of anterior talar translation in the ankle: determination of the most reliable method. Clin Biomech. 2005;20(3):301–6.
- Prado MP, Fernandes TD, Camanho GL, Mendes AAM, Amodio DT. Mechanical instability after acute ankle ligament injury: randomized prospective comparison of two forms of conservative treatment. Rev Bras Ortop. 2013;48:307–16.
- Ellis SJ, Williams BR, Pavlov H, Deland J. Results of anatomic lateral ankle ligament reconstruction with tendon allograft. HSS J. 2011;7(2):134–40.
- Teo TKB, Tay KH, Lin SE, Tan SG, Lo RH, Taneja M, Irani FG, Sebastien MG, Lim KH, Tan BS. Endovenous Laser Therapy in the Treatment of Treatment of Lower-limb Venous Ulcers. J Vasc Interv Radiol. 2010;21(5):657–662.
- de Bie RA, de Vet HC, Lenssen TF, van den Wildenberg FA, Kootstra G, Knipschild PG. Low-level laser therapy in ankle sprains: a randomized clinical trial. Arch Phys Med Rehabil. 1998;79(11):1415–20.
- Kingsley JD, Demchak T, Mathis R. Low-level laser therapy as a treatment for chronic pain. Front Physiol. 2014;5:306.
- Cotler HB, Chow RT, Hamblin MR, Carroll J. The use of low level laser therapy (LLLT) for musculoskeletal pain. MOJ Orthop Rheumatol. 2015;2(5):00068.
- Irnich D, Behrens N, Molzen H, Konig A, Gleditsch J, Krauss M, Natalis M, Senn E, Beyer A, Schops P. Randomised trial of acupuncture compared with conventional massage and "sham" laser acupuncture for treatment of chronic neck pain. Bmj. 2001;322(7302):1574–8.
- Reid SA, Rivett DA, Katekar MG, Callister R. Sustained natural apophyseal glides (SNAGs) are an effective treatment for cervicogenic dizziness. Man Ther. 2008;13(4):357–66.
- Pellow JE, Brantingham JW. The efficacy of adjusting the ankle in the treatment of subacute and chronic grade I and grade II ankle inversion sprains. J Manip Physiol Ther. 2001;24(1):17–24.
- Merlin DJ, McEwan IM, Thom JM. Mulligan's mobilisation with movement technique for lateral ankle pain and the use of magnetic resonance imaging to evaluate the "positional fault" hypothesis; 2005.
- Vicenzino B, Paungmali A, Teys P. Mulligan's mobilization-with-movement, positional faults and pain relief: current concepts from a critical review of literature. Man Ther. 2007;12(2):98–108.

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

